This document is made available electronically by the Minnesota Legislative Reference Library as part of an ongoing digital archiving project. http://www.leg.state.mn.us/lrl/lrl.asp EXECUTION COPY

# CONSTRUCTION SERVICES AGREEMENT BETWEEN THE MINNESOTA SPORTS FACILITIES AUTHORITY AND THE CONSTRUCTION MANAGER

# TABLE OF CONTENTS

	Page
ARTICLE 1 GENERAL PROVISIONS	
1.1 Basic Definitions	
1.2 Execution, Correlation and Intent	
1.3 Construction Manager's Authorized Representatives	
1.3.1 Key Personnel	
ARTICLE 2 CONSTRUCTION MANAGER'S RESPONSIBILITIES	
2.1 Basic Services and Responsibilities	7
2.1.1 Construction Services	7
2.2 Additional Services	8
2.3 Authorization to Proceed	8
ARTICLE 3 AUTHORITY'S RESPONSIBILITIES	10
ARTICLE 4 CONTRACT TIME	12
4.2 Liquidated Damages	12
ARTICLE 5 Guaranteed Maximum Price	16
5.1 Guaranteed Maximum Price (GMP)	16
5.2 Construction Manager Corporate Guaranty	18
ARTICLE 6 DISCOUNTS, REBATES, AND REFUNDS	19
ARTICLE 7 ACCOUNTING RECORDS	20
ARTICLE 8 COSTS TO BE REIMBURSED	22
8.1.1 Construction Manager's Fee	22
8.1.2 Labor Costs	22
8.1.3 Subcontract Costs	23
8.1.4 Costs of Materials and Equipment Incorporated in the Completed Construction	23
8.1.5 Cost of Other Materials and Equipment, Temporary Facilities and Related Items	23
8.1.6 Miscellaneous Costs	25
8.1.7 Other Costs	25
8.1.8 Emergencies: Repairs to Damaged, Defective or Nonconforming Work	26
8.1.9 Value Added Taxes, Including Goods & Services Tax, Sales Taxes	26
8.1.10 Affiliated Entity Transaction	
ARTICLE 9 COSTS NOT TO BE REIMBURSED	27
ARTICLE 10 SUBCONTRACTS AND OTHER AGREEMENTS	29
10.9 Subcontractors and Suppliers	
ARTICLE 11 PAYMENTS	
ARTICLE 12 FINAL PAYMENT	
ARTICLE 13 PROTECTION OF PERSONS AND PROPERTY	
ARTICLE 14 CHANGES IN THE WORK	
14.1 Construction Services	

14.2	Regulatory Changes	39
ARTICLI	E 15 CORRECTION OF WORK	40
ARTICLI	E 16 INSURANCE, DISPUTE RESOLUTION, TERMINATION, AND INDEMNITY	41
16.1	Insurance	41
16.2	Dispute Resolution	41
16.3	Termination by the Authority	41
16.4	Termination by the Construction Manager	41
16.5	Indemnification	41
ARTICLI	E 17 OTHER PROVISIONS	42
17.1	Nondiscrimination	42
17.2	Successors and Assigns	42
17.3	Legal Relations	42
17.4	Third Party Beneficiaries	43
17.5	Governing Law, Venue, and Jurisdiction	43
17.6	Time is of the Essence	44
17.7	Lender Approval	44
17.8	Construction Manager Performance	44
17.9	Payment Bond	44
17.10	Performance Bond	44
17.11	Warranty of Construction	45
17.12	Environmental and Other Responsibilities During Construction	45
17.13	Entire Agreement	45
17.14	Notice Provisions	45
17.23	Design/Build Option	48
17.24	Minnesota False Claims Act	48
17.25	Bond Capacity	48
17.26	Structure Consultants	
EXHIBIT	1 DESCRIPTION OF DESIGNATED SERVICES	50
EXHIBIT	2 [RESERVED]	58
	3 CONSTRUCTION MANAGEMENT PLAN INCLUDING CONSTRUCTION ER'S GUARANTEED MAXIMUM PRICE AND CONSTRUCTION SCHEDULE	59
	4 MASTER PROJECT SCHEDULE	
	5 PROJECT STAFFING	
	6 GENERAL CONDITIONS OF THE CONTRACTS FOR DESIGN AND	=0
	UCTION	
	7 EQUITY PLAN FOR CONSTRUCTION SERVICES AGREEMENT	
	8 CONSTRUCTION MANAGER PARENT COMPANY GUARANTY	
	9 CONSTRUCTION MANAGER'S BONDS	
LAHIBII	11 CONTRACT REVISION AND CHANGE ORDER FORMS	03

EXHIBIT 12 ENVIRONMENTAL AND OTHER RESPONSIBILITIES DURING	
CONSTRUCTION	
EXHIBIT 13 MINNESOTA MULTIPURPOSE STADIUM, MINNEAPOLIS, MINNESO	
NOTES OF CLARIFICATION, CONSTRUCTION SERVICES AGREEMENT	76
EXHIBIT 14 PROJECT COST IDENTIFICATION MATRIX	85

# CONSTRUCTION SERVICES AGREEMENT BETWEEN THE MINNESOTA SPORTS FACILITIES AUTHORITY AND THE CONSTRUCTION MANAGER

# **CONSTRUCTION SERVICES AGREEMENT**

THIS CONSTRUCTION SERVICES AGREEMENT ("Construction Services Agreement") is made as of the fifteenth day of February in the year of 2013.

**BY AND BETWEEN** 

The AUTHORITY:

MINNESOTA SPORTS FACILITIES AUTHORITY 900 South Fifth Street Minneapolis, MN 55415

and the

**CONSTRUCTION MANAGER:** 

M. A. Mortenson Company 700 Meadow Lane North Minneapolis, Minnesota 55422

For the following **PROJECT:** The Stadium and Stadium Infrastructure at the Stadium Site.

The Architecture / Engineering Services for the Project shall be provided by HKS, Inc. (and is defined as the "Architect" for the purposes hereof).

\_1

# TERMS AND CONDITIONS OF CONSTRUCTION SERVICES AGREEMENT

# **RECITALS**

WHEREAS, the owner of the Project is the Minnesota Sports Facilities Authority, a public body, corporate and politic and political subdivision of the State of Minnesota (and is defined as the "Authority" for purposes hereof), or its assigns or designees; and

WHEREAS, the Authority was established to construct, finance, and operate a multi-purpose stadium and related infrastructure as a venue for the National Football League and a broad range of other civic, community, athletic, educational, cultural, and commercial activities in the City of Minneapolis, Minnesota, and desires to develop certain real estate as a public project described herein as the Stadium Site; and

WHEREAS, the Authority and Minnesota Vikings Football, LLC, a Delaware limited liability company authorized and doing business in the State of Minnesota (and is defined as the "Team" for purposes hereof) have entered into a Preliminary Development Agreement ("PDA"); and

WHEREAS, the Authority and the Team will enter into a Development Agreement ("DA"); and

WHEREAS, the Project is being developed pursuant to 2012 Minnesota Laws, Chapter 299, as enacted or hereafter amended or supplemented, and including any successor law, providing for, among other things, the construction, financing, operation, and long-term use of a multi-purpose stadium and related infrastructure as a venue for the National Football League and a broad range of other civic, community, athletic, educational, cultural, and commercial activities in the City of Minneapolis, Minnesota (and is defined as the "Act" for purposes hereof); and

WHEREAS, the Act provides that the Authority shall serve as the Stadium Developer except that, prior to the time the Authority enters into a construction contract with the Construction Manager that contains a certified maximum price and completion date, and at the request of the Team, the Authority may authorize the Team to serve as Stadium Developer for the management of the construction of the Stadium and Stadium Infrastructure (collectively defined as the "**Project**" for purposes hereof and as the term "Project" is further clarified in the definitions that are attached to **Exhibit 6** of this Construction Services Agreement); and

WHEREAS, the Authority has engaged the services of Hammes Company Sports Development, Inc., as its representative to assist in the planning, development, design, and construction of the Project. The Team has engaged the services of ICON Venue Group, LLC, as its representative to assist in the planning, development, design, and construction of the Project; and

**WHEREAS**, the Team and the Authority desire to use a collaborative process to reach consensus on key elements of the Project design, as is detailed in the PDA and as will be more fully described in the DA; and

WHEREAS, to effectuate the collaborative process, the Authority and the Team have established the Stadium Design and Construction Group (and is defined as the "SDC Group" for purposes hereof) to manage the design and oversee construction of the Project as required by the Act and as further specified in the PDA or DA; and

WHEREAS, the SDC Group, subject to certain approvals of the Authority and the Team as established in the PDA, shall direct the Construction Manager under this Construction Services Agreement during the Pre-Construction Phase or until otherwise directed by the Authority pursuant to the PDA or the DA; and

WHEREAS, whenever the Authority's approval, direction, or decision is required pursuant to this Construction Services Agreement during the Pre-Construction Phase, the Construction Manager acknowledges and agrees that such approval shall be provided by the SDC Group as specified in the PDA or the DA or until otherwise directed by the Authority pursuant to the PDA or the DA; and

WHEREAS, Construction Manager acknowledges and agrees that whenever the Authority's written authorization is required under this Construction Services Agreement, such written authorization shall be signed and delivered by the Authority to the Construction Manager, including execution of the Construction Services Agreement; and

**WHEREAS**, the Authority desires to retain the Construction Manager to provide Pre-Construction Phase Services and Construction Phase Services and accomplish the Work pursuant to the terms and conditions of this Construction Services Agreement; and

**WHEREAS**, the Construction Manager's Work for the Project shall be provided by Persons lawfully licensed to perform such Work in the State of Minnesota; and

WHEREAS, the Construction Manager acknowledges and agrees that it shall be required to negotiate and implement a Project Labor Agreement as approved by the Authority; and

WHEREAS, the Architect shall prepare Design Documents from which the Construction Manager shall prepare estimates of Construction Cost and a guaranteed maximum price ("GMP") for the Work that will not exceed the Fixed Construction Budget (as the term "Fixed Construction Budget" is further clarified in the definitions that are attached to <u>Exhibit 6</u> of this Construction Services Agreement), as established between the Authority and Construction Manager, which Fixed Construction Budget is currently set at \$690,000,000; and

**WHEREAS**, the Construction Manager acknowledges that the Authority has a not-to-exceed Fixed Construction Budget for the Work and that there is a Construction Schedule including a Substantial Completion Date by which the Construction Manager must achieve Substantial Completion; and

WHEREAS, the Construction Manager represents that it will complete the Work for an amount not to exceed the certified GMP and achieve Substantial Completion at or before the certified completion date as set forth in the Construction Schedule; and

WHEREAS, the Construction Manager acknowledges and agrees that the intent of this Construction Services Agreement is to provide all aspects of the Work in order to provide the Authority with a Project that is of high quality, fully functional and operation-ready on the Date of Substantial Completion in strict accordance with the GMP and the Construction Schedule, and the Construction Manager acknowledges and agrees to this primary intent while recognizing that the Contract Documents will continue to be developed, including completion and coordination of any and all Drawings, Specifications, Shop Drawings, Product Data, Samples, Bulletins, or answers to Requests for Information prepared by the Architect and/or other member of the Project Team; and

WHEREAS, the Construction Manager is strictly responsible to coordinate the Work with the work of all members of the Project Team, including the coordination of any and all Shop Drawings, Product Data, Samples, or Requests for Information to the Architect in order to meet the Authority's objectives as stated herein; and

WHEREAS, the Construction Manager acknowledges and agrees that the Authority, at its sole discretion, may elect to convert this Construction Services Agreement to a design/build delivery system on terms mutually agreeable to the Authority and Construction Manager; and

**WHEREAS**, the Parties desire to set forth the terms of their agreements in this Construction Services Agreement to be effective as of the date set forth above.

**NOW, THEREFORE**, for valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Construction Manager and the Authority hereby agree as follows:

# [THIS SPACE LEFT INTENTIONALLY BLANK]

#### GENERAL PROVISIONS

#### **1.1 Basic Definitions**

- **1.1.1** The above Recitals are incorporated herein.
- **1.1.2** The definitions set forth in <u>Exhibit 1</u> hereof and in <u>Appendix A</u> of the General Conditions included as <u>Exhibit 6</u> hereto shall define certain terms in this Construction Services Agreement.
- **1.1.3** <u>Exhibits 1</u> through <u>13</u> are attached to and are incorporated as part of this Construction Services Agreement for all purposes, and their terms apply to and obligate the Construction Manager as if fully set forth herein.
- **1.1.4** Whenever the terms GMP, Project Schedule, Substantial Completion (or Date of Substantial Completion), or Final Completion (or Date of Final Completion) are used in the Construction Services Agreement and its Exhibits, those terms also mean "as may be adjusted as allowed by the Contract Documents."

#### **1.2** Execution, Correlation and Intent

- **1.2.1** The Construction Manager and Authority shall sign this Construction Services Agreement in not less than triplicate.
- **1.2.2** It is the intent of the Construction Manager and Authority that the Contract Documents include all items necessary for proper execution and completion of the Work. The Contract Documents are complementary, and what is required by any one shall be as binding as if required by all. Work not covered in the Contract Documents, but reasonably inferable from the Contract Documents, will be included as part of the Work. Words and abbreviations that have well-known technical or trade meanings are used in the Contract Documents in accordance with such recognized meanings.
- **1.2.3** The Construction Manager will review the Design Documents as they are being produced, and provide comments based on constructability, Value Engineering, and known inconsistencies, ambiguities, or errors, especially before it provides a Construction Management Plan containing its GMP Proposal as provided in <u>Exhibit 3</u> of the Construction Services Agreement; as a result, any inconsistencies among and between the Contract Documents shall be governed by the most strict or stringent requirements included therein and as determined by the Authority.
- **1.2.4** The Construction Manager accepts a duty of trust and confidence toward the Authority created by this Construction Services Agreement, and covenants with the Authority to furnish the Construction Manager's skill and judgment and to cooperate with the Project Participants in furthering the interests of the Authority. Consistent with its Standard of Care, the Construction Manager shall furnish construction administration and management services and use the Construction Manager's expertise and best efforts to perform the Work in an expeditious and economical manner and in furtherance of the interests of the Authority.
- **1.2.5** The Construction Manager hereby acknowledges and agrees that it is the intent of Construction Manager and Authority under this Construction Services Agreement that Construction Manager shall fully advise and consult with the Authority, the Team, their representatives, and the Architect from the early stages of design through completion of

construction with respect to, and shall be solely responsible for the Pre-Construction Phase Services and Construction Phase Services and completion of the Work in strict accordance with the Contract Documents, within the GMP, and by the Date of Substantial Completion in the Construction Schedule. The Construction Manager shall lead the efforts of the Project Team in determining the most expeditious Construction Schedule for completion of the Work, appropriate construction means and methods for assembly and constructability of the Project; appropriate materials to fulfill the performance and Program objectives of the Authority during development of the Design Documents or during construction when the Design Documents do not specify the exact materials to be used. and other matters related to the Work. Consistent with its Standard of Care, the Construction Manager is responsible for continuous and thorough review of the Construction Documents for clarity, consistency, and completeness to ensure that the Construction Costs will not exceed the Fixed Construction Budget. Except to the extent the Contract Documents may expressly make the Construction Manager responsible for design or expressly identify a specification as a performance specification, nothing in this Construction Services Agreement or elsewhere in the Contract Documents shall be construed to create any responsibility of or liability upon the Construction Manager to third parties not under the control or responsibility of the Construction Manager for the accuracy, adequacy, sufficiency, or safety of the design of the Project; provided, however, that the Construction Manager shall not make a Claim against the Authority or the Team for an adjustment to the GMP or Project Schedule due to the accuracy, adequacy, sufficiency, or safety of the design except as may be expressly allowed under the Contract Documents.

#### 1.3 Construction Manager's Authorized Representatives

The Construction Manager's representative authorized to act on behalf of the Construction Manager with respect to the Project is John V. Wood, or his/her successor designated in writing and approved by the Authority.

#### 1.3.1 Key Personnel

The Construction Manager's Key Personnel and Project staffing in connection with the Project are identified in Exhibit 5 hereto. Key Personnel may not be removed from the Project, be reassigned or be assigned to additional projects so long as they remain in the Construction Manager's employ. If Key Personnel die, become disabled or otherwise leave the employ of the Construction Manager prior to completion of the Work, the Construction Manager shall provide a replacement with another individual of comparable education and experience, who shall likewise be subject to the Authority's written approval. The Construction Manager shall provide staffing for the Project at not less than the minimum levels set forth on **Exhibit 5**. The Authority shall have the right to approve the Construction Manager's appointment of the Construction Manager's supervisors, with such approval not to be unreasonably withheld. The Authority shall have, in its sole discretion, the right to request the dismissal of the Construction Manager's supervisors or any other personnel employed by the Construction Manager or its Subcontractors of any tier and shall have approval rights over any such replacement supervisors with such approval not to be unreasonably withheld. The Construction Manager shall immediately comply with any such requests by the Authority.

#### **CONSTRUCTION MANAGER'S RESPONSIBILITIES**

#### 2.1 Basic Services and Responsibilities

#### 2.1.1 Construction Services

- .1 The Construction Manager shall provide all services, work, labor, materials, equipment and other necessary payments as are required to complete the Work as set forth in the Contract Documents, including any and all Pre-Construction Phase Services and Construction Phase Services (collectively, **"Basic Services"**) as are set forth on **Exhibit 1**. The Work is intended to be inclusive of all construction, services, work, labor, materials, and equipment required by, or reasonably inferable from, the Construction Documents necessary to complete the Work pursuant to the Construction Schedule and within the GMP.
- .2 The Work is more completely described in the Contract Documents, including all of the Work as defined in the Exhibits hereto.
- .3 The Work consists of those services performed by the Construction Manager, its employees, Subcontractors, Suppliers, and Sub-subcontractors of all tiers. The Construction Manager is solely and strictly responsible for the performance of all Persons performing the Work, including Subcontractors, Suppliers, and Sub-subcontractors of all tiers. It is the responsibility of the Construction Manager to ensure that all Subcontractors and Suppliers, including Sub-subcontractors at every tier, that is performing any portion of the Work are familiar with all the terms and conditions of this Construction Services Agreement and the Contract Documents and that their performance is in accordance therewith.
- .4 It shall be the duty of the Construction Manager to review, direct, and supervise construction of the Work. The Construction Manager is responsible to complete the Work in strict accordance with the Contract Documents. In the event the Construction Manager undertakes construction activities, the Construction Manager shall continually inspect its Work. The Construction Manager shall cooperate in all respects with the Design Services of the Architect or the services or work of any other Project Participant involved in the Work or Project. The Construction Manager shall cooperate in all respects with the Authority, which may include lenders, any lender's inspecting architect or representative, insurers, the Team, sponsors, the National Football League, television broadcast networks, radio broadcasters, cellular/distributed antennae providers, and Governmental Authorities.
- .5 The Construction Manager shall be solely responsible for the means, methods, techniques, and sequences it chooses to perform the Work and shall ensure that when finished, the Work complies completely with the Contract Documents. No inspection or approval by others, including the Authority, Team, Architect, SDC Group, and/or any Project Participant or Governmental Authority, shall relieve the Construction Manager from this responsibility. Pursuant to its Standard of Care, the Construction Manager shall supervise and direct the Work efficiently, shall at all times enforce good order among Persons under its control, and shall ensure that all Persons under its control have the skills necessary to perform the portion of the Work assigned to them. The Construction Manager shall be responsible for the safe and complete performance of all of its Subcontractors, Suppliers, and Sub-Subcontractors of every tier.

- .6 Without diminishing the Construction Manager's obligation to meet all requirements of this Construction Services Agreement, the Construction Manager agrees that its four most important requirements are as follows:
  - .1 to achieve Substantial Completion of all or part of the Work on or before the dates set forth in the Construction Schedule,
  - .2 to minimize the Construction Costs and to absolutely complete the Work within the GMP,
  - .3 to complete the Work with the quality required by the Contract Documents, and
  - .4 to maintain the highest standards of safety on the Project Site.

#### 2.2 Additional Services

- **2.2.1** Other than the Basic Services and responsibilities set forth in **Paragraph 2.1** hereof, the Construction Manager shall not be required to provide any Additional Services, except when ordered to do so in writing.
- **2.2.2** Neither the Authority nor Construction Manager anticipates any Additional Services at the time this Construction Services Agreement is executed.

#### 2.3 Authorization to Proceed

- 2.3.1 From and after the date hereof, the Construction Manager is authorized to perform only those services that involve Pre-Construction Phase Services required to develop the Contract Documents for the purpose of establishing a Construction Management Plan between the Authority and Construction Manager. The Construction Manager acknowledges and agrees that the Construction Manager shall not, without written Notice to Proceed from the Authority, proceed with any Construction Phase Services. Unless otherwise directed by the Authority pursuant to Subparagraph 2.3.2 herein, the Authority will only issue a Notice to Proceed with the Construction Phase Services upon the Authority's receipt and written approval of the Construction Manager's Construction Management Plan, including its GMP Proposal and Construction Schedule. The Construction Manager will proceed with the Construction Phase Services only after the Authority has provided written acceptance of the Construction Manager's Construction Management Plan, which shall be determined to be acceptable in the sole and absolute discretion of the Authority and Team.
- **2.3.2** Notwithstanding the requirements in **Subparagraph 2.3.1** herein, the Authority and Construction Manager expressly acknowledge and agree that the Authority may provide the Construction Manager with a limited Notice to Proceed in writing for specific portions of the Work to expedite the construction of the Project. Any part of the Work that the Construction Manager is authorized to perform before acceptance of the Construction Manager's Construction Management Plan shall be performed pursuant to this Construction Services Agreement and such other terms and conditions as may be reasonably required by the Authority. If the Authority accepts the Construction Manager's Construction Manager will become part of the GMP. Notwithstanding the issuance of any form of Notice to Proceed or under **Subparagraphs 2.3.1** or **2.3.2** hereof, no work on the Stadium foundations may be commenced prior to the publication of the Authority's determination that the final environmental impact statement ("**EIS**") for the Project is adequate.

**2.3.3** The Construction Manager expressly acknowledges and agrees that the Authority may, at any time during the Pre-Construction Phase, terminate development of the Project or this Construction Services Agreement for cause or the convenience of the Authority pursuant to **Paragraph 16.3** hereof. The Authority does not guarantee, represent, or warrant that the Construction Manager will be retained to perform the Construction Phase Services, and the Construction Manager agrees and acknowledges that it shall have no rights, Claims, or entitlement of any kind to perform or receive any compensation or payment of any kind on account of the Construction Phase Services unless and until the Construction Manager to proceed with all or part of the Construction Phase Services. Issuance of a written Notice to Proceed shall be an express condition to the Authority's obligations regarding the Construction Phase Services portion of this Construction Services Agreement.

#### **AUTHORITY'S RESPONSIBILITIES**

- **3.1** The Authority shall have the obligations and responsibilities set forth herein and in the Contract Documents. Whenever the Contract Documents require action by a member of the Project Team, the Authority shall endeavor to use reasonable efforts to cause the Architect, Construction Manager, or other member of the Project Team to take appropriate action and to render decisions in a timely manner, although failure by the Authority to do so shall not absolve the Architect, Construction Manager, or other member of the Project Team from their respective responsibilities under their respective agreements to take such actions and to render decisions in a timely manner in accordance with the Contract Documents. The Construction Manager is responsible to cooperate and coordinate with the Architect to complete the Work and assist the Architect in its completion of the Design Services.
- **3.2** The Authority shall examine documents submitted by the Architect, Construction Manager, or other Project Team members and shall promptly render decisions pertaining to such documents.
- **3.3** Notwithstanding any other provision in the Contract Documents, during the Pre-Construction Phase, whenever the Authority or the Architect are required or permitted under the Contract Documents to review any submittal by the Construction Manager or to observe any of the Work or to give any approval, consent, decision, or certificate, or exercise any discretion or make any determination, the Authority shall act with the approval of the SDC Group as provided in the PDA or DA. The Authority and Team shall designate the Person that during the Construction Phase will provide approvals, consents, decisions, determinations, or any other act on behalf of the Authority under this Construction Services Agreement, and the Authority will advise the Construction Manager of the identity of that Person at or before the time the Construction Manager will commence its Construction Phase Services.
- **3.4** The Authority shall fully cooperate with the Construction Manager who shall be responsible to secure all building and other permits, licenses, and inspections necessary for the Work. The Authority shall pay directly for primary building permit(s) (including Mechanical and Electrical permits) which costs shall be included in the Fixed Construction Budget and shall be deemed a Project Cost. Construction Manager shall include in the GMP costs for all other fees such as permits, licenses, and inspections, which shall be reimbursed as a Cost of the Work.
- **3.5** The Authority shall furnish the services of land surveyors or geotechnical engineers and other consultants for subsoil, air, and water conditions or other professional consultants when such services are deemed necessary by the Architect to properly carry out the Work and only after such services are agreed to by the Authority. The Authority shall furnish, directly or through the Architect, structural, mechanical, chemical, geotechnical, and other laboratory or on-site testing and reports to the extent set forth in the Contract Documents and as required by Applicable Laws.
- **3.6** Other than as stated in **Paragraph 3.5** above, the Authority or Indemnitees shall not be responsible for inspecting or testing any portion or progress of the Work, and any tests or inspections performed by the Authority, the Architect, an Indemnitee, or any of their agents shall not relieve the Construction Manager from properly and timely performing the Work.

#### 3.7 [RESERVED]

#### 3.8 [RESERVED]

**3.9** The Authority and Indemnitees are relying on the Construction Manager's GMP, which contains a Construction Manager Contingency as provided in <u>Exhibit 3</u> to the Construction Services Agreement, to pay for all Costs of the Work and Claims for additional payment (other than for approved Contract Revisions) above the GMP. The Act permits the Authority to enter into this

Construction Services Agreement only on the express condition that the Construction Manager is responsible for cost overruns above the GMP; as a result, the Construction Manager shall be responsible to pay all Costs of the Work and other costs or expenses that exceed the GMP. The Authority and Indemnitees (except the Architect Indemnitees with respect to design errors or omissions or breaches of the Design Services Agreement) are not, and shall in no event be, responsible or liable to any member of the Project Team including the Construction Manager for any aspect of the Design Services, including inspections, quality control or design administration services that will be provided by the Architect under its Design Services Agreement, other than as provided in Paragraph 4.3 herein. Likewise, the Authority and the Indemnitees (except the Architect Indemnitees with respect to design errors or omissions or breaches of the Design Services Agreement) are not and shall in no event be, responsible or liable to the Construction Manager or other Project Team members for any Claims for payment of additional money or costs above the GMP arising out of or related to any aspect of the Construction Manager's performance of the Work, except with respect to the Authority's obligations under Paragraph 3.4 of Exhibit 6 hereto. In no event shall the Authority or Indemnitees have any responsibility or liability for construction means, methods, techniques, sequences, or procedures, or for safety precautions and programs in connection with the Work, notwithstanding any of the rights and authority granted the Authority and Indemnitees in the Contract Documents.

#### CONTRACT TIME

**4.1** The Construction Manager shall perform the Work expeditiously, efficiently, and safely in accordance with the Construction Schedule. The Construction Manager shall commence its Work upon receipt of a Notice to Proceed as provided in **Subparagraph 2.3.1** hereof and, subject to adjustment in accordance with and allowed by the Contract Documents, Construction Manager shall achieve the Milestones and Substantial Completion of the Work within the time frames specifically described in **Exhibit 4** hereof. The Construction Manager shall be considered to have achieved Final Completion only after the Construction Manager completes all of the Work and the Work is accepted by the Authority and Architect. The staging of the Work is more definitively described in **Exhibit 4** hereof.

#### 4.2 Liquidated Damages

The Authority and Construction Manager acknowledge and agree that the Authority and Team are making a considerable investment and the damages from the risks inherent in the Project are difficult to forecast or predict, but that the Authority is relying principally on the Construction Manager's experience and expertise to complete the Work on schedule and to mitigate the risks of late completion to the Authority. For purposes of this Paragraph 4.2, the Authority and Construction Manager have established a structure to address the Construction Manager's liability for damages for delays in Substantial Completion according to the liquidated amounts described below. The liquidated damages agreed to below are the Authority's sole and exclusive remedy for delay-related damages caused by Construction Manager's unexcused failure to achieve Substantial Completion by the Date of Substantial Completion, and shall be the sole remedy for any such delay (and, for clarity, the liquidated damages described in this Paragraph 4.2 are the sole remedy of the Indemnitees in the aggregate for delay-damages caused by the Construction Manager's unexcused failure to achieve Substantial Completion by the Date of Substantial Completion); notwithstanding the foregoing, the Authority shall also have the right to undertake Extraordinary Measures under Paragraph 4.10.7 of Exhibit 6. Nothing contained in this Paragraph 4.2 or elsewhere in the Contract Documents (except the following five sentences), including Exhibit 4 hereof, shall in any manner limit the Authority's rights or ability to recover damages for the Construction Manager's breach of its contractual duty to perform its Work according to the Construction Schedule or to recover liquidated damages for the Construction Manager's failure to achieve Substantial Completion of its Work by the Date of Substantial Completion, or to recover direct and consequential damages due to repair of Construction Manager's Work or damages due to defective construction that are observed after Final Completion. Notwithstanding anything to the contrary in this **Paragraph 4.2** or elsewhere in the Contract Documents, the liability of the Construction Manager for consequential damages (including those described in **Paragraph 6.2.10** of **Exhibit 6** hereto) payable to the Authority and the Indemnitees (other than liquidated damages payable under this Paragraph 4.2), in the aggregate, shall not exceed (a) consequential damages recovered by insurance to the extent of the limits required under the CSA (including any project-specific insurance procured), and (b) in addition to any amounts recovered under part (a), additional consequential damages in the amount of the higher of either (i) Twenty Million Dollars (\$20,000,000) or (ii) ten percent of the aggregate limits of coverage in any project-specific liability or professional liability insurance procured for the Project, with a maximum liability under subpart (b) of Thirty Million Dollars (\$30,000,000). For clarity, deductibles on insurance described in subpart (a) in the previous sentence and uninsured consequential damages shall be included in the amounts recoverable by the Authority in subpart (b). Actual recovery from insurance under subpart (b) is not a condition precedent to the Authority's right to recover said deductibles and uninsured consequential damages from the Construction Manager, but other than making a Claim asserting a right to said deductibles and uninsured consequential damages, the Authority will defer proceeding to dispute resolution against the Construction Manager under Paragraph 16.2 hereof to recover said deductibles and uninsured consequential damages for a period of one year after Authority makes said Claim unless

otherwise ordered by a court or arbitration panel to proceed more quickly. Each party is required to maintain those policies of insurance set forth in the Contract Documents, and to take reasonable steps to obtain coverage for any claim for consequential damages from such policies. Any project specific liability policy and any property insurance procured by the Authority shall include a waiver of subrogation in favor of the Construction Manager and contain customary deductibles for projects of comparable size; provided, however, that if the Authority's property policy does not contain a waiver of subrogation in favor of the Construction Manager (or that would honor this waiver of subrogation that includes the Construction Manager), then the Authority will use commercially reasonable efforts to request its property insurer to include a waiver of subrogation in favor of the Construction Manager as part of the Authority's property insurance, but if the property insurer refuses to offer a waiver of subrogation as part of the Authority's property insurance, then the Authority will so advise the Construction Manager at which time the Construction Manager may assist the Authority to obtain property insurance with said waiver, and if the addition of said waiver would cost the Authority additional premium, then the Construction Manager will have the opportunity to either pay the additional premium, in which case the Authority will obtain said waiver, or the Construction Manager may choose not to pay the additional premium in which case the Authority will have the option not to obtain said waiver, and the requirement that the Authority obtain said waiver of subrogation shall no longer be applicable.

## 4.2.1 Liquidated Damages

The Construction Manager acknowledges that the Authority has advised the Construction Manager regarding the following:

- .1 the Authority, the Team, and their respective Affiliates are incurring substantial and unprecedented costs to construct the Project;
- .2 the Authority, the Team, and their respective Affiliates will be incurring significant indebtedness to pay such costs, which indebtedness must be serviced whether or not Substantial Completion occurs by the Date of Substantial Completion;
- .3 to justify incurring such costs and the risks being undertaken in connection with the design and construction of the Project, the Authority, the Team, and their respective Affiliates have developed a business plan that will generate substantial revenues and provide substantial new business opportunities that cannot be fully realized if Substantial Completion does not occur by the Date of Substantial Completion;
- .4 as an essential part of the consideration from the Construction Manager under this Construction Services Agreement, the Construction Manager has committed to achieve Substantial Completion not later than the Date of Substantial Completion, which will allow the Authority and the Team, among other things, the opportunity to realize the revenues and benefits expected from the Project after that date, which is essential to the public purpose and the financial success of the Project; and
- .5 if Substantial Completion of the Project does not occur by the Date of Substantial Completion, the Authority, the Team, and their respective Affiliates will suffer substantial harm, the damages for which may be incapable of being determined with certainty, very difficult to estimate accurately, likely entail substantial cost and inconvenience, and difficult to prove and calculate. The elements of such harm are expected to include: (1) lost or substantially diminished revenues from contracts that cannot be performed fully (or at all); (2) business opportunities that cannot be realized or exploited fully (or at all); (3) substantial harm and damage to the reputation of the Authority and its

relationships with the public, users of the Stadium, advertisers, sponsors, and other related Persons; (4) substantial harm and damages to the reputation of the Team and its relationships with the public, ticket holders, advertisers, sponsors, and other related Persons; (5) diversion of management time; and (6) loss of full and timely use of the Project and its components to their maximum potential from and after the scheduled Date of Substantial Completion.

.6 in light of the acknowledgements set forth above, and including the potential difficulty in calculating precisely the damages that the Authority, the Team, and their related affiliates would incur, the Parties have agreed that if all or parts of the Work is not Substantially Complete by the Date of Substantial Completion in the Construction Schedule, the Construction Manager shall pay the following amounts to the Authority as Liquidated Damages, which are agreed to be a reasonable forecast, as of the date of this Construction Services Agreement, of the damages that the Authority, the Team, and their respective Affiliates will suffer due to a delay in the scheduled Date of Substantial Completion; it is expressly agreed that the following amounts are forecasts estimated in good faith and not penalties for late completion. The Liquidated Damages shall be calculated from the scheduled Date of Substantial Completion in the Construction Schedule, as may be revised by Contract Revisions, up through and including the day when the Date of Substantial Completion is actually achieved:

Activity and Required Date	Liquidated Damage
Substantial Completion: July 1, 2016	\$20,000 per day
Scheduled Events	\$500,000 per Scheduled Event (maximum of one prior to the first scheduled NFL Game).
NFL Game [Pre-Season/Regular Season]	\$5,000,000 per NFL Game.

For purposes of this **Paragraph 4.2.1**, a "Scheduled Event" shall mean any amateur or professional sporting event (other than a NFL Game), exhibition, tournament, musical or theatrical performance and other form of live or broadcasted entertainment, convention meeting, market, or show for which (a) the reasonably expected paid attendance is at least 10,000 persons and for which tickets or concessions are sold to those attending, or (b) expenses incurred with loss of such Scheduled Event would be at least \$100,000. The Authority shall endeavor to have the Team make reasonable efforts to have the first two pre-season NFL Games after August 1, 2016 be away games. Liquidated Damages assessed against the Construction Manager shall not exceed an aggregate maximum of Fifteen Million Dollars (\$15,000,000). Notwithstanding anything to the contrary in this **Paragraph 4.2**, liquidated damages shall not be assessed for any Scheduled Events or NFL Games prior to August 1, 2016.

**4.3** If the Construction Manager is delayed in the progress of the Work by events of Force Majeure, fire or other property damage not caused by the Construction Manager or a Subcontractor, Supplier, Sub-subcontractor of any tier, or other Person for whom Construction Manager is legally responsible, unusually severe weather conditions not customarily encountered in the area surrounding the Project which affect the scheduled progress of the Work, or any other cause for which the Construction Manager is not responsible, then the Contract Time, provided for in **Paragraph 4.1** above, shall be extended by a Contract Revision for a reasonable period of time as

agreed to by the Authority; provided, however, that delays by the Architect in responding to submittals or RFI's or delays in the Architect's schedule for generating Design Documents (unless materially prejudicial) shall not entitle Construction Manager to an extension of Contract Time. No adjustments to the Contract Time are acceptable without the prior written approval of the Authority as a Contract Revision.

- **4.4** Notwithstanding anything to the contrary in this Construction Services Agreement, the Construction Manager agrees that, regardless of the cause of delay and whether or not any extension of Contract Time may be granted therefore, Construction Manager shall continue to prosecute all Work not directly affected by said cause of delay and, with respect to such portion or portions of the Work as may be so affected, shall take all reasonable measures to minimize the effect of said cause of delay. It is a condition precedent to the consideration or prosecution of any Claim for an extension of Contract Time that the foregoing provisions be strictly adhered to in each instance and, if the Construction Manager fails to comply, Construction Manager shall be deemed to have waived the Claim to the extent that proper mitigation would have shortened the period of delay.
- 4.5 The Construction Manager shall recommend to the Authority a schedule for procurement of long lead-time items, which will constitute part of the Work as required to meet the Milestone Dates in the Construction Schedule and the Date of Substantial Completion. If such long lead-time items are procured by the Authority or a Trade Contractor or Vendor, they shall be procured on terms and conditions reasonably acceptable to the Construction Manager, such acceptance not to be unreasonably withheld. Subject to Paragraph 10.3, the Construction Manager agrees that all contracts for such items may be assigned by the Authority or a Trade Contractor or Vendor to the Construction Manager as more fully described in <a href="https://www.example.com"><u>Exhibits 3</u> and 6</a> hereto, in which case the assigned contract(s) will become part of the GMP. The Construction Manager shall expedite the delivery of long lead-time items whether such long lead-time items are procured by the Pre-Construction Phase or Construction Phase.

#### **Guaranteed Maximum Price**

#### 5.1 Guaranteed Maximum Price (GMP)

**5.1.1** As part of its Pre-Construction Phase Services described in <u>Exhibit 1</u> hereto, the Construction Manager will be asked to prepare a Construction Management Plan including a GMP Proposal for the Work and a Construction Schedule based on the GMP Pricing Documents. If the Construction Management Plan is acceptable to the Authority and Team, which decision shall be in their sole discretion, then the Construction Manager's Construction Management Plan shall be added to this Construction Services Agreement as a Contract Revision in the form of a completed <u>Exhibit 3</u> hereto. The Authority may require that the GMP be separated into two or more separate sums and be tracked separately throughout the course of the Work.

5.1.1.1 The GMP shall include an aggregate line item for Construction Manager's General Conditions which shall itself be a Guaranteed Maximum Price for that line item (the "General Conditions GMP" or "GCGMP"). Construction Manager will not apply for payment for any Cost of the Work that it incurs in excess of the GCGMP or be entitled to any increase in the GCGMP except for reasons that would entitle Construction Manager to a Contract Revision under this Construction Services Agreement increasing the GMP; provided, however, that the GCGMP will not increase if the GMP increases to \$740,000,000 as long as the increase in the Cost of the Work causing such increase in the GMP is for Work performed contiguous to or within the Stadium Site including pedestrian connections to the Stadium and any build-out of site offices for the Authority and Team and a potential marketing center for the Team in a nearby building ("Contiguous Work"), but not including Work at the TCF Bank Stadium at the University of Minnesota or for parking garages that are not integral to the Stadium. The Construction Manager will charge for General Conditions for non-Contiguous Work up to the said GMP limit of \$740,000,000 at its incremental additional actual cost of General Conditions pursuant to Article 8 hereof not to exceed a total of eight and one half percent (8.5%) of the Cost of the Work and with no additional Fee for General Conditions. The amount of the GCGMP shall be Thirty Eight Million Four Hundred Twenty Thousand Five Hundred and Twenty Three Dollars (\$38,420,523), covering all items identified in Exhibit 14 (and any qualifications or modifications contained in said Exhibit 14 shall take precedence over any other contrary language in this Subparaghraph 5.1.1).

**5.1.1.2** If and when appropriate and as approved by the Authority and Team consistent with the Act, the Construction Manager may be allowed to perform a portion of the Work with its own forces as provided pursuant to **Exhibit 6** hereof, in which case the Construction Manager will provide either an agreed upon lump sum price (**"Self Performed Lump Sum"** or **"SPLS"**) or a Guaranteed Maximum Price for that line item (the **"Self Performed GMP"** or **"SPGMP"**). Construction Manager will not apply for payment for any Cost of the Work that it incurs in excess of the SPLS or SPGMP or be entitled to any increase in the SPLS or SPGMP except for the same reasons that would otherwise entitle Construction Manager to a Contract Revision under this Construction Services Agreement increasing the GMP.

**5.1.1.3** Construction Manager's GMP shall contain a Construction Manager Contingency as more fully described in **Exhibit 3** hereof to cover the risks inherent in providing a GMP for the Work. Construction Manager may not use the Construction Manager Contingency established for the GMP to fund any Cost of the Work applicable to and in excess of the GCGMP, SPGMP, or SPLS; Construction Manager acknowledges that the GCGMP and the SPGMP or SPLS each have their own contingencies that are already part of the GCGMP, SPGMP, or SPLS and that are separate from the

Construction Manager Contingency. Up to a GMP limit of \$740,000,000, the Construction Manager Contingency shall be the lesser of four and one half percent (4.5%) of the Cost of the Work or Thirty Million Dollars (\$30,000,000).

5.1.1.4 Allowances included in the GMP, if any, are listed in **Exhibit 3** hereto.

**5.1.1.5** The Construction Manager's Fee for Pre-Construction Phase Services including expenses shall be a fixed amount of One Million Dollars (\$1,000,000), which shall be included in the GMP. Half of this amount shall be at risk and tied to the Authority's acceptance of Construction Management Plan as provided in **Exhibit 3** hereto. Fifty percent of the this Pre-Construction Fee amount will be invoiced in equal monthly installments, and the remaining fifty percent shall be payable in the first pay application during the Construction Phase.

**5.1.1.6** As described in **Exhibit 3** hereto, for a GMP up to Seven Hundred Forty Million Dollars (\$740,000,000), the Construction Manager's Fee for Construction Phase Services shall be a fixed amount of Twelve Million Five Hundred Thousand Dollars (\$12,500,000) If the Authority orders changes to the scope of the Work increasing the GMP to more than Seven Hundred Forty Million (\$740,000,000), then the Construction Manager shall be entitled to an adjustment to its Fee of one and ninety-five hundredths percent (1.95 %) of the amount of the increase in the Cost of the Work that results in the GMP exceeding \$740,000,000. But, if the total aggregate value of the Changes in the Work are less than such amount, then the Construction Manager shall not be entitled to an increase of the Construction Manager's Fee.

5.1.2 The GMP will be calculated and Contract Time established on the basis of the GMP Pricing Documents given to the Construction Manager at the time the GMP Proposal is requested. The Construction Manager acknowledges that, as of the date it will be asked to execute **Exhibit 3** hereto, all the Design Documents necessary for the Work will not be finalized, but Construction Manager agrees that the GMP Pricing Documents will be sufficiently complete to enable Construction Manager to guarantee and agree to perform its Work for the GMP based on the final Construction Documents that will be developed by the Architect. Construction Manager knows and understands the Program, understands that the GMP Pricing Documents are intended to satisfy the Program, and agrees that by executing **Exhibit 3**, it will timely complete the Work contemplated by the Program as revised for the GMP, based on the GMP Pricing Documents, and as required by the final Construction Documents. In determining the GMP and Contract Time, the Construction Manager will take into account the level of completeness of the GMP Pricing Documents and will exercise the best skill and efforts of the Construction Manager to make (1) all appropriate judgments and inferences about how the Documents will be completed as Construction Documents based on the Program and Applicable Law, and (2) all inquiries of and discussions with the Authority, Team, and Architect as necessary in the Construction Manager's judgment to clarify the content and intent of the GMP Pricing Documents and the expected and intended scope of the Construction Documents as described in this Subparagraph 5.1.2 in order to reliably calculate and establish both the GMP and the Contract Time.

**5.1.3** By executing **Exhibit 3** to this Construction Services Agreement, the Construction Manager agrees that the Contract Documents (including the GMP Pricing Documents), materials, and information furnished to the Construction Manager as of the date of the execution of said **Exhibit 3** and the ongoing discussions and meetings between the Construction Manager and the Authority, Team, and the Architect up to said execution will have described the scope, construction requirements, and design intent of the Work in detail sufficient to enable the Construction Manager shall not be permitted to claim

any adjustment in either the GMP or Contract Time in connection with the completion of final Construction Documents, unless allowed pursuant to **Article 14.** 

- **5.1.4** The GMP as described in **Subparagraph 5.1.1** hereof is guaranteed by the Construction Manager not to exceed the amount to be described in <u>Exhibit 3</u> hereto, subject to additions and deductions in accordance with and allowed by the Contract Documents. The GMP may also be referred to in the Contract Documents as the Contract Sum. Costs which would cause the GMP to be exceeded, other than for additions and deletions in accordance with the Contract Documents shall be paid by the Construction Manager without reimbursement by the Authority or Team. By virtue of providing the GMP, the Construction Manager shall, as required by the Act, accept the risk of cost overruns in excess of the GMP , as modified in accordance with and allowed by the Contract Documents, and such cost overruns shall not be the responsibility of the Authority, State, or Team.
- **5.1.5** The Construction Manager acknowledges and agrees that the intent of this Construction Services Agreement is for the Construction Manager to provide the timely completion of all Work required by the Contract Documents within the Contract Time required by the Construction Schedule for an amount not to exceed the GMP.
- **5.1.6** The GMP includes a Construction Manager Contingency to cover any and all foreseen and unforeseen costs as more specifically described in <u>Exhibit 3</u> hereof. The Construction Manager hereby acknowledges and agrees that the intent of the GMP is to include all costs associated with the complete construction of the Work.

# 5.2 Construction Manager Corporate Guaranty

**5.2.1** Subject to receipt of payment from the Authority of all payments that the Authority is obligated and required to make under this Construction Services Agreement, the Construction Manager absolutely and unconditionally guarantees to the Authority and the Indemnitees that all of the Work under this Construction Services Agreement will be performed and completed in accordance with the Contract Documents and will not be defectively constructed. The Construction Manager shall pay to the Authority all costs incurred, including reasonable attorney fees, in enforcing the performance and payment of this guaranty. If the Construction Manager is an Affiliate of another parent entity or Person, or if the Construction Manager is a joint venture partner in a joint venture, the Affiliate parent entity or joint venture partner must sign the Guaranty that is attached hereto as **Exhibit 8**.

#### DISCOUNTS, REBATES, AND REFUNDS

6.1 Cash discounts obtained on payments made to Subcontractors or Suppliers shall accrue to the Authority during or after the term of this Construction Services Agreement. Trade discounts, rebates, refunds, and amounts received from sales of surplus material and equipment shall accrue to the Authority. The Construction Manager shall notify the Authority of provisions that will be required of the Authority to secure trade discounts, rebates, and refunds as described in this **Paragraph 6.1**. If it elects to do so, the Authority shall advance such funds in a timely manner as are required to take advantage of discounts. The Construction Manager shall not obtain for its own benefit any discounts, rebates, or refunds in connection with the Work prior to providing the Authority with seven (7) days' prior written notice of the potential discount, rebate, or refund and an opportunity to furnish funds or take other actions necessary to obtain such discount, rebate, or refund on behalf of the Authority in accordance with the requirements of this **Paragraph 6.1**.

#### **ACCOUNTING RECORDS**

- 7.1 The Construction Manager shall keep full and detailed accounts and exercise such controls as may be necessary for proper financial management under this Construction Services Agreement and as directed by the Authority and Team. The accounting and cost control systems shall be reviewed and approved by the Authority and Team, which approval shall not be unreasonably withheld. Upon receiving seven (7) days notice, the Construction Manager shall afford the Authority or Team and the Authority's or Team's accountants reasonable access to appropriate personnel for interviews and the necessary and relevant Construction Manager's records, books, cost reports, labor rates, labor schedules, correspondence, instructions, drawings, receipts, Subcontracts, purchase orders, vouchers, memoranda, and other electronic and hard copy data to the extent relating to the Construction Manager's right to payment under and the Construction Manager's compliance with the terms and conditions of this Construction Services Agreement, and the Construction Manager shall preserve this information for a period of twelve (12) years after final payment, or for such longer period as may be required by law. The Construction Manager shall account for and keep track of the Cost of the Work, including all materials, equipment and labor utilized in the performance of the Work and shall keep such full and detailed accounts as may be necessary for proper financial management under this Construction Services Agreement. All records shall be maintained in accordance with Generally Accepted Accounting Procedures, consistently applied. Subcontractors retained by the Construction Manager on a "cost-plus" basis shall have the same obligations to retain records and cooperate with audits as are required of the Construction Manager under this Article 7. In addition to all other rights under the Contract Documents, the Authority and Team representatives shall have the right to conduct full and complete audits of the Construction Manager with respect to the Work and also to the extent relating to the Construction Manager's right to payment under and the Construction Manager's compliance with the terms and conditions of this Construction Services Agreement. If any inspection by the Authority or Team of the Construction Manager's records, books, correspondence, instructions, drawings, receipts, vouchers, memoranda, and any other data relating to the Contract Documents reveals an overcharge, including, without limitation, any untimely request for payment, the Construction Manager shall pay the Authority upon demand an amount equal to such overcharge, for reimbursement for said overcharge, and all administrative and legal expenses incurred in determining or recovering the overcharge. The Authority's rights under this provision shall survive termination of this Construction Services Agreement or Final Completion.
- 7.2 The Construction Manager shall provide the Authority and Team copies of all documents and records associated with the Work that the Authority or Team deem necessary. The Construction Manager shall provide all Subcontracts, material purchase orders, and insurance certificates to the requesting party in an executed format for all Subcontractors and Suppliers prior to any such Person commencing work on the Project. The Construction Manager will strictly enforce this requirement.
- **7.3** The Construction Manager is obligated to utilize a Subcontract template for all its Subcontracts that is submitted to the Authority for its approval prior to such use, which approval shall not be unreasonably withheld.
- 7.4 All changes to the Work must be approved in strict accordance with the requirements of the Contract Documents, including the procedures and use of the Contract Request Form and Contract Revision forms that will be added as <u>Exhibit 11</u> to this Construction Services Agreement as the Project Manual is developed in the future. No change in the Work, in any form, will be approved for payment without the Construction Manager providing the Authority with all reasonably required documentation, including a properly completed Contract Request Form, and in accordance with the procedures that will be set forth in <u>Exhibit 11</u>. The Contract Request Form,

and all required contractual documentation associated therewith and required by the Authority, must be provided to and approved by the Authority in writing before commencement of any proposed changes in any part of the Work. The Authority will administer a Pending Change Report to identify and track any and all potential changes in the Work, and the Construction Manager is obligated to provide all supporting documentation reasonably required by the Authority to administer the Pending Change Report. The Authority will use the Pending Change Report to provide the Architect and Construction Manager with written authorization to proceed with commencing changes to the Work. The Construction Manager is responsible to immediately provide the Authority with timely notification of any pending change associated with the Work in accordance with the requirements of this Paragraph 7.4 and the Contract Documents and to provide projected cost impacts within five (5) days of notification of such change. The Authority and Construction Manager must agree on the allocation of the cost of the pending change (e.g. Construction Manager Contingency) prior to including any such pending change on the Pending Change Report. The Construction Manager is responsible to provide final costs associated with any pending change within two (2) weeks of inclusion on the Pending Change Report, or such other timeframe as reasonably agreed to in writing by the Authority. If the Construction Manager performs any alleged change in the Work in the absence of proper notice required by the Contract Documents, the Construction Manager then assumes all liability for such Work, understands and agrees that the Construction Manager will not be paid for such Work, and agrees that the Authority is not obligated to compensate the Construction Manager for such alleged change as a Cost of the Work or as an addition to the GMP. The Authority reserves the right to reasonably amend this procedure in the event an electronic method is developed for these functions or a webbased project management system is utilized by the Authority.

#### COSTS TO BE REIMBURSED

8.1 The term "Cost of the Work" shall mean costs reasonably and necessarily incurred by the Construction Manager for the proper performance of the Work. Such costs shall be at rates not higher than the standard paid in the Minneapolis/St. Paul metropolitan area for similar work, services, labor, equipment, or materials except with prior written consent of the Authority. The Cost of the Work shall include only the items set forth in this Article 8. Except for rates and costs that are stated on attached Exhibits, the Cost of the Work as defined herein, including stipulated labor burdens, equipment rates, and insurance costs, shall and are represented by the Construction Manager to be actual costs paid or incurred by the Construction Manager, less all discounts, rebates, and salvage or scrap values that shall be taken by the Construction Manager, subject to Article 6 of the Construction Services Agreement. All payments made by the Authority pursuant to this Article 8, whether those payments are actually made before or after the execution of Exhibit 3 to the Construction Services Agreement, are included within the GMP specified in Article 5 above. In no event shall the Authority be required to reimburse the Construction Manager for any portion of the Cost of the Work incurred prior to execution of this Construction Services Agreement.

#### 8.1.1 Construction Manager's Fee

- .1 For all Pre-Construction Phase Services set forth in <u>Exhibit 1</u> hereto, the Construction Manager will charge only the fixed Fee for Pre-Construction Phase Services stated in **Subparagraph 5.1.1.5**. The Fee for Pre-Construction Phase Services is guaranteed by the Construction Manager and shall not increase unless pursuant to a Contract Revision agreed to by the Authority. If Construction Manager believes it has a Claim for additional payment for Pre-Construction Phase Services, the Construction Manager must provide the Authority and Team with notice of such Claim pursuant to **Article 14** and **Paragraph 6.2** of <u>Exhibit 6</u> before such services are performed or such Claim shall be deemed waived.
- .2 For all Construction Phase Services set forth in <u>Exhibit 1</u> hereto and required for the proper performance of the Work, the Construction Manager shall be compensated in the amount of the Construction Manager's Fee stated in **Subparagraph 5.1.1.6** hereto and as described in <u>Exhibit 3</u> of this Construction Services Agreement.

#### 8.1.2 Labor Costs

- .1 Actual wages of construction workers directly employed by the Construction Manager to perform the Work at the Project Site or, with the Authority's written agreement, at off-site workshops. Construction Manager shall be responsible for paying prevailing wages as will be provided in the Contract Documents.
- .2 Cost of Construction Manager's supervisory and administrative personnel, whether working on or off the Project site, at the rates set forth in **Exhibit 5**. The number and type of supervisory personnel that the Construction Manager assigns to this Project shall be subject to the approval of the Authority, such approval not to be unreasonably withheld. If administration and supervisory personnel are not stationed on the site, then only the specific people, not the category or type of work descriptions identified on **Exhibit 5** shall be considered as a Cost of the Work; **provided, however**, that additional administrative and supervisory personnel not stationed on the site may be added to **Exhibit 5** at the request of the Construction Manager and with the approval of the Authority,

which approval shall not unreasonably withheld. The rates requested are subject to audit and must be substantiated by the Construction Manager to the satisfaction of the Authority, which audit and approval must completed prior to Authority's acceptance of the Construction Management Plan.

- .3 Actual wages and salaries of the Construction Manager's supervisory and administrative personnel engaged at factories, workshops, or on the road in expediting the production or transportation of materials or equipment required for the Work, but only for that portion of their time required for such Work.
- .4 Costs actually paid or incurred by the Construction Manager for non-supervisory or non-administrative personnel for taxes, insurance, contributions, assessments and benefits required by law, fringe benefit funds, or collective bargaining agreements, and, for personnel not covered by such agreements, customary benefits such as sick leave, medical and health benefits, holidays, vacations and pensions, provided such costs are based on wages and salaries included in the Cost of the Work under **Subparagraphs 8.1.2.1** through **8.1.2.3**. The cost of such taxes, insurance, contributions, and assessments shall be charged to the Project at their actual audited rate required by the appropriate collective bargaining agreement.

# 8.1.3 Subcontract Costs

Payments made to Subcontractors in accordance with the requirements of the Contract Documents and the subcontracts properly approved by the Authority between the Construction Manager and Subcontractors, for the Work and in strict accordance with the Contract Documents.

# 8.1.4 Costs of Materials and Equipment Incorporated in the Completed Construction

- .1 Costs, including transportation, of materials and equipment incorporated or to be incorporated in the completed construction.
- .2 Costs of materials described in the preceding **Subparagraph 8.1.4.1** in excess of those actually installed but required to provide reasonable allowance for waste and for spoilage. Unused excess materials or equipment, if any, shall be handed over to and become property of the Authority at the completion of the Work or, at the Authority's option, shall be returned, sold, recycled, or scrapped by the Construction Manager; amounts realized, if any, from such returns, sales, recycling, or scrap shall be credited to the Authority as a deduction from the Cost of the Work. Materials that are determined to be waste by the Authority shall be properly disposed of or recycled by the Construction Manager. The Construction Manager shall maintain a complete record of all materials and equipment purchased by the Construction Manager for the Work.

## 8.1.5 Cost of Other Materials and Equipment, Temporary Facilities and Related Items

.1 Costs, including transportation, installation, maintenance, dismantling, and removal of materials, supplies, temporary facilities, office equipment for the field office, machinery, equipment, and hand tools not customarily owned by the construction workers, which are provided by the Construction Manager at the Project Site and fully consumed in the performance of the Work; and cost or value at the time of use at the Project Site less salvage value on such items when no longer used at the Project Site if not fully consumed, whether sold to others or retained by the Construction Manager. For clarity, some such costs may be covered instead by rates established under **Paragraph 8.1.5.2**. Cost for items

previously used by the Construction Manager shall be based on their fair market value at the time they are used at the Project Site.

- .2 Rental charges for temporary facilities, machinery, equipment, and hand tools not customarily owned by the construction workers, which are provided by the Construction Manager at the Project Site, whether rented from the Construction Manager or others, and costs of transportation, installation, minor repairs, replacements, dismantling and removal thereof. Costs of machinery and equipment shall require approval of the Authority. The Construction Manager shall provide a written equipment schedule with proposed costs for the equipment for review and approval by the Authority. Authority shall determine which items are purchased and/or leased and shall receive all purchased items at the conclusion of the Project Rental rates for equipment owned by the Construction Manager or an Affiliated Entity shall not exceed seventy-five percent (75%) of the rates in the current edition of "Compilation of Rental Rates for Construction Equipment," prepared by Associated Equipment Distributors, Oak Brook, Illinois or those generally prevailing in the location of the Project, whichever is less, and total no more in the aggregate than eighty percent (80%) of the fair market value of the equipment in question as of the date that such machinery or equipment is first put into service in connection with the Work. The Construction Manager shall obtain bids for all machinery and equipment to be rented from no less than two (2) responsible suppliers other than the Construction Manager itself or an Affiliated Entity. The Authority shall, with the advice of the Construction Manager, determine which bid is to be accepted. In no event shall the Construction Manager be entitled to reimbursement for any cumulative total of rental charges in connection with any single piece of machinery or equipment, whether owned or rented by Construction Manager in excess of eighty percent (80%) of its fair market value as of the date that such machinery or equipment is first put into service in connection with the Work. The Construction Manager shall pay any excess rental charges which will not be considered a Cost of the Work. In the alternative, the Construction Manager may propose an equipment rate schedule at the time it submits its Construction Plan that may be approved by the Authority in its sole discretion after negotiation and audit.
- .3 Costs of removal of debris from the Project Site and its proper legal disposal, salvage or recycling, and costs of on-site salvage or recycling of material. The value of recycled and salvaged material will be credited to the Authority.
- .4 Costs of the Construction Manager's Project Site office and costs of the Architect's and Authority's and Team's Representatives' Project Site offices provided by the Construction Manager, including long-distance telephone calls, postage and parcel delivery charges, office supplies, parking, telephone service, internet service, website charges, photography, cell phones, radios, and ordinary vehicle use attributable to the Project as approved by the Authority.
- .5 Except for personnel who reside within one-hundred (100) miles of the Project Site, that portion of the reasonable out-of-town travel and subsistence expenses of the Construction Manager's personnel incurred while traveling in discharge of duties connected with the Construction Manager's Work when approved in writing by the Authority. Costs of out-of-town travel and subsistence expenses will be part of Construction Manager's General Conditions GMP.
- .6 Costs of materials and equipment suitably stored off the site at a mutually acceptable location, subject to the Authority's prior approval and pursuant to Subparagraph 11.3.2 of Exhibit 6 hereto

# 8.1.6 Miscellaneous Costs

- .1 That portion directly attributable to this Construction Services Agreement of premiums for bonds from the Construction Manager; provided, however, that such costs shall not be included in the Cost of the Work for purposes of calculating the Construction Manager's Fee. The cost of any bonds from first tier Subcontractors shall be a Cost of the Work if the Subcontractor's Subcontract has a value in excess of \$250,000 but also including any first tier Subcontractors involving any operable system, structural integrity, water proofing integrity, or any other agreeable Subcontractor. Any rebates, refunds or credits received by Construction Manager in connection with any Subguard or similar program shall be subject to audit by the Authority and credited to the Insurance will be reimbursed at their actual cost to Construction Authority. Manager assuming there will be no Owner Controlled Insurance Program. Upon substantiation, the cost of insurance may be reduced to a rate sheet and made an Exhibit to this Construction Services Agreement by mutual agreement. This rate shall be subject to annual adjustment.
- .2 Fees and assessments for the permits, licenses, and inspections for which the Construction Manager is required by the Contract Documents to pay.
- .3 Fees of testing laboratories for tests required by the Contract Documents, except those related to defective or nonconforming Work for which the General Conditions or other provisions of the Contract Documents exclude reimbursement or deemed prudent by the Construction Manager under its quality program.
- .4 Deposits lost for causes other than the Construction Manager's fault or negligence.
- .5 Compensation for third party consultants engaged by Construction Manager to perform design and quality peer review of matters relating to structural integrity, building enclosure for water tightness and moisture protection, roof enclosures, and operable systems up to \$300,000 in the aggregate (but excluding the Structure Consultants defined in **Paragraph 17.26** hereof) with the prior written approval of the Authority, which shall not be unreasonably withheld.
- .6 The deductible portion of any losses under any insurance policies to the extent allowed in Article 13 of <u>Exhibit 6</u> hereto.
- .7 Legal, mediation and arbitration costs, including attorneys' fees, other than those arising from disputes between the Authority and the Construction Manager, reasonably incurred by the Construction Manager after the execution of this Construction Services Agreement in the performance of the Work, and with the approval of the Authority, which approval and amount of expenditure shall be in the sole discretion of the Authority. Said costs shall be subject to audit by the Authority and not include those relating to or arising out of any Claim that Construction Manager may bring against the Authority and Team for adjustment to the GMP or Contract Time or to defend claims by third parties not under the control or responsibility of the Construction Manager, including Governmental Authorities.

## 8.1.7 Other Costs

.1 Other costs incurred in the performance of the Work if and to the extent approved in advance in writing by the Authority.

# 8.1.8 Emergencies: Repairs to Damaged, Defective or Nonconforming Work

- .1 Costs reasonably incurred in taking action to prevent threatened damage, injury or loss in case of an emergency affecting the safety of persons and property, as provided in the Contract Documents.
- .2 Costs reasonably incurred in repairing or correcting damage to the Construction Manager's Work provided such damage did not result from the fault or negligence of the Construction Manager's supervisory personnel as described in Subparagraph 9.1.7, and only to the extent that the cost of such repairs is not recoverable by the Construction Manager from others and the Construction Manager is not compensated therefore by insurance or otherwise.

# 8.1.9 Value Added Taxes, Including Goods & Services Tax, Sales Taxes

.1 Where applicable, value added taxes and similar taxes imposed by any Governmental Authority that are related to the Work and for which the Construction Manager is liable are included in the GMP. These costs are to be tracked and identified by the Construction Manager for potential refunds to the Authority by the appropriate Governmental Authorities.

# 8.1.10 Affiliated Entity Transaction

.1 If any of the costs to be reimbursed arise from a transaction between the Construction Manager and an Affiliated Entity, the Construction Manager shall notify the Authority of the specific nature of the contemplated transaction, including the identity of the Affiliated Entity and the anticipated cost to be incurred, before any such transaction is consummated or cost incurred. If the Authority, after such notification, authorizes the proposed transaction, then the cost incurred shall be included as a Cost of the Work to be reimbursed, and the Construction Manager shall procure the Work, equipment, goods or service from the Affiliated Entity, as a Subcontractor or Supplier. If the Authority fails to authorize the transaction, the Construction Manager shall procure the Work, equipment, goods or service from some Person other than an Affiliated Entity.

#### COSTS NOT TO BE REIMBURSED

- **9.1** The Construction Manager shall be responsible for but shall not be paid, or under any circumstance entitled to receive, any of the following as a Cost of the Work or otherwise:
  - **9.1.1** Salaries and other compensation of the Construction Manager's personnel stationed at the Construction Manager's principal office or offices other than the Project Site office, except as specifically provided in **Subparagraphs 8.1.2.1** and **8.1.2.2** or as may be provided in **Article 14.**
  - **9.1.2** Expenses of travel and subsistence of the Construction Manager's personnel other than for specific Project-related travel as described in **Subparagraph 8.1.5.5**.
  - **9.1.3** Expenses of the Construction Manager's principal office and offices other than the Project office at the Project Site.
  - 9.1.4 Overhead and general expenses, except as may be expressly included in Article 8.
  - **9.1.5** The Construction Manager's capital expenses, including interest on the Construction Manager's capital employed for the Work.
  - **9.1.6** Rental or other costs of machinery and equipment, except as specifically provided in **Subparagraph 8.1.5** and only when approved in advance in writing by the Authority.
  - **9.1.7** Except as provided in **Subparagraph 8.1.8.2** hereof, costs due to the fault or negligence of the Construction Manager's personnel and to the extent not otherwise able to be recovered from others or insurance, including costs for the correction of damaged, defective, or nonconforming Work, delays, disposal and replacement of materials and equipment incorrectly ordered or supplied, or damaging property not forming part of the Work.
  - 9.1.8 Costs and fees of legal or other professional services of the Construction Manager, except as permitted under Paragraph 8.1.6.7. Attorneys' fees and costs associated with Construction Manager's compliance with its indemnity and defense obligations under Paragraph 4.18 of Exhibit 6 are not reimbursable.
  - **9.1.9** Costs of overtime, lost productivity or loss of profit associated with the Work and the Construction Manager's responsibility to complete the Work in an amount that exceeds the GMP, unless approved in advance in writing by the Authority on the approved Contract Request Form, in the Authority's discretion, as a Cost of the Work to be reimbursed under **Article 8** hereof. Failure to provide notice of such a Claim pursuant to the Contract Documents and receive advance written approval from the Authority in the form of a Contract Revision shall result in a waiver of any Claims for reimbursement of such costs.
  - **9.1.10** Costs associated with the rescheduling or re-sequencing of any Phase of the Work including the Work of Subcontractors, Suppliers, and Sub-subcontractors of all tiers, unless approved as a proper use of the Construction Manager Contingency by the Authority in writing before such costs are incurred by the Construction Manager, such approval not to be unreasonably withheld, in which case said costs will be considered to be a Cost of the Work to be reimbursed under **Article 8** hereof.

- **9.1.11** Any cost of the Construction Manager, Subcontractor, Supplier, or Sub-subcontractors of all tiers incurred in the performance of the Work that was not properly bid, awarded, and administered in strict accordance with the Contract Documents.
- **9.1.12** Costs involving the Construction Manager, Subcontractors, Supplier, or Subsubcontractors of all tiers associated with changes involving the means, methods, materials, techniques, equipment and related costs, unless otherwise approved as a proper use of the Construction Manager Contingency by the Authority in writing before such costs are incurred by the Construction Manager, such approval not to be unreasonably withheld, in which case said costs will be considered to be a Cost of the Work to be reimbursed under **Article 8** hereof.
- **9.1.13** Costs resulting from coordination problems or delays that are the responsibility of the Construction Manager in the development and processing of the Contract Documents including Submittals unless approved as a proper use of the Construction Manager Contingency by the Authority in writing before such costs are incurred by the Construction Manager, such approval not to be unreasonably withheld, in which case said costs will be considered to be a Cost of the Work to be reimbursed under **Article 8** hereof.
- **9.1.14** Costs of governmental fines, damages, penalties, or sanctions that may be imposed on the Construction Manager, Subcontractors, Suppliers or Sub-subcontractors of all tiers due to a violation of Applicable Law or their performance of the Work. This includes any fines, damages, penalties, or sanctions associated with the failure of the Construction Manager to comply with the Project's applicable Targeted Business and workforce participation plan or Applicable Laws.
- 9.1.15 Costs, if any, which would cause the GMP to be exceeded.
- **9.1.16** Costs associated with any penalties, legal fees, claims, or losses associated with the Project Safety Program.
- **9.1.17** Costs associated with administration of the insurance policies and any losses, deductibles, penalties or claims associated with the insurance for the Project. Insurance deductibles shall be paid in accordance with **Article 13** of **Exhibit 6** hereof.
- 9.1.18 Any cost not specifically and expressly described in Article 8 hereof.
- **9.1.19** Other than as set forth in **Subparagraph 5.1.1** herein and <u>Exhibit 3</u> hereto, any markup fees for General Conditions.

#### SUBCONTRACTS AND OTHER AGREEMENTS

- 10.1 The Work shall be bid or solicited as required by the Authority and awarded to Subcontractors in strict accordance with the Contract Documents and performed pursuant to the terms and conditions of the approved form of Subcontract to be executed between the Construction Manager and all Subcontractors. For Work not vet awarded, the Construction Manager shall obtain bids or solicit competitive proposals as determined by the Authority from Subcontractors and from Suppliers and shall deliver all such bids to the Authority for review and approval by the Authority. Any bid or proposal shall include a detailed accounting for any sales/use taxes that Construction Manager, Subcontractor, or Supplier expects to incur or include in Costs of the Work. Such accounting shall include type of tax (sales, use, excise, etc.), tax amount, tax rate, tax jurisdiction and shall be provided prior to commencing work on the Project. The Construction Manager will recommend to the Authority which Subcontractors and Suppliers it proposes to use, which selection shall be subject to the approval of the Authority, such approval not to be unreasonably withheld. The Authority may designate specific persons or entities from which the Construction Manager shall obtain bids or proposals; however, if the GMP has been approved by the Authority, the Authority may not prohibit the Construction Manager from obtaining bids or proposals from other firms that meet the qualification requirements established in the Contract Documents to perform that portion of the Work. The Construction Manager shall not be required to contract with anyone to whom the Construction Manager has reasonable objection.
- 10.2 Each Subcontract shall: (1) preserve and protect the rights of the Authority under this Construction Services Agreement with respect to the Work to be performed under the Subcontract so that the subcontracting thereof will not prejudice such rights; (2) require the Subcontractor to represent and warrant that it is knowledgeable, to the extent required by the Subcontractor's portion of the Work to be performed, of the Applicable Laws, NFL Rules and Regulations, and applicable codes, standards, rules and regulations applicable to a Project of this type in Minnesota and agree to comply with each of the foregoing; (3) require the Subcontractor to represent and warrant that it is experienced and fully qualified to perform the portion of the Work contemplated to be performed by the Subcontractor; (4) require the Subcontractor to represent and warrant that it is properly licensed, certified, registered and organized to perform such Work under Applicable Laws or any similar requirements in the State of Minnesota; (5) require the Subcontractor to assume toward the Construction Manager and Authority all of the obligations and responsibilities that Construction Manager by the terms of this Construction Services Agreement assumes toward the Authority; (6) require the Subcontractor to acknowledge and agree that the services performed by the Subcontractor are for the benefit of the Authority and Team and the Authority and Team shall have the right to enforce the obligations, responsibilities, and duties of the Subcontractor directly against the Subcontractor; (7) require the Subcontractor to name the Authority and Team as an intended third-party beneficiary to the duties, requirements and obligations of the Subcontractor; (8) require the Subcontractor to acknowledge that the Subcontractor shall have no direct claim, right or cause of action against the Authority or Team by virtue of its third-party beneficiary status; (9) include a provision allowing the Construction Manager to assign the Subcontract to the Authority, a Lender, the Team, or any of the foregoing parties' designees without the Subcontractor's consent or change in the contract price or other terms of compensation; (10) require that such Work be performed in strict accordance with the requirements of the Contract Documents; (11) require a waiver of all Claims for additional costs or damages for delays with respect to subcontracted portions of the Work unless notice for such Claims is provided pursuant to the Contract Documents and agreed in writing by the Construction Manager and Authority before the work that is the subject of the Claim commences; (12) waive all rights the contracting parties may have against one another for damages caused by fire or other perils to the extent covered by the property insurance specified in the Contract Documents; (13) obligate each Subcontractor specifically to consent to all of the terms and conditions of this Construction Services Agreement; (14) require that each Subcontractor agree to indemnify and hold harmless the Authority and all Indemnitees for any and all claims, damages, losses and expenses, including,

but not limited to attorney fees, arising out of or resulting from Subcontractor's negligence or breach of Subcontract; and (15) require each Subcontractor to purchase insurance naming the Authority and all Indemnitees as additional insureds and such insurance be provided to Authority prior to any Subcontractor commencing any of the Work. All insurance provided by any Subcontractor pursuant to this **Paragraph 10.2** shall be primary over and above any insurance provided by the Authority. The Construction Manager shall require each Subcontractor to enter into similar agreements with all Sub-subcontractor, prior to the execution of the Subcontract, copies of the Contract Documents to which the Subcontractor will be bound.

- 10.3 The Authority may elect to award portions of the Project to Trade Contractors and Vendors pursuant to the terms and conditions of a Trade Contract or Vendor Contract to be executed between the Authority and the Trade Contractor or Vendor. The Construction Manager will be responsible for coordinating, scheduling, and integrating the Trade Contractor Work or Vendor Work with the Work of the Construction Manager in accordance with the terms and conditions of the Contract Documents. The Authority reserves the right to assign the rights and obligations of any Trade Contract or Vendor Contract to the Construction Manager, which assignment the Construction Manager agrees to accept to the extent the Construction Manager has previously approved the applicable Trade Contract or Vendor Contract, otherwise, the Authority must obtain the Construction Manager's written consent to any such assignment, which consent shall not unreasonably be withheld; provided, however, that the Construction Manager's approval of any particular Trade Contract or Vendor Contract is not required before the Authority can enter into said contracts. Upon the assignment of a Trade Contract or Vendor Contract to the Construction Manager, the Trade Contractor Work included in the Trade Contract or Vendor Work in included in the Vendor Contract shall become a part of the Construction Manager's Work, and the Trade Contractor or Vendor shall become a Subcontractor of the Construction Manager. Any Trade Contract or Vendor Contract that is assigned to the Construction Manager shall increase the GMP by the remaining amount to be paid under said contract, plus an equitable increase to the GCGMP and an equitable increase to the Construction Manager's Fee, subject to the terms of Exhibit 3 hereof.
- 10.4 If the GMP has been established and a specific bidder among those whose bids are delivered by the Construction Manager to the Authority (1) is recommended to the Authority by the Construction Manager; (2) is qualified to perform that portion of the Work; and (3) has submitted a bid which conforms to the requirements of the Contract Documents without material reservations or exceptions, but the Authority unreasonably refuses to approve the Subcontractor recommended by the Construction Manager, then a Contract Revision shall be issued to adjust the GMP by the difference between the bid of the person or entity recommended to the Authority by the Construction Manager and the amount of the Subcontract or other agreement actually signed with another Person selected by the Construction Manager and approved by the Authority.
- 10.5 The Construction Manager shall be responsible to provide the Authority with a final form of Subcontract (in strict accordance with the terms and conditions set forth in the Contract Documents) and a final form of Purchase Order for materials proposed by the Construction Manager for the Project that shall be reviewed and approved by the Authority, which approval shall not be unreasonably withheld. Once approved by the Authority, the Construction Manager shall use the approved form of Subcontract and Purchase Order for all Work. The Construction Manager is required under the Contract Documents to provide the Authority with a complete Contract Request Form and Bid Analysis, in a format provided by or approved by the Authority, as a basis for recommending the award of any Subcontract, Purchase Order for materials, or Contract Revision. The Construction Manager will be required to provide the Authority with copies of all proposed Subcontracts and Purchase Orders prior to execution with any Subcontractor or Supplier for review and approval by the Authority. The Authority shall have a minimum of five (5) working days for the review and approval of all Subcontracts or Purchase Orders to ensure that they are executed on agreed upon forms and contain appropriate project specific terms. The Construction Manager must provide the Authority with copies of all executed

Subcontracts, Purchase Orders for materials, and Contract Revisions for the Project file. All Subcontracts or other agreements shall conform to the provisions of this Construction Services Agreement and the Contract Documents. Subcontracts shall not be awarded on the basis of cost plus a fee without the prior written consent of the Authority. The Construction Manager will not allow any Subcontractor to proceed with the Work without the Subcontract signed by the Construction Manager and Subcontractor. The Construction Manager is solely responsible for any and all costs resulting from failure to strictly adhere to the contracting requirements set forth in the Contract Documents.

- **10.6** The Construction Manager warrants and represents that the procurement and award of all Subcontracts will conform to the direction of the Authority as allowed by the Act and the Construction Services Agreement. If best value procurement is used to select Subcontractors, the Construction Manager represents and warrants that all of its personnel involved in the procurement of Subcontractors are knowledgeable about the administration of best value procedures. In addition, the criteria and weighting of criteria to be used for any best value procurement shall be subject to the approval of the Authority.
- 10.7 Except as otherwise allowed by the Act, all of the Work, including any Work that is self-performed by the Construction Manager, must be bid and awarded under the terms and conditions of Paragraph 4.20 of Exhibit 6 hereof, and that Work must be contracted for under the form of Subcontract referenced in Paragraph 10.5 of the Construction Services Agreement.
- **10.8** The Construction Manager shall comply with all Applicable Laws and any special requirements in the Contract Documents regarding equal employment opportunity, Targeted Business, and workforce participation initiatives.
  - The Construction Manager shall demonstrate good faith efforts to utilize minority 10.8.1 ("MBE") and women-owned ("WBE") business enterprises (hereinafter referred to as "Targeted Businesses"). The Authority has retained the services of a Targeted Business Coordinator to assist the Construction Manager in meeting its Targeted Business and workforce participation goals. The Construction Manager shall submit to the Authority a Targeted Business Enterprise Participation Plan within ten (10) calendar days after executing this Construction Services Agreement. Subcontractors shall submit a Targeted Business Enterprise Participation Plan with any proposal or bid related to the Project. The Authority requires that the Construction Manager utilize good faith efforts to achieve the goals for MBE and WBE participation set forth in the Authority's Equity Plan for the Construction Services Agreement. The Authority also requires that the Construction Manager utilize good faith efforts to achieve the workforce participation goals for minority and women workers on the Project. The Construction Manager shall utilize the Targeted Business Coordinator to the fullest extent possible to accomplish the following activities:
    - .1 Identify the trades, services and suppliers needed for the Project.
    - .2 Identify Targeted Businesses that have the resources and capabilities to participate in the Project.
    - .3 Contact Targeted Businesses to solicit bids for work on the Project.
    - .4 Certify currently uncertified but qualified companies as Targeted Businesses for participation in the Project.
    - .5 Develop the Targeted Business Enterprise Participation Plan for submittal with any bid or proposal from a Subcontractor.
    - .6 Track participation of Targeted Businesses.
    - .7 Prepare Targeted Business participation reports required by the Authority.
    - .8 Comply with workforce utilization requirements of the Authority's Equity Plan for the Construction Services Agreement, including requirements established pursuant to Section 17, subd.1 of the Act.

Construction Manager's failure to use good faith efforts to achieve the Authority's Targeted Business participation goals or to follow the requirements of the Authority's Equity Plan for the Construction Services Agreement may result in delay or withholding of Construction Manager's payments and the assessment of appropriate damages. The Authority's Equity Plan for the Construction Services Agreement is attached to this Construction Services Agreement as **Exhibit 7.** 

# **10.9** Subcontractors and Suppliers

**10.9.1** The Construction Manager, as soon as practicable after execution of this Construction Services Agreement, and from time to time thereafter as required by the staging of the Work, shall furnish to the Authority, in writing, the names of the Persons the Construction Manager proposes to engage as Subcontractors and Suppliers for the Work.

## PAYMENTS

- 11.1 Based upon Applications for Payment submitted by the Construction Manager and Certificates for Payment issued by the Architect, the Authority shall make payment to the Construction Manager as set forth in the General Conditions included as <u>Exhibit 6</u> hereto. Construction Manager will hold payments it receives in trust to be used for Costs of the Work incurred on this Project only and shall require by contract its Subcontractors and Suppliers to similarly hold payments received in trust. For clarity, nothing in this **Paragraph 11.1** shall be deemed to limit Construction Manager in the enforcement of its rights and its agreements with Subcontractors and Suppliers. Construction Manager agrees that any payments not so used shall be subject to equitable tracing and returned to the Authority.
- **11.2** The period covered by each Application for Payment shall be one (1) calendar month ending on the last day of the month, or as otherwise agreed to between the Authority and Construction Manager in writing.
- **11.3** Provided that a Construction Manager Application for Payment is approved for full or partial payment, the Authority shall make payment of the approved amount to the Construction Manager not later than thirty (30) days after the Authority has received the Architect's Certificate for Payment. The Authority will endeavor to use commercially reasonable efforts to expedite and shorten this payment cycle.
- 11.4 Each Application for Payment shall be based upon the most recent Schedule of Values submitted by the Construction Manager and approved by the Authority and Architect in accordance with the Contract Documents. The Schedule of Values shall allocate the entire GMP among the various portions of the Work as required by the Authority. The Schedule of Values shall be prepared in such form and supported by such data to substantiate its accuracy as the Authority may require. The Schedule of Values, unless objected to by the Authority, shall be used as a basis for reviewing the Construction Manager's Application for Payment. The Construction Manager's Application for Payment will be rejected if the Schedule of Values is inconsistent between months or if the Schedule of Values is "front-end loaded" or otherwise unbalanced at the time the Construction Manager Application for Payment is submitted.
- 11.5 The Construction Manager Application for Payment shall show the percentage completion of each portion of the Work as of the end of the period covered by the Construction Manager Application for Payment. The percentage completion shall be the lesser of (1) the percentage of that portion of the Work which has actually been completed or (2) the percentage obtained by dividing (a) the Cost of the Work which has actually been incurred on account of that portion of the Work for which payment has been made or is intended to be made prior to the next Construction Manager Application for Payment by (b) the share of the GMP allocated to that portion of the Work in the Schedule of Values.
- **11.6** Subject to other provisions of the Contract Documents, the amount of each progress payment shall be computed as follows:
  - **11.6.1** Determine the portion of the GMP properly allocable to completed Work as determined by Subcontractor Applications for Payment, Construction Manager invoices and certified payroll summaries and as more particularly described in the Schedule of Values.
  - **11.6.2** Add that portion of the GMP properly allocable to materials and equipment delivered and suitably stored at the Project Site for subsequent incorporation in the Work, or, if approved in advance by the Authority, suitably stored off the Project Site at a location agreed upon in writing.

- **11.6.3** Add the Construction Manager's Fee. The Construction Manager's Fee shall be computed upon the Cost of the Work in accordance with <u>Exhibit 3</u> hereof, or, if the Construction Manager's Fee is stated as a fixed sum, shall be an amount which bears the same ratio to that fixed-sum as the Cost of the Work bears to a reasonable estimate of the probable Cost of the Work upon its completion.
- **11.6.4** Subtract retainage of five percent (5%). The Authority will in its discretion approve any deviation from this requirement.
- **11.6.5** Add Construction Manager's General Conditions costs.
- **11.6.6** Subtract the aggregate of previous payments made by the Authority.
- **11.6.7** Subtract the shortfall, if any, indicated by the Construction Manager required to substantiate prior Construction Manager Applications for Payment, or resulting from errors subsequently discovered by the Authority's accountants in such documentation.
- **11.6.8** Subtract amounts, if any, for which the Authority has reasonably withheld or denied a Construction Manager Application for Payment as provided in the Contract Documents.
- **11.7** Except with the Authority's prior written approval, payments to Subcontractors included in the Construction Manager's Application for Payment shall not exceed an amount for each Subcontractor calculated as follows:
  - **11.7.1** Determine the portion of the Subcontract Amount properly allocable to completed Work as determined by multiplying the percentage completion of each portion of the Subcontractor's Work by the share of the total Subcontract Amount allocated to that portion in the Subcontractor's Schedule of Values, less retainage.
  - **11.7.2** Add that portion of the Subcontract Amount properly allocable to materials and equipment delivered and suitably stored at the Project Site for subsequent incorporation in the Work or, if approved in advance by the Authority and Construction Manager, suitably stored off the Project Site at a location agreed upon in writing.
  - **11.7.3** Subtract the aggregate of previous payments made by the Authority, through the Construction Manager, to the Subcontractor.
  - **11.7.4** Subtract amounts, if any, for which the Authority or Construction Manager has withheld or denied a Subcontractor Application for Payment for reasons that are the fault of the Subcontractor unless the Work for which payment is requested has been performed by others in accordance with the Contract Documents.
  - **11.7.5** Add, upon Substantial Completion of the entire Work of the Subcontractor, a sum sufficient to increase the total payments to the Subcontractor to one-hundred percent (100%) of the Subcontract Amount, less amounts, if any, for incomplete Work and unsettled claims; and, if final completion of the entire Work is thereafter materially delayed through no fault of the Subcontractor, add any additional amounts payable on account of Work of the Subcontractor in accordance with the Contract Documents. The Subcontract Amount is the total amount stipulated in the Subcontract to be paid to the Subcontractor for the Subcontractor's performance of the Subcontract.
- **11.8** Except with the Authority's prior written approval, the Construction Manager shall not authorize advance payments to Suppliers for materials that have not been delivered to and properly stored at the Project Site.

- **11.9** In taking action on the Construction Manager's Application for Payment, the Authority and Architect shall be entitled to rely on the accuracy and completeness of the information furnished by the Construction Manager and shall not be deemed to represent that the Authority or Architect have made a detailed examination, audit or arithmetic verification of the documentation submitted by the Construction Manager or other supporting data, that the Authority or Architect have made exhaustive or continuous on-site inspections, or that the Authority or Architect have made examinations to ascertain how or for what purposes the Construction Manager has used amounts previously paid. If requested by the Authority, the Authority's accounting representative, acting in the sole interest of the Authority, will perform such examinations, audits and verifications.
- **11.10** The Construction Manager is required to certify that all amounts owed and requested under the Construction Manager Application for Payment are due to the Construction Manager, are accurate and correct, and that all of the Work completed to the date of the Construction Manager Application for Payment has been completed in accordance with the Contract Documents. The Construction Manager will be responsible for any errors or inaccuracies in the Construction Manager Application for Payment. The Construction Manager, and all Subcontractors or other Persons included for payment under the Construction Manager's Application for Payment, will be required to certify that there are no Claims outstanding for any of the Work completed to the date of the Construction Manager's Application for Payment with the Construction Manager's Application for Payment, will be required to certify that there are no Claims outstanding for any of the Work completed to the date of the Construction Manager's Application for Payment with the Contract Documents.
- 11.11 Notwithstanding the provisions of **Paragraph 11.10** and notwithstanding the wording of such certificates, the Construction Manager shall ensure that each Subcontractor's Work is satisfactory and in good order pursuant to the Contract Documents pending the issuance of a final certificate of payment and the Construction Manager shall be responsible for ensuring that the correction of defects or Work not performed regardless of whether or not such defects were apparent when such certificates were issued

#### FINAL PAYMENT

- 12.1 Final payment shall be made by the Authority to the Construction Manager when (1) the Construction Manager has achieved Final Completion, except for the Construction Manager's responsibility to correct defective or nonconforming Work, as provided in <u>Exhibit 6</u>, and to satisfy other requirements, if any, which necessarily survive final payment; (2) a final Application for Payment and a final accounting for the Cost of the Work have been submitted by the Construction Manager and reviewed by the Authority's accountants; and (3) a final Application for Payment has then been approved in writing by the Authority and the Architect. The Authority shall make such final payment, including any remaining retainage, if all of the Construction Manager's responsibilities have been satisfied (otherwise any remaining retainage shall be released upon satisfaction of all requirements) not more than ninety (90) days after approval by the Authority of the final Application for Payment, or as required by law. No final payment shall be issued until all final Contract Revisions have been issued and executed with each final Change Order and Construction Manager has provided the Authority with a detailed list of all Claims that are still unresolved at the time final payment is requested.
- **12.2** The amount of the final payment shall be calculated as follows:
  - **12.2.1** Take the sum of the Cost of the Work substantiated by the Construction Manager's final accounting and the Construction Manager's Fee, but not more than the GMP.
  - **12.2.2** Subtract amounts that the Authority withholds in accordance with the General Conditions or other provisions of the Contract Documents.
  - **12.2.3** Subtract the aggregate of the previous payments made by the Authority, but not in excess of the GMP.
- 12.3 The Authority's accountants will review and report in writing on the Construction Manager's final accounting within ninety (90) days after delivery of the final Application for Payment to the Authority by the Construction Manager, or as required by law. Based upon such Cost of the Work as the Authority's accountants report to be substantiated by the Construction Manager's final accounting and provided the other conditions of **Paragraph 12.1** have been met, the Authority will, pursuant to **Paragraph 12.1**, process Final Payment, or notify the Construction Manager in writing within seven (7) days after receipt of the written report from the Authority's accountants the Authority's reasons for withholding Final Payment as provided in the Contract Documents.
- 12.4 If, subsequent to Final Payment and at the Authority's request, the Construction Manager incurs costs described as reimbursable in Article 8 hereof and not required by Construction Manager's warranty obligations or excluded by Article 9 hereof to correct defective or nonconforming Work, the Authority shall reimburse the Construction Manager such costs and the Construction Manager's Fee applicable thereto on the same basis as if such costs had been incurred prior to Final Payment, but not in excess of the GMP.

#### **PROTECTION OF PERSONS AND PROPERTY**

- **13.1** The Construction Manager acknowledges and understands that the Authority has established safety of all Project Participants and employees involved in the Project as one of the major priorities for this Project. The Construction Manager must emphasize the importance of safety and a safe working environment to all its employees and Subcontractors and Suppliers of all tiers. Project safety must never be compromised.
- **13.2** The Construction Manager understands and agrees that the Construction Manager shall be responsible to the Authority for, and will initiate, maintain, and provide supervision of safety precautions and administer a comprehensive Project Safety Program in connection with the performance of the Work. Furthermore, the Construction Manager agrees that the Construction Manager's Project Safety Program will include reasonable precautions for safety of: (1) its employees on the Project and other Persons who may be affected thereby; and (2) all Project Participants. The Construction Manager will also take reasonable precautions for the protection of the Work, third party materials and equipment stored on the Project site in accordance with the Construction Manager's direction, and other property at or adjacent to the Project Site. The Authority will require its personnel, and will include a provision in its contracts with other members of the Project Team requiring that they shall comply with Construction Manager's Project Safety Program.
- **13.3** The Construction Manager is responsible for developing and administering a Project Safety Program for the Work that addresses, at a minimum, all of the requirements that will be developed in the Project Manual under Division 00800. The Construction Manager will be responsible for the performance and actions of all Persons, including all members of the Project Team, involved in the Work while present at the Project Site.
- 13.4 Notwithstanding anything herein to the contrary, as between the Authority and the Construction Manager, the Construction Manager is responsible for any and all safety issues relating to the Work on the Project. The Construction Manager shall administer and manage the safety program. This will include review of the safety programs of each Subcontractor, Vendor and Trade Contractor. The Construction Manager shall monitor the establishment and execution of effective safety practices, as applicable to the Work, and compliance with all applicable regulatory and advisory agency construction safety standards. The Construction Manager's responsibility for review, monitoring and coordination of the Subcontractors, Vendor and Trade Contractor's safety programs shall not relieve the Subcontractor, Vendor and Trade Contractor from controlling their safety programs. Notwithstanding the Construction Manager's safety obligations, each individual Subcontractor, Vendor and Trade Contractor shall also have controlling employer responsibility for the safety programs and precautions applicable to its own work and the activities of other work in areas designated as within the control of such Subcontractor, Vendor and Trade Contractor.

#### **CHANGES IN THE WORK**

#### 14.1 Construction Services

- 14.1.1 A Contract Revision related to the Work is a written order signed by the Authority and Construction Manager, and issued after execution of this Construction Services Agreement, stating their agreement with respect to a Change in the Work and which may include an adjustment in the GMP, and/or the extent of the adjustment in Milestone Dates and the date required for Substantial Completion. The Contract Revision form that will be used on the Project will be developed and included as an amendment to Exhibit 11 hereto. Costs to the Authority resulting from a Change in the Work shall be determined in writing between the Authority and the Construction Manager before issuance of any Contract Revision. The Construction Manager shall not proceed with any Changes in the Work either without a prior written authorization from the Authority, or if the Authority does not consider the Work in question to be a Change in the Work, then without a written notice of Claim as provided in Paragraph 6.2 of Exhibit 6 hereto. The Construction Manager agrees that it will not be paid for and waives any Claim for payment associated with any alleged Change in the Work or any Contract Revision unless the Construction Manager fully complies with the terms of this Article 14 and the notice of Claims provision in Exhibit 6 hereto. The Construction Manager further assumes any and all risks or liabilities associated with the Construction Manager proceeding with any alleged Change to the Work or any pending Contract Revision that is not administered, or preserved by a written notice of Claim, in strict accordance with the Contract Documents.
- 14.1.2 A Change Order related to the Work is a written order signed by the Construction Manager and Subcontractor, and issued after execution of a Subcontract, authorizing a change in a Subcontract. Costs to the Authority resulting from a change in any Subcontract must be approved in accordance with Subparagraph 14.1.1 hereof before issuance of any Change Order with a Subcontractor. The Construction Manager shall not proceed with any Changes in the Work, including the Work of any Subcontractor or Supplier, without prior written authorization from the Authority. The Construction Manager agrees that it will not be paid for and waive any Claim for payment associated with any alleged Change in the Work or any pending Change Order unless the Construction Manager fully complies with the terms of this Article 14. The Construction Manager proceeding with any alleged Change to the Work or pending Change Order that is not administered in strict accordance with the Contract Documents.
- **14.1.3** All Changes in the Work must be administered under the requirements of Division 1 of the Project Manual, which will be developed by the Authority and Architect, and which will be consistent with this **Article 14** and **Exhibit 6** hereto.
- 14.1.4 The provisions of the Contract Documents shall not be changed, amended, waived, or otherwise modified in any respect except by a writing signed by the Authority. No Person is authorized on behalf of the Authority to orally change, amend, waive, or otherwise modify the terms of the Contract Documents or any of the Construction Manager's duties or obligations under or arising out of the Contract Documents. Any change, waiver, approval, or consent granted to the Construction Manager shall be limited to the specific matters stated in writing signed by the Authority, and shall not relieve Construction Manager of any other duties and obligations under the Contract Documents. No "constructive" changes shall be allowed unless preceded with a specific notice of Claim addressing the alleged constructive change at issue in strict accordance with Subparagraph 6.2 of Exhibit 6 hereto.

# 14.2 Regulatory Changes

**14.2.1** The Construction Manager shall be entitled to an equitable Contract Revision, where applicable, for Changes in the Work necessitated by the enactment or revisions to Applicable Laws, which may be enacted from time to time after the execution of this Construction Services Agreement. In such instances, if any, the Construction Manager shall provide the Authority, in writing, the justification for such Changes in the Work. At the time of execution of this Construction Services Agreement, the Construction Manager is not aware of any regulatory changes that would necessitate a Change in the Work.

#### **CORRECTION OF WORK**

- **15.1** The Construction Manager shall promptly give notice to the Authority and SDC Group of any defective or nonconforming Work of the Architect or Construction Manager discovered during the term of this Construction Services Agreement and within a period of two (2) years from the Date of Substantial Completion of the Work or designated portion thereof, or within such longer period provided by any applicable special or extended warranty.
- **15.2** The Construction Manager shall be responsible to promptly make corrections in the Work when the Construction Manager's Work is found to be damaged, defective, or nonconforming with the Contract Documents. Corrections to the Work that are not remedied within seven (7) days of delivery of a notice to the Construction Manager, or sooner if deemed necessary by the Authority to maintain sequencing of the Work or operation of any portion of the Project, may be completed by the Authority and back charged to the Construction Manager, including all costs of administering such Work without further notice to the Construction Manager. Construction Manager on its behalf and on behalf of its insurers, agents, and attorneys hereby waives any right to spoliation notice or right to be present during any work performed by the Authority pursuant to this provision. Costs associated with corrections of the Work shall be borne by the Construction Manager pursuant to **Paragraphs 8.1** and **8.2** and **Subparagraph 9.1.7** of this Construction Services Agreement.
- **15.3** The Construction Manager shall be responsible for all reasonable and direct costs of the Authority resulting from corrections in the Work for which Construction Manager is responsible and shall be further responsible for any Design Services which are caused or necessitated by errors, omissions, inconsistencies or other damages caused by the Construction Manager and for which the Construction Manager is responsible.

#### INSURANCE, DISPUTE RESOLUTION, TERMINATION, AND INDEMNITY

#### 16.1 Insurance

- 16.1.1 The Construction Manager shall be required to purchase and maintain, at its own expense, the insurance coverages described in the Contract Documents and as specifically set for in Article 13 of <u>Exhibit 6</u> hereto. Certificates of the Construction Manager's insurance shall be made available to the Authority prior to execution of this Construction Services Agreement.
- **16.1.2** The Authority shall be responsible for purchasing and maintaining the insurance required under the Contract Documents and as specifically set forth in Article 13 of <u>Exhibit 6</u> hereto. Certificates of the Authority's insurance shall be made available to the Construction Manager upon written request.

#### 16.2 Dispute Resolution

- **16.2.1** Claims, disputes, and other matters in question between the Parties to this Construction Services Agreement or related to the Work and arising out of or relating to the formation or performance of this Construction Services Agreement or the Work shall be submitted and resolved as provided in **Paragraphs 6.2, 6.3** and **6.4** of **Exhibit 6** hereto.
- **16.2.2** Subparagraph 16.2.1 hereof shall survive completion or termination of this Construction Services Agreement.

#### **16.3** Termination by the Authority

**16.3.1** As provided more fully in **Article 16** of <u>**Exhibit 6**</u> hereto, the Authority shall have the right to suspend, terminate for cause, or terminate for convenience this Construction Services Agreement.

#### **16.4** Termination by the Construction Manager

16.4.1 The Construction Manager shall have the right to suspend its performance of the Work or terminate this Construction Services Agreement in accordance with Article 16 of <u>Exhibit 6</u> hereto.

#### 16.5 Indemnification

16.5.1 The Construction Manager is required to indemnify, hold harmless and defend the Authority and all of the Indemnitees as specifically set for in Paragraph 4.18 of <u>Exhibit 6</u> hereto.

#### **OTHER PROVISIONS**

#### 17.1 Nondiscrimination

In connection with the performance of its Work, the Construction Manager agrees not to illegally discriminate against any employee or applicant for employment for any reason, including: age, race, sex, national origin, ancestry, religion, or color. This provision shall include the following: employment, upgrading, demotion or transfer, recruitment or recruitment advertising, layoff or termination, rates of pay or other forms of compensation, and selection for training. Nothing contained in this **Paragraph 17.1** shall be considered a limitation of the Authority's rights of termination pursuant to **Paragraph 16.4** hereof.

#### 17.2 Successors and Assigns

The Authority and Construction Manager each bind themselves, their partners, successors, permitted assigns, and legal representatives to the other Party to this Construction Services Agreement and to the partners, successors, permitted assigns, and legal representative of such other Party with respect to covenants of this Construction Services Agreement.

**17.2.1** Notwithstanding anything to the contrary in this Construction Services Agreement, it is acknowledged and agreed that the Authority shall have the right to assign all its rights and duties under this Construction Services Agreement without the consent of Construction Manager to third parties including the Team. The Construction Manager shall execute all consents reasonably required to facilitate such assignment. This Construction Services Agreement may not be assigned by Construction Manager without the prior written approval of the Authority, which approval may be withheld in the Authority's sole discretion.

## 17.3 Legal Relations

- **17.3.1** Nothing contained in this Construction Services Agreement shall create or is intended to create a contractual relationship with, or a cause of action in favor of, any third party against the Authority or any of the Indemnities (except the Architect Indemnitees). Notwithstanding the foregoing, the Parties hereto acknowledge and agree to the third party beneficiary rights in favor of the Authority established in **Paragraph 17.4** herein.
- 17.3.2 Should Construction Manager have a Claim against a Person involved with the Project, other than the Persons described in this **Paragraph 17.3**, then the Construction Manager shall make such Claim directly against such other Person and not against the Authority or any of the Indemnitees(except the Architect Indemnitees with respect to design errors or omissions or breaches of the Design Services Agreement).
- 17.3.3 Construction Manager agrees and acknowledges that the Act requires the Authority to bid project construction in a manner that any cost overruns that exceed the GMP are the responsibility of the successful bidder and not the Authority or the State. Accordingly, Construction Manager agrees and acknowledges that as the successful bidder, Construction Manager is solely responsible for any cost overruns that may occur on the Project in excess of the GMP as modified in accordance with and as allowed by the Contract Documents, however caused, as the Authority has no authority to accept liability for cost overruns in contravention of the Act. Therefore, notwithstanding anything to the contrary in this Construction Services Agreement except as provided in Paragraph 4.3 hereof and Subparagraph 10.3.1 of Exhibit 6 hereto, to the fullest extent permitted by Applicable Law, Construction Manager hereby waives any and all Claims against Authority and any of the Indemnitees (except the Architect Indemnitees) arising from or

relating to (1) the Architect's negligent acts, errors or omissions; (2) any implied or express warranty as to the completeness, constructability, accuracy, suitability, or timeliness of the completion of any Drawings, Specifications, or other Contract Documents; and (3) any other Claim the result of which would be to impose liability upon the Authority for a cost overrun in violation of the Act. The Construction Manager hereby agrees that its sole and exclusive remedy for any Claims arising out of the negligent acts, errors, or omissions of the Architect, the completeness, constructability, accuracy, suitability, or timeliness of the completion of any Drawings, Specifications, or other Contract Documents, or any implied or express warranties applicable by law to such documents, shall be a claim made directly against the Architect, the Architect's Subconsultants, or any insurer of a project specific insurance policy who may provide coverage for such claim, but only to the extent the Construction Manager is entitled to directly enforce the insurance policy. Neither the Authority nor the Team makes any promise or representation of the rights, Claims, or causes of action that the Construction Manager may have against the Architect or Architect Indemnitees. Notwithstanding anything in this Subparagraph 17.3.3 to the contrary, the Construction Manager is not assuming liability for design defects that are discovered after Final Completion (except for those about which Construction Manager had knowledge before Final Completion).

17.3.4 The Construction Manager expressly acknowledges that it is an independent contractor and that it is not the representative or agent of the Authority or Team. Nothing contained in this Construction Services Agreement shall be construed as constituting a joint venture or partnership between the Construction Manager and the Authority or the Team. The Construction Manager shall have the authority to act on behalf of the Authority only to the extent expressly provided in this Construction Services Agreement unless otherwise modified by a subsequent written instrument. Under no circumstances shall Construction Manager contract, negotiate or make commitments concerning the Project with any Governmental Authority or other authority with jurisdiction over the Project without the Authority's prior written authorization. Construction Manager shall not order or direct any corrective work on the Project to any structural enclosure, mechanical, electrical, or plumbing issue, or any other item of correction work in excess of \$10,000 for any one item of work in which case prior written authorization from the Authority is required. The Construction Manager is not authorized to act on the Authority's behalf, and shall not act on the Authority's behalf, in such a manner as to result in change(s) to (1) the cost or compensation to be paid to any Project Participant or other Persons, or (2) the time for completing any portion of the Design Services or the Work as required and agreed to in this Construction Services Agreement or the Contract Documents, or (3) the scope of the Design Services or the Work, unless such representation is specifically provided for, set forth and authorized in this Construction Services Agreement. The Authority will not assume, accept or ratify any obligation, commitment, responsibility or liability which may result from representation by the Construction Manager not specifically provided for and authorized as stated in this Construction Services Agreement.

## 17.4 Third Party Beneficiaries

It is expressly agreed and understood that all of the Indemnitees (except Architect Indemnitees) are intended third party beneficiaries of all the representations, warranties and covenants made by the Construction Manager in the Contract Documents, including without limitation the promises of the Construction Manager to perform and complete the Work; provided, however, that the Architect Indemnitees are third party beneficiaries of **Paragraph 4.18** of **Exhibit 6** hereto. The rights of third party beneficiaries as described herein shall be consistent with the Contract Documents. The responsibilities and liabilities of the Construction Manager toward such third party beneficiaries shall be consistent with the Contract Documents.

## 17.5 Governing Law, Venue, and Jurisdiction

The laws of the State of Minnesota, without giving any effect to conflicts of law provisions, shall govern and construe this Construction Services Agreement. Each Party to this Construction Services Agreement (1) agrees that except for those Claims or disputes which are subject to dispute resolution requirements set forth in this Construction Services Agreement, any suit, action, or other legal proceeding arising out of this Construction Services Agreement, any related agreements, or any of the transactions contemplated hereby or thereby shall be brought in the courts of the State of Minnesota, Fourth Judicial District, Hennepin County District Court; (2) consents to the jurisdiction of such court in any such suit, action, or proceeding; and (3) waives any objection which it may have to the venue of any such suit, action, or proceeding in such court.

#### 17.6 Time is of the Essence

Time is of the essence with respect to this Construction Services Agreement and the performance of obligations hereunder.

#### 17.7 Lender Approval

The Construction Manager hereby acknowledges that the Authority may be required to provide this Construction Services Agreement to the Team's lender(s) for review and approval. The Construction Manager agrees to modify the terms and conditions of this Construction Services Agreement as may be reasonably requested by the Authority to satisfy the requirements of the Team's lender(s), so long as any such modification does not materially increase the Construction Manager's risk or obligations under the Contract Documents as may be determined by the Authority in its discretion.

#### 17.8 Construction Manager Performance

The Construction Manager understands and agrees that the Authority retains, at its choice, any and all rights under law and under this Construction Services Agreement, including injunctive relief, specific performance, and the right to recover damages against the Construction Manager caused directly or indirectly by the Construction Manager's breach of the Construction Services Agreement; **provided**, **however**, that the Authority's rights to recover the Liquidated Damages as provided in **Paragraph 4.2** herein shall limit the Authority's and Team's rights to recover from the Construction Manager due to its unexcused failure to achieve the Date of Substantial Completion in the Construction Schedule, and provided further that the provisions of **Subparagraph 6.2.10** of **Exhibit 6** hereto shall determine the parties' rights to recover consequential damages.

## 17.9 Payment Bond

Pursuant to Minnesota Statutes §§ 574.26 - 574.32 and the Act, at the time the GMP is certified, the Construction Manager is required to provide a Payment Bond in an amount equal to one-hundred percent (100%) of the GMP and in the form of **Exhibit 9** hereto and from a surety or sureties acceptable to the Authority. In the event the Construction Manager self performs any Work, that portion of the Work must still be 100% bonded, and a copy of the Construction Manager's Payment Bond will be provided and included within said **Exhibit 9**. If after the giving of the bond the GMP is increased, for any reason, the Authority may require additional bonds, the amount of which shall be not less than the amount of the increase in the GMP, and if the additional bonds are not furnished within ten (10) calendar days after demand, the Work shall cease until the additional bonds are furnished.

## 17.10 Performance Bond

Pursuant to Minnesota Statutes §§ 574.26 - 574.32 and the Act, at the time the GMP is certified, the Construction Manager is required to provide a Performance Bond in an amount equal to one-hundred percent (100%) of the GMP and in the form of **Exhibit 9** hereto and from a surety or sureties acceptable to the Authority. In the event the Construction Manager self performs any Work, that portion of the Work must still be 100% bonded, and a copy of the Construction Manager's Performance Bond will be provided and included within said **Exhibit 9**. If after the giving of the bond the GMP is increased, for any reason, the Authority may require additional bonds, the amount of which shall be not less than the amount of the increase in the GMP, and if the additional bonds are not furnished within ten (10) calendar days after demand, the Work shall cease until the additional bonds are furnished.

#### 17.11 Warranty of Construction

The Construction Manager's Warranty of Construction is included herein as **Exhibit 10**.

#### 17.12 Environmental and Other Responsibilities During Construction

The Construction Manager is responsible to administer the Construction Manager's Work in accordance with the Environmental and Other Responsibilities as outlined in <u>Exhibit 12</u> hereto.

## 17.13 Entire Agreement

This Construction Services Agreement represents the entire agreement between the Authority and Construction Manager and supersedes any prior negotiations, representations, promises, or agreements whether written or oral. This Construction Services Agreement may be amended only by written instrument signed by both Authority and the Construction Manager.

#### **17.14** Notice Provisions

Every notice, demand, request, consent, approval, or other communication, including notice of breach, indemnity, suspension, termination, or default, that either Party hereto is required or desires to give or make to the other Party hereto shall, notwithstanding any other provisions of the Construction Services Agreement, be effective only if given in writing and delivered by hand and receipted for, or by registered or certified mail, postage-prepaid, return receipt requested, or by overnight mail as follows:

**17.14.1** If to the Construction Manager, addressed to:

M. A. Mortenson Company 700 Meadow Lane North Minneapolis, Minnesota 55422 Attention: John V. Wood Senior Vice President Fax: 763-287-5507

with a copy to: M. A. Mortenson Company

700 Meadow Lane North Minneapolis, Minnesota 55422 Attention: Dwight Larson Vice President Senior Counsel Fax: 763-287-5224

Faegre Bakers Daniels 2200 Wells Fargo Center 90 S. Seventh Street Minneapolis, Minnesota 55402-3901 Attention: Peter Halls Fax: 612-766-1600

#### **17.14.2** If to the Authority, addressed to:

	Metropolitan Sports Facility Authorities 900 South Fifth Street Minneapolis, MN 55415 Attention: Ted Mondale CEO/Executive Director Fax: 612-332-8334
with a copy to:	Dorsey & Whitney, LLP Suite 1500 50 South Sixth Street Minneapolis, MN 55402-1498 Attention: Jay L. Lindgren Fax: 612-340-2868
with a copy to:	Fabyanske, Westra, Hart & Thomson, PA 800 LaSalle Avenue, Suite 1900 Minneapolis, MN 55402 Attention: Dean B. Thomson Fax: 612-359-7602
with a copy to:	Minnesota Vikings Football, LLC 9520 Viking Drive Eden Prairie, MN 55344 Attention: Kevin Warren Vice President of Legal Affairs & Chief Administrative Officer Fax No.: 952-828-6514
with a copy to:	Briggs and Morgan 2200 IDS Center 80 South Eighth Street Minneapolis, MN 55402 Attention: Brian Wenger, Mike Grimes, and Matt Slaven Fax No.: 612-977-8650

Or to such other address or addresses or via other means as Authority and Construction Manager shall from time to time designate by written agreement and notice given and delivered as aforesaid.

17.15 The Construction Manager shall cooperate with efforts made by the Authority and other members of the Project Team to ensure that, to the extent practicable, steel used in the construction of the Stadium is American-made steel that is made from Minnesota iron ore. To the extent practicable, the Construction Manager shall ensure that twenty five percent (25%) of other materials, supplies, and equipment used in the construction of the Stadium and Stadium Infrastructure are made or produced by Minnesota businesses.

- **17.16** Construction Manager agrees and acknowledges that the Project is a public project and the Project will be used for public purposes and all of the Work hereunder is in furtherance of a public project.
- **17.17** The Construction Manager shall cooperate with efforts made by the Authority and other members of the Project Team to build the Stadium, to the extent practicable, so that it is eligible to receive the Leadership in Energy and Environmental Design (LEED) certification or the Green Building Initiatives Green Globes certification for environmental design.
- 17.18 In the event that, after the determination of the Performance Compensation under Paragraph 7 of Exhibit 3 and just prior to Final Completion of the Work, the total Cost of the Work, plus the Construction Manager's Fee for Construction Phase Services, plus the Construction Manager's stipulated lump sum Fee for Pre-Construction Phase Services, plus any Performance Compensation, is less than the final GMP, as adjusted during the Project by applicable Contract Revisions (including the Contract Revision that reduces the Construction Manager's Contingency to no more than 0.5% immediately before Final Completion), then the difference will be considered savings and then the GMP shall be reduced by the amount of said savings and (1) half of the savings shall be paid to the Construction Manager; and (2) the other half of the savings shall be paid to the State for transfer to the Authority for capital reserves.
- **17.19** This Construction Services Agreement may be executed by the Authority and Construction Manager separately in counterparts which, taken together, shall constitute one original. This Construction Services Agreement may also be executed by signatures sent by facsimile or email (in .pdf format), or by electronic signatures, which shall be deemed to have the same force and effect as an original signature.
- **17.20** The Authority will, by subsequent Contract Revision, specifically state the prevailing wage rates, prevailing hours of labor, and hourly basic rates of pay once they are determined for this Project, and will incorporate into this Construction Services Agreement the applicable wage determinations for Work along with language to be provided by the commissioner of the state department of Labor and Industry that will notify the Construction Manager and its Subcontractors of the applicability of Minn. Stat. §§177.41 to 177.44 to this Project. The Construction Manager shall comply with prevailing wage requirements under Minn. Stat. §§ 177.41 to 177.43 or as otherwise required by the Contract Documents or Applicable Laws. The Authority shall demand and the Construction Manager and its Subcontractors shall furnish to the Authority copies of any or all payrolls no more than fourteen (14) days after the end of each pay period and said payrolls must contain all the data required by Minn. Stat. §177.30. The Authority may also examine all records relating to wages paid to laborers or mechanics on work to which Minn. Stat. §§ 177.41 to 177.43 apply.
- 17.21 The Construction Manager shall keep strictly confidential all Confidential Information concerning and relating to the Project, in accordance with the requirements set forth in <u>Exhibit 6</u> hereto. The Construction Manager, its officers, agents, owners, partners, employees, volunteers, and Subcontractors shall abide by the provisions of the Minnesota Government Data Practices Act, Minnesota Statutes, ch. 13 (the "MGDPA") and all other Applicable Laws relating to data privacy or confidentiality, and as any of the same may be amended. The Construction Manager agrees to defend, indemnify, and hold harmless the Indemnitees from and against any claims resulting from the unlawful disclosure and/or use of such protected data by the Construction Manager or the officers, agents, owners, partners, employees, volunteers, assignees, or Subcontractors of the Construction Manager, or other noncompliance with the requirements of this **Paragraph 17.21**. The Construction Manager agrees to promptly notify the Authority and Team if it becomes aware of any potential claims, or facts giving rise to such, under the MGDPA. The terms of this **Paragraph 17.21** shall survive the cancellation, suspension or termination of this Construction Services Agreement.

- **17.22** To the extent practicable, the Construction Manager will hire up to ten (10) current Authority maintenance employees during the Construction Phase of the Project to function as security, watch persons, and clean up workers for the Stadium Site.
- **17.23 Design/Build Option**. The Construction Manager acknowledges and agrees that the Authority, at its sole discretion, may elect to convert this Construction Services Agreement to a design/build delivery system on terms mutually agreeable to the Authority and Construction Manager.
- **17.24 Minnesota False Claims Act ("FCA").** Minnesota Statutes §§15C.01 *et seq.* are applicable to this Project, and claims under the FCA include payment applications, Claims for additional Contract Time and increases to the GMP, and certifications regarding the Equity Plan for the Project. The Authority considers that the application of and obligation to comply with the FCA not only flows down from the Construction Manager to all its Subcontractors and Suppliers, but is also directly applicable to Subcontractors and Suppliers. The Construction Manager shall provide in its Subcontracts and Purchase Orders that its Subcontractors and Suppliers shall comply with the FCA in regards to claims made against the Construction Manager to the same extent that the Construction Manager must comply with the FCA in regards to its claims made to the Authority, and that its Subcontractors and Suppliers shall defend and indemnify the Construction Manager from any such claims.
- 17.25 Bond Capacity. The Authority shall have a period of not less than sixty (60) days following execution of this Construction Services Agreement to verify the Construction Manager's bonding capacity. The Construction Manager shall use all commercially reasonable efforts to assist the Authority in its review and shall endeavor to cause its surety to cooperate with the Authority. If at the time the Construction Manager submits its Construction Plan, including its GMP Proposal, it is unable to provide the bonds required under **Paragraphs 17.9** and **17.10**, the Authority shall have the right to terminate this Construction Services Agreement or to cause the Construction Manager to form a joint venture capable of providing the bonds required under **Paragraphs 17.9** and **17.10**.
- 17.26 Structure Consultants. The Construction Manager has retained the services of Walter P. Moore and Associates, Inc., L.P.R. Construction Co., and Justin Waldron, P.E. (collectively "Structure Consultants") to support and reinforce the performance of the Construction Manager's obligations and responsibilities under the Construction Services Agreement and specifically provide construction management related advice and experience related to planning, procurement, constructability, pricing, and construction of the long span roof structure and any operable roof or wall components, including advice and experience to the Construction Manager in support of Construction Manager's efforts to provide pricing and constructability assistance relating to the Architect's evaluation of such long span roof structure and evaluation and comparison of fixed and operable roof and wall models as set forth in **Exhibit 1** to the Design Services Agreement. The Construction Manager will use the Structure Consultants as appropriate pursuant to its Standard of Care and to the extent appropriate to perform both its Pre-Construction Phase Services and its Construction Phase Services. The Construction Manager shall pay Structure Consultants from its GCGMP and submit the charges for Structure Consultants' services incurred during the Pre-Construction Phase in the Construction Manager's first Application for Payment for the Construction Phase of the Work. Should the Construction Manager not develop a Construction Management Plan that is accepted by the Authority, the Construction Manager will not be entitled to reimbursement for the Structure Consultants' services performed during the Pre-Construction Phase.

## [THIS SPACE LEFT INTENTIONALLY BLANK]

## [SIGNATURE PAGES FOLLOW]

THIS CONSTRUCTION SERVICES AGREEMENT is entered into as of the day and year first written above.

**AUTHORITY:** 

Michele Kelm-Helgen By: Title: Its Chair

**AUTHORITY:** 

By: Ted Mondale Title: Its CEO/Executive Director

**CONSTRUCTION MANAGER:** 

By: John V. Wood Title: Senior Vice President

## EXHIBIT 1 DESCRIPTION OF BASIC SERVICES

The following Description of Basic Services is included in the Construction Services Agreement as **Exhibit 1**:

The Basic Services to be provided by the Construction Manager and its Subcontractors, Suppliers, consultants and subconsultants of any tier and associated with the Construction Manager's Work for the Project are described below, without limitation and will be performed by the Construction Manager in accordance with the terms and conditions of the Construction Services Agreement.

It is the intention of this Construction Services Agreement that the Construction Manager shall provide all work, services, labor, materials, and equipment to complete the Project in accordance with this Construction Services Agreement, including the Construction Documents and all applicable legal requirements. The Services required to deliver the Project in such form include the following, all of which shall be Basic Services under this Construction Services Agreement.

#### RESPONSIBILITIES

#### 1.1 CONSTRUCTION MANAGER'S BASIC SERVICES

- **1.1.1** The Construction Manager's Basic Services consist of those services performed by the Construction Manager, Construction Manager's employees, and Construction Manager's Subcontractors, Suppliers, consultants, and subconsultants at any tier. This description of Basic Services (**"Basic Services"**) is intended to provide a general description of the Construction Manager's Basic Services but shall not in any way be considered a limitation of the services or the responsibilities of the Construction Manager.
- **1.1.2** The Construction Manager shall provide sufficient organization, personnel and management to carry out the requirements of this Construction Services Agreement in an expeditious and economical manner consistent with the interests of the Authority.

#### SCOPE OF CONSTRUCTION MANAGER'S BASIC SERVICES

## 2.1 **DEFINITION**

2.1.1 The Construction Manager's Basic Services consist of those activities described in Article 2 of the Construction Services Agreement and all services identified herein.

# 2.2 **PRE-CONSTRUCTION PHASE**

- **2.2.1** The Construction Manager shall review the Program furnished by the Authority to ascertain the requirements of the Project and shall arrive at a mutual understanding of such requirements with the Authority.
- **2.2.2** The Construction Manager shall provide a preliminary evaluation of the Authority's Program, schedule and construction budget requirements, each in terms of the other.
- **2.2.3** Based on the Design Documents and other design criteria prepared by the Architect, the Construction Manager shall prepare preliminary estimates of Construction Cost for Program requirements using area, volume or similar conceptual estimating techniques. The Construction Manager shall provide cost evaluations of alternative materials and systems.

- **2.2.4** The Construction Manager shall be responsible to notify the Authority when the Design Documents being developed by the Architect deviate from the Authority's Fixed Construction Budget and/or the Master Project Schedule. The Fixed Construction Budget shall be evaluated in summary and in particular line items. The Construction Manager shall provide the Authority with detailed value engineering evaluations and recommendations to modify the Project design and Design Documents to correspond to the Fixed Construction Budget and Master Project Schedule.
- **2.2.5** The Construction Manager shall expeditiously review Design Documents during their development and advise on proposed site use and improvements, selection of means, methods, materials, building systems and equipment, and methods of Project delivery. The Construction Manager shall provide recommendations on relative feasibility of construction methods, availability of materials and labor, time requirements for procurement, installation and construction, and factors related to construction cost and schedule including, but not limited to, costs of alternative designs or materials, preliminary budgets, sequencing and schedule and possible economies.
- 2.2.6 The Construction Manager shall prepare and periodically update a Construction Schedule for the Authority's review and approval. The Construction Manager shall obtain the Architect's input and approval for the portion of the Construction Schedule relating to the performance of the Architect's services. In the Construction Schedule, the Construction Manager shall identify, coordinate, and integrate the Construction Manager's Services, the Design Services, Trade Contractor Work, Vendor Work, and the Authority's responsibilities with anticipated construction schedules, highlighting critical and long lead-time items. The Construction Schedule must identify and provide adequate time for Authority reviews and approvals. The Construction Manager is responsible to coordinate its Work with the Design Services and in accordance with the Construction Schedule to avoid any delays in completion of the Work in accordance with the Master Project Schedule. The Construction Schedule shall provide for the components of the Construction Manager's Work, including phasing of construction, times of commencement and completion required of each Subcontractor ordering and delivery of products requiring long lead time, and the occupancy requirements of the Authority.
- **2.2.7** As the Architect progresses with the preparation of the Design Documents, the Construction Manager shall prepare and update, at 50% Schematic Design, final Schematic Design, 50% Design Development Documents, and final Design Development Documents, estimates of Construction Cost of increasing detail and refinement. The estimated cost for each Division of the Work shall be indicated with supporting detail. Such estimates shall be provided for the Authority's review and approval. The Construction Manager shall advise the Authority if it appears that the Construction Cost may exceed the latest approved Fixed Construction Budget and make recommendations for corrective action.
- **2.2.8** The Construction Manager shall consult with the Authority and Architect regarding the Design Documents as they are being developed and make recommendations whenever design decisions or details adversely affect constructability, cost and schedules.
- **2.2.9** The Construction Manager shall provide recommendations and information to the Authority regarding the assignment of responsibilities for temporary Project facilities and equipment, materials and services for common use of the Subcontractors. The Construction Manager shall verify that such requirements and assignment of responsibilities are included in the proposed Construction Documents.
- **2.2.10** The Construction Manager shall prepare and manage a comprehensive safety program in compliance with workplace safety regulations and all other federal, state or local rules

and regulations to be administered by the Construction Manager for all activities at the Project Site. The Construction Manager shall be responsible to enforce the safety program with all Subcontractors, Suppliers, Trade Contractors and Vendors of all tiers.

- **2.2.11** The Construction Manager shall advise on the division of the Project into individual Subcontractors for various categories of Work, including the method to be used for selecting Subcontractors and awarding Subcontracts. The Construction Manager shall review the Contract Documents and make recommendations as required to provide and ensure that (1) the Work of the Subcontractors, Suppliers, Trade Contractors, and Vendors is coordinated, (2) all requirements for the Construction Manager's Work have been assigned to either the Construction Manager or the appropriate Subcontractor or Supplier, (3) the Contract Documents are complete and in sufficient detail for bidding and award of Subcontracts and the Construction Documents are representative of the full and complete scope of the Work, (4) the likelihood of jurisdictional disputes has been minimized, and (5) proper coordination has been provided for phased construction.
- **2.2.12** The Construction Manager shall review the Design Documents as they are being prepared by the Architect to determine that they are complete and include all necessary detail and information sufficient to allow the Construction Manager to secure competitive bids and proposals appropriate to the phase of design development.
- **2.2.13** The Construction Manager shall provide the current Construction Schedule for each set of bidding documents.
- **2.2.14** The Construction Manager shall expedite and coordinate the ordering and delivery of materials requiring long lead time.
- **2.2.15** The Construction Manager shall assist the Authority in selecting, retaining and coordinating the professional services of surveyors, special consultants and testing laboratories required for the Project. The Construction Manager shall be responsible for surveying necessary for the proper performance of the Work.
- **2.2.16** The Construction Manager shall provide an analysis of the types and quantities of labor required for the Work and review the availability of appropriate categories of labor required for critical Phases. The Construction Manager shall make recommendations for actions designed to minimize adverse effects of labor shortages.
- **2.2.17** The Construction Manager shall be responsible to comply with requirements for equal employment opportunity programs and Targeted Business and workforce utilization plans applicable to the Project in accordance with the Authority's requirements.
- **2.2.18** Following the Authority's review of the final Design Development Documents, the Construction Manager shall update and submit the latest estimate of Construction Cost and the Construction Schedule for the Authority's review and approval.
- **2.2.19** The Construction Manager shall submit the list of prospective bidders for all Phases of the Work for the Authority's review and approval.
- **2.2.20** The Construction Manager shall develop bidders' interest in the Project and establish bidding schedules and qualification procedures. The Construction Manager, with the assistance of the Architect, shall issue bidding documents to bidders and conduct pre-bid conferences with prospective bidders. The Construction Manager shall assist the Architect with regard to questions from bidders and with the issuance of addenda.
- **2.2.21** The Construction Manager shall receive bids, prepare bid analyses and make recommendations to the Authority for the award of Subcontracts or rejection of bids. The Construction Manager shall prepare detailed bid summaries in a format acceptable to the

Authority describing the proposed bids, exclusions to the bids and proposed alternates or substitutions. The Construction Manager's bid summary shall include a detailed accounting for any sales/use taxes that Construction Manager, Subcontractor, or Supplier expects to incur or include in Costs of the Work. Such accounting shall include type of tax (sales, use, excise, etc.), tax amount, tax rate, tax jurisdiction and shall be provided prior to commencing work on the Project.

- **2.2.22** The Construction Manager shall prepare Subcontracts and advise the Authority on the acceptability of Subcontractors and material suppliers proposed by Subcontractors.
- **2.2.23** The Construction Manager shall obtain building permits and special permits for the Work, except for permits required to be obtained directly by the various Subcontractors. The Construction Manager shall verify that the Authority has paid applicable fees and assessments. The Construction Manager shall assist the Authority and Architect in connection with the Authority's responsibility for filing documents required for the approvals of Governmental Authorities having jurisdiction over the Project. The Authority shall pay directly for primary building permit(s). Construction Manager shall include in the Cost of the Work costs for all other permits, licenses and inspections.
- 2.2.24 The Construction Manager shall provide the Authority with a detailed Construction Management Plan upon completing the Pre-Construction Phase Services in this Paragraph 2.2 that describes in detail the Construction Manager's approach to the construction of the Project. The Construction Management Plan will include: (1) Proposed GMP and Detailed Construction Cost Estimates, (2) Subcontractor Bids and Proposals, (3) Summary of Construction Documents, (4) Cost Control Procedures, (5) Construction Schedule, (6) Communication and Document Control Procedures.

# 2.3 CONSTRUCTION PHASE - ADMINISTRATION OF THE CONSTRUCTION CONTRACT

- **2.3.1** The Construction Phase Services will commence upon the Authority accepting the Construction Manager's Construction Management Plan and a certified GMP and completion date for final completion of the Work. The Authority may require that the GMP be separated into two (2) or more separate sums and be tracked separately throughout the course of the Project.
- **2.3.2** The Construction Manager shall provide administration, management, and related services to coordinate the scheduled activities and responsibilities of the Subcontractors, Trade Contractors, and Vendors with each other and with those of the Construction Manager, Trade Contractors, Vendors, the Authority, and Architect to manage the Project in accordance with the GMP, the Construction Schedule, and the Contract Documents.
- **2.3.3** The Construction Manager shall schedule and conduct meetings to discuss such matters as procedures, progress, and scheduling. The Construction Manager shall prepare and promptly distribute minutes to the Authority, Architect, Subcontractors, Trade Contractors, and Vendors.
- **2.3.4** Utilizing the construction schedules provided by the Subcontractors, Trade Contractors, and Vendors for informational purposes only, the Construction Manager shall update the Construction Schedule incorporating the activities of all Subcontractors, Trade Contractors, and Vendors on the Project, including activity sequences and durations, allocation of labor and materials, processing of Shop Drawings, Project Data, and Samples, and delivery of products requiring long lead time and procurement. The Construction Schedule shall include the Authority's occupancy requirements showing portions of the Project having occupancy priority. The Construction Manager shall update and reissue the Construction Schedule as required, but no less than monthly, to

show current conditions. If an update indicates that the previously approved Construction Schedule may not be met, the Construction Manager shall recommend corrective action to the Authority.

- **2.3.5** Consistent with the various bidding documents and utilizing input and information from the Subcontractors for informational purposes only, the Construction Manager shall be responsible to coordinate the sequence of construction and assignments of responsibilities and space in areas where the Subcontractors are performing Work. The Construction Manager shall be responsible for the means, methods, materials, sequencing, and techniques used in construction.
- **2.3.6** The Construction Manager shall be responsible for the satisfactory performance of each Subcontractor.
- **2.3.7** The Construction Manager shall monitor the approved Cost of the Work and GMP in a format acceptable to the Authority. The Construction Manager shall show actual costs for activities in progress and estimates for uncompleted tasks by way of comparison with such approved estimate.
- **2.3.8** The Construction Manager shall develop a Cost Control Report for the review and approval of the Authority. The Cost Control Report will provide cash flow reports and forecasts for the Project and advise the Authority as to variances between actual and budgeted or estimated Costs of the Work.
- **2.3.9** The Construction Manager shall maintain accounting records on the Work performed by the Construction Manager and Subcontractors on the basis of contract amounts and/or actual Costs of the Work. The Construction Manager shall also review accounting records related to the Trade Contractor Work and Vendor Work and shall assist the Authority to account for the costs of the Trade Contractor Work or Vendor Work, where a Trade Contract or Vendor Contract is based on a guaranteed maximum price or other cost-reimbursable terms.
- **2.3.10** The Construction Manager shall develop and implement procedures for the review and processing of Applications for Payment by Subcontractors for progress and final payments. The Construction Manager shall develop the project accounting procedures in accordance with the requirements set forth by the Authority.
  - .1 Based on the Construction Manager's inspections and evaluations of each Subcontractor's Application for Payment, the Construction Manager shall review and certify the amounts due the respective Subcontractors.
  - .2 The Construction Manager shall prepare a complete Application for Payment based on Subcontractors' Applications for Payment. The Authority shall have the right to accept or reject a Subcontractor's Application for Payment in order to obtain properly allocated and manageable Subcontractor Application for Payments.
  - .3 The Construction Manager's Application for Payment shall constitute a representation to the Authority, based on the Construction Manager's determinations and inspections at the Project Site as provided herein and on the data comprising the Subcontractors' Applications for Payment, that, to the best of the Construction Manager's knowledge, information and belief, the Construction Manager's Work has progressed to the point indicated and the quality of the Work is in

accordance with the Contract Documents. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to minor deviations from the Contract Documents correctable prior to completion and to specific qualifications expressed by the Construction Manager. The issuance of the Construction Manager's Application for Payment shall further constitute a representation by the Construction Manager that the Subcontractor's Work is complete for that Phase and in conformance with the Contract Documents and that the Subcontractors are entitled to payment in the amount certified. The Construction Manager shall provide the Authority with a full Lien Waiver for costs included with each Application for Payment.

- The issuance of an Application for Payment shall be a representation that the Construction Manager has (1) made continuous on-site inspections to check the quality and quantity of the Work, (2) reviewed construction means, methods, materials, techniques, sequences, and procedures for the Subcontractors' Work, (3) reviewed copies of requisitions received from Subcontractors and Suppliers and other data requested by the Authority to substantiate the Subcontractor's right to payment, (4) confirmed that Subcontractors have properly applied and accounted for payments previously made, and (5) determined that there are no outstanding Claims by the Construction Manager or its Subcontractors, for which the Authority has not previously received written notice according to the requirements of the Construction Services Agreement.
- **2.3.11** The Construction Manager shall review the safety programs developed by each of the Subcontractors for coordination with the safety program administered by the Construction Manager.
- **2.3.12** The Construction Manager shall determine that the Work of each Subcontractor is being performed in accordance with the requirements of the Contract Documents, and shall guard the Authority against defects and deficiencies in the Work. As appropriate, the Construction Manager shall have authority, upon written authorization from the Authority, to require additional inspection or testing of the Work in accordance with the provisions of the Contract Documents, whether or not such Work is fabricated, installed or completed. The Construction Manager shall not permit Work that does not conform to the requirements of the Contract Documents.
- **2.3.13** The Construction Manager shall schedule and coordinate the sequence of construction in accordance with the Contract Documents and the latest approved Construction Schedule.
- **2.3.14** The Construction Manager shall transmit to the Architect requests for interpretations of the meaning and intent of the Drawings and Specifications in the form of a request for information (**"RFI"**), and assist in the resolution of questions that may arise.
- **2.3.15** The Construction Manager shall review requests for Changes or Change Orders, assist in negotiating Subcontractor proposals, submit recommendations to the Authority, and, if they are accepted, prepare Change Orders and Construction Change Directives which incorporate the Architect's modifications to the Construction Documents.
- **2.3.16** The Construction Manager shall assist the Authority in the review, evaluation and documentation of Claims.

.4

- **2.3.17** The Construction Manager is responsible to administer all insurance requirements of the Construction Manager and Subcontractors involved in the performance of the Work. The Construction Manager shall be responsible to verify and administer insurance requirements for all aspects of construction.
- **2.3.18** In collaboration with the Architect, the Construction Manager shall establish and implement procedures for expediting the processing and approval of Shop Drawings, Product Data, Samples, and other submittals. The Construction Manager shall review and approve all Shop Drawings, Product Data, Samples, and other submittals from the Subcontractors. The Construction Manager shall coordinate submittals with information contained in related documents and transmit to the Architect those that have been approved by the Construction Manager. The Construction Manager's actions shall be taken with such reasonable promptness as to cause no delay in the Work or in the activities of the Authority or Subcontractors.
- **2.3.19** The Construction Manager shall record the progress of the Work. The Construction Manager shall submit written progress reports to the Authority at least once a month including information on each Subcontractor's Work, as well as the entire Project, showing percentages of completion. The Construction Manager shall keep and transmit to Authority a daily log containing a record of weather, each Subcontractor's Work on the Project Site, number of workers, identification of equipment, Work accomplished, problems encountered, and other similar relevant data as the Authority may require.
- 2.3.20 The Construction Manager shall maintain at the Project Site for the Authority one record copy of all Subcontracts, Drawings, Specifications, Addenda, Change Orders and other Modifications, in good order and marked currently to record changes and selections made during construction, and in addition, approved Shop Drawings, Product Data, Samples and similar required submittals. The Construction Manager shall maintain records, in duplicate, of principal building layout lines, elevations of the bottom of footings, floor levels and key elevations certified by a qualified surveyor or professional engineer. The Construction Manager shall make all such records available to the Architect upon completion of the Project shall deliver them to the Authority. The Construction Manager is responsible to prepare all As-Built Drawings for the Project in an electronic format acceptable to the Authority. The Construction Manager shall provide CADD based As-Built Drawings for any Work that was planned or detailed using CADD.
- **2.3.21** The Construction Manager shall arrange for the delivery, storage, installation, protection, and security of Authority-purchased materials, systems, and equipment that are a part of the Project.
- **2.3.22** With the Architect and the Authority's designated representative(s), the Construction Manager shall observe the Subcontractors' final testing and start-up of utilities, operational systems and equipment.
- **2.3.23** When the Construction Manager considers each Subcontractor's Work or a designated portion thereof substantially complete, the Construction Manager shall, jointly with the Subcontractor, prepare for the Authority a list of incomplete or unsatisfactory items and a schedule for their completion. The Construction Manager shall assist the Authority in conducting inspections to determine whether the Work or designated portion thereof is substantially complete.
- **2.3.24** The Construction Manager shall coordinate the correction and completion of the Work. Following issuance of a Certificate of Substantial Completion of the Work or a designated portion thereof, the Construction Manager shall evaluate the completion of the Work of the Subcontractors and make recommendations to the Authority when the Work

is ready for final inspection. The Construction Manager shall assist the Authority in conducting final inspections.

- **2.3.25** The Construction Manager shall secure and transmit to the Authority warranties and similar submittals required by the Contract Documents for delivery to the Authority and deliver all keys, manuals, record drawings and maintenance stocks to the Authority. The Construction Manager shall forward to the Authority a final Application for Payment upon compliance with the requirements of the Contract Documents.
- **2.3.26** Duties, responsibilities and limitations of authority of the Construction Manager as set forth in the Contract Documents shall not be restricted, modified or extended without written consent of the Authority.

# EXHIBIT 2 [RESERVED]

Exhibit 2 is reserved.

# **EXHIBIT 3**

## CONSTRUCTION MANAGEMENT PLAN INCLUDING CONSTRUCTION MANAGER'S GUARANTEED MAXIMUM PRICE AND CONSTRUCTION SCHEDULE

This **Exhibit 3** constitutes the Construction Manager's Construction Management Plan in which the Construction Manager proposes and commits pursuant to the terms of this **Exhibit 3** to a GMP to perform all the Work required in and by the Contract Documents not only as they currently exist, but also as they will be developed in the future by the Architect, Authority, and Team as the final Construction Documents are completed. This **Exhibit 3** also incorporates by reference the Construction Schedule by which the Construction Manager will perform the Work pursuant to critical Milestone Dates, the Date of Substantial Completion, and the Date of Final Completion. Once the Authority and Team accept and execute this **Exhibit 3** after it has been completed and executed by the Construction Manager, this **Exhibit 3** will become part of the Construction Services Agreement as a Contract Revision.

The GMP as set forth below is guaranteed by the Construction Manager in accordance with Article 5 of this Construction Services Agreement:

## 1. GUARANTEED MAXIMUM PRICE (GMP)

The Construction Manager hereby certifies that it will complete the Work for an amount equal to or less than a guaranteed maximum price (GMP) of \$ [INSERT] Dollars.

## 2. COST OF THE WORK

The Cost of the Work are the costs to be reimbursed as described in **Paragraph 8.1** of the Construction Services Agreement.

## 3. CONSTRUCTION MANAGER FEE

The Construction Manager's Fee is described below in this **Exhibit 3**. The Authority and Construction Manager agree that the Construction Manager's Fee may be amended from time to time as mutually agreed to by the Authority and Construction Manager.

- **3.1** The Construction Manager's Fee is included as part of the GMP and is considered a Cost of the Work to be Reimbursed in accordance with **Subparagraph 8.1.1** of the Construction Services Agreement and as outlined below:
- **3.2** The Construction Manager's Fee for Construction Phase Services is Twelve Million Five Million Thousand (\$12,500,000) Dollars up to and until the GMP would exceed \$740,000,000. The Construction Manager's Fee is a capped amount. The Construction Manager has agreed to this fixed fee which shall not be increased or decreased for changes in the scope of the Work involving increases to the GMP; **provided**, **however**, that
  - .1 Up to and until the GMP would exceed \$740,000,000, the Authority may assign Trade Contracts or Vendor Contracts to the Construction Manager without an adjustment to Construction Manager's Fee for Construction Phase Services, but if the assigned Cost of the Work would cause the GMP to exceed \$740,000,000 in the aggregate,, then the Construction Manager shall be entitled to an equitable adjustment to its Fee of 1.95 % of the value of the Trade Contracts or Vendor Contracts assigned in excess such amount.

.2 If the Authority orders changes to the scope of the Work increasing the GMP to more than Seven Hundred Forty Million Dollars (\$740,000,000) in aggregate, then the Construction Manager shall be entitled to an adjustment to its Fee of 1.95 % of the value of the increase in the GMP in excess of such amount. But, if the total aggregate value of the Changes in the Work is less than such amount, then the Construction Manager shall not be entitled to an increase of the Construction Manager's Fee.

## 4. GENERAL CONDITIONS

The GMP shall include an aggregate line item for Construction Manager's General Conditions which shall itself be a Guaranteed Maximum Price for that line item (the "General Conditions GMP" or "GCGMP"). Pursuant to Subparagraph 5.1.1.1 of the Construction Services Agreement, the GCGMP is a capped amount of Forty Three Million Eight Hundred Seventy-three Thousand Four Hundred and Sixty-two Dollars (\$43,873,462), as such amount may be adjusted pursuant to the Agreement. General Conditions include the items listed in <u>Exhibit 14</u> of the Construction Services Agreement.

## 5. CONSTRUCTION MANAGER SELF PERFORMED WORK

If and when appropriate and as approved by the Authority and Team, the Construction Manager may be allowed to perform a portion of the Work with its own forces as provided pursuant to **Exhibit 6** hereto, in which case the Construction Manager will provide either an agreed upon lump sum price ("Self Performed Lump Sum" or "SPLS") or a Guaranteed Maximum Price for that line item (the "Self Performed GMP" or "SPGMP"). Construction Manager will not apply for payment for any Cost of the Work that it incurs in excess of the SPLS or SPGMP or be entitled to any increase in the SPLS or SPGMP except for the same reasons that would otherwise entitle Construction Manager to a Contract Revision under this Construction Services Agreement increasing the GMP.

## 6. CONSTRUCTION MANAGER CONTINGENCY

Construction Manager's GMP contains a Construction Manager Contingency to cover the risks inherent in providing a GMP for the Work. Construction Manager may not use the Construction Manager Contingency established for the GMP to fund any Cost of the Work applicable to and in excess of the GCGMP or SPGMP, if any; Construction Manager acknowledges that the GCGMP and the SPGMP each have their own contingencies that are already part of the GCGMP and SPGMP and that are separate from the Construction Manager Contingency. Up to a GMP of \$740,000,000, the Construction Manager Contingency shall be the lesser of four and one half percent (4.5%) of the Cost of the Work or Thirty Million Dollars (\$30,000,000).

- **6.1** The Authority and Construction Manager acknowledge and agree that a fundamental premise of the Construction Services Agreement is to complete the Work in an amount that will not exceed the GMP. The Construction Manager acknowledges and agrees it has had adequate time to review the scope of the Project and the Work, the Project Site, existing conditions, market conditions, Contract Documents, and other conditions and factors that may impact the GMP. The Construction Manager acknowledges and agrees that the Construction Manager Contingency provides an adequate means of accounting for the foreseen and unforeseen costs associated with the completion of the Work including those additional costs required as a result of the completion of the final Construction Documents by the Architect.
- 6.2 The Construction Manager agrees and acknowledges that, as of the date that it executes this <u>Exhibit 3</u>, all the Design Documents necessary for the Work have not been finalized, but Construction Manager agrees that the GMP Pricing Documents are sufficiently complete to enable Construction Manager to guarantee and agree to perform its Work for the GMP based on the final Construction Documents that will be developed by the Architect. Construction Manager knows and understands the Program, understands that

the GMP Pricing Documents are intended to satisfy the Program, and agrees that it can timely complete the Work contemplated by the Program as revised for the GMP, based on the GMP Pricing Documents, and as required by the final Construction Documents. In determining the GMP and Contract Time, the Construction Manager has taken into account the level of completeness of the GMP Pricing Documents and has exercised the best skill and efforts of the Construction Manager to make (1) all appropriate judgments and inferences about how the Documents will be completed as Construction Documents based on the Program and Applicable Law, and (2) all inquiries of and discussions with the Authority, Team, and Architect as necessary in the Construction Manager's judgment to clarify the content and intent of the GMP Pricing Documents and the expected and intended scope of the Construction Documents in order to reliably calculate and establish both the GMP and the Contract Time.

- **6.3** By executing this **Exhibit 3**, the Construction Manager agrees that the Contract Documents (including the GMP Pricing Documents), materials, and information furnished to the Construction Manager as of the date of the execution of said **Exhibit 3** and the ongoing discussions and meetings between the Construction Manager and the Authority, Team, and the Architect up to said execution have described the scope, construction requirements, and the expected and intended scope of the Work in detail sufficient to enable the Construction Manager to certify the GMP and the Construction Schedule, including the Date of Substantial Completion. The Construction Manager shall not be permitted to claim any adjustment in either the GMP or Contract Time in connection with the completion of final Construction Documents, unless allowed pursuant to **Article 14**.
- **6.4** The purpose of the negotiated Construction Manager Contingency is to allow the Construction Manager to cover foreseen and unforeseen costs in the performance of the Pre-Construction Phase Services and Construction Phase Services and the Work for which the Construction Manager is not entitled to an increase in the GMP under this Agreement. The Construction Manager has assumed the risk hereunder that the Construction Manager Contingency is sufficient to cover these foreseen and unforeseen costs and, accordingly, the Construction Manager shall not be entitled to an increase in the GMP in the event that the Construction Manager incurs such costs in excess of the Construction Manager Contingency.
- **6.5** For illustrative purposes only, the Construction Manager acknowledges and agrees that the Construction Manager Contingency may be used by the Construction Manager, with approval of the Authority, to cover costs foreseen and unforeseen costs that could include the following:
  - .1 costs associated with the completion of the Contract Documents, including clarifications in scope of the Contract Documents, by the Architect that may not have been anticipated when calculating the GMP;
  - .2 costs associated with the development or completion of Submittals that may not have been anticipated when calculating the GMP;
  - .3 finalization, clarification, and/or coordination of the Contract Documents consistent with the intent of the Authority;
  - .4 costs of any Construction Phase Services inadvertently omitted by the Construction Manager when calculating the GMP;
  - .5 additional costs incurred as a result of the buyout of any Subcontractor in excess of the Schedule of Values;

- .6 costs associated with default by a Subcontractor;
- .7 costs related to changes in construction means and methods of construction;
- .8 costs of the Construction Manager and other unanticipated costs or cost overruns in connection with the performance of the Work ;
- .9 costs associated with accelerating or re-sequencing the Work as necessary to maintain the Construction Schedule; and
- .10 any costs associated with price escalation in materials or labor rates.
- **6.6** The Construction Manager understands and agrees that according to the Act, the Authority's contribution to the GMP cannot exceed a certain amount and that the Authority can only accept financial obligations relating to cost overruns associated with the construction of the Project provided that any cost overruns are the responsibility of the Construction Manager and not the Authority; as a consequence, the Construction Manager agrees, represents, and warrants that the Construction Manager's Contingency is intended to cover any and all cost overruns arising out of or related to its performance of the Work and that the Construction Manager will not make any Claims for cost overruns as the risk of any such overruns are covered by and assumed in the Construction Manager's Contingency.
- **6.7** The Authority will continue to evaluate possible cost reduction opportunities that may provide additional money for the Construction Manager's Contingency. Such cost reduction opportunities and resulting decisions shall be at the sole and absolute discretion of the Authority. The Construction Manager's Schedule of Values must allocate all uncommitted costs to the Construction Manager Contingency. Any such cost reduction opportunities acceptable to the Authority will further fund the Construction Manager Contingency up to an additional amount not to exceed Six Million Dollars (\$6,000,000) but not to exceed a maximum contingency amount of Thirty-Six Million Dollars (\$36,000,000). Any amount in excess of the maximum allowed Construction Manager Contingency will be immediately deducted from the GMP by Contract Revision. The Construction Manager hereby agrees to release to the Authority amounts in the Construction Manager Contingency as set forth in its GMP Proposal, which at a minimum shall require:
  - .1 The Construction Manager will establish a One and One Half Million Dollar sub-contingency within the Construction Manager Contingency (without limiting the use of the remainder of the Construction Manager Contingency) that is to be used to address conditions relating to demolition, excavation, and caissons that do not qualify as a differing site condition pursuant to **Subparagraph 6.2.7** of **Exhibit 6** to the Construction Services Agreement. The remainder of One and One Half Million Dollar (\$1,500,000) contingency that is not used to address conditions relating to demolition, excavation, and caissons that do not qualify as a differing site condition pursuant to **Subparagraph 6.2.7** of **Exhibit 6** to the Construction pursuant to **Subparagraph 6.2.7** of **Exhibit 6** to the Construction services Agreement shall be released to the Authority after the conclusion of demolition, excavation, and deep caissons; and
  - .2 100% of Construction Manager Contingency (net of the sub-contingency referenced in **Subparagraph 6.7.1** hereof) over 4.25% of the Cost of the Work on that date Forty-five (45) days after delivery by the Architect of the final Construction Documents; and
  - .3 100% of Construction Manager Contingency (net of the sub-contingency referenced in **Subparagraph 6.7.1** hereof) over 3.5% of the Cost of the Work

on the earlier of that date when Ninety Percent (90%) of the Work to be performed by Subcontractors has been awarded or One Hundred Twenty (120) days after delivery by the Architect of the final Construction Documents; and

- .4 100% of Construction Manager Contingency over 2.5% of the Cost of the Work on the earlier of completion of the superstructure or July 1, 2015; and; and
- .5 100% of Construction Manager Contingency over 2% of the Cost of the Work at Substantial Completion (or earlier as agreed to by Authority and Construction Manager); and
- .6 100% of Construction Manager Contingency over 0.5% of the Cost of the Work at Final Completion.

Construction Manager Contingency released pursuant to this <u>**Paragraph 6.7**</u> shall reduce the GMP by a sum equal to the amount released. Said reduction of the GMP shall be implemented by Contract Revision.

## 7. COMPENSATION ADJUSTMENTS

The Construction Manager's Fee shall be adjusted for Performance Compensation as follows:

The Construction Manager's Fee may be decreased \$250,000 for each individual as described below if the Construction Manager does not maintain the continuous involvement of the following personnel on the Project, as determined solely by the Authority, substantially in accordance with **Exhibit 5** hereto:

- 1. John Wood
- 2. Ken Sorensen
- 3. Brent Leiter
- 4. Greg Clark
- 5. Dave Mansell
- **6.** Steve Laurila

The decrease in Construction Manager Fee described above will not apply if the employees described above discontinue employment with the Construction Manager. The Authority shall have the absolute right to approve any replacement of key staff of the Construction Manager.

The Construction Manager may be compensated for performance fees as follows in the reasonable discretion of the Authority based on the Construction Manager's performance:

- 7. One Million Dollars (\$1,000,000) for achieving Substantial Completion pursuant to the Construction Schedule and at or under the GMP.. This performance fee will only be considered if no liens, litigation, or arbitration proceedings exist at the Final Completion of the Work.
- **8.** Five Hundred Thousand Dollars (\$500,000) if the Construction Manager completes the Work so as to achieve a loss ratio jointly determined and agreed to by the Authority and the Authority's insurance broker.
- **9.** Two Hundred Fifty Thousand Dollars (\$250,000) if Construction Manager achieves or exceeds the Workforce goals in the Equity Plan for the Construction Services Agreement.
- **10.** Two Hundred Fifty Thousand Dollars (\$250,000) if Construction Manager achieves or exceeds the Targeted Business goals in the Equity Plan for the Construction Services Agreement.

- Five Hundred Thousand Dollars (\$500,000) if the Construction Manager qualifies for and the Authority awards the Construction Manager all other performance fees in Subsections 7.7, 7.8, 7.9, and 7.10 hereof.
- 12. Performance fees, if any, shall be paid only to the extent that sufficient funds remain in the Construction Manager Contingency immediately prior to Final Completion (and prior to the reduction of the Contingency described in Paragraph 6.7.6 herein) to pay said performance fees. Additionally, these performance fees will be made with further reasonable consideration given to quality of construction, job-site safety, and the Construction Manager's ability to avoid Claims in completing the Work against the Authority or Team or third party claims against the Authority or Team.

# 8. ALLOWANCES

Allowances will be included in the GMP as described in **Attachment 1** to this **Exhibit 3**. Construction Manager represents that allowances are reasonably accurate cost estimates, but not guaranteed. The Guaranteed Maximum Price will be adjusted for any items as an allowance when the actual cost for each allowance item is determined. For items listed as allowances, where the allowance is greater than the actual Cost of the Work or where the allowance is less than the actual Cost of the Work, the Guaranteed Maximum Price will be decreased or increased respectively by the amount of the difference of some or all of the allowances in the aggregate, as applicable, through a Contract Revision. Any cost savings below the amounts of Allowances described above shall be returned to the Authority and not the Construction Manager Contingency. Before an allowance is exceeded, the Authority and Team shall be given written notice, and Owner and Construction Manager shall cooperate to explore options of obtaining the desired work for the allowance price.

# 9. ALTERNATES

Construction Manager has reviewed and developed pricing for the Alternates identified in <u>Attachment 2</u> of this **Exhibit 3**, and said Attachment shall identify the amount that shall be added to or deducted from the GMP if the Authority elects to make the changes described in the Alternates. All requirements of the Specifications shall govern the Alternate Work, except as specifically modified herein. The Work required under the Alternates shall be complete, and the Construction Manager has included the work of all trades necessary to complete the Alternate Work

# 10. UNIT PRICES

Unit prices, if any are set forth in the "Schedule of Unit Prices" attached hereto as <u>Attachment 3</u>, if applicable. Such unit prices are considered complete and include (1) all materials, equipment, labor, delivery, installation, overhead, and profit and (2) any other costs or expenses in connection with, or incidental to, the performance of that portion of the Work to which unit prices apply.

# 11. SCHEDULE OF VALUES

The Schedule of Values in <u>Attachment 4</u> to this <u>Exhibit 3</u> allocates the various portions of the Work and identifies the individual line items that have either been awarded as self-perform work or capped.:

# **12.** CONSTRUCTION SCHEDULE

The Construction Schedule by which Construction Manager will perform the Work for the GMP is attached hereto as **Attachment 5**. The Construction Manager hereby certifies that it shall achieve Substantial Completion of the Work on or before [INSERT DATE]. The Construction Manager shall be deemed to have achieved Final Completion of the Work only upon the satisfactory completion of all the Work as described in the Contract Documents.

# 13. CONTRACT DOCUMENTS AND ASSUMPTIONS UPON WHICH THE GMP IS BASED

<u>Attachment 6</u> to this <u>Exhibit 3</u> contains the Contract Documents upon which the GMP is based and is incorporated into this <u>Exhibit 3</u> for all purposes. <u>Attachment 7</u> to this <u>Exhibit 3</u> contains the Assumptions upon which the GMP is based and is incorporated into this <u>Exhibit 3</u> for all purposes.

This <u>Exhibit 3</u> is Agreed to and Accepted as a Contract Revision to the Construction Services Agreement this \_\_\_\_\_\_, of \_\_\_\_\_\_, 2013 by and between:

The Minnesota Sports Facilities Authority

By:\_\_\_\_\_

Its:\_\_\_\_\_

The Minnesota Sports Facilities Authority

By:\_\_\_\_\_

Its:\_\_\_\_\_

The Construction Manager

By: \_\_\_\_\_

Its:\_\_\_\_\_

## EXHIBIT 4 MASTER PROJECT SCHEDULE

The Construction Manager shall perform its Work expeditiously and consistent with its contractual obligations and Standard of Care to further the orderly progress of the Work. The Construction Manager's Pre-Construction Phase Services shall be commenced, subject to **Paragraph 2.3** of the Construction Services Agreement, on the date hereof, and, subject to authorized adjustments and excusable delays as allowed by **Paragraph 4.3**, Construction Manager shall achieve Project Milestone Dates and Substantial Completion in accordance with this **Exhibit 4** of this Construction Services Agreement.

Final Completion of the Work shall be deemed to have occurred only after completion of all the Work and acceptance of it by the Authority and Architect.

The Date of Substantial Completion is described in more detail below:

#### Substantial Completion

July 1, 2016

Project Milestones Dates of the Work that must be complete in accordance herewith are outlined on the Outline of Construction Schedule below.

The following Outline of Construction Schedule highlights critical components of the Project and mandatory Milestone Dates that must be completed, without exception, by the Construction Manager in order to meet the requirements of the Construction Schedule and Substantial Completion. The Parties hereto acknowledge and agree, in accordance with **Paragraph 4.10** of **Exhibit 6** to the Construction Services Agreement, that the Construction Manager will undertake Extraordinary Measures if the Authority determines that the performance of the Work has not progressed or reached the level of completion required by the Contract Documents.

# **OUTLINE OF CONSTRUCTION SCHEDULE**

# **PROJECT MILESTONE ACTIVITIES**

## **COMPLETION DATE**

[INSERT]

[INSERT]

EXHIBIT 5 PROJECT STAFFING
-------------------------------

The following Project Staffing diagram is included in this Construction Services Agreement as **Exhibit 5**.

Title	Name of	2013	2014	2015	2016											
1120	Individu al	Rate/Hr	Rate/Hr	Rate/Hr	Rate/Hr	2013	2013	2013	2013	2013	2013	2013	2013	2013	2013	2
			3.0%	3.0%	3.0%	Jan	Feb	Mar	Apr	May	June	July	Avg	Sept	Oct	,
reconstruction Services																
esign Phase Manager	Jason Hopper						100%	100%	100%	100%	100%	100%	100%	100%	100%	1
ead Estimator	Eric Kloss						100%	100%	100%	100%	100%	100%	100%	100%	100%	1
rchitectural Estimator	David Green						100%	100%	100%	100%	100%	100%	100%	100%	100%	1
echanical Estimator	Tom Sheehan					<del> </del>	100%	100%	100%	100%	100%	100%	100%	100%	100%	1
ectrical Estmator	Tom McCarthey					<u> </u>	100%	100%	100%	100%	100%	100%	100%	100%	100%	1
ructural Estimator	Jim Block						100%	100%	100%	100%	100%	100%	100%	100%	100%	1
cheduling Manager	Tim Johnson					<u> </u>	100%	100%	100%	100%	100%	100%	100%	100%	100.70	<u> </u>
roject Scheduler	Matt London						100%	100%	100%	100%	100%	100%	100%	100%	<u> </u>	⊢
r. Construction Manager	Kevin Dalager					<u> </u>	50%	50%	50%	100%	100%	100%	100.70	100.74	<u> </u>	⊢
reject Engineer	TBD						50%	50%	50%	50%	50%	50%	50%	50%	<u> </u>	⊢
reconstruction Community Participation																
ommunity Participation Director	Lynn Litlejohn						50%	50%	50%	50%	50%	50%	50%	50%		
ommunity Participation Director	Ravi Norman						25%	25%	25%	25%	25%	25%	25%	25%		$\vdash$
ommunicity Participation Coordinator	Jessie Volal															$\square$
roject Executive Management																
rincipal-in-Charge	John Wood	250.00	257.50	265.23	273.18		80%	80%	80%	80%	80%	80%	80%	80%	80%	
onstucton Executve	Brent Leiter	165.00	169.95	175.05	180.30		100%	100%	100%	100%	100%	100%	100%	100%	100%	1
eneral Superintendent	Dave Mansel	155.00	159.65	164.44	169.37		100%	100%	100%	100%	100%	100%	100%	100%	100%	1
rector of Operations	Allen Troshinsky	178.00	181.28	186.72	192.32		20%	20%	20%	20%	20%	20%	20%	20%	20%	
<b>DC</b>																
DC Services Manager	Ricardo Khan	115.00	118.45	122.00	125.66										25%	
enior VDC Coordinator	Michai Wojtak	90.00	92.70	95.48	98.35										100%	1
DC Coordinator	Eric Keleny	90.00	92,70	95.48	98.35								100%	100%	100%	1
DC Coordinator	Nathan House	80.00	82,40	84.87	87.42											
alety .																
afety Manager	Jennifer Hume-Jorgeson	96.00	97.85	100.79	103.81											1
afety Engineer	Pedro Lopez	70.00	72.10	7426	76.49											Г
ublic Safety Engineer	Dalton Harrison	70.00	72.10	74.26	76.49											
wellty																
ually Manager	Mke Mason	95.00	97.85	100.79	103.81										100%	
EP/Commissioning Manager	Craig Sharpe	100.00	103.00	108.09	109.27											
ually Engineer	Greg Lien	80.00	82.40	84.87	87.42											
roject Management Controls																
reject Controis Manager	Joe Prouix	100.00	103.00	108.09	109.27											Γ
reject Scheduler	Matt London	80.00	82.40	84.87	87.42										100%	Г

67

Page 1 of 12

FEBRUARY 15, 2012				Exhib	vit 5											
Title	Name of	2013	2014	2015	2016											
	Individu al	Rate/Hr	Rate/Hr	Rate/Hr	Rate/Hr	2013	2013	2013	2013	2013	2013	2013	2013	2013	2013	2013
Office Manager	Karen Wilenbring	45.00	46.35	47.74	49.17								70%	100%	100%	100%
Project Accountant	Lloyd Kimbail	75.00	77.25	79.57	81.95								50%	60%	100%	100%
Accounting Clark/Admin	TBD	40.00	41.20	42.44	43.71											50%
Administrative Assistant	TBD	40.00	41.20	42.44	43.71											
ReceptionistAdmin	TBD	30.00	30.90	31.83	32,78											
Community Particip ation																
Community Participation Manager	Lynn Littlejchn	100.00	103.00	108.09	109.27										50%	50%
Community Participation Manager	Ravi Norman	105.00	108.15	111.39	114.74										88 %	25%
Community Participation Coordinator	Khailah Chandler	50.00	51.50	53.05	54.64										100%	100%
Structural / Site																
Sr Construction Manager	Kevin Dalager	115.00	118.45	122.00	125.66								100%	100%	100%	100%
Structural Project Manager	Greg Huber	105.00	108.15	111.39	114.74								100%	100%	100%	100%
Structural Superintendent	Scot Amundson	95.00	97.85	100.79	103.81											50%
Structural Project Engineer	Tom Clark	85.00	87.55	90.18	92.88											
PrecastProject Engineer	Ryan Res	80.00	82.40	84.87	87.42											
Site Project Engineer	Justin Utpadel	80.00	82,40	84.87	87.42											
Site SuperIntendent	Hannah Patterson	75.00	77.25	79.57	81.95											
Fleid Engineer	TBD	80.00	61.80	63.65	65.56											
Roof / Enclosure																
Sr. Construction Manager	Steve Laurila	115.00	118.45	122.00	125.66		20%	20%	20%	20%	20%	80%	100%	100%	100%	100%
Project Manager / Roof	Ron Stodoka	100.00	103.00	108.09	109.27									100%	100%	100%
Senior Superintendent	Jason Brown	115.00	118.45	122.00	125.66											
Superintendent	John Carter	95.00	97.85	100.79	103.81											
Project Engineer	Art Gardner	80.00	82,40	84.87	87.42											
Project Engineer	Lindsey Cabel	80.00	82.40	84.87	87.42											
Fleid Engineer	TBD	60.00	61.80	63.65	65.56											
Mech / Elec / Low Voltage Systems																
Construction Manager	Matt Nelson	110.00	113.30	116.70	120.20										100%	100%
Mechanical Project Manager	Kevin Boyce	100.00	103.00	108.09	109.27											$ \longrightarrow $
MEP Project Engineer	Brian Windingland	80.00	82.40	84.87	87.42											$ \longrightarrow $
Electrical Project Manager	Dave McNeal	100.00	103.00	108.09	109.27											
Interfors																
Construction Manager	Gabe Stofferahn	100.00	103.00	108.09	109.27											$ \longrightarrow $
Project Manager	Chad Passch	100.00	103.00	108.09	109.27											$ \longrightarrow $
Project Manager	Joe Pritzkow	100.00	103.00	106.09	109.27											$\vdash$
Food Service/FF&E	Leslie Seppamaki	100.00	103.00	108.09	109.27											$ \longrightarrow $
Area Superintendent	Brendan Moore	95.00	97.85	100.79	103.81											<b>└──┤</b>
Area Superintendent	Tom Schwab	95.00	97.85	100.79	103.81											$ \longrightarrow $
Area Superintendent	John Oliver	95.00	97.85	100.79	103.81											$\vdash$
Area Superintendent	Martin Mathery	115.00	118.45	122.00	125.66											$ \longrightarrow $
Area Superintendent	Russ Brown	95.00	97.85	100.79	103.81											$ \longrightarrow $

68

Page 2 of 12

FEBRUARY 15, 2012				Exhib	it 5											
Title	Name of Individual	2013 Rate/Hr	2014 Rate/Hr	2015 Rate/Hr	2016 Rate/Hr	2013	2013	2013	2013	2013	2013	2013	2013	2013	2013	2013
Sports Systems																
Construction Manager	Heid Brown	110.00	113.30	118.70	120.20											
Seating/Playing Field Project Manager	Brad Hjernvick	100.00	103.00	108.09	109.27											
Scoreboard and Signage Project Manager	Emily Roland	95.00	97.85	100.79	103.81											
General Staffing Rates																
Design Phase Manager		105.00	108.15	111.39	114.74	1										
Sr. Estimator		105.00	108.15	111.39	114.74	1										
Estimator		85.00	87.55	90.18	92.88											
Scheduling Manager		80.00	82.40	84.87	87.42											
Associate Counsel		237.00	244.11	251.43	258.98	]										
Vice President, Senior Counsei		253.00	260.59	268.41	276.46	]										
Director of Planning & Scheduling		124.00	127.72	131.55	135.50	]										
Director of Quality		124.00	127.72	131.55	135.50	1										
Director of Salety		124.00	127.72	131.55	135.50	1										
Manager of Environmental Health & Safety		114.00	117.42	120.94	124.57	1										
Document Control Engineer		60.00	61.80	63.65	65.56	1										
OCIP Manager		75.00	77.25	79.57	81.95	1										
Asst. Project Manager		85.00	87.55	90.18	92,88	1										
Sr. Project Manager		105.00	108.15	111.39	114.74	1										
TSO Engineer		80.00	82,40	84.87	87.42	1										
Integrated Construction Manager/VDC/BIM Mgr.		90.00	92.70	95.48	98.35	1										
Director of Integrated Construction		115.00	118.45	122.00	125.66	1										

NOTES: 1. Individuals shall be billed on an hearly basis not to exceed 40 hears per week. 2. Hearly rates and individuals are subject to Authority approval consistent with Article 6 of the CSA.

69

Page 3 of 12

FEBRUARY 15, 2012					Exh	nibit 5												
Title	Name of																	
	Individual	2013	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2015	2015	2015	2015
		Dec	Jan	Feb	Mar	Apr	May	June	July	Aug	Sept	Oct	Nov	Dec	Jan	Feb	Mar	Apr
Preconstruction Services																		
Design Phase Manager	Jason Hopper	100%	100%	100%	100%													
Lead Estimator	Eric Kloss	100%	100%	100%	100%	100%	100%	100%										
Architectural Estimator	David Green	100%																
Mechanical Estimator	Tom Sheehan	100%																$ \rightarrow $
Electrical Estimator	Tom McCartney	100%																<b>⊢</b> − +
Structural Estimator	Jim Block	100%																
Scheduling Manager	Tim Johnson																	<b>├</b> ──┦
Project Scheduler	Mett London																	
Sr. Construction Manager	Kevin Dalager	<u> </u>																
Project Engineer	TBD																	
Preconstruction Community Participation																		
Community Participation Director	Lynn Littleichn																	
Community Participation Director	Ravi Norman																	
Communicity Participation Coordinator	Jessie Volal																	
Project Executive Management																		
Principal-in-Charge	John Wood	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%
Construction Executive	Brent Leiter	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
General Superintendent	Dave Mansel	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Director of Operations	Allen Troshinsky	20%																
VDC																		
VDC Services Manager	Ricardo Khan	25%	25%	25%	25%	25%	25%	25%										
Senior VDC Coordinator	Mohal Wojtak	100%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%
VDC Coordinator	Eric Keleny	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
VDC Coordinator	Nathan House		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Salety																		
Safety Manager	Jannifer Hume-Jargeson	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Safety Engineer	Pedro Lopez			100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Public Safety Engineer	Dalton Harrison			100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Quality																		
Qualty Manager	Mke Mason	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
MEP/Commissioning Manager	Craig Sharpe																	
Qually Engineer	Greg Lien			75%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Project Management Controls																		
Project Controls Manager	Joe Prouix		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Project Scheduler	Matt London	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Accounting and Administration																		

70

Page 4 of 12

FEDROART 15, 2012		_			100	inone o												
Title	Name of																	
	Individual	2013	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2015	2015	2015	2015
Office Manager	Karen Wilenbring	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Project Accountant	Lloyd Kimbail	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Accounting Clark/Admin	TBD	50%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Administrative Assistant	TBD		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Reception is tA dmin	TBD			100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Community Participation																		
Community Participation Manager	Lynn Littlejchn	50%	50%	50%	50%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%
Community Participation Manager	Ravi Norman	25%	25%	25%	25%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%
Community Participation Coordinator	Khailah Chandler	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Structural / Site																		
Sr .Construction Manager	Kevin Dalager	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Structural Project Manager	Greg Huber	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Structural Superintendent	Scott Amundson	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Structural Project Engineer	Tom Clark		50%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
PrecastProject Engineer	Ryan Res		50%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Site Project Engineer	Justin Utpadal		100%	100%	100%	100%	100%	100%										
Site SuperIntendent	Hannah Patterson		100%	100%	100%	100%	100%	50%										
Fleid Engineer	TBD		50%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Roof / Enclosure																		
Sr. Construction Manager	Steve Laurila	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Project Manager / Rod	Ron Stodoka	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Senior Superintendent	Jason Brown												100%	100%	100%	100%	100%	100%
Superintendent	John Carter														100%	100%	100%	100%
Project Engineer	Art Gardher		50%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Project Engineer	Lindsey Cabel		50%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Fleid Engineer	TBD		50%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Mech / Bec / Low Voltage Systems																		
Construction Manager	Matt Nelson	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Mechanical Project Manager	Kevin Boyce		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
MEP Project Engineer	Brian Windingland		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Electrical Project Manager	Dave McNeal		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Interfors																		
Construction Manager	Gabe Stofferahn		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Project Manager	Chad Pasech														100%	100%	100%	100%
Project Manager	Joe Pritzkow														100%	100%	100%	100%
Food Service/FF&E	Leslie Seppamaki																	
Area Superintendent	Brendan Moore																	
Area Superintendent	Tom Schwab	1																
Area Superintendent	John Oliver																	
Area Superintendent	Martin Mathery																	
Area Superintendent	Russ Brown																	
vide ouper montgene	Pues brown																	<b></b>

Exhibit 5

FEBRUARY 15, 2012

EE	D D I	JARY	15	201	12
	DIVU	- AL		203	

Exhibit 5

PEDROART 15, 2012					6,01	IIDIL 3												
Title	Name of Individual	2013	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2015	2015	2015	2015
Sports Systems	Incrysolati	2013	2014	2014	2014	2/14	4/14	2/14	2014	2014	2014	2014	2014	2014	2019	2019	2019	2013
Construction Manager	Heid Brown												100%	100%	100%	100%	100%	100%
Seating Playing Field Project Manager	Brad Hjernvick																	
Scoreboard and Signage Project Manager	Emily Roland																	
General Staffing Rates																		
Design Phase Manager																		
Sr. Estimator																		
Estimator																		
Scheduling Manager		T																
Associate Counsel		1																
Vice President, Senior Counsei		T																
Director of Planning & Scheduling		Τ																
Director of Quality		Т																
Director of Safety		T																
Manager of Environmental Health & Safety		Т																
Document Control Engineer		T																
OCIP Manager		Т																
Asst. Project Manager		1																
Sr. Project Manager		1																
TSO Engineer		1																
Integrated Construction Manager/VDC/BIM Mgr.		1																
Director of Integrated Construction		1																

NOTES: 1. Individuals shall be billed on an hearly basis not to exceed 40 hears per week. 2. Hearly rates and individuals are subject to Authority approval consistent with Article 6 of the CSA.

72

FEBRUARY 15, 2012					Exh	nibit 5												
Tide	Name of																	
	Individual	2015	2015	2015	2015	2015	2015	2015	2015	2016	2016	2016	2016	2016	2016	2016	2016	2016
		May	June	July	Avg	Sept	Oct	Nov	Dec	Jan	Feb	Mar	Apr	May	June	July	Aveg	Sept
Preconstruction Services																		
Design Phase Manager	Jason Hopper																	
Lead Estimator	Eric Kloss																	
Architectural Estimator	David Green																	
Mechanical Estimator	Tom Sheehan	<u> </u>																
Electrical Estimator	Tom McCartney	<u> </u>												<u> </u>				
Structural Estimator	Jim Block																	
Scheduling Manager	Tim Johnson	<u> </u>																
Project Scheduler	Matt London	<u> </u>																
Sr. Construction Manager	Kevin Dalager																	
Project Engineer	TBD																	
Preconstruction Community Participation																		
Community Participation Director	Lynn Littleichn																	
Community Participation Director	Ravi Norman																	
Communicity Participation Coordinator	Jessie Volal																	
Project Executive Management																		
Principal-in-Charge	John Wood	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	
Construction Executive	Brent Leiter	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
General Superintendent	Dave Mansel	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Director of Operations	Allen Troshinsky																	
VDC																		
VDC Services Manager	Ricardo Khan																	
Senior VDC Coordinator	Michai Wojtak	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%			
VDC Coordinator	Eric Keleny	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%			
VDC Coordinator	Nathan House	100%	100%	100%	100%	100%	100%	100%	100%	100%								
Safety																		
Safety Manager	Jennifer Hume-Jorgeson	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%			
Safety Engineer	Pedro Lopez	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	60%			
Public Safety Engineer	Dalton Harrison	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	60%			
Quality																		
Qually Manager	Mke Mason	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%			
MEP/Commissioning Manager	Craig Sharpe									50%	50%	100%	100%	100%	100%	100%		
Qually Engineer	Greg Lien	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Project Management Controls																		
Project Controls Manager	Joe Prouix	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Project Scheduler	Matt London	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%			
Accounting and Administration																		

73

Page 7 of 12

FEBRUARY 15, 2012					Exh	nibit 5												
Title	Name of																	
	Individual	2015	2015	2015	2015	2015	2015	2015	2015	2016	2016	2016	2016	2016	2016	2016	2016	2016
Office Manager	Karen Wilenbring	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Project Accountant	Lloyd Kimbail	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Accounting Clerk/Admin	TBD	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Administrative Assistant	TBD	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
ReceptionistAdmin	TBD	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Community Participation																		
Community Participation Manager	Lynn Littleichn	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%			
Community Participation Manager	Ravi Norman	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%			
Community Participation Coordinator	Khailah Chandler	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%			
Structural / Site																		
Sr. Construction Manager	Kevin Dalacer	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Structural Project Manager	Greg Huber	100%	50%															
Structural Superintendent	Scot Amundson	100%	50%															
Structural Project Engineer	Tom Clark	100%	50%															
Precast Project Engineer	Ryan Res	100%	100%	100%	100%	100%	100%	100%	100%	100%	50%							
Site Project Engineer	Justin Utpadal									100%	100%	100%	100%	100%	100%	100%		
Site Superintendent	Hannah Patterson											100%	100%	100%	100%	100%		
Fleid Engineer	TBD	100%	50%															
Roof / Enclosure																		
Sr. Construction Manager	Steve Laurila	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Project Manager / Roof	Ron Stodok a	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%							
Senior Superintendent	Jason Brown	100%	100%	100%	100%	100%	100%	100%	100%									
Superintendent	John Carter	100%	100%	100%	100%	100%	100%	50%										
Project Engineer	Art Gardher	100%	100%	100%	100%	100%	100%	100%										
Project Engineer	Lindsey Cabel	100%	100%	100%	100%	100%	100%	100%	100%	100%								
Fleid Engineer	TBD	100%	100%	100%	100%	100%	100%	100%	100%	100%								
Mech / Elec / Low Voltage Systems																		
Construction Manager	Matt Nelson	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Mechanical Project Manager	Kevin Boyce	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%		
MEP Project Engineer	Brian Windingland	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%		
Electrical Project Manager	Dave McNeal	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%		
Interfors																		
Construction Manager	Gabe Stofferahn	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Project Manager	Chad Passch	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Project Manager	Joe Pritzkow	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Food Service/FF&E	Lesle Sepamaki							100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Area Superintendent	Brendan Moor e						100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Area Superintendent	Tom Schwab			100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%		
	John Oliver			100.10	100.10	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	50%		
Area Superintendent Area Superintendent	Martin Mathery			100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	50%		

74

Page 8 of 12

FEBRUARY 15, 2012					Exh	ibit 5											
Title	Name of																
	Individual	2015	2015	2015	2015	2015	2015	2015	2015	2016	2016	2016	2016	2016	2016	2016	2016
Sports Systems																	
Construction Manager	Heid Brown	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Seating Playing Fleid Project Manager	Brad Hjernvick				100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Scoreboard and Signage Project Manager	Emily Roland				100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
General Staffing Rates																	
Design Phase Manager		1															
Sr. Estimator		Ι															
Estimator		Ι															
Scheduling Manager		T															
Associate Counsel		I															
Vice President, Senior Counsel		Ι															
Director of Planning & Scheduling		Ι															
Director of Quality		T															
Director of Salety		Ī															
Manager of Environmental Health & Safety		T															
Document Control Engineer		I															
OCIP Manager		Ι															
Asst. Project Manager		Ι															
Sr. Project Manager		T															
TSO Engineer		Ι															
Integrated Construction Manager/VDC/BIM Mgr.		Ι															
Director of integrated Construction		Ι															

NOTES: 1. Individuals shall be billed on an hourly basis not to exceed 40 hours per week. 2. Hourly rates and individuals are subject to Authority approval consistent with Article 8 of the CSA.

Page 9 of 12

2016

FEBRUARY 15, 2012

FEBRUARY 15, 2012				
Title	Name of			
	Individual	2016	2016	2016
		Oct	Nov	Dec
Preconstruction Services				
Design Phase Manager	Jason Hopper			
Lead Estimator	Eric Kloss			
Architectural Estimator	David Green			
Mechanical Estimator	Tom Sheehan			
Electrical Estimator	Tom McCartney			
Structural Estimator	Jim Block			
Scheduling Manager	Tim Johnson			
Project Scheduler	Matt London			
Sr. Construction Manager	Kevin Dalager			
Project Engineer	TBD			
Preconstruction Community Participation				
Community Participation Director	Lynn Littlejchn			
Community Participation Director	Ravi Norman			
Communicity Participation Coordinator	Jessie Volal			
Project Executive Management				
Principal-in-Charge	John Wood			
Construction Executive	Brent Leiter	100%		
General Superintendent	Dave Mansel			
Director of Operations	Allen Troshinsky			
VDC				
VDC Services Manager	Ricardo Khan			
Senior VDC Coordinator	Michai Wojtak			
VDC Coordinator	Eric Keleny			
VDC Coordinator	Nathan House			
Salety				
Safety Manager	Jannifer Hume-Jargeson			
Safety Engineer	Pedro Lopez			
Public Safety Engineer	Dalton Harrison			
Quality				
Qually Manager	Mke Mason			
MEP/Commissioning Manager	Craig Sharpe			
Qually Engineer	Greg Lien			
Project Management Controls				
Project Controls Manager	Joe Prouix	100%		
Project Scheduler	Matt London	I	I	

Exhibit 5

Page 10 of 12

\_76

FEBRUARY 15, 2012

FEBRUARY 15, 2012				
Tide	Name of			
	Individual	2016	2016	2016
Office Manager	Karen Willenbring			
Project Accountant	Lloyd Kimbail	100%		
Accounting Clark/Admin	TBD			
Administrative Assistant	TBD			
Reception is tA dmin	TBD			
Community Participation				
Community Participation Manager	Lynn Littlejohn			
Community Participation Manager	Ravi Norman			
Community Participation Coordinator	Khailah Chandler			
Structural / Site				
Sr .Construction Manager	Kevin Dalager			
Structural Project Manager	Greg Huber			
Structural Superintendent	Scott Amundson			
Structural Project Engineer	Tom Clark			
Precast Project Engineer	Ryan Res			
Site Project Engineer	Justin Utpadel			
Site Superintendent	Hannah Patterson			
Fleid Engineer	TBD			
Roof / Enclosure				
Sr. Construction Manager	Steve Lauria			
Project Manager / Roof	Ron Stodok a			
Senior Superintendent	Jason Brown			
Superintendent	John Carter			
Project Engineer	Art Gardher			
Project Engineer	Lindsey Cabel			
Fleid Engineer	TBD			
Mech / Bec / Low Voltage Systems				
Construction Manager	Matt Nelson			
Mechanical Project Manager	Kevin Boyce			
MEP Project Engineer	Brian Windingland			
Electrical Project Manager	Dave McNeal			
Interiors				
Construction Manager	Gabe Stofferahn			
Project Manager	Chad Pasach			
Project Manager	Joe Pritzkow			
Food Service/FF&E	Leslie Seppamaki			
Area Superintendent	Brendan Moor e	100%	100%	100%
Area Superintendent	Tom Schwab			
Area Superintendent	John Oliver			
Area Superintendent	Martin Mathery			
Area Superintendent	Russ Brown			

Exhibit 5

Page 11 of 12

Construction Services Agreement

LL

FEBRUARY 15, 2012

_		_		_
- 6	xh	ан.	it.	ε.
	~	-	•	•

Title	Name of Individual	2016	2016	2016
Sports Systems		2010	2010	2010
Construction Manager	Heid Brown			
Seating Playing Field Project Manager	Brad Hjernvick			
Scoreboard and Signage Project Manager	Emily Roland			
General Staffing Rates				
Design Phase Manager				
Sr. Estimator				
Estimator				
Scheduling Manager				
Associate Counsel				
Vice President, Senior Counsel				
Director of Planning & Scheduling				
Director of Quality				
Director of Safety				
Manager of Environmental Health & Safety				
Document Control Engineer				
OCIP Manager				
Asst. Project Manager				
Sr. Project Manager				
TSO Engineer				
Integrated Construction Manager/VDC/BIM Mgr.				
Director of Integrated Construction				

NOTES:

1. Individuals shall be billed on an hourly basis not to exceed 40 hours per week

2. Hourly rates and individuals are subject to Authority approval conditions with Article 8 of the CSA.

Page 1.2 of 12

78

FEBRUARY 15, 2012				Exhib	oit 5											
Title	Name of	2013	2014	2015	2016											
	Individual	Rate/Hr	Rate/Hr	Rate/Hr	Rate/Hr	2013	2013	2013	2013	2013	2013	2013	2013	2013	2013	2013
			3.0%	3.0%	3.0%	Jan	Feb	Mar	Apr	May	June	July	Aug	Sept	Oct	Nov
Preconstruction Services																
Design Phase Manager	Jason Hopper						100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Lead Estimator	Eric Kloss						100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Architectural Estimator	David Green						100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Mechanical Estimator	Tom Sheehan						100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Electrical Estimator	Tom McCartney						100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Structural Estimator	Jim Block			1			100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Scheduling Manager	Tim Johnson						100%	100%	100%	100%	100%	100%	100%	100%	10070	10070
Project Scheduler	Matt London						100%	100%	100%	100%	100%	100%	100%	100%		
Sr. Construction Manager	Kevin Dalager						50%	50%	50%	100%	100%	100%				
Project Engineer	TBD						50%	50%	50%	50%	50%	50%	50%	50%		
Preconstruction Community Participation																
Community Participation Director	Lynn Littlejohn						50%	50%	50%	50%	50%	50%	50%	50%		
Community Participation Director	Ravi Norman						25%	25%	25%	25%	25%	25%	25%	25%		
Communicty Participation Coordinator	Jessie Volal															
Project Executive Management	•						1				1	1				
Principal-in-Charge	John Wood	250.00	257.50	265.23	273.18		80%	80%	80%	80%	80%	80%	80%	80%	80%	80%
Construction Executive	Brent Leiter	165.00	169.95	175.05	180.30		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
General Superintendent	Dave Mansell	155.00	159.65	164.44	169.37		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Director of Operations	Allen Troshinsky	176.00	181.28	186.72	192.32		20%	20%	20%	20%	20%	20%	20%	20%	20%	20%
VDC																
VDC Services Manager	Ricardo Khan	115.00	118.45	122.00	125.66										25%	25%
Senior VDC Coordinator	Michal Wojtak	90.00	92.70	95.48	98.35										100%	100%
VDC Coordinator	Eric Keleny	90.00	92.70	95.48	98.35								100%	100%	100%	100%
VDC Coordinator	Nathan House	80.00	82.40	84.87	87.42											
Safety							-									
Safety Manager	Jennifer Hume-Jorgeson	95.00	97.85	100.79	103.81											100%
Safety Engineer	Pedro Lopez	70.00	72.10	74.26	76.49											
Public Safety Engineer	Dalton Harrison	70.00	72.10	74.26	76.49											
Quality	1			1			1				1	r				
Quality Manager	Mike Mason	95.00	97.85	100.79	103.81										100%	100%
MEP/Commissioning Manager	Craig Sharpe	100.00	103.00	106.09	109.27	ļ										
Quality Engineer	Greg Lien	80.00	82.40	84.87	87.42							L				
Project Management Controls																
Project Controls Manager	Joe Proulx	100.00	103.00	106.09	109.27											
Project Scheduler	Matt London	80.00	82.40	84.87	87.42										100%	100%
Accounting and Administration																

FEBRUARY 15, 2012				Exhib	oit 5											
Title	Name of	2013	2014	2015	2016											
	Individual	Rate/Hr	Rate/Hr	Rate/Hr	Rate/Hr	2013	2013	2013	2013	2013	2013	2013	2013	2013	2013	2013
Office Manager	Karen Willenbring	45.00	46.35	47.74	49.17								70%	100%	100%	100%
Project Accountant	Lloyd Kimball	75.00	77.25	79.57	81.95								50%	60%	100%	100%
Accounting Clerk/Admin	TBD	40.00	41.20	42.44	43.71											50%
Administrative Assistant	TBD	40.00	41.20	42.44	43.71											1
Receptionist/Admin	TBD	30.00	30.90	31.83	32.78											
Community Participation																
Community Participation Manager	Lynn Littlejohn	100.00	103.00	106.09	109.27										50%	50%
Community Participation Manager	Ravi Norman	105.00	108.15	111.39	114.74										25%	25%
Community Participation Coordinator	Khalilah Chandler	50.00	51.50	53.05	54.64										100%	100%
Structural / Site																
Sr .Construction Manager	Kevin Dalager	115.00	118.45	122.00	125.66								100%	100%	100%	100%
Structural Project Manager	Greg Huber	105.00	108.15	111.39	114.74								100%	100%	100%	100%
Structural Superintendent	Scott Amundson	95.00	97.85	100.79	103.81											50%
Structural Project Engineer	Tom Clark	85.00	87.55	90.18	92.88											
Precast Project Engineer	Ryan Ries	80.00	82.40	84.87	87.42											
Site Project Engineer	Justin Utpadel	80.00	82.40	84.87	87.42											
Site Superintendent	Hannah Patterson	75.00	77.25	79.57	81.95											
Field Engineer	TBD	60.00	61.80	63.65	65.56											1
Roof / Enclosure																
Sr. Construction Manager	Steve Laurila	115.00	118.45	122.00	125.66		20%	20%	20%	20%	20%	80%	100%	100%	100%	100%
Project Manager / Roof	Ron Stodolka	100.00	103.00	106.09	109.27									100%	100%	100%
Senior Superintendent	Jason Brown	115.00	118.45	122.00	125.66											1
Superintendent	John Carter	95.00	97.85	100.79	103.81											1
Project Engineer	Art Gardner	80.00	82.40	84.87	87.42											
Project Engineer	Lindsey Cabel	80.00	82.40	84.87	87.42											
Field Engineer	TBD	60.00	61.80	63.65	65.56											
Mech / Elec / Low Voltage Systems																
Construction Manager	Matt Nelson	110.00	113.30	116.70	120.20										100%	100%
Mechanical Project Manager	Kevin Boyce	100.00	103.00	106.09	109.27											1
MEP Project Engineer	Brian Windingland	80.00	82.40	84.87	87.42											1
Electrical Project Manager	Dave McNeal	100.00	103.00	106.09	109.27											
Interiors																
Construction Manager	Gabe Stofferahn	100.00	103.00	106.09	109.27											
Project Manager	Chad Paasch	100.00	103.00	106.09	109.27											
Project Manager	Joe Pritzkow	100.00	103.00	106.09	109.27											
Food Service/FF&E	Leslie Seppamaki	100.00	103.00	106.09	109.27	1						1		1		1
Area Superintendent	Brendan Moore	95.00	97.85	100.79	103.81	Î						1				
Area Superintendent	Tom Schwab	95.00	97.85	100.79	103.81	Î						1				
Area Superintendent	John Oliver	95.00	97.85	100.79	103.81	Î						1				
Area Superintendent	Martin Mathery	115.00	118.45	122.00	125.66	1						1		1		1
				122.00	125.00											

FEBRUARY 15, 2012				Exhib	oit 5											
Title	Name of	2013	2014	2015	2016											
	Individual	Rate/Hr	Rate/Hr	Rate/Hr	Rate/Hr	2013	2013	2013	2013	2013	2013	2013	2013	2013	2013	2013
Sports Systems																
Construction Manager	Heidi Brown	110.00	113.30	116.70	120.20											
Seating/Playing Field Project Manager	Brad Hjemvick	100.00	103.00	106.09	109.27											
Scoreboard and Signage Project Manager	Emily Roland	95.00	97.85	100.79	103.81											
General Staffing Rates							·	·								
Design Phase Manager		105.00	108.15	111.39	114.74											
Sr. Estimator		105.00	108.15	111.39	114.74											
Estimator		85.00	87.55	90.18	92.88											
Scheduling Manager		80.00	82.40	84.87	87.42											
Associate Counsel		237.00	244.11	251.43	258.98											
Vice President, Senior Counsel		253.00	260.59	268.41	276.46											
Director of Planning & Scheduling		124.00	127.72	131.55	135.50											
Director of Quality		124.00	127.72	131.55	135.50	1										
Director of Safety		124.00	127.72	131.55	135.50											
Manager of Environmental Health & Safety		114.00	117.42	120.94	124.57	1										
Document Control Engineer		60.00	61.80	63.65	65.56											
OCIP Manager		75.00	77.25	79.57	81.95	1										
Asst. Project Manager		85.00	87.55	90.18	92.88	1										
Sr. Project Manager		105.00	108.15	111.39	114.74	1										
TSO Engineer		80.00	82.40	84.87	87.42	1										
Integrated Construction Manager/VDC/BIM Mgr.		90.00	92.70	95.48	98.35	1										
Director of Integrated Construction		115.00	118.45	122.00	125.66	]										

1. Individuals shall be billed on an hourly basis not to exceed 40 hours per week.

2. Hourly rates and individuals are subject to Authority approval consistent with Article 8 of the CSA.

FEBRUARY 15, 2012					Exh	nibit 5												
Title	Name of																	
	Individual	2013	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2015	2015	2015	2015
	marviada	Dec	Jan	Feb	Mar	Apr	May	June	July	Aug	Sept	Oct	Nov	Dec	Jan	Feb	Mar	Apr
		Dec	Jan	TED	IVIDI	Дрі	iviay	June	July	Aug	Jehr	001	NOV	Dec	5411	1 CD	IVICI	Арі
Preconstruction Services	-													-			-	
Design Phase Manager	Jason Hopper	100%	100%	100%	100%													
Lead Estimator	Eric Kloss	100%	100%	100%	100%	100%	100%	100%										
Architectural Estimator	David Green	100%																
Mechanical Estimator	Tom Sheehan	100%																
Electrical Estimator	Tom McCartney	100%																
Structural Estimator	Jim Block	100%																
Scheduling Manager	Tim Johnson	1																
Project Scheduler	Matt London																	
Sr. Construction Manager	Kevin Dalager																	
Project Engineer	TBD																	
Preconstruction Community Participation						1	1				1					1		
Community Participation Director	Lynn Littlejohn																	
Community Participation Director	Ravi Norman																	
Communicty Participation Coordinator	Jessie Volal																	
Project Executive Management																		
Principal-in-Charge	John Wood	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%
Construction Executive	Brent Leiter	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
General Superintendent	Dave Mansell	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Director of Operations	Allen Troshinsky	20%																
VDC																		
VDC Services Manager	Ricardo Khan	25%	25%	25%	25%	25%	25%	25%										
Senior VDC Coordinator	Michal Wojtak	100%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%
VDC Coordinator	Eric Keleny	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
VDC Coordinator	Nathan House		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Safety																		
Safety Manager	Jennifer Hume-Jorgeson	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Safety Engineer	Pedro Lopez			100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Public Safety Engineer	Dalton Harrison			100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Quality																		
Quality Manager	Mike Mason	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
MEP/Commissioning Manager	Craig Sharpe																	
Quality Engineer	Greg Lien			75%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Project Management Controls																		
Project Controls Manager	Joe Proulx		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Project Scheduler	Matt London	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Accounting and Administration																		

FEBRUARY 15, 2012					Exh	ibit 5												
Title	Name of																	
	Individual	2013	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2015	2015	2015	2015
Office Manager	Karen Willenbring	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Project Accountant	Lloyd Kimball	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Accounting Clerk/Admin	TBD	50%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Administrative Assistant	TBD		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Receptionist/Admin	TBD			100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Community Participation												1	1					
Community Participation Manager	Lynn Littlejohn	50%	50%	50%	50%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%
Community Participation Manager	Ravi Norman	25%	25%	25%	25%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%
Community Participation Coordinator	Khalilah Chandler	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Structural / Site												1	1					
Sr .Construction Manager	Kevin Dalager	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Structural Project Manager	Greg Huber	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Structural Superintendent	Scott Amundson	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Structural Project Engineer	Tom Clark		50%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Precast Project Engineer	Ryan Ries		50%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Site Project Engineer	Justin Utpadel		100%	100%	100%	100%	100%	100%										
Site Superintendent	Hannah Patterson		100%	100%	100%	100%	100%	50%										
Field Engineer	TBD		50%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
De of / En ele euro																		
Roof / Enclosure																		
Roof / Enclosure Sr. Construction Manager	Steve Laurila	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
	Steve Laurila Ron Stodolka	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%	100% 100%
Sr. Construction Manager																		
Sr. Construction Manager Project Manager / Roof	Ron Stodolka												100%	100%	100%	100%	100%	100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent	Ron Stodolka Jason Brown												100%	100%	100% 100%	100% 100%	100% 100%	100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent	Ron Stodolka Jason Brown John Carter		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100% 100%	100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer	Ron Stodolka Jason Brown John Carter Art Gardner		100% 50%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer	Ron Stodolka Jason Brown John Carter Art Gardner Lindsey Cabel		100% 50% 50%	100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager	Ron Stodolka Jason Brown John Carter Art Gardner Lindsey Cabel		100% 50% 50%	100% 100% 100%	100% 100% 100% 100%	100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100%	100% 100% 100% 100%	100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager Mechanical Project Manager	Ron Stodolka Jason Brown John Carter Art Gardner Lindsey Cabel TBD	100%	100% 50% 50% 50%	100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100%	100%           100%           100%           100%           100%           100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager	Ron Stodolka Jason Brown John Carter Art Gardner Lindsey Cabel TBD Matt Nelson	100%	100% 50% 50% 50% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager Mechanical Project Manager	Ron Stodolka Jason Brown John Carter Art Gardner Lindsey Cabel TBD Matt Nelson Kevin Boyce	100%	100% 50% 50% 50% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager Mechanical Project Manager MEP Project Engineer	Ron Stodolka Jason Brown John Carter Art Gardner Lindsey Cabel TBD Matt Nelson Kevin Boyce Brian Windingland	100%	100% 50% 50% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager Mechanical Project Manager MEP Project Engineer Electrical Project Manager	Ron Stodolka Jason Brown John Carter Art Gardner Lindsey Cabel TBD Matt Nelson Kevin Boyce Brian Windingland	100%	100% 50% 50% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager Mechanical Project Manager MEP Project Engineer Electrical Project Manager Interiors Construction Manager Project Manager	Ron Stodolka Jason Brown John Carter Art Gardner Lindsey Cabel TBD Matt Nelson Kevin Boyce Brian Windingland Dave McNeal Gabe Stofferahn Chad Paasch	100%	100% 50% 50% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager Mechanical Project Manager MEP Project Engineer Electrical Project Manager Interiors Construction Manager Project Manager Project Manager Project Manager	Ron Stodolka         Jason Brown         John Carter         Art Gardner         Lindsey Cabel         TBD         Matt Nelson         Kevin Boyce         Brian Windingland         Dave McNeal         Gabe Stofferahn         Chad Paasch         Joe Pritzkow	100%	100% 50% 50% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager Mechanical Project Manager MEP Project Engineer Electrical Project Manager Interiors Construction Manager Project Manager Project Manager Project Manager Project Manager Project Manager Project Manager Project Manager	Ron Stodolka         Jason Brown         John Carter         Art Gardner         Lindsey Cabel         TBD         Matt Nelson         Kevin Boyce         Brian Windingland         Dave McNeal         Gabe Stofferahn         Chad Paasch         Joe Pritzkow         Leslie Seppamaki	100%	100% 50% 50% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager Mechanical Project Manager MEP Project Engineer Electrical Project Manager Interiors Construction Manager Project Manager Project Manager Project Manager Project Manager Project Manager Project Manager Project Manager Project Manager Project Manager	Ron Stodolka         Jason Brown         John Carter         Art Gardner         Lindsey Cabel         TBD         Matt Nelson         Kevin Boyce         Brian Windingland         Dave McNeal         Gabe Stofferahn         Chad Paasch         Joe Pritzkow         Leslie Seppamaki         Brendan Moore	100%	100% 50% 50% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager Mechanical Project Manager MEP Project Engineer Electrical Project Manager Interiors Construction Manager Project Manager	Ron Stodolka         Jason Brown         John Carter         Art Gardner         Lindsey Cabel         TBD         Matt Nelson         Kevin Boyce         Brian Windingland         Dave McNeal         Gabe Stofferahn         Chad Paasch         Joe Pritzkow         Leslie Seppamaki         Brendan Moore         Tom Schwab	100%	100% 50% 50% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager Mechanical Project Manager MEP Project Engineer Electrical Project Manager Interiors Construction Manager Project Manager Project Manager Project Manager Project Manager Project Manager Project Manager Project Manager Project Manager Project Manager	Ron Stodolka         Jason Brown         John Carter         Art Gardner         Lindsey Cabel         TBD         Matt Nelson         Kevin Boyce         Brian Windingland         Dave McNeal         Gabe Stofferahn         Chad Paasch         Joe Pritzkow         Leslie Seppamaki         Brendan Moore	100%	100% 50% 50% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%
Sr. Construction Manager Project Manager / Roof Senior Superintendent Superintendent Project Engineer Project Engineer Field Engineer Mech / Elec / Low Voltage Systems Construction Manager Mechanical Project Manager MEP Project Engineer Electrical Project Manager Interiors Construction Manager Project Manager	Ron Stodolka         Jason Brown         John Carter         Art Gardner         Lindsey Cabel         TBD         Matt Nelson         Kevin Boyce         Brian Windingland         Dave McNeal         Gabe Stofferahn         Chad Paasch         Joe Pritzkow         Leslie Seppamaki         Brendan Moore         Tom Schwab	100%	100% 50% 50% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%	100% 100% 100% 100% 100% 100% 100% 100%

FEBRUARY 15, 2012					Exh	nibit 5												
Title	Name of																	
	Individual	2013	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2014	2015	2015	2015	2015
Sports Systems																		
Construction Manager	Heidi Brown												100%	100%	100%	100%	100%	100%
Seating/Playing Field Project Manager	Brad Hjemvick																	
Scoreboard and Signage Project Manager	Emily Roland																	
General Staffing Rates						·			·	·								
Design Phase Manager																		
Sr. Estimator																		
Estimator																		
Scheduling Manager																		
Associate Counsel																		
Vice President, Senior Counsel																		
Director of Planning & Scheduling																		
Director of Quality																		
Director of Safety																		
Manager of Environmental Health & Safety																		
Document Control Engineer																		
OCIP Manager																		
Asst. Project Manager																		
Sr. Project Manager																		
TSO Engineer		1																
Integrated Construction Manager/VDC/BIM Mgr.		1																
Director of Integrated Construction		1																

1. Individuals shall be billed on an hourly basis not to exceed 40 hours per week.

2. Hourly rates and individuals are subject to Authority approval consistent with Article 8 of the CSA.

FEBRUARY 15, 2012					Ext	nibit 5												
Title	Name of																	
	Individual	2015	2015	2015	2015	2015	2015	2015	2015	2016	2016	2016	2016	2016	2016	2016	2016	2016
		May	June	July	Aug	Sept	Oct	Nov	Dec	Jan	Feb	Mar	Apr	May	June	July	Aug	Sept
		.,		,	5								r	. ,		,	1	[ .
Preconstruction Services																		
Design Phase Manager	Jason Hopper					1		1					I	1				
Lead Estimator	Eric Kloss																	
Architectural Estimator	David Green																	
Mechanical Estimator	Tom Sheehan																	
Electrical Estimator	Tom McCartney																	
Structural Estimator	Jim Block				-										-	-		
Scheduling Manager	Tim Johnson				-										-	-		
Project Scheduler	Matt London																	
Sr. Construction Manager	Kevin Dalager				-										-	-		
Project Engineer	TBD																	
Preconstruction Community Participation	100																	
Community Participation Director	Lynn Littlejohn																	
Community Participation Director	Ravi Norman																	
Communicty Participation Coordinator	Jessie Volal																	
Project Executive Management			I	I			I										I	
Principal-in-Charge	John Wood	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	80%	
Construction Executive	Brent Leiter	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
General Superintendent	Dave Mansell	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Director of Operations	Allen Troshinsky																	
VDC																		
VDC Services Manager	Ricardo Khan																	
Senior VDC Coordinator	Michal Wojtak	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%	15%			
VDC Coordinator	Eric Keleny	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%			
VDC Coordinator	Nathan House	100%	100%	100%	100%	100%	100%	100%	100%	100%								
Safety																		
Safety Manager	Jennifer Hume-Jorgeson	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%			
Safety Engineer	Pedro Lopez	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	60%			
Public Safety Engineer	Dalton Harrison	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	60%			
Quality																		
Quality Manager	Mike Mason	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%			
MEP/Commissioning Manager	Craig Sharpe									50%	50%	100%	100%	100%	100%	100%		
Quality Engineer	Greg Lien	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Project Management Controls																		
Project Controls Manager	Joe Proulx	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Project Scheduler	Matt London	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%			
Accounting and Administration																		

FEBRUARY 15, 2012					Ext	nibit 5												
Title	Name of																	
	Individual	2015	2015	2015	2015	2015	2015	2015	2015	2016	2016	2016	2016	2016	2016	2016	2016	2016
Office Manager	Karen Willenbring	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Project Accountant	Lloyd Kimball	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Accounting Clerk/Admin	TBD	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Administrative Assistant	TBD	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Receptionist/Admin	TBD	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Community Participation					1								1	1		1		
Community Participation Manager	Lynn Littlejohn	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%	25%			
Community Participation Manager	Ravi Norman	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%	10%			
Community Participation Coordinator	Khalilah Chandler	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%			
Structural / Site																		
Sr .Construction Manager	Kevin Dalager	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Structural Project Manager	Greg Huber	100%	50%															
Structural Superintendent	Scott Amundson	100%	50%															
Structural Project Engineer	Tom Clark	100%	50%															
Precast Project Engineer	Ryan Ries	100%	100%	100%	100%	100%	100%	100%	100%	100%	50%							
Site Project Engineer	Justin Utpadel									100%	100%	100%	100%	100%	100%	100%		
Site Superintendent	Hannah Patterson											100%	100%	100%	100%	100%		
Field Engineer	TBD	100%	50%															
Roof / Enclosure																		
Sr. Construction Manager	Steve Laurila	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Project Manager / Roof	Ron Stodolka	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%							
Senior Superintendent	Jason Brown	100%	100%	100%	100%	100%	100%	100%	100%									
Superintendent	John Carter	100%	100%	100%	100%	100%	100%	50%										
Project Engineer	Art Gardner	100%	100%	100%	100%	100%	100%	100%										
Project Engineer	Lindsey Cabel	100%	100%	100%	100%	100%	100%	100%	100%	100%								
Field Engineer	TBD	100%	100%	100%	100%	100%	100%	100%	100%	100%								
Mech / Elec / Low Voltage Systems																		
Construction Manager	Matt Nelson	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Mechanical Project Manager	Kevin Boyce	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%		
MEP Project Engineer	Brian Windingland	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%		
Electrical Project Manager	Dave McNeal	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%		
Interiors																		
Construction Manager	Gabe Stofferahn	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Project Manager	Chad Paasch	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Project Manager	Joe Pritzkow	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Food Service/FF&E	Leslie Seppamaki							100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Area Superintendent	Brendan Moore						100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
Area Superintendent	Tom Schwab	1		100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%		
Area Superintendent	John Oliver					100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	50%		
Area Superintendent	Martin Mathery			100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	50%		
Area Superintendent	Russ Brown						100%	100%	100%	100%	100%	100%	100%	100%	100%	100%		

FEBRUARY 15, 2012					Exh	ibit 5												
Title	Name of																	
	Individual	2015	2015	2015	2015	2015	2015	2015	2015	2016	2016	2016	2016	2016	2016	2016	2016	2016
Sports Systems																		
Construction Manager	Heidi Brown	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	
Seating/Playing Field Project Manager	Brad Hjemvick				100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%		
Scoreboard and Signage Project Manager	Emily Roland				100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%		
General Staffing Rates											2			2	•	2		
Design Phase Manager																		
Sr. Estimator																		
Estimator																		
Scheduling Manager																		
Associate Counsel																		
Vice President, Senior Counsel																		
Director of Planning & Scheduling																		
Director of Quality																		
Director of Safety																		
Manager of Environmental Health & Safety																		
Document Control Engineer																		
OCIP Manager																		
Asst. Project Manager																		
Sr. Project Manager																		
TSO Engineer																		
Integrated Construction Manager/VDC/BIM Mgr.																		
Director of Integrated Construction																		
v v	•																	

1. Individuals shall be billed on an hourly basis not to exceed 40 hours per week.

2. Hourly rates and individuals are subject to Authority approval consistent with Article 8 of the CSA.

FEBRUARY 15, 2012				
Title	Name of			
	Individual	2016	2016	2016
		Oct	Nov	Dec
Preconstruction Services				
Design Phase Manager	Jason Hopper			
Lead Estimator	Eric Kloss			
Architectural Estimator				
Mechanical Estimator	David Green Tom Sheehan			
Electrical Estimator	Tom McCartney			
Structural Estimator	Jim Block			
Scheduling Manager	Tim Johnson Matt London			
Project Scheduler Sr. Construction Manager				
Sr. Construction Manager Project Engineer	Kevin Dalager TBD	<u> </u>		
	IBD			
Preconstruction Community Participation Community Participation Director	Lynn Littlejohn			
Community Participation Director	Ravi Norman			
Communicty Participation Coordinator	Jessie Volal			
Project Executive Management	Jessie volai			
Principal-in-Charge	John Wood			
Construction Executive	Brent Leiter	100%		
General Superintendent	Dave Mansell	10070		
Director of Operations	Allen Troshinsky			
VDC	y worth Processing			
VDC Services Manager	Ricardo Khan			
Senior VDC Coordinator	Michal Wojtak	1		
VDC Coordinator	Eric Keleny	1		
VDC Coordinator	Nathan House	1		
Safety	· · · · · · · · · · · · · · · · · · ·		· · · · · ·	
Safety Manager	Jennifer Hume-Jorgeson			
Safety Engineer	Pedro Lopez			
Public Safety Engineer	Dalton Harrison			
Quality			<u> </u>	
Quality Manager	Mike Mason			
MEP/Commissioning Manager	Craig Sharpe	1		
Quality Engineer	Greg Lien			
Project Management Controls				
Project Controls Manager	Joe Proulx	100%		
Project Scheduler	Matt London			
Accounting and Administration				

FEBRUARY 15, 2012				
Title	Name of			
	Individual	2016	2016	2016
Office Manager	Karen Willenbring			
Project Accountant	Lloyd Kimball	100%		
Accounting Clerk/Admin	TBD			
Administrative Assistant	TBD			
Receptionist/Admin	TBD			
Community Participation				
Community Participation Manager	Lynn Littlejohn			
Community Participation Manager	Ravi Norman			
Community Participation Coordinator	Khalilah Chandler			
Structural / Site				
Sr .Construction Manager	Kevin Dalager			
Structural Project Manager	Greg Huber			
Structural Superintendent	Scott Amundson			
Structural Project Engineer	Tom Clark			
Precast Project Engineer	Ryan Ries			
Site Project Engineer	Justin Utpadel			
Site Superintendent	Hannah Patterson			
Field Engineer	TBD			
Roof / Enclosure				
Sr. Construction Manager	Steve Laurila			
Project Manager / Roof	Ron Stodolka			
Senior Superintendent	Jason Brown			
Superintendent	John Carter			
Project Engineer	Art Gardner			
Project Engineer	Lindsey Cabel			
Field Engineer	TBD			
Mech / Elec / Low Voltage Systems				
Construction Manager	Matt Nelson			
Mechanical Project Manager	Kevin Boyce			
MEP Project Engineer	Brian Windingland			
Electrical Project Manager	Dave McNeal			
Interiors			1	
Construction Manager	Gabe Stofferahn			
Project Manager	Chad Paasch	1		
Project Manager	Joe Pritzkow	1		
Food Service/FF&E	Leslie Seppamaki			
Area Superintendent	Brendan Moore	100%	100%	100%
Area Superintendent	Tom Schwab	1		
Area Superintendent	John Oliver			
Area Superintendent	Martin Mathery			
Area Superintendent	Russ Brown	1		

Exhibit 5

FEBRUARY 15, 2012				
Title	Name of			
	Individual	2016	2016	2016
Sports Systems				
Construction Manager	Heidi Brown			
Seating/Playing Field Project Manager	Brad Hjemvick			
Scoreboard and Signage Project Manager	Emily Roland			
General Staffing Rates				
Design Phase Manager				
Sr. Estimator				
Estimator				
Scheduling Manager				
Associate Counsel				
Vice President, Senior Counsel				
Director of Planning & Scheduling				
Director of Quality				
Director of Safety				
Manager of Environmental Health & Safety				
Document Control Engineer				
OCIP Manager				
Asst. Project Manager				
Sr. Project Manager				
TSO Engineer				
Integrated Construction Manager/VDC/BIM Mgr.				
Director of Integrated Construction				

1. Individuals shall be billed on an hourly basis not to exceed 40 hours per week.

2. Hourly rates and individuals are subject to Authority approval consistent with Article 8 of the CSA.

Exhibit 5

## EXHIBIT 6 GENERAL CONDITIONS OF THE CONTRACTS FOR CONSTRUCTION

The following General Conditions of the Contracts for Construction is included in this Construction Services Agreement as **Exhibit 6**:

# EXHIBIT 6

# GENERAL CONDITIONS OF THE CONSTRUCTION SERVICES AGREEMENT BETWEEN AUTHORITY AND CONSTRUCTION MANAGER

	TABLE OF CONTENTS	
		Page
ARTICI	LE 1	
1.1	BASIC DEFINITIONS	
1.2	EXECUTION, CORRELATION AND INTENT	
1.3	CAPITALIZATION	
1.4	CONSTRUCTION OF LANGUAGE	2
1.5	CONFIDENTIALITY	
1.6	TRADE SECRETS, TRADEMARKS AND TRADE NAMES	3
1.7	PRESS RELEASES AND OTHER PROMOTIONAL MATERIALS	4
1.8	TAXATION	4
1.9	NO WAIVER	4
1.10	SEVERABILITY	4
1.11	INTERPRETATION	4
1.12	MULTIPLE COUNTERPARTS; FAXES	4
1.13	SURVIVAL OF REPRESENTATIONS AND WARRANTIES	4
1.14	OWNERSHIP AND USE OF ARCHITECT'S DRAWINGS, SPECIFICATIONS	
	OTHER DOCUMENTS	
_	LE 2 - PROJECT REPRESENTATIVE	
2.2	LIMITATION OF LIABILITY OF PROJECT REPRESENTATIVE(S)	
	LE 3 - AUTHORITY	
3.1	DEFINITION	
3.3	EXTENT OF AUTHORITY RIGHTS	
3.4	DIRECT PURCHASES BY AUTHORITY	
3.5	AUTHORITY DIRECTION OF WORK AND INSPECTIONS	
3.6	AUTHORITY'S RIGHT TO RECORDS	6
ARTICI	LE 4 – CONSTRUCTION MANAGER	7
4.1	DEFINITION	7
4.2	REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONSTRUCTION MANAGER	7
4.3	SUPERVISION AND CONSTRUCTION PROCEDURES	
4.4	LABOR AND MATERIALS	
4.5	WARRANTY	9
4.6	TAXES	

4.7	PERMITS, FEES AND NOTICES	10
4.8	ALLOWANCES	10
4.9	FIELD PERSONNEL	11
4.10	CONSTRUCTION MANAGER'S CONSTRUCTION SCHEDULE	11
4.11	DOCUMENTS AND SAMPLES AT THE PROJECT SITE	12
4.12	SHOP DRAWINGS, PRODUCT DATA, AND SAMPLES	12
4.13	USE OF PROJECT SITE	14
4.14	CUTTING AND PATCHING	14
4.15	CLEANING UP	15
4.16	ACCESS TO WORK	15
4.17	ROYALTIES AND PATENTS	15
4.18	INDEMNIFICATION BY CONSTRUCTION MANAGER	15
4.19	WARRANTIES AND REPRESENTATIONS.	16
4.20	CONSTRUCTION MANAGER SELF PERFORMED WORK	16
4.21	PROGRESS MEETINGS; REPORTS	17
4.22	WEB BASED PROJECT MANAGEMENT SYSTEM	17
ARTIC	JE 5 - ARCHITECT	17
5.1	ARCHITECT	17
ARTICI	LE 6 – ADMINISTRATION OF THE CONTRACT	18
6.1	ADMINISTRATION OF THE CONTRACT	18
6.2	CLAIMS AND DISPUTES	19
6.3	REVIEW OF CLAIMS AND DISPUTES	20
6.4	DISPUTE RESOLUTION	21
ARTICI	LE 7 - SUBCONTRACTORS	22
7.1	DEFINITIONS	22
7.2	AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF TWORK	
7.3	SUBCONTRACTUAL RELATIONS	22
7.4	CONTINGENT ASSIGNMENT OF SUBCONTRACTS	23
ARTICI	LE 8 - CONSTRUCTION BY AUTHORITY	23
8.1	AUTHORITY'S RIGHT TO PERFORM CONSTRUCTION	23
8.2	MUTUAL RESPONSIBILITY	24
ARTIC	LE 9 - CHANGES IN THE CONSTRUCTION MANAGER'S WORK	24
9.1	CHANGES	24
9.2	CONTRACT REVISIONS	25
9.3	CONSTRUCTION CHANGE DIRECTIVES	25
9.4	MINOR CHANGES IN THE WORK	26
9.5	AGREED OVERHEAD AND PROFIT RATES	26
ARTIC	LE 10 - TIME	26

10.1	DEFINITIONS	. 26
10.2	PROGRESS AND COMPLETION	. 26
10.3	DELAYS AND EXTENSIONS OF TIME	. 27
ARTICL	E 11 – PAYMENTS AND COMPLETION	. 27
11.1	PAYMENTS	. 27
11.2	SCHEDULE OF VALUES	. 27
11.3	APPLICATIONS FOR PAYMENT	. 27
11.4	APPROVAL OF APPLICATIONS FOR PAYMENT	. 29
11.5	DECISIONS TO WITHHOLD CERTIFICATION	. 29
11.6	PROGRESS PAYMENTS	. 30
11.7	FAILURE OF PAYMENT	. 31
11.8	SUBSTANTIAL COMPLETION	. 31
11.9	PARTIAL OCCUPANCY OR USE	. 32
11.10	FINAL COMPLETION AND FINAL PAYMENT	. 32
ARTICL	E 12 - PROTECTION OF PERSONS AND PROPERTY	. 33
12.1	SAFETY PRECAUTIONS AND PROGRAMS	. 33
12.2	SAFETY OF PERSONS AND PROPERTY	. 33
12.3	EMERGENCIES	. 34
ARTICL	E 13 - INSURANCE AND BONDS	. 34
13.1	CONSTRUCTION MANAGER'S LIABILITY INSURANCE	. 34
13.2	AUTHORITY'S LIABILITY INSURANCE	. 36
13.3	PROPERTY INSURANCE	. 36
13.5	PERFORMANCE BOND AND PAYMENT BOND	. 37
13.5	PERFORMANCE BOND AND PAYMENT BOND	. 37
13.6	GENERAL REQUIREMENTS	. 38
ARTICL	E 14 - UNCOVERING AND CORRECTION OF WORK	. 39
14.1	UNCOVERING OF WORK	. 39
14.2	CORRECTION OF WORK	. 39
14.3	ACCEPTANCE OF NONCONFORMING WORK	. 40
ARTICL	E 15 – MISCELLANEOUS PROVISIONS	. 40
15.1	GOVERNING LAW	. 40
15.2	SUCCESSORS AND ASSIGNS	. 40
15.3	WRITTEN NOTICE	. 40
15.4	RIGHTS AND REMEDIES	. 40
15.5	TESTS AND INSPECTIONS	. 41
15.6	GENERAL PROVISIONS	. 41
ARTICL	E 16 - TERMINATION OR SUSPENSION OF THE CONTRACT	. 42
16.1	TERMINATION BY THE CONSTRUCTION MANAGER	. 42
16.2	TERMINATION BY THE AUTHORITY FOR CAUSE	. 43

16.3	SUSPENSION BY THE AUTHORITY FOR CONVENIENCE 4	.3
16.4	TERMINATION BY THE AUTHORITY FOR CONVENIENCE 4	4

## EXHIBIT 6

# GENERAL CONDITIONS OF THE CONSTRUCTION SERVICES AGREEMENT BETWEEN AUTHORITY AND CONSTRUCTION MANAGER

The General Conditions of the Construction Services Agreement between the Authority and the Construction Manager are included herein as **Exhibit 6**:

#### ARTICLE 1 GENERAL PROVISIONS

## 1.1 BASIC DEFINITIONS

1.1.1 Definitions included in the Construction Services Agreement are included as <u>Appendix A</u>.

## 1.2 EXECUTION, CORRELATION AND INTENT

- 1.2.1 Execution of the Construction Services Agreement by the Construction Manager is a representation that the Construction Manager has visited the Project Site, become familiar with local conditions under which the Work is to be performed and correlated personal observations with requirements of the Contract Documents. The Construction Manager and each Subcontractor shall evaluate and satisfy themselves as to the conditions and limitations under which the Work is to be performed, including (1) the location and layout of the Project Site, (2) prevailing climatic conditions, (3) anticipated labor supply and costs, (4) market conditions that will impact the Guaranteed Maximum Price, (5) the requirement that the Construction Manager has completed a thorough and comprehensive review of the Contract Documents and understands and recognizes the complete scope of the Work, (6) existing conditions that will impact the Work, and (7) availability and cost of materials, tools, and equipment. The Construction Manager shall be solely responsible for preparing and administering a comprehensive safety program for and for all safety regarding the performance of the Work. The Authority shall not be required to make any adjustment in either the Guaranteed Maximum Price or Contract Time in connection with any failure by the Construction Manager to comply with the requirements of this Subparagraph 1.2.1.
- **1.2.2** The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Construction Manager. Execution of the Construction Services Agreement by the Construction Manager is an express representation (1) that the Construction Manager understands the intent stated herein with respect to the Pre-Construction Phase Services, and (2) the Construction Manager's execution of a Contract Revision, including the Construction Management Plan in <u>Exhibit 3</u> of the Construction Services Agreement shall be an express and unqualified representation that the Construction Manager understands the intent stated herein and therein . The Contract Documents are complementary, and what is required by one shall be as binding as if required by all; subject to the Construction Services Agreement, performance by the Construction Manager shall be required only to the extent consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the intended results. In the event of inconsistencies within or between parts of the Construction Manager shall (1) provide the better quality or greater quantity of the Work or (2) comply with the more stringent requirement, either or both in accordance with the Authority's reasonable interpretation. In general, the following rules of interpretation shall apply:
  - .1 On the Drawings, given dimensions shall take precedence over scaled measurements, and large-scale drawings over small-scale drawings. Drawings are intended to show the general arrangement, design and extent of the Work, and are partly diagrammatic; they are not intended to be scaled for roughing in measurements, or to serve as Shop Drawings.
  - .2 Before ordering any materials or doing any Work, the Construction Manager shall verify all existing conditions and measurements at the Project Site and shall be responsible for the correctness of such measurements. Any difference that may be found shall be submitted to the Architect for interpretation before proceeding with the Work.
  - .3 If a minor change in the Work is found necessary due to actual field conditions, the Construction Manager shall so advise the Architect who shall issue detailed drawings of such before making the change.

- 1.2.3 Organization of the Specifications into Divisions, Sections and Articles, and arrangement of Drawings shall not control the Construction Manager in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade. Titles to Specification Divisions, Sections, and Paragraphs in the Contract Documents are for convenience only and shall not be taken as a part of the Specifications or as a correct and complete segregation of the several units of the Work. No responsibility, either direct or implied, is assumed by the Authority for omissions or duplications by Construction Manager or its Subcontractors, due to real or alleged error in the arrangement of matters or numbering of the Contract Documents. Instructions and other information furnished in the Specifications, including, without limitation, items in connection with prefabricated or pre-finished items, are not intended to supersede work agreements between employees and employees. Should the Specifications conflict with such work agreements, the work agreements shall be followed, on the condition that such items are provided and finished as specified. If necessary, such Work shall be performed on the Project Site, instead of at the shop, by appropriate labor and in accordance with the requirements of the Drawings and Specifications.
- **1.2.4** Unless otherwise stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meaning.
  - .1 Whenever a product is specified in accordance with a Federal Specification, an ASTM Standard, an American National Standards Institute Specification, or other association standard, the Construction Manager shall present an affidavit from Construction Manager and the manufacturer, when requested by the Architect or required in the Specifications, certifying that the product complies with the particular standard or Specification. When requested by the Architect or when specified, support test data shall be submitted to substantiate compliance.
  - .2 Whenever a product is specified or shown by describing proprietary items, model number, catalog number, manufacturer, trade names, or similar reference, no substitutions may be made unless authorized in writing by the Architect and the Authority prior to execution of the Construction Services Agreement or if accepted as a Contract Revision to the Construction Services Agreement. When two (2) or more products are shown or specified, the Construction Manager has the option to use any of those shown or specified.

## 1.3 CAPITALIZATION

**1.3.1** Terms capitalized in these General Conditions include those that are (1) specifically defined in <u>Appendix A</u> of the General Conditions, and (2) the titles of numbered articles and identified references to Paragraphs, Subparagraphs and Clauses in the Contract Documents.

#### 1.4 CONSTRUCTION OF LANGUAGE

- 1.4.1 General. In the interest of brevity, the Contract Documents frequently omit modifying words such as "all" and 'any" articles such as "the" and "an" but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement. The terms and phrases used in the Contract Documents, unless the context otherwise requires, are to be interpreted as follows: (i) the words "including," "include" or "includes" mean including without limitation; (ii) reference to any agreement, appendix, schedule, exhibit, instrument or coverage policy means as such is amended, modified or supplemented, including by waiver or consent; (iii) reference to any Applicable Laws means such Applicable Laws, as amended, modified, codified or reenacted, in whole or part, and in effect from time to time; (iv) reference to any Party includes such Party's successors and assigns, to the extent that such successors and assigns are permitted; (v) pronouns in masculine, feminine and neuter genders are to be construed to state and include any other gender; (vi) the words "will" and "shall" have the same meaning; (vii) unless the context otherwise requires, all defined terms in the Contract Documents include the singular and the plural; (viii) "to the extent practicable" means using good faith, commercially reasonable efforts, without the expenditure of significant additional costs or expenses as compared to other compliant alternatives; (ix) the headings and captions contained in the Contract Documents are for convenience and reference only and in no way define, describe, extend or limit the scope or intent of the agreements or the intent of any provision of the agreements; (x) all monetary figures are expressed in currency of the United States of America (US Dollars); and (xi) the language in the Contract Documents shall be construed according to its customary meaning within the construction industry in Minnesota except where a specific definition is provided herein.
- **1.4.2** <u>Approved.</u> When the words "approved," "satisfactory," "proper" or "as directed" are used, approval by the Authority shall be understood.
- **1.4.3 Provide.** When the word "provide," including derivatives thereof, is used, it shall mean to properly fabricate, complete, transport, deliver, install, erect, construct, test and furnish all labor, materials, equipment apparatus, appurtenances, and all items and expenses necessary to properly complete in place, ready for operation or use under the terms of the Contract Documents.

1.4.4 - 1.4.5 [Reserved]

- **1.4.6** <u>Knowledge.</u> The terms "knowledge," "recognize" and "discover," their respective derivatives and similar terms in the Contract Documents, as used in reference to the Construction Manager, shall be interpreted to mean that which the Construction Manager knows, recognizes, or discovers or should have known, recognized, or discovered utilizing the Standard of Care applicable to the Construction Manager. The expression "reasonably inferable" and similar terms in the Contract Documents shall be interpreted by the Authority to mean reasonably inferable by the Construction Manager utilizing the Standard of Care applicable to the Construction Manager.
- **1.4.7 Persistently.** The phrase "persistently fails" and other similar expressions, as used in reference to the Construction Manager, shall be interpreted to mean any combination of acts and omissions, which causes the Authority to reasonably conclude that the Construction Manager will not complete the Work within the Contract Time, for the GMP, or in substantial compliance with the requirements of the Contract Documents.

### 1.5 CONFIDENTIALITY

- 1.5.1 As a result of the Architect, Construction Manager and all other members of the Project Team's participation in the Project, the Project Team will have access to information and materials of a highly sensitive nature, including Confidential Information belonging to the Authority or the Team. The Construction Manager, hereby warrants that its employees, subcontractors, and agents shall not (without in each instance obtaining prior written consent of the other party) disclose, make commercial or other use of, or give or sell to any person, firm, or corporation, any Confidential Information received directly or indirectly from the Authority or the Team, as applicable, or acquired or developed in the course of the performance of the Work unless: (1) required to do so pursuant to Applicable Laws (and then only after the Architect, Construction Manager or other member of the Project Team has given the Authority or the Team, as applicable, prompt written notice of the legal compulsion and, at the Authority's or the Team's, as applicable, expense, provide the Authority or the Team, as applicable, with cooperation in any attempt the Authority or the Team, as applicable, may make to gain a protective order acceptable to the Authority or the Team, as applicable); or (2) it is rightfully in the possession of the Architect, Construction Manager or other Project Team member from a source other than the Authority or the Team prior to the time of disclosure of the information to the Architect, Construction Manager or other Project Team member under the Contract Documents; or (3) it became part of the public domain prior to the time of the Architect's, Construction Manager's or other Project Team member's receipt; or (4) it is supplied to the Architect, Construction Manager, or other Project Team member after the time of the Architect's, Construction Manager's or other Project Team members receipt by a third party who is under no obligation to the Authority or to the Team to maintain such information in confidence; or (5) it was independently developed by the Architect, Construction Manager, or other Project Team member prior to the time of receipt.
- **1.5.2** The Architect's, Construction Manager's, or other Project Team members' confidentiality and non-disclosure obligations shall survive the expiration or earlier termination of their respective agreements and shall continue for a period of ten (10) years following the expiration of earlier termination of their respective agreements. All Confidential Information, regardless of form, shall be the property of the Authority or Team, as applicable, and shall be returned to the Authority or Team upon its request, or in any event, at the expiration or earlier termination of the Construction Services Agreement.

## 1.6 TRADE SECRETS, TRADEMARKS AND TRADE NAMES

- 1.6.1 The Architect, Construction Manager, and Project Team acknowledge that the Authority or Team may provide the Architect, Construction Manager, and Project Team with access to certain information which may qualify as a Trade Secret under Applicable Laws, and the Architect, Construction Manager, and Project Team agree that for all such Trade Secrets that come into their possession, custody or control: (1) such Trade Secrets shall remain the sole property of the Authority or Team, as applicable, and the Architect, Construction Manager, and Project Team shall have no interest in said Trade Secrets; (2) the Architect, Construction Manager, and Project Team shall maintain the secrecy of the Trade Secrets for so long as they remain Trade Secrets under Applicable Laws; and (3) immediately upon the expiration or earlier termination of the Project, the Architect, Construction Manager, and Project Team shall deliver to the Authority or Team, as applicable, all Trade Secret documentation and any and all copies thereof, regardless of form or content.
- 1.6.2 Without prior written approval of the Authority or Team, as applicable, the Architect, Construction Manager, and Project Team shall have no right to use any Trademark or Trade Name of the Authority, Team, or Affiliated Entities. Further, the Architect, Construction Manager, and Project Team shall not refer to the Contract Documents or the Services or Work performed thereunder or to any Confidential Information, directly or indirectly, in connection with any production, promotion, or publication without the prior written consent of the Authority and Team, and the Authority and Team reserve the right in their sole discretion to release such information, time its release, and approve its form and content prior to release.
- **1.6.3** The Architect, Construction Manager, and Project Team may publish information (other than Trade Secrets and Confidential Information, which shall not be published) and images resulting from or relating to their Services or Work only after obtaining the Authority's and Team's prior written approval, which Authority and Team may grant or withhold in their sole discretion. The Authority and Team's approval, if granted, may be conditioned upon changes in the text, the use of different images, or reasonable delay in publication to protect Trade Secrets or Confidential Information and other interests of the Authority or Team.

## 1.7 PRESS RELEASES AND OTHER PROMOTIONAL MATERIALS

- 1.7.1 Both prior to, during, and after completion of the Project, the Authority and Team shall control the issuance of all press releases and all contacts with the press and all other media relating to the Project. The Architect, Construction Manager, and Project Team may not issue any press release, agree to be interviewed by members of the press, or otherwise interact with and/or disseminate information to the press or any media without the prior written consent of the Authority and Team, which consent may be withheld by the Authority and Team in their sole discretion.
- 1.7.2 After completion of the Project, the Architect, Construction Manager, and Project Team may use images of the Project and explanatory text in the Architect's, Construction Manager's, and Project Team member's marketing and promotional materials, subject to the Authority's and the Team's written approval prior to the Architect's, Construction Manager's, or Project Team member's first use of such materials; provided, however, that the Architect's, Construction Manager's, and Project Team member's marketing and promotional materials should not include any of the Authority's or the Team's Confidential Information.
- **1.7.3** No signs advertising the services to be performed by the Architect, Construction Manager, or Project Team, or identifying any person, firm or entity concerned with the Work to be performed by the Architect, Construction Manager, or Project Team shall be allowed at the Project Site or elsewhere unless approved in writing by the Authority and the Team in advance, which approval shall be within Authority's and the Team's sole and exclusive discretion.

## 1.8 TAXATION

The Construction Manager is responsible for any and all costs of taxes, license fees, royalties, and related fees imposed by any Governmental Authority having jurisdiction over the Project and associated with its Work.

#### 1.9 NO WAIVER

The failure of the Authority or Team to insist upon strict performance of the Contract Documents or any of their rights on any occasion shall <u>not</u> be deemed a waiver of any rights under the Contract Documents or otherwise.

#### 1.10 SEVERABILITY

Except as expressly provided to the contrary in the Contract Documents, each section, part, term, and provision of the Contract Documents is severable from each other section, part, term, and provision and if, for any reason, any section, part, term, or provision of the Contract Documents is determined by a court or arbitrator having valid jurisdiction in a decision which becomes final and not subject to appeal to which the parties are bound, to be invalid and contrary to, or in conflict with, any Applicable Laws, the determination that the section, part, term, or provision is invalid will not impair the operation of, or have any other effect on, the other portions, sections, parts, terms, and provisions of the Contract Documents as may remain otherwise enforceable, and all of the remaining sections, parts, terms, and provisions of the Contract Documents will continue to be in full force and effect and be binding. Any sections, parts, terms or provisions so determined to be invalid and contrary to, or in conflict with, any Applicable Laws will be severed from the Contract Documents without any further action of the Architect, Construction Manager, or the Authority to amend the Contract Documents. It is the intention of the Architect, Construction Manager, and the Authority that if any provision of the Contract Documents is susceptible to two or more constructions, one of which would render the provision enforceable and the other or others of which would render the provision unenforceable, then the provision shall have the meaning which renders it enforceable.

## 1.11 INTERPRETATION

The Authority and Construction Manager acknowledge and agree that they have participated jointly in the negotiation and drafting of the Contract Documents to which they are parties. If an ambiguity or question of intent or interpretation arises, the Contract Documents are to be construed as if drafted jointly by the applicable Parities and no presumption or burden of proof will arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of the Contract Documents.

## 1.12 MULTIPLE COUNTERPARTS; FAXES

The Contract Documents may be executed in counterparts. It is not necessary that the signature on behalf of each party appear on each counterpart copy, so long as each party executes the Contract Documents. All counterparts of the Contract Documents collectively constitute a single agreement. Authority is authorized to combine each party's execution sheets into a single document. A facsimile-transmitted signature of the Contract Documents or any document, instrument, or agreement hereinafter executed or given in connection with the Contract Documents shall be considered valid and binding upon the parties as if an original.

#### 1.13 SURVIVAL OF REPRESENTATIONS AND WARRANTIES

The representations, warranties, and indemnifications set forth in the Contract Documents and in any document, instrument, or agreement executed or given in connection herewith, shall survive final completion of the Work or termination of the Contract Documents.

## 1.14 OWNERSHIP AND USE OF ARCHITECT'S DRAWINGS, SPECIFICATIONS AND OTHER DOCUMENTS

The Drawings, Specifications and other documents prepared by the Architect are instruments of the Architect's 1.14.1 service through which the Work to be executed by the Construction Manager is described, and are the property of the Authority. The Construction Manager or any Subcontractor, Sub-Subcontractor, Supplier, Trade Contractor, Vendor, or Consultant shall not own or claim a copyright in the Drawings, Specifications and other documents prepared by the Architect. The Authority will retain all common law, statutory and other reserved rights, in addition to the copyright. The Drawings, Specifications and other documents prepared by the Architect, and copies thereof furnished to the Construction Manager or any Subcontractor, Sub-Subcontractor, Supplier, Trade Contractor, Vendor, or Consultant, are for use solely with respect to this Project. They are not to be used by the Architect, Construction Manager or any Subcontractor, Sub-Subcontractor or Supplier, Trade Contractor, Vendor, or Consultant on other projects or for additions to this Project outside the scope of the Work without the specific written consent of the Authority. The Construction Manager, Subcontractors, Sub-Subcontractors, Suppliers, Trade Contractor, Vendors, or Consultant are granted a limited license to use and reproduce applicable portions of the Drawings, Specifications and other documents prepared by the Architect appropriate to and for use in the execution of their Work. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with this Project is not to be construed as publication in derogation of the Authority's copyright or other reserved rights.

### ARTICLE 2 PROJECT REPRESENTATIVE

#### 2.1 [Reserved]

2.2 LIMITATION OF LIABILITY OF PROJECT REPRESENTATIVE(S) Except as provided in Paragraph 4.3 of the Construction Services Agreement and 10.3.1 herein and notwithstanding any of the rights and authority granted the Authority in the Design Services Agreement or the Contract Documents, the Authority and the Indemnitees (except the Architect Indemnitees) are not and shall, in no event, be responsible or in any manner liable for any aspect of the Design Services, including, without limitation, design, engineering, inspections, quality control, review and coordination of the Construction Documents or design administration services, which shall be provided solely by the Architect under the Design Services Agreement. Likewise, the Authority and the Indemnitees (except the Architect Indemnitees), are not and shall in no event, be responsible or in any manner liable for any aspect of the Construction Manager's Work, including construction management and administration, cost estimating, the GMP, the Contract Time, Value Engineering, safety, quality control, construction of the Construction Documents, construction means, methods, techniques, inspections, safety precautions and programs in connection with the Construction Manager's Work, notwithstanding any of the rights and authority granted the Authority and the Indemnitees in or under the Construction Manager's Work, notwithstanding any of the rights and authority granted the Authority and the Indemnitees in or under the Construction Manager's Work, notwithstanding any of the rights and authority granted the Authority and the Indemnitees in or under the Construction Manager's Work, notwithstanding any of the rights and authority granted the Authority and the Indemnitees in or under the Construction Manager's Work, notwithstanding any of the rights and authority granted the Authority and the Indemnitees in or under the Construction Coursents.

#### ARTICLE 3 AUTHORITY

#### 3.1 DEFINITION

- 3.1.1 [Reserved]
- 3.1.2 [Reserved]
- 3.2 [Reserved]

#### 3.3 EXTENT OF AUTHORITY RIGHTS

- **3.3.1** The **rights** stated in this **Article 3** and elsewhere in the Contract Documents are cumulative and not in limitation of any rights of the Authority (1) granted in the Contract Documents, (2) at law or (3) in equity.
- **3.3.2** Pursuant to **Paragraph 2.2** hereof, it is the clear intent of the Parties that the Authority and the Indemnitees (except the Architect Indemnitees), will have no responsibility or liability for any aspect of the Design Services or Construction Manager's Work as defined in the Design Services Agreement with the Architect and the Construction Services Agreement with the Construction Manager, respectively. The Construction Manager acknowledges and agrees to this provision by execution of its Contract Documents with the Authority.

#### 3.4 DIRECT PURCHASES BY AUTHORITY

**3.4.1** Before the GMP is established, the Construction Manager acknowledges that the Authority reserves the right to purchase directly any equipment and materials for the Work. For any item directly purchased by the Authority, the gross price of such item, plus the sales or use tax, if any, that would have been applicable to such item if not purchased by the Authority, shall not be a part of the GMP to be paid to the Construction Manager. Any discounts, savings or rebates on purchases by Authority of equipment and materials shall belong to the Authority.

- 3.4.2 The Authority shall receive, store, and protect all equipment and materials that it purchases until they are provided to the Construction Manager at the Project Site or delivered to such place as designated by the Construction Manager. Authority may order the supplier to deliver such equipment and materials directly to the Project Site. If the Construction Manager purchases or eventually takes an assignment of the materials and equipment purchased by the Authority and delivered to the site, the Construction Manager will assure that the equipment and materials purchased are acceptable and conform to the Contract Documents. If the Construction Manager has not purchased and will not take an assignment of the materials and equipment purchased but will install the materials and equipment delivered, the Construction Manager will inspect the materials upon delivery and before installation and call to the Authority's attention any visibly damaged or defective equipment or material. If the Construction Manager has not purchased or will not take an assignment of the materials and equipment purchased by the Authority and will not install the materials and equipment delivered, the Construction Manager will make a reasonable visual inspection of such equipment and materials, taking into account protective packaging that may need to remain in place until installation. Any defect or deficiency shall be called to the attention of the Authority immediately upon delivery in order that the Authority may make necessary arrangements for the adjustment or replacement of such equipment and materials. The Construction Manager shall not install any defective equipment or materials. Upon acceptance of such equipment and materials, the Construction Manager is responsible for the installation and incorporation of such equipment and materials into the Work in accordance with the Contract Documents.
- **3.4.3** Authority shall at all times have and possess all incidents of ownership with respect to equipment and materials purchased by it.
- **3.4.4** The Authority shall be responsible for any and all sales or use tax imposed on any direct purchases of materials or supplies made by the Authority. For purchases made by the Authority, the Authority agrees to indemnify, defend, and hold the Construction Manager harmless from any and all sales and use tax imposed or assessed against the Construction Manager, its Subcontractors, and Suppliers, including all losses, expenses, interest, fines, costs and reasonable attorney's fees incurred in any action, suit, or proceeding related to, in connection with, or arising out of any assessment of sales or use taxes for direct purchases made by the Authority. The Construction Manager shall notify the Authority of any assessment or proposed assessment. The indemnification contained in this **Subparagraph 3.4.4** shall survive beyond completion of Work under the Contract Documents.

### 3.5 AUTHORITY DIRECTION OF WORK AND INSPECTIONS

- **3.5.1** In no event shall the Authority have control over, charge of, or any responsibility for construction means, methods, techniques, sequences, or procedures, or for safety precautions and programs in connection with the Work, notwithstanding any of the rights and authority granted the Authority in the Contract Documents.
- **3.5.2** Inspection of the progress, quantity, or quality of the Work done by the Authority, any Authority's representative, any Governmental Agency, the Architect, or any inspector, shall not relieve the Construction Manager of any responsibility for the compliance of the Work with the Contract Documents. The Authority or its approved representative shall have access to the work site and all Work. No supervision or inspection by the Authority's representative, nor right of the Authority to act, nor any other actions taken by the Authority's representative shall relieve the Construction Manager of any of its obligations under the Contract Documents nor give rise to any duty on the part of the Authority.

## 3.6 AUTHORITY'S RIGHT TO RECORDS

- **3.6.1** The Construction Manager's records, which shall include but not be limited to accounting records, written policies and procedures, Subcontractor files (including proposals of successful bidders), original estimates, estimating work sheets, correspondence, schedules, change order files (including documentation covering negotiated settlements), and any other supporting evidence, in all cases solely related to the Construction Manager's performance of the Work or to substantiate charges related to this Construction Services Agreement (all foregoing hereinafter referred to as "**Records**") shall be open to inspection and subject to audit and/or reproduction, during normal working hours, by the Authority, Team and their respective agents or authorized representatives to the extent necessary to adequately permit evaluation and verification of any invoices, payments, or Claims submitted by the Construction Manager or any of its payees.
- **3.6.2** For the purpose of such audits, inspections, examinations, and evaluations, the Authority, Team, and their respective agents, or authorized representatives, shall have access to said Records from the effective date of the Construction Services Agreement for the duration of Work and until twelve (12) years (or longer if required by law) after the date of final payment by Authority to Construction Manager.
- **3.6.3** The Authority, Team, and their respective agents or authorized representatives shall have access during normal business hours to the Construction Manager's facilities, shall have access to all necessary records and shall be provided adequate and appropriate work space, in order to conduct audits in compliance with this **Article 3**. The Authority, Team, and their respective agents or authorized representatives shall give any entity to be audited reasonable advance notice of intended audits.

**3.6.4** Construction Manager shall require all Subcontractors, insurance agents, and Suppliers (payees) with cost plus contracts and not fixed price contracts to comply with the provisions of this **Article 3** by insertion of the requirements hereof in a written contract agreement between Construction Manager and its Subcontractors and Suppliers. Failure to obtain such written contracts which include such provisions shall be reason to exclude some or all of the related payee's costs from amounts payable to the Construction Manager pursuant to this Construction Services Agreement.

### ARTICLE 4 CONSTRUCTION MANAGER

#### 4.1 DEFINITION

4.1.1 [Reserved]

## 4.2 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONSTRUCTION MANAGER

- **4.2.1** The Construction Manager shall carefully study and compare the Contract Documents with each other and with information furnished by the Authority and shall at once report to the Authority and Architect errors, inconsistencies, or omissions discovered. Prior to execution of the Construction Management Plan including its GMP Proposal (Exhibit 3 to the Construction Services Agreement), the Construction Manager and each applicable Subcontractor shall have evaluated and satisfied themselves as to the conditions and limitations under which the Work is to be performed, including, without limitations, (i) the location, condition, layout, and nature of the Project Site and surrounding areas, (ii) generally prevailing climatic conditions, (iii) anticipated labor supply and costs, (iv) availability and cost of materials, tools, and equipment, and (v) other similar issues impacting the Work, and have calculated the GMP or the Contract Time in connection with any failure by the Construction Manager or any applicable Subcontractor to have complied with the requirements of this Subparagraph 4.2.1.
- 4.2.2 The Construction Manager shall take field measurements and verify field conditions and shall carefully compare such field measurements and conditions and other information known to the Construction Manager with the Contract Documents before commencing its Services or the Work. Errors, inconsistencies, or omissions discovered shall be reported to the Authority and Architect at once. In all cases of interconnection of its Work with existing or other work, including Vendor Work or Trade Contractor Work, Construction Manager shall verify at the Project Site all dimensions relating to such existing or other work. Any errors due to the Construction Manager's failure to so verify all such dimensions or locations shall be promptly rectified by the Construction Manager whether such errors are a reimbursable Cost of the Work shall be determined by Article 8 of the Construction Services Agreement without any additional cost to the Authority and such corrections shall not be considered an allowable Cost of Work; nor shall the Construction Manager be permitted to utilize its Contingency to cover the cost of such corrections. The Construction Manager shall promptly rectify any errors due to the Construction Manager's failure to so verify all such grades, elevations, locations or dimensions. The Construction Manager is responsible for coordination of the Work with the field verification of existing conditions to establish conformance with the Contract Documents. The Construction Manager's obligations herein include verifying dimensions and conditions of materials or equipment directly purchased by the Authority. The Construction Manager shall, therefore, satisfy itself as to the accuracy of all dimensions and locations shown on the Drawings and other Contract Documents and those dimensions and locations that are present at the Project Site and compare one to the other before starting the Work in the affected area.
- **4.2.3** The Construction Manager shall perform the Work in accordance with the Contract Documents and submittals approved pursuant to **Paragraph 4.12**.
- 4.2.4 Except as to any reported errors, inconsistencies, or omissions, concealed or unknown conditions defined in Subparagraph 6.2.7 hereof, and Changes to the Work as contemplated by Paragraph 9.2 hereof, by executing the Construction Management Plan in <u>Exhibit 3</u> of the Construction Services Agreement, the Construction Manager shall warrant and represent the following:
  - .1 (1) the Guaranteed Maximum Price is just and reasonable compensation for all the Work, (2) the Construction Manager Contingency is sufficient to address all foreseen and unforeseen conditions that will impact the Work and ability to complete the Work in accordance with the Guaranteed Maximum Price and Contract Time, (3) the Contract Time is adequate for the performance of the Work, and (4) the Construction Manager has a complete understanding of the scope of the Work and has been provided adequate time to establish the Guaranteed Maximum Price in sufficient detail to support the Guaranteed Maximum Price.
  - .2 The Work required by the Contract Documents, including all construction details, and construction means, methods, procedures, and techniques necessary to perform the Work are consistent with: (1) good and sound practices within the construction industry, (2) generally prevailing and accepted industry standards applicable to the Work, and (3) all Applicable Laws.

**4.2.5** The Construction Manager shall make available to the Authority and the Architect the results of any Project Site investigation, test borings, analyses, studies, or other tests conducted by or in possession of the Construction Manager. The Construction Manager represents that it is familiar with the Project Site and has reviewed all of the Contract Documents and other information provided by the Authority and Architect concerning the conditions of the Project Site and the Work. The Construction Manager shall have no claims for surface or subsurface conditions encountered, except as provided in **Subparagraph 6.2.6** hereof. The Construction Manager shall exercise special care in executing subsurface work in proximity of known subsurface utilities, improvements, and easements.

## 4.3 SUPERVISION AND CONSTRUCTION PROCEDURES

- **4.3.1** The Construction Manager shall properly supervise and direct the Work. The Construction Manager shall be solely responsible for and have control over all construction means, methods, techniques, sequences, and procedures and for coordination of the duties of all trades, unless the Contract Documents give other specific instructions concerning these matters.
- **4.3.2** The Construction Manager shall be solely responsible to the Authority for acts and omissions of the Construction Manager's employees, Subcontractors and their agents and employees, and any other Person performing portions of the Work under a contract with the Construction Manager.
- **4.3.3** The Construction Manager shall not be relieved of obligations to perform the Work in accordance with the Contract Documents by activities or duties of the Architect in the Architect's administration of the Work under the Contract Documents or by tests, inspections, or approvals required or performed by persons other than the Construction Manager.
- **4.3.4** The Construction Manager shall be responsible for continuous and thorough inspection of portions of Work already performed under the Construction Services Agreement to determine that such portions are in proper condition to receive subsequent Work.
- **4.3.5** If any of the Work is required to be inspected or approved by any Governmental Authority, the Construction Manager shall cause such inspection or approval to be performed. No inspection performed or failed to be performed by the Authority hereunder shall be a waiver of any of the Construction Manager's obligations hereunder or be construed as an approval or acceptance of the Work or any part thereof.

## 4.4 LABOR AND MATERIALS

- **4.4.1** Unless otherwise provided in the Contract Documents, the Construction Manager shall arrange for payment of labor, materials, equipment, tools, construction equipment and machinery, water, heat, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work. Should the Contract Documents require Work to be performed after regular working hours or should the Construction Manager elect to perform Work after regular working hours, the additional cost of such Work shall be borne by the Construction Manager. The Construction Manager shall check all materials and labor entering into the Project Site or incorporated into the Work and shall keep full detailed accounts thereof. The Construction Manager shall reject any materials that will not conform with the Contract Documents when properly installed.
- **4.4.2** The Construction Manager shall enforce strict discipline and good order among the Construction Manager's employees and other Persons carrying out the Work. The Construction Manager shall not permit employment of unfit Persons not skilled in tasks assigned to them.
- **4.4.3** The Construction Manager shall only employ labor in connection with the Work capable of working harmoniously with all trades, crafts, and any other Persons associated with the Project. The Construction Manager shall also use its best efforts to minimize the likelihood of any strike, work stoppage, or other labor disturbance.
  - .1 If the Work is to be performed by trade unions, the Construction Manager shall make all necessary arrangements to reconcile any conflict between the Contract Documents and any agreements or regulations of any kind at any time in force among members or councils which regulate or distinguish what activities shall not be included in the Work of any particular trade.
  - .2 Construction Manager shall negotiate and implement a Project Labor Agreement as approved by the Authority.
  - .3 In case the progress of the Work is affected by any undue delay in furnishing or installing any items or materials or equipment required under the Contract Documents because of a conflict involving any labor agreement or regulation, the Authority may require that other material or equipment of equal kind and quality be provided pursuant to a Contract Revision, Change Order, or Construction Change Directive.

- **4.4.4** If, after execution of the Construction Services Agreement and prior to submittal of applicable Shop Drawings, the Construction Manager desires to submit an alternate product in lieu of what has been specified or shown in the Contract Documents, the Construction Manager may do so in writing to the Architect, with a copy to the Authority, and setting forth the following:
  - .1 Full explanation of the proposed substitution and submittal of all supporting data including technical information, catalog cuts, warranties, test results, installation instructions, operating procedures, and other like information necessary for a complete evaluation of the substitution.
  - .2 Reasons the substitution is advantageous and beneficial to the Project, the Work, or the Construction Schedule in the event the substitution is acceptable.
  - .3 The adjustment, if any, in the GMP or Construction Schedule in the event the substitution is acceptable.
  - .4 A statement that (1) the proposed substitution conforms to and meets all the requirements of the pertinent Specifications and the requirements shown on the Drawings and (2) the Construction Manager will provide a warranty for the substituted material, equipment, or portion of the Work comparable to the originally specified material, equipment or portion of the Work. Proposals for substitutions shall be submitted in triplicate to the Architect and Authority in sufficient time to allow the Architect and Authority no less than ten (10) working days for review. No substitutions will be considered or allowed without the Construction Manager's submittal of complete substantiating data and information as stated hereinbefore.
- **4.4.5** Substitutions and alternates may be rejected by the Authority or Architect and will be considered only under one or more of the following conditions:
  - .1 Required for compliance with interpretation of code requirements or insurance regulations or suggestions.
  - .2 Unavailability of specified products, through no fault of the Construction Manager.
  - .3 Subsequent information discloses inability of specified products to perform properly or to fit in designated space.
  - .4 Manufacturer/fabricator refuses to certify or guarantee performance or specified product as required.
  - .5 In the judgment of the Authority or the Architect, a substitution would be in the best interests of the Authority to consider in terms of cost, time, or other considerations.
- **4.4.6** Whether or not any proposed substitution is accepted by the Authority, the Construction Manager shall reimburse the Authority for any reasonable fees charged by the Architect or other Consultants for evaluating each proposed substitute.

## 4.5 WARRANTY

- 4.5.1 The Construction Manager warrants to the Authority that the Work will be of good quality and new unless otherwise required or permitted by the Contract Documents, that the Work will be free from defects not inherent in the quality required or permitted, and that the Work will conform with the requirements of the Contract Documents. The Construction Manager represents and warrants to the Authority that its materials and workmanship, including its construction means, methods, procedures, and techniques necessary to perform the Work, and its use of materials, selection of equipment, and requirements of product manufacturers are and shall be consistent with: (1) good and sound practices within the construction industry; (2) generally prevailing and accepted industry standards applicable to the Work; (3) requirements of any warranties applicable to the Work; and (4) all Applicable Laws. Work not conforming to these requirements, including substitutions not properly approved and authorized, shall be considered defective, and promptly after notification of non-conformance shall be repaired or replaced by Construction Manager with Work conforming to this warranty. The Construction Manager's warranty excludes remedy for damage or defect caused by normal wear and tear under normal usage. If required by the Authority, the Construction Manager shall furnish satisfactory evidence as to the kind and quality of materials and equipment.
- **4.5.2** Construction Manager further warrants that all materials that affect the aesthetics of finishes or equipment of a category or classification will be a product of the same manufacturer and such materials or equipment shall be of the same lot, batch, or type and that such materials and equipment will be as specified in Contract Documents or as approved by Addendum or Contract Revision, as applicable.
- **4.5.3** The Construction Manager agrees to assign to the Authority at the time of Substantial Completion of the Work, any and all manufacturer's warranties relating to materials and labor used in the Work and further agrees to perform the Work in such manner so as to preserve any and all such manufacturer's warranties. Construction

Manager hereby agrees that such warranties shall be in full force and effect as of the Date of Substantial Completion, and the warranty periods shall commence as of the Date of Substantial Completion.

- 4.6 TAXES
  - 4.6.1 Except for direct purchases made by the Authority, and unless otherwise exempted by the Act, the Construction Manager shall pay appropriate sales, consumer, use, employment-related taxes, and similar taxes for Work or portions thereof provided by the Construction Manager which are legally enacted when bids are received or negotiations concluded, whether or not yet effective or merely scheduled to go into effect. The Project is exempt from sales tax to the extent provided under Section 6 of the Act, Minn. Stat. § 297A.71, Subd. 43. The Construction Manager is responsible for compliance with Minnesota Department of Revenue procedures to obtain the benefits of such exemption, as well exercising commercially reasonable efforts to qualify transactions under the Act's sales/use tax exemption to avoid or minimize the sales/use tax costs. Any and all tax rebates or refunds applicable to the Work shall accrue to the Authority.
  - **4.6.2** Within a reasonable period of time following a request from the Authority or Team, Construction Manager shall provide the Authority or Team with any information regarding quantities and descriptions of equipment installed on or ordered for the Project and any other information the Authority or Team may deem reasonably necessary in connection with the preparation of the tax returns or other tax documentation.

#### 4.7 PERMITS, FEES AND NOTICES

- **4.7.1** Unless otherwise provided in the Contract Documents, the Construction Manager shall secure, pay for, and, as soon as practicable, furnish the Authority with copies or certificates of all permits and fees, licenses and inspections necessary for the proper execution and completion of the Work. The Construction Manager shall procure all certificates of inspection, occupancy, permits, and licenses, pay all customary charges and fees including sewer access and water access charges, and give all notices necessary and incidental to the lawful prosecution of the Work. The Construction Manager shall deliver certificates of inspection, use, and occupancy to the Authority upon completion of the Work in sufficient time for occupancy of the Project in accordance with the Construction Schedule. The costs of such procurement, payment, and delivery are included within the GMP. Unless due to defects or deficiencies in the Contract Documents, failure to obtain such approvals shall not extend the Contract Time.
- **4.7.2** The Construction Manager shall comply with and give notices required by Applicable Laws. The Construction Manager shall procure and obtain all bonds or other security required of the Construction Manager by the Construction Services Agreement, the Act or as otherwise required by Applicable Laws. In connection with such bonds or other security, the Construction Manager shall prepare all applications, supply all necessary back-up material, and furnish the surety or other insurer with any required personal undertakings. The Construction Manager shall also obtain and pay all charges for all approvals for street closing, parking meter removal, and other similar matters as may be necessary or appropriate from time to time for the performance of the Work, all of which costs and charges are included in the GMP.
- **4.7.3** The Construction Manager shall not knowingly violate any Applicable Laws, including zoning, setback or other location requirements, temporary or conditional use permits, or of any recorded covenants of which the Construction Manager has knowledge. If the Construction Manager observes or recognizes that portions of the Contract Documents are at variance with Applicable Laws, the Construction Manager promptly shall notify the Authority and Architect in writing, and necessary changes shall be accomplished by appropriate Contract Revision.
- **4.7.4** If the Construction Manager performs Work knowing it to be contrary to Applicable Laws without such notice to the Authority and Architect, the Construction Manager shall assume full responsibility for such Work and shall bear any and all costs attributable to correct it and all damages resulting therefrom.

#### 4.8 ALLOWANCES

- **4.8.1** The Construction Manager shall include in the Guaranteed Maximum Price all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as approved by the Authority based on the Construction Manager's recommendation, but the Construction Manager shall not be required to employ persons or entities against which the Construction Manager makes reasonable objection.
- 4.8.2 Unless otherwise provided in **Exhibit 3** of the Construction Services Agreement:
  - .1 materials and equipment under an allowance shall be selected promptly by the Authority or Construction Manager, as the case may be, so as to avoid delay in the Work, and Construction Manager shall identify decisions needed from the Authority on the Construction Schedule and give Authority a reasonable time to make those decisions;

- .2 allowances shall cover the cost to the Construction Manager of materials, equipment, unloading, handling at the Project Site, labor, installation costs, overhead, profit, all required taxes, less applicable trade discounts, and other expenses contemplated and encompassed in the allowance;
- .3 the cost of estimated allowances, while not guaranteed, represent Construction Manager's best estimate; and
- .4 whenever allowable costs are more than or less than allowances, the GMP shall be adjusted accordingly by Contract Revision.

### 4.9 FIELD PERSONNEL

4.9.1 The Construction Manager shall employ a competent superintendent and project executive and, as necessary, competent staff who shall be in attendance at the Project Site during performance of the Work. The superintendent and project executive shall represent the Construction Manager, and communications given by or to the superintendent or project executive shall be as binding as if given by or to the Construction Manager. Important communications shall be similarly confirmed on written request in each case. The Construction Manager's superintendent, project executive, and staff shall be satisfactory to the Authority and Team in all respects, and the Authority and Team shall have the right to require Construction Manager to dismiss from the Project any superintendent, project executive, or staff with a superintendent, project executive, and staff shall be performance is not satisfactory, and to replace such superintendent, project executive, and staff with a superintendent, project executive or staff satisfactory to the Authority and Team in their sole discretion. The Construction Manager shall not replace the superintendent, project executive or staff without the written consent of the Authority and Team.

### 4.10 CONSTRUCTION MANAGER'S CONSTRUCTION SCHEDULE

- **4.10.1** The Construction Manager shall prepare in consultation with the Authority, Team and Architect and submit for the Authority's review and approval a Construction Schedule. The Construction Schedule shall not exceed time limits current under the Contract Documents, shall be updated and revised at appropriate intervals as required by the conditions of the Work and Project, shall be related to the entire Project to the extent required by the Contract Documents, shall provide for expeditious and practicable execution of the Work, and the critical path and Construction Milestone Dates shall not be modified or extended without the prior written approval of the Authority or as provided in the Contract Documents in each instance. The Construction Schedule shall include, but shall not be limited to, a schedule of primary trade activities and the delivery of or installation of critical materials or components, the failure of timely delivery of which will adversely impact the Project's completion as provided in the Contract Documents.
- **4.10.2** The Construction Manager shall prepare a monthly report in a form in sufficient detail, and of a character approved by the Authority (the "Progress Report"). The Progress Report shall specify, among other things, an estimated percentage of completion, whether the Project is in compliance with the Construction Schedule, and if not, the reasons therefor, and the proposed new Construction Schedule, as well as the number of man-days worked for each category of labor and the projected Work to be completed in the next succeeding month. Accompanying the Progress Report shall be an updated current Construction Schedule subject to the approval of the Authority, and a listing and the status of all Contract Revisions, Modifications, Bulletins, and other relevant documents. The Construction Manager shall additionally maintain a daily log (the "Daily Log") containing a record of weather, Subcontractors working on the Project, number of workers, Work accomplished, problems encountered, and other similar relevant data as the Authority may reasonably require. The log shall be available at all times to the Authority or the Architect. The Construction Manager shall prepare such additional reports as the Authority may reasonably request.
- **4.10.3** The Construction Manager shall hold weekly progress meetings at the Project Site or at such other time and frequency as the Authority requests. Progress of the Work shall be reported in detail with reference to the Construction Schedule. Each appropriate Subcontractor shall have present a competent representative to report the condition of its work and to receive information.

### 4.10.4 [Reserved]

- **4.10.5** The Construction Manager shall generally perform according to the most recent Construction Schedule; provided, however, that Construction Manager shall meet Construction Schedule Milestones unless otherwise agreed to in writing by the Authority.
- 4.10.6 The Construction Schedule shall be in a detailed format satisfactory to the Authority which shall also: (i) provide a 3-dimensional, BIM based graphic representation of all activities and events that will occur during performance of the Work; (ii) identify early and late start dates so that all "float" time can be accurately identified; (iii) provide relationships, interdependencies, and constraints among the various line items of Work; (iv) identify each phase of construction and occupancy for each phase; (v) set forth certain major dates by which certain activities must be accomplished that are critical in ensuring the timely and orderly completion of the Work in accordance with the requirements of the Contract Documents (hereinafter referred to as "Milestone Dates"); (vi) identifies, coordinates, and integrates the design and construction schedules; and (vii) identifies dates by which decisions must be made by the Authority and provides a reasonable amount of time for the

Authority to make such decisions. Upon its review and acceptance by the Authority, the Construction Schedule shall be deemed part of the Contract Documents. If not accepted, the Construction Schedule shall be promptly revised by the Construction Manager in accordance with the recommendations of the Authority and the Architect and re-submitted for acceptance by the Authority. The Construction Manager shall monitor the progress of the Work for conformance with the requirements of the Construction Schedule and shall promptly advise the Authority, Team and Architect of any delays or potential delays. The accepted Construction Schedule is each progress report to reflect actual conditions as set forth in **Subparagraphs 4.10.1** and **4.10.2** hereof. In the event any progress report indicates any delays, the Construction Manager shall propose an affirmative plan to correct the delay, including overtime and/or additional labor, if necessary. In no event shall any Progress Report constitute an adjustment in the Contract Time, any Milestone Date, or the GMP unless any such adjustment is agreed to by the Authority and authorized pursuant to a Contract Revision.

- 4.10.7 In the event the Authority determines that the performance of the Work has not progressed or reached the level of completion required by <u>Exhibit 4</u> to the Construction Services Agreement or the Construction Schedule, the Authority shall have the right, but not the obligation, to order the Construction Manager to take corrective measures necessary to expedite the progress of construction including (i) working additional shifts or overtime, (ii) supplying additional manpower, equipment, and facilities, and (iii) other similar measures (hereinafter referred to collectively as "Extraordinary Measures"). Such Extraordinary Measures shall continue until the progress of the Work complies in the Authority's right to require Extraordinary Measures is solely for the purpose of ensuring the Construction Manager's compliance with the Construction Schedule or as required by Paragraph 4.10 hereof.
  - .1 To the extent that the need for Extraordinary Measures is the responsibility or fault of the Contraction Manager, the Construction Manager shall not be entitled to an adjustment in the GMP or Contract Time in connection with Extraordinary Measures pursuant to this **Subparagraph 4.10.7**.
  - .2 The Authority may exercise the rights under or pursuant to this **Subparagraph 4.10.7** as frequently as the Authority deems necessary to ensure that the Construction Manager's performance of the Work will comply with any Milestone Date in the Construction Schedule.
- **4.10.8** The Authority shall have the right to direct a postponement or rescheduling of any date or time for the performance of any part of the Work that may interfere with the operation of the Authority's premises or any tenants or invitees thereof. **4.10.9** Float in the Construction Schedule shall be utilized for the benefit of the Project.

### 4.11 DOCUMENTS AND SAMPLES AT THE PROJECT SITE

**4.11.1** The Construction Manager shall maintain at the Project Site for the Authority one record copy of the Drawings, Specifications, Addenda, Change Orders, and other Modifications, in good order and marked currently to record changes and selections made during construction, and in addition approved Shop Drawings, Product Data, Samples, and similar required Submittals. These shall be available to the Authority, Team and Architect and shall be delivered to the Architect for submittal to the Authority upon completion of the Work.

## 4.12 SHOP DRAWINGS, PRODUCT DATA, AND SAMPLES

- **4.12.1** Shop Drawings are drawings, diagrams, schedules, and other data specially prepared for the Work by the Construction Manager or Subcontractors, Sub-Subcontractors, Supplier, manufacturer, or distributor to illustrate some portion of the Work.
- **4.12.2** Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished to illustrate materials or equipment for some portion of the Work.
- **4.12.3** Samples are physical examples that illustrate materials, equipment, or workmanship and establish standards by which the Work will be judged.
- **4.12.4** Product Data, Samples, and similar Submittals other than Shop Drawings are not Contract Documents. The purpose of the Submittal is to demonstrate for those portions of the Work for which Submittals are required the way the Construction Manager proposes to conform to the information given and the design concept expressed in the Contract Documents.
- **4.12.5** The Construction Manager shall review, approve, and submit to the Authority and Architect Shop Drawings, Product Data, Samples, and similar Submittals required by the Contract Documents with reasonable promptness, but in no event less than ten (10) days after receipt of same, and in such sequence as to cause no delay in the Work or in the activities of the Authority or any Trade Contractors or Vendors. When and as requested by the Authority, Construction Manager and Architect shall tender an agreed-upon schedule describing all Submittals, identifying the dates when Submittals are to be given to the Architect, establishing time for Architect's review of Submittals, and re-submission, if any, of Submittals by Construction Manager to

Architect (the "Submittal Schedule"). The Authority will be provided with a copy of all approved Shop Drawings, Product Data, Samples and similar Submittals for the Project file.

- **4.12.6** The Construction Manager shall perform no portion of the Work requiring submittal and review of Shop Drawings, Product Data, Samples, or similar Submittals until the respective Submittal has been reviewed and approved by the Architect. Such Work shall be in accordance with approved Submittals.
- **4.12.7** By approving and submitting Shop Drawings, Product Data, Samples, and similar Submittals, the Construction Manager represents that the Construction Manager has determined and verified materials, field measurements, and field construction criteria related thereto, or will do so, and has checked and coordinated the information contained within such Submittals with the requirements of the Work and of the Contract Documents.
- **4.12.8** The Construction Manager shall not be relieved of responsibility for deviations from requirements of the Contract Documents by the Architect's approval of Shop Drawings, Product Data, Samples, or similar Submittals unless the Construction Manager has specifically informed the Architect in writing of the existence and nature of such deviation at the time of submittal and the Architect has given written approval to the specific deviation. The Construction Manager shall not be relieved of responsibility for errors or omissions in Shop Drawings, Project Data, Samples, or similar Submittals by the Architect's approval thereof.
- **4.12.9** The Construction Manager shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data, Samples, or similar Submittals, to revisions other than those requested by the Architect, or previous Submittals.
- **4.12.10** Informational Submittals upon which the Architect is not expected to take responsive action may be so identified in the Contract Documents.
- **4.12.11** When professional certification of performance criteria of materials, systems, or equipment is required by the Contract Documents, the Construction Manager shall provide the person or parties providing the certification with full information on the specified performance requirements of the materials, systems, or equipment. The certification shall be based on performance under the operating conditions identified in the Contract Documents. The Authority and Architect shall be entitled to rely upon the accuracy and completeness of such certificates.
- **4.12.12** All Shop Drawings, Product Data, Samples, or similar Submittals must be submitted to, and approved by, the Architect. The Construction Manager shall represent and warrant that all Shop Drawings shall be prepared by persons and entities possessing expertise and experience in the trade for which the Shop Drawing is prepared and, if required by the Contract Documents or Applicable Laws, by a licensed design professional.
- **4.12.13** The Architect shall be responsible to review and approve Shop Drawings, Product Data, Samples, or similar Submittals for conformance with the design requirements and criteria set forth in the Contract Documents. The Architect shall promptly review all such Shop Drawings, Product Data, Samples, or similar Submittals and provide the Construction Manager with written notice of the Architect's approval or rejection, no less than ten (10) days after receipt of same, and in such sequence as to cause no delay to the critical path of the Work as set forth in the Construction Schedule or in the activities of the Authority or separate Trade Contractors or Vendors unless a shorter period of time is required by the Submittal Schedule. It is the responsibility of the Architect and Construction Manager to coordinate the schedule and sequence for review and approval of all Submittals in accordance with the Contract Documents and to work in a cooperative fashion to avoid any delays in the processing of Submittals. The Authority will assume no responsibility or liability associated with delays in the review and/or approval of Submittals by either the Construction Manager, Architect, or any other member of the Project Team.
- 4.12.14 The Architect will review and approve or take other appropriate action upon the Construction Manager's Submittals such as Shop Drawings, Product Data and Samples, for the purpose of: (1) verifying compliance with Applicable Laws and the NFL Rules and Regulations; and (2) confirming that such Submittals are in compliance with the requirements of the Contract Documents. Architect shall be responsible for determining what aspects of the Work shall be the subject of Shop Drawings and Submittals. Architect shall promptly inform the Authority when the Construction Manager, Subcontractor or a Trade Contractor has proceeded with any aspect of the Work in the absence of approved Shop Drawings and Submittals. Review of such Submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Construction Manager as required by the Contract The Architect's review of the Construction Manager's Submittals shall not relieve the Documents. Construction Manager of the obligations under this Paragraph 4.12. The Architect's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Architect, of any construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

## 4.13 USE OF PROJECT SITE

- **4.13.1** The Construction Manager shall confine operations at the Project Site to areas permitted by Applicable Laws, the Contract Documents, or as reasonably determined by the Authority, and shall not unreasonably encumber the Project Site with materials or equipment.
- **4.13.2** Only materials and equipment that are to be used directly in the Work shall be brought to and stored on the Project Site by the Construction Manager. After equipment is no longer required for the Work, it shall be promptly removed from the Project Site. Construction Manager will periodically review its onsite equipment usage with the Authority and if any piece of equipment is frequently idle, the Construction Manager shall remove the equipment in question if the demobilization and remobilization charges are less than the charged rental for said equipment; equipment that are part of or required for the Work and stored at the Project Site is solely the responsibility of the Construction Manager.
- **4.13.3** The Construction Manager shall erect signage, as approved by the Authority, describing the Project and its participants at the Project Site, the cost of which shall be included in the Construction Manager's GMP. No party, including the Construction Manager, Subcontractors, Suppliers, the Architect or its Subconsultants shall erect any signage at the Project Site without the Authority's prior written approval. This includes signage on equipment, office trailers, and storage trailers. The Construction Manager shall be responsible to provide and erect any and all signage required for Project safety and related programs, with such signage being approved by the Authority in writing.
- 4.13.4 Construction Manager shall ensure that the Work, at all times, is performed in a manner that affords reasonable access, both vehicular and pedestrian, to the Project Site of the Work and all adjacent areas. The Work shall be performed, to the fullest extent reasonably possible, in such a manner that public and private areas adjacent to the Project Site of the Work shall be free from all debris, building materials, and equipment likely to cause hazardous conditions. Without limitation of any other provision of the Contract Documents, Construction Manager shall not interfere with the occupancy or beneficial use of (1) any areas and buildings adjacent to the Project Site or (2) the Stadium in the event of partial occupancy, as more specifically described in **Paragraph 11.9**. In no event shall the Construction Manager, Subcontractors, or Suppliers be present or perform Work on the Project Site for a period that is 24 hours in advance of a home football game of the Team or during such game.
- **4.13.5** Without prior approval of the Authority, the Construction Manager shall not permit any workers to use any existing facilities at the Project Site, including without limitation, lavatories, toilets, entrances, and parking areas other than those designated by the Authority. Without limitation of any other provision of the Contract Documents, the Construction Manager shall comply with all rules and regulations promulgated by the Authority in connection with the use and occupancy of the Project Site and the building, as amended from time to time. The Construction Manager shall immediately notify the Authority in writing if during the performance of the Work, the Construction Manager finds compliance with any portion of such rules and regulations to be impracticable, setting forth the problems of such compliance and suggesting alternatives through which the same results intended by such portions of the rules and regulations can be achieved. The Construction Manager shall also comply with all insurance requirements set forth in the Contract Documents and collective bargaining agreements applicable to use and occupancy of the Project Site.
- **4.13.6** The Construction Manager shall at all times, cooperate, coordinate, and integrate with any Trade Contractors and Vendors employed by the Authority with respect to their schedules so that the Construction Manager causes no unreasonable interferences with, delays or disruptions to the Trade Contract Work and Vendor Work, and so that the Work is timely completed to permit the Project to be completed on schedule.

#### 4.14 CUTTING AND PATCHING

- **4.14.1** The Construction Manager shall be responsible for cutting, fitting, or patching required to complete the Work or to make the Work fit together properly. The Construction Manager shall be responsible for cutting, patching, repairing, and cleaning of any and all roads, streets and walkways providing access to the Project Site that may be damaged during construction.
- **4.14.2** The Construction Manager shall not damage or endanger a portion of the Work, a Trade Contractor's Work, Vendor's Work, or any other construction fully or partially completed by the Authority, a Trade Contractors, or a Vendor by cutting, patching, or otherwise altering such construction, or by excavation. The Construction Manager shall not cut or otherwise alter a Trade Contractor's Work, Vendor's Work, or any other such construction by the Authority except with written consent of the Authority and the applicable Trade Contractor or Vendor; such consent shall not be unreasonably withheld. The Construction Manager shall not unreasonably withhold from the Authority, Trade Contractor, or Vendor the Construction Manager's consent to cutting or otherwise altering the Work.
- **4.14.3** Altering or cutting of structural members will not be allowed without written approval by the Architect. The Construction Manager shall in all cases exercise extreme care in cutting operations and perform such operations under adequate supervision by competent mechanics skilled in the applicable trade. Openings shall be neatly

cut and shall be kept as small as possible to avoid unnecessary damage. Careless, avoidable cutting or damage is not permitted and the Construction Manager will be held responsible for such avoidable damage.

# 4.15 CLEANING UP

- **4.15.1** The Construction Manager shall keep the premises and all surrounding areas to the Project Site free from (i) accumulation of waste materials, debris, or rubbish; (ii) the Construction Manager's tools, construction equipment, machinery; and (iii) surplus materials, and any excess mud, gravel, or earth caused by operations under the Construction Services Agreement. At completion of the Work, the Construction Manager shall remove from and about the Project waste materials, debris, rubbish, the Construction Manager's tools, construction equipment, machinery, surplus materials, and any excess mud, gravel, or earth from and around the Project and the surrounding rights of way.
- **4.15.2** If the Construction Manager fails to clean up as provided in the Contract Documents, the Authority may, but shall not be obligated to, perform the Construction Manager's clean up and charge the cost thereof to the Construction Manager.

## 4.16 ACCESS TO WORK

**4.16.1** The Construction Manager shall provide the Authority, Team, their respective agents and representatives, and Architect access to the Work in preparation and progress wherever located.

### 4.17 ROYALTIES AND PATENTS

**4.17.1** The Construction Manager shall pay all royalties and license fees. The Construction Manager shall defend suits or claims for infringement of patent rights and shall hold the Authority, and all Indemnitees harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process, or product of a particular manufacturer or manufacturers is required by the Contract Documents. However, if the Construction Manager has reason to believe that the required design, process, or product is an infringement of a patent, the Construction Manager shall be responsible for such loss unless such information is promptly furnished to the Authority and the Architect.

## 4.18 INDEMNIFICATION BY CONSTRUCTION MANAGER

- 4.18.1 Construction Manager hereby agrees, to the fullest extent permitted by Applicable Laws, to indemnify, hold harmless, and defend the Authority and Indemnitees (except the Architect Indemnitees with respect to design errors or omissions or breaches of the Design Services Agreement) from and against any and all claims, damages, liabilities, losses and expenses (specifically including attorneys' fees and costs, court fees and costs, and arbitration fees and costs incurred to defend the Authority and Construction Manager's liability shall not be limited by the amount or limit of insurance Construction Manager procures to insure its obligations to the Authority or Indemnitees), provided that such claims, damages, losses or expenses arise out of bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), but only to the extent arising out of, resulting from, or incurred in connection with the performance of the Work and to the extent in whole or in part caused or alleged to have been caused by the negligent or grossly negligent acts or omissions, of the Construction Manager, Construction Manager's employees, Subcontractors, Sub-Subcontractors, Suppliers or any Person for whom the Construction Manager is legally responsible in the performance of the Work under this Construction Services Agreement, whether arising before or after completion of the Work and arising out of, resulting from, or occurring in connection with the performance of the Work or any activity associated with the Work, from any activity of the Construction Manager, Construction Manager's employees, its Subcontractors, Sub-Subcontractors, Suppliers, their agents, or employees at the Project Site or elsewhere, and whether in part caused by the active or passive negligence or other fault of a Party or Person indemnified hereunder, excepting only personal injury to person or damage to property found to have been caused by the sole negligence of a Party or Person indemnified hereunder. Such obligations shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity that would otherwise exist as to a Party or Person described in this Subparagraph 4.18.1.
  - .1 Construction Manager acknowledges and agrees that it has an independent obligation under this Construction Services Agreement to procure and maintain such insurance as will insure Construction Manager's obligations in this Construction Services Agreement, and further acknowledges that such insurance is commercially available. Construction Manager's liability shall not be limited by the amount or limit of insurance Construction Manager procures to insure its obligations herein.
- **4.18.2** In addition to its duties under Paragraph 4.18.1 herein, the Construction Manager shall defend, indemnify, and hold harmless the Authority and the Indemnitees from and against any and all alleged claims, damages, liabilities, losses and expenses (specifically including attorneys' fees and costs, court fees and costs, and arbitration fees and costs incurred to defend the Authority and Indemnitees) caused by Construction Manager's breach of this Construction Services Agreement to the extent finally determined. For the purpose of this **Subparagraph 4.18.2**, the Authority's or an Indemnitee's tender shall not be deemed to require proof of the Construction Manager's breach of contract at the time of tender.

- **4.18.3** In the case of claims against any Party or Person indemnified under this **Paragraph 4.18** by an employee of the Construction Manager, a Subcontractor, anyone directly or indirectly employed by them, or anyone for whose acts they may be liable, the indemnification obligation under this **Paragraph 4.18** shall not be limited by a limitation on amount or type of damages, compensation, or benefits payable by or for the Construction Manager or a Subcontractor under workers' compensation acts, disability benefit acts, or other employee benefit acts.
- **4.18.4** The Construction Manager shall indemnify the Authority, Indemnitees, and the respective directors, officers, agents, and employees of any of them from and against any and all costs and expenses, specifically including attorneys' fees and costs, court fees and costs, and arbitration fees and costs, incurred by any of the foregoing Parties or Persons to enforce any of the Construction Manager's defense, indemnity, and hold-harmless obligations under the Construction Services Agreement.
- **4.18.5** The Construction Manager's obligation to defend as set forth in **Subparagraphs 4.18.1 and 4.18.2** includes the obligation to provide and pay for attorneys to defend the Party or Persons entitled to indemnification under **Subparagraphs 4.18.1** and **4.18.2**, which attorneys shall be subject to the approval of such Party and Persons; and if such Party or Persons do not approve the attorneys provided by the Construction Manager, the Construction Manager shall promptly pay the attorneys fees incurred by the Party or Persons entitled to indemnification.

**4.19 WARRANTIES AND REPRESENTATIONS.** The Construction Manager represents and warrants the following to the Authority and Team, in addition to any other representations and warranties contained in the Contract Documents, as an inducement to the Authority to execute the Construction Services Agreement, which representations and warranties shall survive the completion of the Work, any termination of the Construction Services Agreement, and the Final Completion of the Work:

- **4.19.1** that it is financially solvent, able to pay all debts as they mature, and possesses sufficient working capital to complete the Work and perform all obligations hereunder;
- **4.19.2** that it is able to furnish the plant, tools, materials, supplies, equipment, and labor required to complete the Work and perform its obligations hereunder;
- **4.19.3** that it is authorized to do business in the State of Minnesota and properly licensed by all necessary Governmental Authorities;
- **4.19.4** that its execution of the Construction Services Agreement and its performance thereof is within its duly authorized powers;
- **4.19.5** that its duly authorized representative has visited the Project Site, is familiar with the local conditions under which the Work is to be performed, and has correlated on site observations with the requirements of the Contract Documents;
- **4.19.6** that it has examined the Contract Documents carefully and thoroughly and accepts the responsibilities set forth therein and the obligation to complete the Work in accordance with the Contract Documents;and
- **4.19.7** that it possesses a high level of experience and expertise in the business administration, construction, construction management, costing, scheduling and superintendence of projects of the size, complexity, and nature of this particular Project, and that it will perform the Work with the Standard of Care required of the Construction Manager.

The foregoing warranties are in addition to, and not in lieu of, any and all other liability imposed upon the Construction Manager by law with respect to the Construction Manager's duties, obligations and performance hereunder. All representations and warranties set forth in the Construction Services Agreement and these General Conditions, including this **Paragraph 4.19**, shall survive the Final Completion of the Work or the earlier termination of the Construction Services Agreement. The Construction Manager acknowledges that the Authority is relying upon the Construction Manager's skills, experience, and promised Standard of Care in connection with the Work described herein.

# 4.20 CONSTRUCTION MANAGER SELF PERFORMED WORK

- **4.20.1** To the extent permitted by Applicable Laws, Construction Manager or its Affiliates shall be permitted to bid only on those categories of the Work approved in writing by the Authority. Such approval shall generally only be provided for portions of the Work that (1) the Authority, in consultation with the Construction Manager, determines are critical to the Construction Schedule, (2) the Construction Manager demonstrates to the satisfaction of the Authority that the Construction Manager regularly performs such type of work with its own forces, and (3) the Authority determines that the Construction Manager's performance under the Contract Documents, including performance with respect to the Contract Time and GMP, are materially enhanced due to the Construction Manager self-performing such portions of the Work.
- **4.20.2** Construction Manager, or its Affiliates, shall be permitted to submit to the Authority a sealed bid for any Work approved by the Authority as potential Self-Performed Work pursuant to the competitive bidding procedures applicable to all bidders. In such instance, the opening, review and advice with respect to award or rejection of such bids shall be managed by the Authority and its designated representative. In such instances, the following

requirements shall also apply: (1) if Construction Manager, or its Affiliates, desire to bid on Self-Performed Work, then Construction Manager, or its Affiliates, shall review such Work (including the bid packaging plan) with the Authority and Architect prior to finalizing the bid package; (2) the Construction Manager will use a separate crew and supervisory personnel to perform any Self Performed Work and not its supervisory personnel for its duties under the Construction Services Agreement; there shall be a strict separation of the personnel involved with bidding the Self-Performed Work and Construction Manager's other personnel involved in the Project, and Construction Manager shall, by written policy distributed to all affected personnel (a copy of which shall be delivered to the Authority), strictly prohibit any communication prior to bid award among personnel involved with the estimating, bidding, management or other services in connection with the Self-Performed Work and personnel working on other aspects of this Project pursuant to this Agreement (other than such communication as is permitted by all bidders); (3) if fewer than two other bids from responsible bidders are submitted for Self-Performed Work, the Authority, at its option, may disqualify Construction Manager, or its Affiliates, from award of the bid for Self-Performed Work and, in the Authority's discretion, may cause the bid package with respect to the Self-Performed Work to be re-bid; (4) the Authority shall determine the successful bidder for any Self-Performed Work; (5) the Authority will provide the same bid packages to all bidders or proposers for any opportunity for Self-Performed Work; (6) Construction Manager shall not be permitted to use Construction Contingency for Self-Performed Work; (7) Construction Manager, or its Affiliates, shall not, in its bid, use any of the General Conditions GMP to support the Self-Performed Work or use the General Conditions GMP for Self-Performed Work on any terms or conditions different from the terms or conditions on which such General Conditions are made available to all other bidders; and (8) the solicitation for bids on Self-Performed Work shall specifically state that Construction Manager, or its Affiliates, shall have the right to submit a sealed bid on Self-Performed Work. If these procedures are not strictly followed, then the Authority shall have the right to reject the bid of Construction Manager or its Affiliates for Self-Performed Work.

4.20.3 In addition, (1) if fewer than two (2) competitive bids from responsible bidders (in addition to the bid of Construction Manager or its Affiliate) are submitted for the Self-Performed Work, or (2) if the bid by Construction Manager or its Affiliates for any Self-Performed Work is higher than the most recent estimate of the Cost of the Work for such Self-Performed Work, as of the day before the applicable bid opening (which estimate, may at the Authority's discretion, be provided by a third party), then the Authority shall have the right, in its reasonable discretion, to reject the bid of Construction Manager or its Affiliate (even if its bid is the low bid) and require the Self-Performed Work to be re-bid. Any rejection of a bid or required re-bid under this Subparagraph 4.20.3 shall not be the basis for an increase in the GMP or adjustment to the Construction Schedule.

#### 4.21 PROGRESS MEETINGS; REPORTS

**4.21.1** The Construction Manager shall schedule and conduct periodic meetings, at intervals appropriate for the stage of construction, at which the Authority and the Team and their respective agents and representatives, the Architect, the Construction Manager, and appropriate Subcontractors and Suppliers may discuss the status of the Work. The Construction Manager shall prepare and promptly distribute detailed meeting minutes as required by the Authority.

### 4.21.2 [Reserved]

## 4.22 WEB BASED PROJECT MANAGEMENT SYSTEM

- **4.22.1** If the Authority elects, in its sole discretion, to utilize a web-based or other project management system for the Project, the Construction Manager shall each participate to the fullest extent possible in the utilization of the system.
- **4.22.2** All costs for the Construction Manager's, and Subcontractors' use of the project management system shall be included in the GMP and additional compensation shall not be permitted.
- **4.22.3** If the Construction Manager's staff (or its Subcontractors) are unfamiliar with the proper use of the project management system, the Construction Manager (and its Subcontractors) shall provide its staff for training without additional compensation under the Contract Documents.

#### ARTICLE 5 ARCHITECT

### 5.1 ARCHITECT

**5.1.1** Architect's Duties. The duties of the Architect with respect to the Project are found in the Design Services Agreement by and between the Authority and Architect and as otherwise set forth in other Articles of these General Conditions.

## ARTICLE 6 ADMINISTRATION OF THE CONTRACT

### 6.1 ADMINISTRATION OF THE CONTRACT

6.1.1 The Architect will be responsible to coordinate, manage and administer all Design Services of the Architect, the Subconsultants, and any Consultants employed directly by the Authority pursuant to the Design Services Agreement. The Construction Manager will be responsible to coordinate, manage and administer all of the Work, including Work performed by Subcontractors and Suppliers of all tiers. The Construction Manager is further responsible to manage and administer the coordination of the Trade Contractor Work and Vendor Work to be performed by any Trade Contractors or Vendors and to perform services related to such Trade Contractors and Vendors that are set forth in the Construction Manager's and administration of the construction Manager's and administration of the Construction Manager's and administration of the coordination of the Trade Contractors to comply with the Construction Manager's and administration of the coordination of the Trade Contractor Work and Vendor Work.

#### 6.1.2 [Reserved]

- 6.1.3 Communications Facilitating Contract Administration. The Authority will communicate with the Architect and its Subconsultants through its designated representative. Written communications by the Construction Manager with Subconsultants shall be through the Architect. Communications by the Architect with Subcontractors, Sub-Subcontractors and material suppliers shall be through the Construction Manager.
- **6.1.4** Based on the Architect's observations and evaluations of the Construction Manager's Application for Payment, the Authority shall review the amounts due the Construction Manager and shall issue a Certificate of Payment as deemed appropriate by the Authority in such amounts. The Architect will be responsible to review the Applications for Payment submitted by the Construction Manager and to provide the Authority with certification that the Work included in the Construction Manager's Application for Payment has been completed.
- 6.1.5 The Architect shall recommend the rejection of Work that the Architect discovers does not conform to the Contract Documents. However, the Architect shall not have the authority to stop the Work on the Project Site. Whenever the Architect considers it necessary or advisable for implementation of the intent of the Contract Documents, the Architect will have authority to require additional inspection or testing of the Work in accordance with Subparagraphs 15.5.2 and 15.5.3 hereof, whether or not such Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Construction Manager, Subcontractors, Suppliers, their agents or employees, or other Persons performing portions of the Work.
- 6.1.6 As and when directed, the Architect will prepare Construction Change Directives, subject to the approval of the Authority, and may authorize minor changes in the Work as provided in **Paragraph 9.4** hereof.
- **6.1.7** The Architect shall conduct inspections to determine the date or dates of Substantial Completion and the date of Final Completion. The Architect will receive and review any and all records and written warranties and related documents required by the Construction Services Agreement or the Contract Documents and assembled by the Construction Manager. The Architect will forward all such documentation to the Authority upon the Architect's review and acceptance. The Architect shall issue a final Certificate for Payment upon compliance with the requirements of the Contract Documents.
- 6.1.8 The Architect will review and approve or take other appropriate action upon the Construction Manager's Submittals such as Shop Drawings, Product Data and Samples. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Construction Manager as required by the Contract Documents. The Architect's review of the Construction Manager's submittals shall not relieve the Construction Manager of the obligations under Paragraphs 4.3, 4.5 and 4.12 hereof. The Architect's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Architect, of any construction means, methods, techniques, sequences, or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.
- **6.1.9** Upon request of the Authority, Claims, disputes and other matters in question relating to the execution or progress of the Work or the interpretation of the Contract Documents may be referred by the Authority to the Architect only for initial recommendation, which the Architect shall render in writing within a reasonable time, not to exceed fifteen (15) days after the date on which such request is made.
- **6.1.10** Interpretations and recommendations of the Architect will be consistent with the intent of and reasonably inferable from the Contract Documents and will be in writing or in the form of Drawings. When making such interpretations and recommendations, the Architect will endeavor to secure faithful performance by Authority and Construction Manager or Trade Contractor, and will not show partiality to any of them.

#### 6.1.11-6.6.14 [Reserved]

- **6.1.15** The Architect shall not be responsible for, nor have control or charge of, construction means, methods, techniques, sequences, or procedures, or for safety precautions and programs in connection with the Project, and shall not be responsible for the Construction Manager's failure to carry out the Work in accordance with the Contract Documents, provided such failure of the Construction Manager is not caused by the negligent or willful act of the Architect. The Architect shall not be responsible for, nor have control over, the acts or omissions of the Construction Manager, Subcontractors, and any of their agents or employees, or any other Persons performing any of the Work.
- 6.1.16 [Reserved]

### 6.2 CLAIMS AND DISPUTES

- **6.2.1 Definition.** A Claim is a demand or assertion by a Person seeking as a matter of right, adjustment or interpretation of terms of an agreement, payment of money, extension of time, or other relief with respect to the terms of the Contract Documents. The term "Claim" also includes other disputes and matters in question between the Authority and the Architect, between the Architect and the Construction Manager, or between the Authority and the Construction Manager arising out of or relating to the Contract Documents. Claims must be made by written notice provided by the claimant to the Party against whom the Claim is being made, with a copy to the SDC Group and the Authority. The responsibility to substantiate Claims shall rest with the party making the Claim.
- **6.2.2** Claims, including those alleging an error or omission by the Architect or Construction Manager will be referred initially to the SDC Group for action as provided in **Paragraph 6.3** hereof. The recommendation by the SDC Group, if any, in response to a Claim shall not be a condition precedent to arbitration or litigation.
- 6.2.3 Time Limits on Claims. Except as provided in Subparagraph 6.2.6 below, and except for Claims asserted by the Authority, Claims by any Person that are discovered during the performance of the Work must be made within twenty-one (21) days after occurrence of the event giving rise to such Claim or within twenty-one (21) days after the claimant first recognizes the condition giving rise to the Claim, whichever is later, or said Claim shall be deemed waived. In addition to the foregoing, as soon as the claimant recognizes the conditions giving rise to the Claim, claimant shall provide notice of such conditions to the Authority, Architect, and SDC Group and shall use its best efforts to cooperate with the Authority and SDC Group and the Party against whom the Claim is made in any effort to mitigate the alleged or potential damages, delay, or other adverse consequences arising out of the condition which is the cause of such a Claim. Any additional Claim related to the initial Claim that is made after the initial Claim has been adjusted and dealt with by Contract Revision will not be considered. Claims may also be reserved in writing within the time limits set forth in this Subparagraph 6.2.3. If a Claim is reserved, the Resolution of Claims and Disputes procedures described in Paragraph 6.3 and Paragraph 6.4 hereof shall not commence until a written notice from the claimant is received by the Authority. Any notice of Claim or reservation of Claim must clearly identify the alleged cause and the nature of the Claim and include data and information then available to the claimant that will facilitate prompt verification and evaluation of the Claim, and it shall be assumed that the Authority has been prejudiced if it does not receive said information.
- **6.2.4 Continuing Contract Performance.** Pending final resolution of a Claim, unless otherwise agreed in writing, the Architect and Construction Manager shall proceed diligently with performance of their respective Design Services and Work, and the Authority shall continue to make payments in accordance with the Contract Documents. In the event the Architect or Construction Manager fails to diligently proceed with their respective Design Services or Work during a period of dispute resolution, the Authority may procure necessary replacement services or work and back-charge all associated costs to the non-performing party.
- 6.2.5 Waiver of Claims: Progress and Final Payment. The making of a progress payment by the Authority shall constitute a waiver of any right to make Claims by the Construction Manager occurring prior to the time covered by such progress payment except those Claims that have been submitted or reserved pursuant to this **Paragraph 6.2**. The making of final payment shall constitute a waiver of Claims by the Construction Manager except those Claims that may either arise from events occurring after receipt of final payment by the Authority or Claims which have been submitted or reserved pursuant to this **Paragraph 6.2**.
- 6.2.6 Injury or Damage to Person or Property. If during the performance of the Work any member of the Project Team suffers injury or damage to person or property because of an act or omission of any other member of the Project Team, such other party's employees or agents, or others for whose acts such member of the Project Team is legally liable, written notice of such injury or damage, whether or not insured, shall be given to the other member of the Project Team within a reasonable time not exceeding twenty-one (21) days after first observance of the injury or damage. The notice shall provide sufficient detail to enable the other members of the Project Team to investigate the matter.
- **6.2.7** Claims for Concealed or Unknown Conditions. Notwithstanding anything to the contrary in the Contract Documents, if conditions are encountered at the Project Site which are (1) subsurface or otherwise concealed

physical conditions which differ materially from those indicated in the Contract Documents and (2) unknown physical conditions of an unusual nature, which differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, then notice by the observing party shall be given to the other party promptly before conditions are disturbed and in no event later than fourteen (14) days after first observance of the conditions. The Authority, with the assistance of the Architect, shall promptly investigate such conditions and, if they meet the conditions described above and cause a material increase or decrease in the GMP or Contract Time for performance of any part of the Work, the Construction Manager shall be entitled to a Contract Revision providing an equitable adjustment in the GMP or Contract Time, or both. If the Authority, with the assistance of the Architect, determines that the conditions at the Project Site do not meet the conditions described above and that no changer in writing, stating the reasons. If the Authority and Construction Manager cannot agree to an adjustment in the GMP or Contract Time, the adjustment shall be subject to further proceedings pursuant to **Paragraph 6.4** hereof.

6.2.8 Claims for Additional Cost. If the Construction Manager wishes to make a Claim for an increase in the GMP, written notice to the Authority as strictly set forth under this Paragraph 6.2 is required and shall be given before the Construction Manager proceeds to execute the Work or the Construction Manager's Claim shall be deemed waived. Any Work completed by the Construction Manager without the proper notice to and approval by the Authority will not be considered or approved as a change to the GMP. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Paragraph 12.3 hereof. If the Construction Manager believes additional cost above the GMP are involved for reasons including but not limited to (1) written interpretation from the Architect, (2) an order by the Authority to stop the Work where the Construction Manager was not at fault, (3) a written order for a minor change in the Work issued by the Architect, (4) other reasonable grounds, the Claim shall be filed in accordance with the procedure established herein.

## 6.2.9 Claims for Additional Time.

- .1 If the Construction Manager wishes to make a Claim for an increase in the Contract Time, written notice as provided herein shall be given within the time period specified in **Subparagraph 6.2.3** hereof or said Claim shall be deemed waived. The Construction Manager's Claim shall include an estimate of cost and of probable effect of delay on progress of the Work and the Construction Schedule. In the case of a continuing delay only one Claim is necessary.
- .2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time and could not have been reasonably anticipated, and that weather conditions had an adverse effect on the Construction Schedule.
- 6.2.10 Claims for Consequential Damages. The Construction Manager and the Authority waive Claims against each other for consequential damages arising out of or relating to this Contract. This mutual waiver includes
  - .1 damages incurred by the Authority for rental expenses, for losses of use, income, profit, financing, business and reputation, and for loss of management or employee productivity or of the services of such persons; and
  - .2 damages incurred by the Construction Manager for principal office expenses including the compensation of personnel stationed there, for losses of financing, business and reputation, and for loss of profit except anticipated profit arising directly from the Work.

This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination in accordance with **Article 16** below. Nothing contained in this **Subparagraph 6.2.10** shall be deemed to preclude an award of liquidated damages, when applicable, in accordance with the requirements of the Contract Documents. In addition, this **Subparagraph 6.2.10** does not apply to or preclude the Authority's right to recover consequential damages to the extent provided in **Paragraph 4.2** of the Construction Services Agreement. Notwithstanding anything to the contrary, the Authority does not waive and specifically reserves all rights and claims to recover any and direct damages, indirect damages, incidental damages, consequential, damages incurred by the Authority and Indemnitees (except the Architect Indemnitees with respect to design errors or omissions or breaches of the Design Services Agreement), including rental expenses, for losses of use, income, profit, financing, business and reputation, and for loss of management or employee productivity or the services of such persons, and any other type or form of damages ("Reserved Damages") against the Construction Services Agreement.

## 6.3 **REVIEW OF CLAIMS AND DISPUTES**

**6.3.1** The SDC Group, on behalf of the Authority, will review all Claims once submitted by the Construction Manager and take one or more of the following preliminary actions within ten (10) days of receipt of a Claim: (1) request additional supporting data from the claimant, (2) submit a schedule to the parties indicating when the

SDC Group expects to take action, (3) reject the Claim in whole or in part, stating reasons for rejection, (4) recommend approval of the Claim, (5) suggest a compromise, or (6) not take any of the above actions, in which case the Claim shall be deemed denied. The Authority may also, but is not obligated to, notify the Construction Manager's surety, if any, of the nature and amount of the Claim.

- **6.3.2** If a Claim submitted to the SDC Group has been resolved, the Authority will prepare or obtain appropriate documentation to memorialize and process the resolution.
- **6.3.3** If a Claim submitted to the SDC Group has not been resolved after consideration of the foregoing and of further evidence presented by the parties or requested by the SDC Group, the Authority will, in a timely fashion, notify the parties in writing that the SDC Group's recommendation will be made within seven (7) days. Upon expiration of such time period, the SDC Group will render to the parties its final recommendation relative to the Claim including any change in the GMP or Contract Time or both. The parties may then (1) agree to resolve the Claim or (2) proceed pursuant to **Paragraph 6.4** hereof. If the SDC Group's recommendation is not submitted within 30 days of the SDC Group's receipt of the Claim, the Claim shall be denied.

#### 6.4 DISPUTE RESOLUTION

- 6.4.1 Mediation. Claims, disputes, or other matters in controversy arising out of or related to the Contract Documents shall be subject to mediation as a condition precedent to binding dispute resolution. The parties shall submit a written demand for mediation to the Authority, which shall describe in detail the facts and circumstances of the dispute. The Authority shall designate a neutral third party to serve as mediator. The mediator shall hear the matter in Minneapolis, Minnesota, and provide an informal opinion and advice, none of which shall be binding on the parties involved in the dispute. The opinion shall be submitted to the parties in dispute within twenty (20) days of the mediator hearing the dispute. The parties in dispute will then agree within ten (10) days to resolve the dispute pursuant to the mediator's advice or submit the matter to arbitration pursuant to these General Conditions. The mediator's costs shall be shared equally by the parties in the dispute. The request to mediate may be made concurrently with the filing of binding dispute resolution proceedings but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of sixty (60) days from the date of filing, unless stayed for a longer period by agreement of the parties or court Order. If an arbitration is stayed pursuant to this Subparagraph 6.4.1, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.
- 6.4.2 Claims Subject to Arbitration. Any Claim subject to, but not resolved by, mediation shall be subject to arbitration, which shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules ("AAA Rules") in effect as of the date of this Construction Services Agreement. The party filing a notice of demand for arbitration must assert in the demand all Claims then known to that party on which arbitration is permitted to be demanded. All proceedings set forth under this Paragraph 6.4 shall be conducted in Minneapolis, Minnesota.
- **6.4.3** Notices for Arbitration. Notice of Demand for Arbitration shall be filed, in writing, with the other party to the applicable agreement, with any party to be joined pursuant to **Subparagraph 6.4.6** hereof, and with the American Arbitration Association, the Authority, Construction Manager, and the Architect, to the extent they are not otherwise a party to the arbitration. At the Authority's election, and its sole discretion, it may invoke application of Fast Track Procedures under the AAA Rules for any such arbitration proceedings initiated under the Contract Documents. The Project Team acknowledges and agrees that Minn. Stat. § 544.42 does not apply to this Project.
- 6.4.4 Contract Performance During Arbitration. During arbitration proceedings, the Authority, Architect, Construction Manager and any Subcontractor or Subconsultant shall comply with Subparagraph 6.2.4 hereof.
- **6.4.5** When Arbitration May Be Demanded. A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the Claim would be barred by the applicable statute of limitations or repose. For statute of limitations purposes, receipt of a written demand for arbitration by the Person administering the arbitration shall constitute the institution of legal or equitable proceedings based on the Claim.
- **6.4.6 Consolidation or Joinder.** Any arbitration shall include, by consolidation or joinder, the Construction Manager, Architect, Subcontractors, Suppliers, Subconsultants, Trade Contractors, and other Persons bound to arbitrate according to this **Paragraph 6.4** if substantially involved in a common question of fact or law. The foregoing agreement to arbitrate and consolidation and joinder shall be specifically enforceable under Applicable Laws in any court having jurisdiction thereof. Any contract between the Construction Manager and any Subcontractor and Supplier and contracts between Subcontractors and Suppliers of Sub-Subcontractors and suppliers shall include provisions for arbitration, consolidation, and joinder consistent with this **Paragraph 6.4**.
- **6.4.7** Judgment on Final Award. The award rendered by the arbitrator or arbitrators shall be final, and judgment may be entered upon it in accordance with Applicable Laws in any court having jurisdiction thereof.

**6.4.8** Litigation of Third Party Claims. If a third party who is not a member of the Project Team sues a member of the Project Team, the Authority, or Team in a court of law ("Third Party Suit"), then the Authority, or the Team as applicable, shall have the absolute discretion to join in the Third Party Suit members of the Project Team who are involved in common questions of law or fact or whose participation is necessary to provide complete relief in the Third Party Suit, and the Authority, and the Team as applicable, have the absolute discretion to decide to waive and not require arbitration of Claims involving the Third Party Suit as otherwise provided in this Construction Services Agreement. The Project Team hereby consents to jurisdiction and venue in any court in which the suit is commenced or to which such Third Party Suit is removed.

### ARTICLE 7 SUBCONTRACTORS

### 7.1 DEFINITIONS

- 7.1.1 A Subcontractor is a person or entity who has a direct contract with the Construction Manager to perform a portion of the Work at the Project Site. The term "Subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Subcontractor or an authorized representative of the Subcontractor. The term "Subcontractor" does not include a Trade Contractor who is a party that has a direct contract with the Authority to perform a portion of the Work at the Project Site; provided, however, that if the Authority assigns a Trade Contract to the Construction Manager, the Trade Contractor shall then be a Subcontractor.
- 7.1.2 A Sub-Subcontractor is a person or entity of any tier who has a direct or indirect contract with a Subcontractor to perform a portion of the Work at the Project Site. The term **"Sub-Subcontractor"** is referred to throughout the Contract Documents as if singular in number and means a Sub-Subcontractor or an authorized representative of the Sub-Subcontractor.

## 7.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK

- **7.2.1** In accordance with the Construction Schedule, the Construction Manager shall furnish the Authority, in writing, with (1) the name, trade and subcontract amount of all persons and entities proposed as Subcontractors, and (2) the names of all persons or entities proposed as Suppliers of the material or equipment identified in the Specifications (including those who are to furnish materials or equipment fabricated to a special design) and, where applicable, the name of the installing Subcontractor. The Authority will promptly reply to the Construction Manager in writing stating whether or not the Authority, after due investigation, has reasonable objection to any such proposed person or entity.
- **7.2.2** The Construction Manager shall not contract with a proposed person or entity to whom the Authority has made reasonable and timely objection. The Construction Manager shall not be required to contract with anyone to whom the Construction Manager has made reasonable objection.
- **7.2.3** If the Authority has reasonable objection to a person or entity proposed by the Construction Manager, the Construction Manager shall propose another to whom the Authority has no reasonable objection.
- **7.2.4** The Construction Manager shall not change a Subcontractor or Supplier previously selected without the prior written approval of the Authority.

#### 7.3 SUBCONTRACTUAL RELATIONS

- 7.3.1 By written agreement, the Construction Manager shall require each Subcontractor or Supplier, to the extent of the Work to be performed by the Subcontractor, to be bound to the Construction Manager by all of the terms of Contract Documents, and to assume toward the Construction Manager all the obligations and responsibilities which the Construction Manager, by these Contract Documents, assumes toward the Authority and Architect. Each Subcontract shall preserve and protect the rights of the Authority and Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor or Supplier so that subcontracting thereof will not prejudice such rights, and shall allow the Subcontractor or Supplier, unless specifically provided otherwise in the Subcontract, the benefit of all rights, remedies, and redress against the Construction Manager that the Construction Manager has against the Authority under the Contract Documents. The Construction Manager shall require each Subcontractor and Supplier to enter into similar agreements with Sub-Subcontractors and Sub-Suppliers. The Construction Manager shall make available to each proposed Subcontractor and Supplier, prior to the execution of the Subcontract, copies of the Contract Documents to which the Subcontractor and Supplier will be bound, and, upon written request of the Subcontractor or Supplier, identify to the Subcontractor or Supplier terms and conditions of the proposed Subcontract which may be at variance with the Contract Documents. Subcontractors and Suppliers shall similarly make copies of applicable portions of such documents available to their respective proposed Sub-Subcontractors and Sub-Suppliers.
- **7.3.2** All Subcontracts shall be in writing and shall specifically provide that the Authority and the Indemnitees are intended third party beneficiaries of such subcontracts.

7.3.3 The Authority shall have the right to cause the Construction Manager to terminate for convenience any Subcontract with a Subcontractor, Supplier, Sub-Subcontractor, or any other Person with whom the Construction Manager has a contractual relationship for the Work. Termination by the Authority as described in this Subparagraph 7.3.3 shall be for the convenience of the Authority and shall be in the sole and absolute discretion of the Authority. Such termination shall occur with five (5) days written notice by the Authority to the Construction Manager, and the Authority shall compensate the Construction Manager only for actual expenditures of the Subcontractor, Supplier, Sub-Subcontractor, or other Person with whom the Construction Manager had a contractual relationship incurred in the interest of the Work to the date of termination and shall not include any costs for profit, overhead, damages, or demobilization beyond the date of written notice of termination. If such a termination for convenience has an impact on Construction Manager's GMP or Contract Time, the Construction Manager shall be entitled to an equitable adjustment to the GMP or the Contract Time.

## 7.4 CONTINGENT ASSIGNMENT OF SUBCONTRACTS

- **7.4.1** Each Subcontract for a portion of the Work is assigned by the Construction Manager to the Authority provided that:
  - .1 assignment is effective only after termination of the Construction Services Agreement by the Authority or stoppage of Work by the Authority as provided in the Contract Documents and only for those Subcontracts which the Authority accepts by notifying the Subcontractor in writing in the Authority's sole discretion; and
  - .2 assignment is subject to the prior rights of the Construction Manager's surety, obligated under bonds relating to the Construction Services Agreement.
- **7.4.2** If the Work in connection with a Subcontract has been suspended for more than sixty (60) days after termination of the Construction Services Agreement by the Authority pursuant to Contract Documents, and the Authority accepts assignment of such Subcontract, the Subcontractor's compensation shall be equitably adjusted for any increase in direct costs incurred by such Subcontractor as a result of the suspension.
- **7.4.3** Each Subcontract shall specifically provide that the Authority shall only be responsible to the Subcontractor for those obligations of the Construction Manager that accrue subsequent to the Authority's exercise of any rights under this conditional assignment.

# ARTICLE 8 CONSTRUCTION BY AUTHORITY

# 8.1 AUTHORITY'S RIGHT TO PERFORM CONSTRUCTION

### 8.1.1 [Reserved].

- **8.1.2** The Authority reserves the right to perform construction or operations related to the Project with the Authority's own forces and to award separate Trade Contracts or Vendor Contracts in connection with other portions of the Project or other construction or operations on the Project Site under conditions identical or substantially similar to these General Conditions, including those portions related to insurance and waiver of subrogation.
- **8.1.3** When Trade Contracts or Vendor Contracts are awarded for different portions of the Project or other construction or operations on the Project Site, the term "Construction Manager" in the Contract Documents in each case shall mean the applicable Trade Contractor or Vendor.
- 8.1.4 The Construction Manager shall be responsible for managing and administering the coordination and integration of the Trade Contractor Work or Vendor Work being performed at the Project Site by Trade Contractors or Vendors with the Work. The Construction Manager shall establish a complete Construction Schedule, which shall be coordinated with the schedules provided by the Trade Contractors and Vendors and include all logical relationships between the Work and the Trade Contractor Work or Vendor Work, if any. The Construction Manager shall review and periodically update the Construction Schedule to monitor, track and control the progress of the Work, and to facilitate the coordination of and assist timely performance of the Trade Contractor Work or Vendor Work where the Work is logically dependent upon such timely performance. The Construction Manager shall make any revisions to the Construction Schedule deemed necessary for the Work or the Trade Contractor Work or Vendor Work after a joint review and mutual agreement with the Authority and affected Trade Contractor or Vendor. The resulting Construction Schedule shall then constitute the Construction Schedule to be used by the Construction Manager, Trade Contractors, Vendors, and the Authority until subsequently revised. The Trade Contractors, Vendors and Construction Manager shall cooperate with each other so as to avoid delay, interference or disruption to either the Work or the Trade Contractor Work or Vendor Work.
- 8.1.5 Assignment of Vendor or Trade Contracts. In order to maintain the Master Project Schedule, the Authority may purchase materials or equipment directly from Vendors or Trade Contractors before the Construction Manager's GMP Proposal has been accepted by the Authority and the Construction Manager has been approved

to start the Construction Phase Services. In the Authority's sole discretion, the Authority may assign to the Construction Manager, and, subject to the terms of Paragraph 8.3 of the Construction Services Agreement, the Construction Manager shall accept such assignment, at which time the Trade Contractor or Vendor shall become a Subcontractor of the Construction Manager, and the Construction Manager shall assume, and shall solely fulfill and be responsible for, all to the liabilities and obligations of the Trade Contractor or Vendor that the Authority had prior to the assignment. Except as set forth in Paragraph 3.4 hereof, the Construction Manager shall be responsible and liable for such assigned Vendor or Trade Contracts, if any, as if the Construction Manager were the original purchaser of the Vendor Work or Trade Contractor Work. If the Authority accepts the Construction Manager's GMP Proposal, said acceptance will simultaneously accomplish and effect the assignment of all executed Vendor Contracts and Trade Contracts to the Construction Manager, except as otherwise stated in the Authority's acceptance of the GMP Proposal. Except as otherwise provided in the Contract Documents, the GMP includes, without limitation, all costs and expenses in connection with delivery, handling, storage, insurance, installation and testing of items covered in any assigned Vendor Contracts or Trade Contracts. All warranty and correction of the Work obligations under the Contract Documents shall also apply to any Vendor Contracts or Trade Contracts assigned to the Construction Manager, unless the Contract Documents specifically and expressly provide otherwise.

### 8.2 MUTUAL RESPONSIBILITY

- **8.2.1** The Construction Manager shall afford the Authority, Trade Contractors and Vendors reasonable opportunity for introduction and storage of their materials and equipment and performance of the Trade Contractor Work or Vendor Work at the Project Site. The Construction Manager shall properly and fully connect and coordinate the Work with the Trade Contractor Work or Vendor Work as required by the Contract Documents. The Construction Manager, Trade Contractors and Vendors are required to schedule and coordinate, each with respect to the other, their respective activities at the Project Site so as not to interfere with or to delay the Work or the work of such Trade Contractors or Vendors. The Construction Manager agrees to provide such coordination and scheduling without increase in the GMP.
- **8.2.2** If part of the Construction Manager's Work depends for proper execution or results upon a Trade Contractor's Work or Vendor's Work, the Construction Manager shall, prior to proceeding with that portion of the Construction Manager's Work, inspect and promptly report in writing to the Authority and any such Trade Contractor or Vendor apparent discrepancies or defects in the Trade Construction Manager's Work. Failure of the Construction Manager so to report in writing shall constitute an acknowledgment that the Trade Contractor's Work or Vendor's Work is fit and proper to receive the Work of the Construction Manager, except as to latent defects not then reasonably discoverable.
- **8.2.3** The Construction Manager shall reimburse the Authority for costs incurred by the Authority which are payable to a Trade Contractor or Vendor because of delays, improperly timed activities, defective Work or other errors or omissions of the Construction Manager.
- **8.2.4** The Construction Manager shall promptly remedy damage caused by the Construction Manager to completed or partially completed construction or to property of the Authority or Work of a Trade Contractor or Vendor as provided in **Subparagraph 12.2.5**.
- **8.2.5** Claims and other disputes and matters in question between the Construction Manager and a Trade Contractor or Vendor shall be subject to the provisions of **Paragraphs 6.2 6.4** provided the Trade Contractor or Vendor has reciprocal obligations.

### ARTICLE 9 CHANGES IN THE CONSTRUCTION MANAGER'S WORK

### 9.1 CHANGES

- **9.1.1** Changes in the Work shall only be accomplished after execution of the Construction Services Agreement, and without invalidating the Contract Documents, by Contract Revision, Construction Change Directive, or order for a minor change in the Work, subject to the limitations stated in this **Article 9** and elsewhere in the Contract Documents.
- **9.1.2** A Contract Revision shall be based upon agreement between the Authority and Construction Manager. A Construction Change Directive must be signed and issued by the Authority and Architect. An order for a minor change in the Work may be issued by the Architect subject to approval of the Authority.
- **9.1.3** Changes in the Work shall be performed under applicable provisions of the Contract Documents, and the Construction Manager shall proceed promptly, unless otherwise provided in the Contract Revision, Construction Change Directive, or order for a minor change in the Work.
- **9.1.4** If unit prices are stated in the Contract Documents or subsequently agreed upon in writing, and if quantities originally contemplated are so changed in a proposed Contract Revision or Construction Change Directive that

application of such unit prices to quantities of Work proposed will cause substantial inequity to the Authority or Construction Manager, the applicable unit prices shall be equitably adjusted.

# 9.2 CONTRACT REVISIONS

- **9.2.1** A Contract Revision is a written instrument prepared by the Authority and signed by the Authority, Architect, and Construction Manager, stating their agreement upon all of the following:
  - .1 a change in the Construction Manager's Work;
  - .2 the amount of the adjustment in the GMP, if any;
  - .3 the extent of the adjustment in the Contract Time, if any; and
  - .4 the updated Schedule of Values.
- 9.2.2 Methods used in determining adjustments to the GMP may include those listed in Subparagraph 9.3.2 hereof.
- **9.2.3** Except as otherwise set forth therein, any executed Contract Revision shall constitute a final settlement of all matters relating to or arising out of the change in the Work which is the subject of the Contract Revision, including all direct and indirect costs associated with such change and any and all adjustments to the GMP and the Construction Schedule.
- **9.2.4** Except as permitted in **Article 6, Paragraph 9.3,** and **Subparagraph 11.7.2** hereof, a change in the GMP or the Contract Time shall be accomplished only by Contract Revision. Accordingly, unless a written notice of a Claim has been properly made pursuant to **Article 6** before the allegedly extra Work is started, no course of conduct or dealings between the parties, nor express or implied acceptance of alterations or additions to the Work, and no claim that the Authority has been unjustly enriched by an alteration of or addition to the Work, shall be the basis of any Claim to an increase in any amounts due under the Contract Documents or a change in the Contract Time.

## 9.3 CONSTRUCTION CHANGE DIRECTIVES

- **9.3.1** A Construction Change Directive is a written order prepared by the Architect, after full consultation with and signed by the Authority, directing a change in the Work. The Authority may by Construction Change Directive, without invalidating the Contract Documents, order changes in the Work within the general scope of the Contract Documents consisting of additions, deletions, or other revisions, the GMP and Contract Time being adjusted if appropriate. If any Construction Change Directive proposes an adjustment, if any, in the GMP or Contract Time, or both, then Construction Manager shall advise the Authority and the Architect if Construction Manager plans to use the Construction Manager's Contingency to fund said adjustment. A Construction Change Directive shall be used in the absence of total agreement on the terms of a Contract Revision. No adjustment to the Contract Time shall be made or allowed if the Construction Change Directive does not cause a delay to the critical path of the Construction Schedule.
- **9.3.2** If the Construction Change Directive provides for an adjustment to the GMP, the adjustments shall be based on one of the following methods:
  - .1 mutual acceptance of a lump sum properly itemized and supported by efficient substantiating data to permit evaluation;
  - .2 unit prices stated in the Contract Documents or subsequently agreed upon;
  - .3 Costs of the Work as provided in the Construction Services Agreement; or
  - .4 as provided in Subparagraph 9.3.5 hereof.
- **9.3.3** Upon receipt of a properly authorized Construction Change Directive, the Construction Manager shall promptly proceed with the change in the Work involved and advise the Authority of the Construction Manager's agreement or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the GMP or Contract Time.
- **9.3.4** A Construction Change Directive signed by the Construction Manager indicates the agreement of the Construction Manager therewith, including adjustment in GMP and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Contract Revision.
- **9.3.5** If the Construction Manager does not respond promptly or disagrees with the method for adjustment in the GMP, the method and the adjustment shall be determined by the Authority on the basis of reasonable Costs of the Work and savings attributable to the change, including, in case of an increase in the GMP, overhead and profit as provided in **Subparagraph 9.5.1** hereof. Deductive changes shall be determined by subtracting from the Guaranteed Maximum Price, the greater of the Construction Manager's bid associated with the deducted

Work or the Cost of the Work associated with the deducted Work plus, in either case, overhead and profit as provided in **Subparagraph 9.5.1** hereof. In such case, and also under **Subparagraph 9.3.2.3** hereof, the Construction Manager shall keep and present, in such form as the Authority may prescribe, an itemized accounting together with appropriate supporting date of any requested change. Costs for the purposes of this **Subparagraph 9.3.5** shall be determined in accordance with the provisions of the Construction Services Agreement for determining the Cost of the Work.

- **9.3.6** Pending final determination of cost, amounts not in dispute may be included in Applications for Payment. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.
- **9.3.7** If the Construction Manager does not agree with the adjustment in Contract Time or the method for determining it, the adjustment or the method shall be subject to the procedure for Claims under **Paragraph 6.3 and 6.4** hereof.
- **9.3.8** When the Construction Manager agrees with the determination made by the Authority concerning the adjustments in the GMP and Contract Time, or otherwise reaches agreement upon the adjustments, such agreement shall be effective immediately and shall be recorded by preparation and execution of an appropriate Contract Revision.
- **9.3.9** Overtime, when specifically requested authorized by the Authority and not as an Extraordinary Measure, shall be paid for by the Authority on the basis of premium payment only, plus the cost of insurance and taxes based on the premium payment period. Overhead and profit will not be paid by the Authority for overtime.

#### 9.4 MINOR CHANGES IN THE WORK

**9.4.1** The Architect may, with the prior approval of the Authority, order minor changes in the Work not involving adjustment in the GMP, the quality of the Work, or extension of the Contract Time and not inconsistent with the intent of the Contract Documents. Such changes shall be effected by written order and shall be binding on the Authority, Architect and Construction Manager. The Construction Manager shall carry out such written orders promptly.

### 9.5 AGREED OVERHEAD AND PROFIT RATES

**9.5.1** For any adjustments to the GMP which are based on other than the unit prices method, the Construction Manager agrees to charge, and accept, as payment for overhead and profit, the Construction Manager Fee as described in **Exhibit 3** of the Construction Services Agreement. For any adjustments to the GMP which are based on the unit prices method, no overhead or profit shall be charged.

#### ARTICLE 10 TIME

#### 10.1 DEFINITIONS

- **10.1.1** Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work.
- **10.1.2** The date of commencement of the Work is the date established in the Construction Services Agreement. The date shall not be postponed by the failure to act of the Construction Manager or Persons for whom the Construction Manager is responsible.
- **10.1.3** The date of Substantial Completion is the date certified by the Architect in accordance with **Paragraph 11.8** hereof.

### 10.2 PROGRESS AND COMPLETION

- **10.2.1** Time limits stated in the Contract Documents are of the essence. By executing the Construction Services Agreement the Construction Manager confirms that the Contract Time is a reasonable period for performing the Work.
- **10.2.2** The Construction Manager shall not knowingly, except by agreement or instruction of the Authority in writing, prematurely commence the Work prior to the effective date of insurance required by **Article 13** hereof to be furnished by the Construction Manager. The date of commencement of the Work shall not be changed by the effective date of such insurance.
- **10.2.3** The Construction Manager shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time.

# 10.3 DELAYS AND EXTENSIONS OF TIME

- **10.3.1** Except as otherwise provided in **Paragraph 4.3** of the Construction Services Agreement, if the Construction Manager is delayed at any time in progress of the Work by changes properly authorized and ordered by the Authority in the Work, or by industry wide labor disputes, fire, unusual delay in deliveries of materials outside the control of a Supplier or Vendor, unavoidable casualties or other causes beyond the Construction Manager's control, or by other causes which the Authority determines may justify delay, then the Contract Time shall be equitably adjusted by Contract Revision to the extent such delay extends the critical path of the Construction Schedule and the performance of the Work is not, was not, or would not have been delayed by any other cause for which the Construction Manager further acknowledges and agrees that adjustments in the Contract Time will be permitted for a delay only to the extent such delay (i) is not caused, or could not have been reasonably avoided, by the Construction Manager, (ii) could not be reasonably limited or avoided by the Construction Manager's timely notice to the Authority of the delay or reasonable likelihood that a delay will occur, or (iii) is of a duration not less than one (1) day.
- 10.3.2 Claims relating to time shall be made in accordance with applicable provisions of Paragraph 6.3 hereof.
- **10.3.3** This **Paragraph 10.3** does not preclude recovery of damages for delay by any party except as may otherwise be provided under other provisions of the Contract Documents.
- **10.3.4** If the Construction Manager submits a progress report indicating, or otherwise expresses an intention to achieve, completion of the Work prior to any completion date required by the Contract Documents or expiration of the Contract Time, no liability of the Authority to the Construction Manager for any failure or inability of the Construction Manager to so complete the Work shall be created or implied.
- **10.3.5** The Authority shall have the right to occupy, without prejudice to rights of either party, any completed or Substantially Completed portions of the Work, notwithstanding the fact that time for completion of entire Work, or portions thereof, may not be expired. Occupancy and use by Authority shall not constitute, in itself, acceptance of the Work.

#### ARTICLE 11 PAYMENTS AND COMPLETION

#### 11.1 PAYMENTS

**11.1.1** The GMP is the amount stated in the Construction Management Plan for the Work and is the maximum total amount payable to the Construction Manager for performance of the Work under the Construction Services Agreement.

### 11.2 SCHEDULE OF VALUES

- **11.2.1** Upon execution of the Construction Management Plan, the Construction Manager shall submit to the Authority and Architect a Schedule of Values allocated to various portions of the Work, prepared in such form and supported by such data to substantiate its accuracy as the Authority deems necessary. The Authority shall have the right to approve the Construction Manager's format of the Schedule of Values and all subsequent forms of the Schedule of Values. The Schedule of Values shall be used by the Authority and Architect as a basis for reviewing the Construction Manager's Applications for Payment and approving award of subcontracts.
- **11.2.2** Any Schedule of Values that fails to include sufficient detail, is unbalanced, or exhibits "front-end loading" of the value of the Work shall be rejected. If a Schedule of Values had been initially approved and subsequently used, but later found improper for any reason, sufficient funds shall be withheld from future Applications for Payment to ensure an adequate reserve, exclusive of normal retainage, to complete the Work.

### 11.3 APPLICATIONS FOR PAYMENT

11.3.1 The Construction Manager shall submit to the Authority and Architect an itemized Application for Payment for completed portions of the Work in accordance with the Schedule of Values. Such Application for Payment shall be notarized, if required, and supported by such data substantiating the Construction Manager's right to payment as the Authority may require, such as copies of requisitions from Subcontractors and Suppliers of any tier, and reflecting retainage if provided for elsewhere in the Contract Documents. Such Applications for Payment may include requests for payment on account of changes in the Work that have been properly authorized by Construction Change Directives but not yet included in a Contract Revision. Such Applications for Payment shall not intentionally include requests of a dispute or other reason unless the Construction Manager does not intend to pay to a Subcontractor or Supplier because of a dispute or other reason unless the Construction or Supplier. By signing and submitting an Application for Payment, Construction Manager shall represent and warrant that Construction Manager has no knowledge of and waives any Claims, specifically including Claims for additional Time or adjustments to the GMP, based on events or circumstances existing or occurring before

the date of the subject Application for Payment, except as specifically stated in said Application or as may have been properly preserved by timely written notice of a Claim as required by the Contract Documents. Each Application for Payment shall be accompanied by the following, all in form and substance satisfactory to the Authority:

- .1 A current Construction Manager's waiver of lien or claim for payment and duly executed and acknowledged sworn statement showing all Subcontractors and Suppliers with whom the Construction Manager has entered into Subcontracts, the amount of each such Subcontract, the amount requested for any Subcontractor and Supplier in the requested progress payment, and the amount to be paid to the Construction Manager from such progress payment, together with similar sworn statements from all such Subcontractors and Suppliers;
- .2 Duly executed waivers of liens and claims for payment from all Subcontractors, Suppliers, and lower tier Sub-Subcontractors establishing payment or satisfaction of all amounts requested by the Construction Manager on behalf of such entities or persons in the previous Application for Payment;
- .3 Invoices and back-up documentation from any Subcontractor, Supplier, Sub-Subcontractor, or other party to whom the Construction Manager proposes to make payment under that Application for Payment;
- .4 Certified payroll receipts for the Construction Manager's personnel, equipment charge or rental logs, invoices for material purchases, and other documentation requested by the Authority for self-performed Work of the Construction Manager; and
- .5 All information and materials required to comply with the requirements of the Contract Documents, requested by the Authority or reasonably requested by any Lender, Lender's representative and any escrow or disbursement agent or title company, if any, retained by the Authority or any Lender to disburse payments.
- **11.3.2** Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the Project Site for subsequent incorporation in the Work. If approved in advance by the Authority, payment may be made for materials and equipment suitably stored off the Project Site at a location agreed upon in writing. Payment for materials and equipment stored on or off the Project Site shall be conditioned upon compliance by the Construction Manager with procedures satisfactory to the Authority to establish the Authority's title to such materials and equipment or otherwise protect the Authority's interests, and shall include applicable insurance, storage, and transportation to the Project Site for such materials and equipment stored off the Project Site. Construction Manager shall also comply with the following specific requirements:
  - .1 The aggregate cost of materials stored off site shall not exceed Five Million and 00/100 Dollars (\$5,000,000.00) at any time without written approval of the Authority;
  - .2 Title to such materials shall be vested in the Authority, as evidenced by documentation satisfactory in form and substance to the Authority, including recorded financing statements, UCC filings, and UCC searches;
  - .3 With each Application for Payment, the Construction Manager shall submit to the Authority a written list identifying each location where materials are stored off the Project Site and the value of materials at each location. The Authority (or the Construction Manager, if the Construction Manager is procuring the builder's risk insurance) shall procure insurance for materials stored off the Project Site in an amount not less than the total value thereof and shall provide the Construction Manager (or Authority if procured by Construction Manager) with satisfactory evidence of such insurance coverage along with the Application for Payment for such materials;
  - .4 The consent of any surety shall be obtained to the extent required prior to payment for any materials stored off the Project Site;
  - .5 Representatives of the Authority and the Architect shall have the right to make inspections of the storage areas at any time; and
  - .6 Such materials shall be (1) protected from diversion, destruction, theft, and damage to the satisfaction of the Authority and the Architect, (2) specifically marked for use in the Work, and (3) segregated from other materials at the storage facility.
- **11.3.3** The Construction Manager warrants that title to all Work covered by an Application for Payment will pass to the Authority no later than the time of payment. The Construction Manager further warrants that upon submittal of an Application for Payment, all Work for which Certificates for Payment have been previously issued and payments received from the Authority shall be free and clear of liens, Claims, security interests, or encumbrances in favor of the Construction Manager, Subcontractors, Suppliers, Sub-Subcontractors, or other Persons able to make a Claim by reason of having provided labor, materials, and equipment relating to the

Work, except as specifically stated in said Application or as may have been properly preserved by timely written notice of a Claim as required by the Contract Documents.

- .1 To the extent of payments received from the Authority, the Construction Manager further expressly undertakes to defend and indemnify the Authority and the Indemnitees, at the Construction Manager's sole expense with independent counsel of the Authority's choice, against any actions, lawsuits, or proceedings brought against the Authority or any Indemnitee as a result of Claims or liens filed against the Construction Manager, the Work, the Project Site, any improvements thereon, or any portion of the property of the Authority or any Indemnitee, referred to collectively as Liens in Subparagraphs 11.3.3 and 11.6.6 hereof. To the extent of payments received from the Authority, the Construction Manager hereby agrees to indemnify and hold the Authority and all Indemnitees harmless against any such Liens or claims of Lien and agrees to pay any judgment or Lien resulting from any such actions, lawsuits or proceedings.
- .2 The Authority shall release any payments withheld due to a Lien if the Construction Manager obtains security acceptable to the Authority or a bond which is: (1) issued by a surety acceptable to the Authority, (2) in form and substance satisfactory to the Authority, and (3) in an amount not less than One Hundred Fifty percent (150%) of such Lien claim. By posting a bond or other acceptable security, however, the Construction Manager shall not be relieved of any responsibilities or obligations under this **Paragraph 11.3**, including, without limitation, the duty to defend and indemnify the Authority and all Indemnitees. The cost of any premiums incurred in connection with such bonds or security shall be the responsibility of the Construction Manager and shall not be reimbursable as a Cost of the Work.

## 11.4 APPROVAL OF APPLICATIONS FOR PAYMENT

- 11.4.1 The Construction Manager shall submit a complete and properly prepared Application for Payment to the Authority and Architect no later than the twenty-fifth (25th) day of the month (or as agreed to between the Authority and Construction Manager in writing). If an Application for Payment is received at any time later than the twenty-fifth (25th) day of the month it will be processed the following month. The Authority, Architect, and Construction Manager will meet during the last five (5) days of the month to review the Construction Manager's Application for Payment. The Construction Manager will make corrections to the Construction Manager's Application for Payment as agreed to by the Authority and Construction Manager and re-submit the Application for Payment by the last business day of the month when required by the Authority. By the fifteenth (15th) day of the following month, the Architect shall issue a Certificate for Payment to the Authority in such amounts as deemed appropriate by the Architect and Authority or notify the Construction Manager in writing of the reasons for withholding approval in whole or in part as provided in **Paragraph 11.5** hereof.
- **11.4.2** The issuance of approval of the Application for Payment by the Authority is based on the Architect's observations at the Project Site and the data comprising the Application for Payment that the Work has progressed to the point indicated and that the quality of the Work is in accordance with the Contract Documents. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents and to specific qualifications expressed by the Architect. The issuance of approval of the Application for Payment will further constitute a representation and certification by the Architect to the Authority that the Construction Manager is entitled to payment in the amount approved. However, the issuance of a Certificate for Payment will not be a representation that the Authority has (1) made exhaustive or continuous on-site observations to check the quality or quantity of the Work or (2) reviewed construction means, methods, techniques, sequences, or procedures.

### 11.5 DECISIONS TO WITHHOLD CERTIFICATION

- 11.5.1 The Architect may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Authority, if in the Architect's opinion the representations to the Authority required by Subparagraph 11.4.2 hereof cannot be made. If the Architect is unable to approve payment in the amount of the Application for Payment, the Architect will notify the Construction Manager and Authority as provided in Subparagraph 11.4.1 hereof. If the Construction Manager and Authority cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount of which the Authority is able to approve. The Architect may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Architect's opinion to protect the Authority from loss because of:
  - .1 Defective Work not remedied;
  - .2 Third party claims filed or reasonable evidence indicating probable filing of such claims;
  - .3 Failure of the Construction Manager to properly pay Subcontractors or Suppliers;
  - .4 Reasonable evidence that the Work cannot be completed for the unpaid balance of the GMP;

- .5 Loss or damage incurred by the Authority or another party;
- .6 Reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay; or
- .7 Persistent failures to carry out the Work in accordance with the Contract Documents.
- **11.5.2** When the above reasons for withholding payment are removed as approved in writing by the Authority, a Certificate for Payment will be issued by the Architect for amounts previously withheld.
- **11.5.3** If the Construction Manager disputes any determination by the Authority or Architect with regard to any Application for Payment, the Construction Manager nevertheless expeditiously shall continue to prosecute the Work.
- **11.5.4** The Authority shall not be deemed to be in breach of the Contract Documents by reason of the withholding of any payment pursuant to any provision of the Contract Documents provided that the Work for which payment is being withheld shall have been rejected by the Authority and Architect due to non-compliance with the Contract Documents.

# 11.6 PROGRESS PAYMENTS

- 11.6.1 The Authority shall make payment to the Construction Manager for an approved portion of the Construction Manager's Fee and other approved Costs of the Work to be reimbursed, less all applicable retentions and withholdings, not later than thirty (30) days after the Authority has received the Architect's Certificate for Payment. To the extent necessary to protect the Authority from Claims, the Authority may make payment by joint payment to the Construction Manager and the Subcontractors and Suppliers for Work performed by such Subcontractors and Suppliers. Payments by the Authority shall not constitute approval or acceptance of any item of cost in the Application for Payment. The Authority shall pay interest of one and one-half percent (1.5%) per month (or any part of a month) to the Construction Manager on an undisputed amount not paid to the Construction Manager within thirty (30) days after the Authority has received the Architect's Certificate for Payment. The minimum monthly interest penalty payment for an unpaid balance of \$100 or more is \$10. For an unpaid balance of less than \$100, the Authority shall pay the actual interest penalty due to the Construction Manager in accordance with Minnesota Statutes.
- 11.6.2 If the Authority makes payment directly to the Construction Manager for Work performed by Subcontractors, then the Construction Manager shall promptly pay each Subcontractor, in accordance with the approved Application for Payment submitted by the Construction Manager, on account of such Subcontractor's portion of the Work, the amount to which said Subcontractor is entitled, reflecting percentages actually retained from payments to the Subcontractor on account of such Subcontractor's portion of the Work. The Construction Manager must pay any Subcontractor within ten (10) days of the Construction Manager's receipt of payment from the Authority for undisputed services provided by the Subcontractor. The Construction Manager must pay interest of one and one-half percent (1.5%) per month (or any part of a month) to the Subcontractor on any undisputed amount not paid on time to the Subcontractor. The minimum monthly interest penalty payment for an unpaid balance of \$100 or more is \$10. For an unpaid balance of less than \$100, the Construction Manager shall pay the actual penalty due to the Subcontractor in accordance with Minnesota Statutes. If the Authority makes joint payment to Subcontractors as provided in Subparagraph 11.6.1 hereof, then the Authority, through the Construction Manager, shall promptly pay each Subcontractor through a joint payment to Construction Manager and each Subcontractor, in accordance with the approved Application for Payment submitted by the Construction Manager, on account of such Subcontractor's portion of the Work, the amount to which said Subcontractor is entitled, reflecting percentages actually retained from payments to the Subcontractor on account of such Subcontractor's portion of the Work. The Construction Manager shall, by appropriate agreement with the Construction Manager and each Subcontractor, require each Subcontractor to make payments to Sub-Subcontractors in similar manner upon receipt of payment from the Construction Manager.

#### 11.6.3 [Reserved]

- **11.6.4** As the Project is a public project, the Authority shall have no obligation to pay or to see to the payment of money to a Subcontractor or Supplier.
- 11.6.5 Payment to Suppliers shall be treated in a manner similar to that provided in Subparagraphs 9.6.2 and 9.6.4.
- **11.6.6** An approved Application for Payment, Certificate for Payment, a progress payment or partial payment of any kind, or partial or entire use or occupancy of the Project by the Authority shall not constitute, nor shall it be construed as, acceptance or approval of any portion of the Work not in accordance with the Contract Documents or to otherwise relieve the Construction Manager of any of its obligations hereunder with respect thereto.
- 11.6.7 If required by the Authority or any Lender, all payments to the Architect, Construction Manager, Subcontractors, or Suppliers shall be made through a construction escrow established with an Escrow Agent

mutually acceptable to the Authority and such Lender. The Construction Manager hereby agrees to execute an Escrow Agreement which shall be (1) consistent with the requirements of the Contract Documents, except as the commercially reasonable standard procedures of the Escrow Agent may otherwise require, (2) structured to provide that the Escrow Agent may disburse funds directly to Subcontractors (with the Construction Manager's approval) or to the Construction Manager and Subcontractors payable jointly, if so directed by the Authority, and (3) otherwise in a form and substance reasonably satisfactory to the Authority.

**11.6.8** If the Authority is entitled to reimbursement or payment from the Construction Manager under or pursuant to the Contract Documents such payment shall be made promptly by the Construction Manager upon demand from the Authority. Notwithstanding anything contained in the Contract Documents to the contrary, if the Construction Manager fails to promptly make any payment to the Authority which is due to the Authority, or the Authority incurs any costs and expenses to cure any default of the Construction Manager or to correct defective Work, the Authority shall have an absolute right to offset such amount against the GMP or any progress payment otherwise due and may, in the Authority's sole discretion, elect either to: (1) deduct an amount equal to that which the Authority is entitled from any payments then or thereafter due the Construction Manager, or (2) issue a written notice to the Construction Manager reducing the GMP by an amount equal to that to which the Authority is entitled.

### 11.7 FAILURE OF PAYMENT

11.7.1 If the Architect does not issue a Certificate for Payment, through no fault of the Construction Manager, by the fifteenth (15) day of the month following Construction Manager's timely submission of a complete and properly completed Application for Payment, or if the Authority does not make payment of the amount approved by the Architect within thirty (30) days after issuance of the Certificate for Payment, or if the Authority does not make payment of an arbitration award within thirty (30) days of its confirmation after conclusion of any appeal, then the Construction Manager may, within ten (10) additional days' written notice to the Authority, stop the Work until payment of the amount owing has been received. The Construction Manager's reasonable costs of shut-down, delay, and start-up which shall be accomplished as provided in Article 9 hereof.

### 11.8 SUBSTANTIAL COMPLETION

- **11.8.1 "Substantial Completion"** or **"Substantially Complete"** shall mean the Work (or separable units or phases as provided in the Contract Documents) is essentially and satisfactorily complete in accordance with the Contract Documents, such that the Project is ready for opening to the general public and full occupancy or use by Authority (it being understood that, without limitation of the foregoing, all suites, concessions and other income-generating areas and all areas serving the general public shall be ready for full operation without material inconvenience or discomfort). A minor amount of Work, as determined by and at the discretion of Authority, such as installation of minor accessories or items, a minor amount of painting, minor replacement of defective work, minor adjustment of controls or sound systems, or completion or correction of minor exterior work that cannot be completed due to weather conditions, will not delay determination of Substantial Completion. For purposes of Substantial Complete. In no event shall Substantial Completion be deemed to have occurred unless such certificates or licenses as required for opening of the Project to the general public have been issued to Authority.
- 11.8.2 When the Construction Manager considers that the Work, or a portion thereof which the Authority and Architect agree to accept separately, is Substantially Complete, the Construction Manager shall prepare and submit to the Architect and the Authority a comprehensive list (the "punch list") of items to be completed or corrected. The Construction Manager shall proceed promptly to complete and correct items on the punch list. Failure to include an item on the punch list does not alter the responsibility of the Construction Manager to complete all Work in accordance with the Contract Documents. The Authority and the Architect reserve the right to add to the punch list within thirty (30) days after receipt of an acceptable punch list from the Construction Manager. Upon receipt of the Construction Manager's punch list, the Architect and Authority will make an inspection to determine whether the Work or designated portion thereof is Substantially Complete. If the inspection discloses any item, whether or not included on the Construction Manager's punch list, which is not in accordance with the requirements of the Contract Documents, the Construction Manager shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Architect or Authority. The Construction Manager shall then submit a request for another inspection by the Architect and Authority to determine Substantial Completion. When the Work or designated portion thereof is Substantially Complete, the Architect, with assistance from the Authority, will prepare a Certificate of Substantial Completion. The date of Substantial Completion shall establish responsibilities of the Authority and the Construction Manager for security, maintenance, heat, utilities, damage to the Work, and insurance, and shall fix the time within which the Construction Manager shall finish all items on the punch list accompanying the Certificate of Substantial Completion which shall identify all non-conforming, defective, and incomplete Work. Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion. The Certificate of Substantial Completion shall be submitted to the Authority and the Construction Manager for their written acceptance of responsibilities assigned to them in such Certificate.

11.8.3 Upon Substantial Completion of the Work or designated portion thereof and upon submission of an Application of Payment by the Construction Manager and its certification by the Architect, as provided in Paragraphs 11.3, 11.4 and 11.5 hereof, the Authority shall in turn make payment to the Construction Manager for all sums due upon Substantial Completion. Construction Manager agrees that the Authority shall hold back one-hundred fifty percent (150%) of the cost of the punch list items as determined by the Architect and Authority as retainage for performance of such punch list work.

### 11.9 PARTIAL OCCUPANCY OR USE

- 11.9.1 The Authority may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated to be Substantially Complete by the Construction Schedule, provided such occupancy or use is consented to by the insurer as required under **Subparagraph 13.3.11** and authorized by Governmental Authorities. In addition, such partial occupancy or use may commence whether or not the portion is Substantially Complete, provided the Authority and Construction Manager have accepted in writing the responsibilities assigned to each of them for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work, and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Construction Manager shall prepare and submit a punch list to the Architect and the Authority as provided under **Subparagraph 11.8.2** hereof. Consent of the Construction Manager to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by decision of the Authority.
- **11.9.2** Immediately prior to such partial occupancy or use, the Authority, Construction Manager, and Architect shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.
- **11.9.3** Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents.

## 11.10 FINAL COMPLETION AND FINAL PAYMENT

- 11.10.1 Upon receipt of written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such inspection and, when the Architect and Authority find the Work acceptable under the Contract Documents and the Work fully performed, the Architect will promptly issue a final Certificate for Payment stating that to the best of the Architect's knowledge, information and belief, and on the basis of the Architect's observations and inspections, the Work has been completed in accordance with terms and conditions of the Contract Documents and that the entire balance found to be due the Construction Manager and noted in said final Application for Payment will constitute a further representation that conditions listed in **Subparagraph 11.10.2** hereof as conditions precedent to the Construction Manager's being entitled to final payment shall be provided as part of the final Application for Payment. The final Certificate for Payment will not be issued by the Architect until all warranties and guarantees have been received and accepted by the Authority.
- 11.10.2 Neither final payment nor any remaining retained percentage shall become due until the Construction Manager submits to the Authority (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Authority or the Authority's property might be responsible or encumbered, less amounts withheld by Authority, have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect and that Construction Manager will give the Authority and Team not be canceled or allowed to expire until at least thirty (30) days' prior written notice before they are allowed to expire or be cancelled (except in the event of non-payment of premium, in which case seven (7) days notice will be provided, (3) a written statement that the Construction Manager knows of no substantial reason that the insurance will not be renewable to cover the period required by the Contract Documents, (4) consent of surety, if any, to final payment, and (5) if required by the Authority, other data establishing payment or satisfaction of obligations, such as receipts, releases and waivers of liens, Claims, security interests or encumbrances arising out of the Contract Documents, to the extent and in such forms as may be designated by the Authority. If the Authority has paid the Construction Manager what is due a Subcontractor, but a Subcontractor refuses to furnish a release or waiver required by the Authority, the Authority may require the Construction Manager to furnish a bond satisfactory to the Authority and to indemnify the Authority against such Lien or Claim for payment. If such Lien or Claim for payment remains unsatisfied after payments are made, the Construction Manager shall pay to the Authority all money that the Authority may be compelled to pay in discharging such Lien or Claim, including all costs and reasonable attorneys' fees. The Authority shall have the right to hold back one hundred fifty percent (150%) of the cost of completing any punch list items outstanding as of the final Certificate for Payment.
- **11.10.3** If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Construction Manager or by issuance of Contract Revisions affecting final completion, and the Authority so confirms, the Authority shall, upon application by the Construction Manager and certification by the Architect,

and without terminating the Construction Services Agreement, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance of the Work not fully completed or corrected is less than retainage stipulated in the Contract Documents, and if bonds have been furnished, the written consent of surety to payment of the balance due for that portion of Work fully completed and accepted shall be submitted by the Construction Manager to the Authority prior to certification of such payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of Claims by the Authority. The making of final payment shall constitute a waiver of claims against the Authority and the Indemnitees to the extent provided in **Subparagraph 6.2.5** hereof.

#### ARTICLE 12 PROTECTION OF PERSONS AND PROPERTY

### 12.1 SAFETY PRECAUTIONS AND PROGRAMS

- **12.1.1** The Construction Manager shall be solely responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the performance of the Construction Services Agreement. The Construction Manager is responsible to prepare and administer the Project Safety Program. The Architect shall report any violations of the Project Safety Program that it observes to the Construction Manager and the Authority. If the Authority adopts an Project specific insurance program, the Construction Manager shall coordinate and conform its Project Safety Program with the safety requirements of the Project specific insurance program; **provided, however,** that compliance with the safety requirements in any Project specific insurance program will not relieve the Construction Manager from its obligation to be solely responsible for safety precautions and preventing Project Site accidents.
- 12.1.2 In addition to its defined meaning, for purposes of this **Paragraph 12.1**, "**Hazardous Materials**" also means any element, compound, mixture, solution, particle or substance which is or may become dangerous, or harmful to the health and the welfare of life or the physical environment if not used, stored or disposed of in accordance with Applicable Laws, including explosives, petroleum products, radioactive materials, hazardous wastes, toxic substances, any "**PCBs**" or "**PCB items**," as defined in 40 CFR 761.3, and any "**asbestos**," as defined in 40 CFR 763.63.
- 12.1.3 In the event the Construction Manager encounters on the Project Site existing material reasonably believed to be Hazardous Materials which has not been rendered harmless and that presents an unknown and differing site condition pursuant to **Subparagraph 6.2.6** hereof, the Construction Manager shall immediately stop Work in the area affected and report the condition to the Authority in writing. The Work in the affected area shall not thereafter be resumed except by written agreement of the Authority, and Construction Manager, if in fact the material is a Hazardous Material and has not been rendered harmless. The Work in the affected area shall be resumed in the absence of Hazardous Materials, or when it has been rendered harmless. If the existing Hazardous Material is not an unknown, differing site condition pursuant to **Subparagraph 6.2.6** hereof, the Construction Manager shall be responsible for rendering the Hazardous Material harmless.

#### 12.1.4 [Reserved]

### 12.2 SAFETY OF PERSONS AND PROPERTY

- **12.2.1** The Construction Manager shall develop and administer a Project Safety Program that will include all reasonable precautions for safety of laborers and employees performing the Work and other persons who may be affected thereby. The Construction Manager will also take, and require its Subcontractors to take, all reasonable precautions for the protection of;
  - .1 the Work and materials and equipment to be incorporated therein, whether in storage on or off the Project Site, under care, custody, or control of the Construction Manager or the Construction Manager's Subcontractors or Sub-Subcontractors s; and
  - .2 other property at the Project Site or adjacent thereto including Trade Contractor Work, Vendor Work, trees, shrubs, lawns, walks, pavements, roadways, structures, and utilities not designated for removal, relocation, or placement in the course of construction.
- **12.2.2** The Construction Manager shall give notices and comply with all Applicable Laws bearing on safety of Persons or property or their protection from damage, injury, or loss.
- **12.2.3** The Construction Manager shall erect and maintain, as required by existing conditions and the Construction Services Agreement, reasonable safeguards for the safety and protection of persons and property, including posting danger signs and other warnings against hazards, promulgating safety regulations, and notifying owners and users of adjacent sites and utilities. The Construction Manager shall also be responsible, as a Cost of the Work, for all measures necessary to protect any property adjacent to the Project and improvements therein.
- 12.2.4 When use or storage of explosives or other hazardous materials or equipment or unusual methods are necessary for execution of the Work, the Construction Manager shall exercise utmost care and carry on such activities

under supervision of properly qualified personnel. When use or storage of explosives or other hazardous materials or equipment or unusual construction methods are necessary, the Construction Manager shall give the Authority and the Architect reasonable advance notice in writing.

- 12.2.5 The Construction Manager shall promptly remedy damage and loss to property referred to in Subparagraphs 12.2.1.2 and 12.2.1.3 hereof caused in whole or in part by the Construction Manager, a Subcontractor, a Sub-Subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Construction Manager is responsible under Subparagraphs 12.2.1.2 and 12.2.1.3, hereof, except damage or loss attributable to acts or omissions of the Authority or Architect or anyone directly or indirectly employed by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Construction Manager. The foregoing obligations of the Construction Manager is obligations under Paragraph 4.18 hereof.
- **12.2.6** The Construction Manager shall designate a responsible member of the Construction Manager's organization at the Project Site whose duty shall be the prevention of accidents. This person shall be the Construction Manager's representative with specific responsibility and experience to manage the Project Safety Program as the primary responsibility in accordance with the provisions of **Subparagraph 4.9.1** hereof.
- **12.2.7** The Construction Manager shall not load or permit any part of the construction or Project Site to be loaded so as to endanger safety.
- **12.2.8** When all or a portion of the Work is suspended for any reason, the Construction Manager shall securely fasten down all coverings and protect the Work, as necessary, from injury or damage by any cause.
- **12.2.9** The Construction Manager shall promptly report in writing to the Authority and Architect all accidents arising out of or in connection with the Work which cause death, personal injury, or property damage, giving full details and statements of any witnesses. In addition, if death, serious personal injuries, or serious property damages are caused, the accident shall be reported immediately by telephone or messenger to the Authority and the Architect.

### 12.3 EMERGENCIES

12.3.1 In an emergency affecting safety of persons or property, the Construction Manager shall act, at the Construction Manager's discretion, to prevent threatened damage, injury, or loss. Additional compensation or extension of time claimed by the Construction Manager on account of an emergency shall be determined as provided in Paragraph 6.2 and Article 9. As soon as practicable after taking such action, the Construction Manager shall give notice thereof to the Authority and Architect in writing.

### ARTICLE 13 INSURANCE AND BONDS

### 13.1 CONSTRUCTION MANAGER'S LIABILITY INSURANCE

- **13.1.1** The Construction Manager shall obtain and maintain the Required Insurance as set forth in <u>Appendix B</u>, and fully comply with all provisions thereof unless additional insurance is required by a Contract Revision and/or the insurance referenced is provided through a controlled insurance program. In addition, Construction Manager shall obtain and maintain, in a company or companies lawfully authorized to do business in the State of Minnesota, such insurance as will protect the Construction Manager's operations and Work under the Construction Services Agreement and for which the Construction Manager may be legally liable, whether such operations be by the Construction Manager or by a Subcontractor, Sub-Subcontractor or by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable:
  - .1 claims under workers' compensation, disability benefit and other similar employee benefit acts which are applicable to the Construction Manager's Work;
  - .2 claims for damages because of bodily injury, occupational sickness, disease or death of the Construction Manager's employees;
  - .3 claims for damages because of bodily injury, sickness or disease, or death of any person other than the Construction Manager's employees;
  - .4 claims for damages insured by usual personal injury liability coverage which are sustained (1) by a person as a result of an offense directly or indirectly related to employment of such person by the Construction Manager, or (2) by another person;
  - .5 claims for damages, other than to the Work itself, because of injury to or destruction of tangible and intangible property, including loss of use resulting therefrom whether or not the tangible or intangible property is physically injured or destroyed;

- .6 claims for damages because of bodily injury, death of a person or property damage arising out of ownership, maintenance or use of a motor vehicle or equipment;
- .7 claims involving contractual liability insurance applicable to the Construction Manager's obligations under **Paragraph 4.18** hereof, subject to industry standard policy terms and conditions;
- .8 claims for professional liability (errors & omissions) should professional design services be included as part of the Construction Services Agreement; and
- .9 claims for damages because of loss of capital equipment and tools that are owned or rented by the Construction Manager or its Subcontractors.
- 13.1.2 The Construction Manager shall, for the protection and benefit of the Authority, the Indemnitees (except the Architect Indemnitees, who shall only be an Additional Insured to the extent of the Construction Manager's negligence), and the Construction Manager, and as part of the Construction Manager's efforts to satisfy the obligation set forth in Subparagraph 13.1.1 hereof, procure, pay for and maintain in full force and effect, at all times during the performance of the Work until Final Completion and acceptance of the Work or for such duration thereafter as required, policies of insurance issued by a responsible carrier or carriers acceptable to the Authority and in form and substance reasonably satisfactory to the Authority which afford the coverages set forth in <u>Appendix B</u> hereto. The Authority does not in any way represent that the insurance or limits of insurance specified herein are sufficient or adequate to protect the Construction Manager, but are merely minimums. In addition to what is specified in this Article 13 and <u>Appendix B</u>, Contractor shall take such steps as it may deem necessary to insure itself against all special and unusual hazards peculiar to this Project.
- 13.1.3 The costs of all insurance premiums required under this Paragraph 13.1 and Appendix B, shall be included in the Construction Manager's Cost of Work and GMP as set forth in Exhibit 3 of the Construction Services Agreement, unless the Authority elects to implement an owner controlled insurance program pursuant to Appendix B. The costs of all associated deductibles required under this Paragraph 13.1 and Appendix B, shall be included in the Construction Manager's Fee as set forth in Exhibit 3 of the Construction Services Agreement. The Construction Manager hereby agrees to deliver to the Authority within ten (10) days of the date of the Construction Services Agreement and prior to any equipment or personnel being utilized in connection with the Project or the Work or brought onto the Project Site, certified copies of all insurance policies procured by the Construction Manager under or pursuant to this Paragraph 13.1 and evidencing the required coverages with limits not less than those specified in Appendix B. The receipt by the Authority of such policies does not constitute approval or agreement by the Authority that the insurance requirements of this Construction Services Agreement have been met or complied with. The Authority and the Indemnitees (except the Architect Indemnitees, who shall only be an Additional Insured to the extent of the Construction Manager's negligence) shall be included as additional insureds as set forth in Appendix B. The Commercial General Liability Policy, Automobile Policy, and Umbrella/Excess Liability Policy obtained under or pursuant to this Paragraph 13.1 and Appendix B, shall each be primary and non-contributory to any valid and collectible insurance carried separately the Authority, or any of the Indemnitees (except the Architect Indemnitees, who shall only be an Additional Insured to the extent of the Construction Manager's negligence), except for Builders Risk insurance. Further, the Construction Manager warrants and represents that it will provide that no less than thirty (30) days' prior written notice to the Authority and the Indemnitees before any material alteration, cancellation, non-renewal or expiration of the coverage to the insurance it obtains pursuant to this Paragraph 13.1 and Appendix B.
- 13.1.4 In no event shall any approval or lack of approval of insurance or receipt or lack of receipt by the Authority of certified copies or certificates of policies required under this Paragraph 13.1 and <u>Appendix B</u> or failure to demand receipt of such certified copies or certificates prior to the Construction Manager commencing the Work be construed as a waiver by the Authority or any of the Indemnitees of the Construction Manager's obligations to obtain insurance pursuant to this Paragraph 13.1 and <u>Appendix B</u>. The obligation to procure and maintain any insurance required by this Paragraph 13.1 and <u>Appendix B</u> is a separate responsibility of the Construction Manager and independent of the duty to furnish a certified copy or certificate of such insurance policies.
- 13.1.5 [Reserved]
- **13.1.6** Thirty days (30) days prior to the expiration or non renewal of any insurance required by this **Paragraph 13.1** and **Appendix B**, the Construction Manager shall supply the Authority with Certificates of Insurance and amendatory riders or endorsements that clearly evidence the continuation of all coverage in the same manner, limits of protection, and scope of coverage as is required in this **Paragraph 13.1**. In the event any renewal or replacement policy, for whatever reason obtained or required, is written by a carrier other than that with whom the coverage was previously placed, or the subsequent policy differs in any way from the previous policy, the Construction Manager shall also furnish the Authority with a certified copy of the renewal or replacement policy unless the Authority provides the Construction Manager with prior written consent to submit only a Certificate of Insurance for any such policy. All renewal and replacement policies shall be in form and substance satisfactory to the Authority and written by carriers acceptable to the Authority.

- **13.1.7** Any general aggregate limit under the Construction Manager's Commercial General Liability Policy and Umbrella/Excess Liability Policy shall apply to this Project separately and shall not require exhaustion of any policies providing Construction Manager coverage as an additional insured.
- **13.1.8** The Construction Manager shall cause each first-tier Subcontractor to (1) provide insurance in accordance with the terms and conditions of Construction Manager's standard Subcontract Agreement and reasonably satisfactory to the Authority, unless otherwise approved in writing by the Authority, and (2) name the Construction Manager, Authority and the Indemnitees (except the Architect Indemnitees, who shall only be and Additional Insured to the extent of the Subcontractor's negligence) as additional insureds under the Subcontractor's Commercial General Liability Policy. The additional insured endorsement included on the Subcontractor's Commercial General Liability Policy shall state that coverage is afforded the additional insured with respect to claims arising out of ongoing and completed operations performed by or on behalf of the Subcontractor. With the exception of any coverage provided under an Owner Controlled Insurance Program, if the additional insured shave other insurance which is applicable to the Project, such other insurance shall be, for the purposes hereof, on an excess or contingent basis. The amount of the insurer's liability under Subcontractor's insurance policy shall not be reduced by the existence of such other insurance.
- 13.1.9 [Reserved]
- 13.1.10 [Reserved]
- 13.1.11 [Reserved]

### 13.2 AUTHORITY'S LIABILITY INSURANCE

**13.2.1** The Authority shall be responsible for purchasing and maintaining the Authority's liability insurance in an amount to be determined by the Authority in its sole discretion.

#### 13.2.2 [Reserved]

**13.2.3** With the exception of any deductibles payable pursuant to an Owner Controlled Insurance Program, any deductibles required to be paid for claims made by the Construction Manager or Subcontractors against the Authority's liability insurance shall be paid by the claiming Construction Manager or claiming Subcontractor. Construction Manager's payments of the Authority's deductible for claims against the Authority's Liability insurance may be funded from the Construction Manager's Contingency if approved by the Authority.

### 13.3 PROPERTY INSURANCE

- **13.3.1** Before any portion of the Construction Phase Services is commenced, or in the sole discretion of the Authority before any orders for long lead-time items are placed, the Authority shall purchase and maintain property insurance and/or builder's risk insurance in an amount and with coverages to be determined by the Authority for the Work at the Project Site or portions thereof and materials stored off the Project Site or being shipped to the Project Site. The Authority will consult with the Construction Manager before procuring said copy of the property insurance and/or builder's risk insurance, but the Authority will retain the discretion to determine the types and limits of said insurance.
- 13.3.2 [Reserved]
- 13.3.3 [Reserved]
- **13.3.4** If the Construction Manager requests in writing that insurance for risks other than those described herein or for other special hazards be included in the property insurance policy, the Authority shall, if possible, include such insurance, and the cost thereof shall be charged to the Construction Manager by appropriate Contract Revision.
- 13.3.5 [Reserved]
- 13.3.6 [Reserved]
- 13.3.7 Waivers of Subrogation. The Authority, Architect, and Construction Manager waive all rights of subrogation and claims of any sort or kind against each other and any of their respective Subconsultants, Subcontractors, Suppliers, Sub-Subcontractors of all tiers, agents, and employees, whether directly or indirectly through any means through their Subconsultants, Subcontractors, Sub-Subcontractors of all tiers, agents, sub-Subcontractors of all tiers, agents, and employees, for loss or damages caused by fire or other named perils to the extent covered by property insurance obtained pursuant to this Paragraph 13.3 or other property insurance applicable to the Work, except such rights as they have to proceeds of such insurance held by the Authority as fiduciary. The Contract Documents hereby require of the Authority, Architect, Construction Manager, and Trade Contractors, if any, and their Subconsultants, Subcontractors of all tiers, agents, agente, agreement, to include similar waivers of subrogation each in favor of other parties enumerated herein. All insurance policies required in the Contract Documents shall provide such waivers of subrogation by endorsement or otherwise. A waiver of subrogation shall be effective as to a Person even though that Person

would otherwise have a duty of indemnification, contractual or otherwise, did not pay the insurance premium directly or indirectly, and whether or not the person or entity had an insurable interest in the property damaged. The Architect and Construction Manager shall provide to the Authority, prior to the commencement of the Work, evidence that all of the insurance policies required herein contained an appropriate endorsements as provided in this **Subparagraph 13.3.7**.

- **13.3.8** A loss insured under the Authority's property insurance shall be adjusted by the Authority and made payable to the Authority for the benefit of the insureds, as their interests may appear, subject to requirements of any applicable mortgagee clause and of **Subparagraph 13.3.10** hereof. Any costs, including attorneys' fees, involved in adjusting the loss shall be reimbursed to the Authority if not covered by the property insurance. The Construction Manager shall pay Subcontractors their just shares of insurance proceeds received by the Construction Manager, and by appropriate agreements, written where legally required for validity, shall require Subcontractors to make payments to their Sub-Subcontractors in similar manner.
- 13.3.9 [Reserved]
- 13.3.10 [Reserved]
- **13.3.11** Partial occupancy or use in accordance with **Paragraph 11.9** hereof shall not commence until the insurance company or companies providing property insurance have consented to such partial occupancy or use by endorsement or otherwise. The Authority and the Construction Manager shall take reasonable steps to obtain consent of the insurance company or companies and shall, without mutual written consent, take no action with respect to partial occupancy or use that would cause cancellation, lapse of reduction of insurance.
- 13.3.12 Any deductibles required to be paid for claims made by the Construction Manager or Subcontractors against Authority's Property and/or Builder's Risk insurance shall be paid by the claiming Construction Manager or claiming Subcontractor. Construction Manager's payments of the Authority's deductible for Property/Builder's Risk claims may be funded from the Construction Contingency if approved by the Authority.

### 13.5 PERFORMANCE BOND AND PAYMENT BOND

- **13.5.1** Prior to the commencement of the Work, the Construction Manager shall furnish a Performance Bond and Labor and Material Payment Bond meeting all statutory requirements of the Act and the State of Minnesota, in a form and substance satisfactory to the Authority and the Team, and complying with the following specific requirements:
  - .1 Except as otherwise required by statute, the form and substance of such bonds shall be satisfactory to the Authority, in its sole discretion;
  - .2 Bonds shall be executed by a responsible surety licensed and authorized to do business in the State of Minnesota, with a Best's Key Rating Guide's rating of no less than A:X and shall remain in effect for a period not less than two (2) years following the date of Substantial Completion or the time required to resolve any items of incomplete Work and the payment of any disputed amounts, whichever time period is longer;
  - .3 The Performance Bond and the Labor and Material Payment Bond shall each be in an amount equal to the GMP and the Performance Bond shall guarantee Contractor will satisfactorily perform each and every part of the Construction Services Agreement, including all guarantees and warranties required;
  - .4 The Construction Manager shall require the attorney in fact who executes the required bonds on behalf of the surety to affix thereto a certified and current copy of his/her power of attorney indicating the monetary limit of such power;
  - .5 Late completion shall be considered a failure of performance and any liquidated damages owed by the Construction Manager as a result of late completion shall be covered by Construction Manager's performance bond; and
  - .6 Every Bond under this **Subparagraph 13.5.1** must display the Surety's Bond Number. A rider including the following provisions shall be attached to each Bond:
    - .1 Surety hereby agrees that it consents to and waives notice of any addition, alteration, omission, change, or other modification of the Contract Documents which, singularly or in the aggregate, does not exceed One Hundred Fifty Percent (150%) of the GMP. Except as to increases in the GMP in excess of the percentage set for the above in this Subparagraph 13.5.1.6(i), any other addition, alteration, change, extension of time, or other modification of the Contract Documents, or a forbearance on the part of the Authority or the Construction Manager to each other, shall not release the Surety of its obligations hereunder and notice to the Surety of such matters is hereby waived.

- .2 Surety further agrees that in event of any default by the Authority in the performance of the Authority's obligations to the Construction Manager under the Construction Services Agreement, the Construction Manager or Surety shall cause written notice of such default, specifying said default in detail, to be given to the Authority, and the Authority shall have thirty (30) days after the receipt of such notice within which to cure such default, or such additional reasonable period of time as may be required if the nature of such default is such that it cannot be cured within thirty (30) days. Such Notice of Default shall be sent by certified or registered U.S. Mail, return receipt requested, first class postage prepaid, to the Authority.
- .3 Surety agrees that it is obligated under the bonds to the Authority and to any successor, grantee or assignee of the Authority.
- .4 The costs of all bonds required under this **Paragraph 13.5** shall be a Cost of the Work.
- .5 In any Claim involving the surety and the Construction Manager, the surety shall be bound by and agrees to be a party to the dispute resolution provisions in **Article 6** of these General Conditions, including arbitration.
- **13.5.2** Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Construction Services Agreement, the Construction Manager shall promptly furnish a copy of the bonds or shall permit a copy to be made.
- **13.5.3** The Construction Manager shall keep the surety informed of the progress of the Work, and, where necessary, obtain the surety's consent to, the waiver of: (1) notice of changes in the Work; (2) request for reduction or release of retention; (3) request for final payment; and (4) any other material required by the surety. The Authority shall be notified by the Construction Manager, in writing, of all significant communications with the surety regarding the Project. The Authority may inform surety of the progress of the Work and obtain consents as necessary to protect the Authority's rights, interest, privileges and benefits under pursuant to any bond issued in connection with the Work.
- **13.5.4** Construction Manager shall advise the Authority of the commission the Construction Manager will have pay to procure the bonds required by **Paragraph 13.5**, and if Authority considers the commission unreasonable, the Authority shall have the right to use its broker to procure the required bonds for the Construction Manager.

## 13.6 GENERAL REQUIREMENTS

- **13.6.1** All insurance coverages required under this **Article 13** shall be provided by insurance companies having policyholder ratings no lower than "A" and financial ratings not lower than "X" in the Best's Key Rating Guide, latest edition in effect as of the date of the respective contracts contemplated hereunder, and subsequently in effect at the time of renewal of any policies required by the Contract Documents.
- 13.6.2 If the Authority or any Indemnitee is damaged by the failure of any other party to purchase or maintain insurance required under this Article 13, then the party required to obtain or maintain, or to cause to be obtained or maintained, the insurance shall bear all damages and costs, including attorneys' fees and court and settlement expenses, attributable thereto.

#### 13.6.3 [Reserved]

- **13.6.4** By requiring insurance, the Authority does not represent that coverage and limits will necessarily be adequate to protect the Construction Manager's or Architect's interests. The types and limits of insurance procured by the Construction Manager are in the control of the Construction Manager, and the Construction Manager's failure, if any, not to obtain adequate or appropriate insurance to insure shall not serve to reduce or limit Construction Manager's contractual insurance obligations including its contractual obligation to indemnify and hold harmless the Authority or the Indemnitees.
- **13.6.5** If any insurance required herein is to be issued or renewed on a claim-made form as opposed to the occurrence form, the retroactive date for coverage will be no later than the commencement date of the Project and will state that in the event of cancellation or nonrenewal, the discovery period for insurance claims (tail coverage) will be at least sixty (60) months.
- **13.6.6** All policies, except for the Professional Liability Policy, and Workers Compensation policies (if such waivers are not available), shall be endorsed to provide a waiver of subrogation in favor of the Authority and Indemnitees.

#### ARTICLE 14 UNCOVERING AND CORRECTION OF WORK

# 14.1 UNCOVERING OF WORK

- 14.1.1 If a portion of the Work is covered contrary to the Architect's or Authority's request or to requirements specifically expressed in the Contract Documents, it must, if required in writing by the Architect, the Authority, or any Governmental Authority, be uncovered for their observation, and be replaced at the Construction Manager's expense without change in the Contract Time or the GMP. If prior to the date of Substantial Completion, the Construction Manager, a Subcontractor, Supplier, or anyone for whom either is responsible uses or damages any portion of the Work, including mechanical, electrical, plumbing, or other building systems, machinery, equipment, the Construction Manager shall cause such item to be restored to "like new" condition at no expense to the Authority.
- **14.1.2** If a portion of the Work has been covered which the Architect has not specifically requested to observe prior to its being covered, the Architect or Authority may request to see such Work and it shall be uncovered by the Construction Manager. If such Work is in accordance with the Contract Documents, costs of uncovering and replacement shall, by appropriate Contract Revision, be charged to the Authority. If such Work is not in accordance with the Construction Manager shall pay such costs.

#### 14.2 CORRECTION OF WORK

- 14.2.1 The Construction Manager shall promptly correct Work rejected by the Architect or the Authority for failing to conform to the requirements of the Contract Documents, whether observed before or after Substantial Completion and whether or not fabricated, installed or completed. The Construction Manager shall bear all costs of correcting such rejected Work, including additional testing and inspections and compensation for the Architect's or a Consultant's services and expenses made necessary thereby and such costs shall not be considered a Cost of the Work. If prior to the date of Substantial Completion, the Construction Manager, a Subcontractor, or anyone for whom either is responsible uses or damages any portion of the Work, including, without limitation, mechanical, electrical, plumbing, or other building systems, machinery, or equipment, the Construction Manager shall cause such item to be restored to "like new" condition.
- 14.2.2 If, within two (2) years after the date of Substantial Completion of the Work or designated portion thereof, or after the date for commencement of warranties established under Subparagraph 11.9.1 hereof, or by terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Construction Manager shall correct it promptly after receipt of written notice from the Authority to do so unless the Authority has previously given the Construction Manager a written acceptance of such condition. This period of two (2) years shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual performance of the Work. This obligation under this Subparagraph 14.2.2 shall survive Final Completion and acceptance of the Work under the Construction Services Agreement and termination of the Construction Services Agreement. The Authority shall provide such notice promptly after discovery of the improper conditions.
- **14.2.3** The Construction Manager shall remove from the Project Site portions of the Work which are not in accordance with the requirements of the Contract Documents and are neither corrected by the Construction Manager nor accepted by the Authority.
- 14.2.4 If the Construction Manager fails to correct nonconforming Work within a reasonable time, the Authority may correct it in accordance with **Subparagraph 15.4.4**. hereof. If the Construction Manager does not proceed with correction of such nonconforming Work within a reasonable time fixed by written notice from the Authority, the Authority may remove it and store the salvageable materials or equipment at the Construction Manager's expense. If the Construction Manager does not pay costs of such removal and storage within ten (10) days after written notice, the Authority may upon ten (10) additional days' written notice sell such materials and equipment at auction or at private sale and shall account for the proceeds thereof, after deducting costs and damages that should have been borne by the Construction Manager, including compensation for the Architect's and the Construction Manager should have borne, the GMP shall be reduced by the deficiency. If payments then or thereafter due the Construction Manager are not sufficient to cover such amount, the Construction Manager shall pay the difference to the Authority.
- **14.2.5** The Construction Manager shall bear the cost of correcting destroyed or damaged construction, whether completed or partially completed, of the Authority or Trade Contractors caused by the Construction Manager's correction or removal of Work or Trade Contractor Work or Vendor Work which is not in accordance with the requirements of the Contract Documents.
- **14.2.6** The Architect shall promptly give notice to the Authority of any defective or nonconforming Work of the Construction Manager or defective or nonconforming Design Services of the Architect when such defective or nonconforming Work or Design Services are discovered or observed by the Architect, Architect's employees, or Architect's Subconsultants. The Architect shall be responsible to promptly make corrections in the Architect's

Design Services when the Architect's Design Services are found to contain errors or omissions by the Architect, Architect's employees, or Architect's Subconsultants. Costs associated with corrections in the Architect's Design Services, including damages or corrections involving the Work, resulting from errors or omissions in the Design Services of the Architect, its employees, or Subconsultants, shall be borne by the Architect.

14.2.7 Nothing contained in this Paragraph 14.2 shall be construed to establish a period of limitation with respect to other obligations that the Construction Manager or Architect might have under the Contract Documents. Establishment of the time period of two (2) years as described in Subparagraph 14.2.2 hereof relates only to the specific obligation of the Construction Manager or Architect to correct its respective Work or Design Services, and has no relationship to the time within which the obligation of the Construction Manager or Architect to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Construction Manager's or Architect's liability with respect to the Construction Manager's or Architect's liability with respect to the Construction Manager's or Architect's liability with respect to the Construction Manager's or Architect's liability with respect to the Construction Manager's or Architect's liability with respect to the Construction Manager's or Architect's liability with respect to the Construction Manager's or Architect's liability with respect to the Construction Manager's or Architect's liability to correct their respective Work or Design Services.

# 14.3 ACCEPTANCE OF NONCONFORMING WORK

**14.3.1** If the Authority prefers to accept Work which is not in accordance with the requirements of the Contract Documents, the Authority may do so instead of requiring its removal and correction, in which case the GMP or Costs of the Work will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

#### ARTICLE 15 MISCELLANEOUS PROVISIONS

## 15.1 GOVERNING LAW

- **15.1.1** The Construction Services Agreement shall be governed by the law of the State of Minnesota, without giving any effect to conflicts of law provisions.
- **15.1.2** Historical lack of enforcement of any Applicable Laws shall not constitute a waiver of the Construction Manager's responsibility for compliance with such law in a manner consistent with the Contract Documents unless and until the Construction Manager has received written consent for the waiver of such compliance from the Authority and the Governmental Authority responsible for the enforcement of Applicable Laws.

# 15.2 SUCCESSORS AND ASSIGNS

**15.2.1** The Authority, Architect and Construction Manager respectively bind themselves, their partners, successors, assigns, and legal representatives to the other parties hereto and to partners, successors, assigns, and legal representatives of such other party in respect to covenants, agreements, and obligations contained in the Contract Documents. The Architect and the Construction Manager may not assign their rights or obligations under the Contract Documents. The Authority may assign its rights and obligations under the Construction Services Agreement, and the Construction Manager agrees to complete the Work according to such assignment upon appropriate provision for payment of the balance of the payments due the Construction Manager. The Construction Manager acknowledges and agrees that the Team is a third party beneficiary under the Construction Services Agreement. Any entity which shall succeed to the rights of the Authority shall be entitled to enforce its rights under the Construction Services Agreement. If the Construction Manager makes an assignment with the consent of the Authority, the Construction Manager shall nevertheless remain legally responsible for all obligations under the Construction Services Agreement.

## 15.3 WRITTEN NOTICE

**15.3.1** Written notice shall be deemed to have been duly served if delivered in person or sent by registered or certified mail to the individuals at the business address set forth on <u>Appendix E</u> hereof, or to any other address subsequently identified in writing by any party hereto.

# 15.4 RIGHTS AND REMEDIES

- **15.4.1** Except as expressly provided herein, duties and obligations imposed by the Contract Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights, and remedies otherwise imposed or available by law or equity.
- **15.4.2** No action or failure to act by the Authority, Architect, or Construction Manager shall constitute a waiver of a right or duty afforded them under the Contract Documents, nor shall such action or failure to act constitute approval of or acquiescence in a breach thereunder, except as may be specifically agreed in writing.
- **15.4.3** If the Construction Manager fails to carry out or correct any Work that is not in accordance with the requirements of the Contract Documents, the Authority, by written order, may order the Construction Manager

to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Authority to stop the Work shall not give rise to a duty on the part of the Authority to exercise this right for the benefit of the Construction Manager or any other Person, except to the extent required by **Subparagraph 8.1.3** hereof. This right shall be in addition to the rights under **Paragraph 14.2** and **Paragraph 16.3** hereof.

15.4.4 If the Construction Manager defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a seven (7) day period after receipt of written notice from the Authority to commence and continue correction of such default or neglect with diligence and promptness, the Authority may, without prejudice to other remedies the Authority may have, commence and continue to carry out the Work. In such case an appropriate Contract Revision shall be issued deducting from payments then or thereafter due the Construction Manager the cost of correcting such deficiencies, including compensation for any additional services and expenses made necessary by such default, neglect, or failure. The right of the Authority to stop the Work pursuant to this **Subparagraph 15.4.4** shall not give rise to any duty on the part of the Authority to exercise this right for the benefit of the Construction Manager or any other Person.

#### 15.5 TESTS AND INSPECTIONS

- **15.5.1** Tests, inspections, and approvals of portions of the Work required by the Contract Documents or by Applicable Laws shall be made and paid for by the Construction Manager at an appropriate time. Unless otherwise provided, the Construction Manager shall make arrangements for such tests, inspections, and approvals with an independent testing laboratory or entity designated by the Authority, or with the appropriate Governmental Authorities, and if such tests, inspections, and approvals are not included in the Construction Manager shall give the Authority shall bear all related costs of tests, inspections, and approvals. The Construction Manager shall give the Architect timely notice of when and where tests and inspections are to be made so the Architect may observe such procedures. The Authority shall bear costs of tests, inspections, or approvals that do not become requirements until after bids are received or negotiations concluded.
- 15.5.2 If the Architect, Authority, or Governmental Authorities determine what portions of the Work require additional testing, inspection, or approval not included under Subparagraph 15.5.1 hereof, the Architect will, upon written authorization from the Authority, instruct the Construction Manager to make arrangements for such additional testing, inspection, or approval by an entity designated by the Authority, and the Construction Manager shall give timely notice to the Architect of when and where tests and inspections are to be made so the Architect may observe such procedures. The Authority shall bear such costs except as provided in Subparagraph 15.5.3 hereof.
- 15.5.3 If such procedures for testing, inspection, or approval under **Subparagraphs 15.5.1** and **15.5.2** hereof reveal failure of the portions of the Work to comply with requirements established by the Contract Documents, the Construction Manager shall bear all costs made necessary by such failure including those of repairing or replacing the affected Work and compensation for the Architect's services and expenses. The Construction Manager also agrees that the cost of testing services required for the convenience of the Construction Manager in its scheduling and performance of the Work, and the cost of testing services related to remedial operations performed to correct deficiencies in the Work shall be borne by the Construction Manager.
- **15.5.4** Required certificates of testing, inspection, or approval shall, unless otherwise required by the Contract Documents, be secured by the Construction Manager and promptly delivered to the Architect.
- **15.5.5** If the Architect is to observe tests, inspections, or approvals required by the Contract Documents, the Architect will do so promptly and, where practicable, at the normal place of testing.
- **15.5.6** Tests or inspections conducted pursuant to the Contract Documents shall be made promptly to avoid unreasonable delay in the Work.

#### 15.6 GENERAL PROVISIONS

#### 15.6.1 [Reserved]

- **15.6.2** Whenever possible, each provision of the Contract Documents, the Construction Services Agreement, and the Design Services Agreement shall be interpreted in a manner as to be effective and valid under Applicable Laws. If, however, any provision of the Contract Documents, or portion thereof, is prohibited by law or found invalid under any law, any such provision or portion thereof shall be ineffective, without in any manner invalidation or affecting the remaining provisions of the Contract Documents, the Construction Services Agreement and the Design Services Agreement or valid portions of such provision, which are hereby deemed severable.
- **15.6.3** Each Party hereto agrees to do all acts and things and to make, execute and deliver such written instruments, as shall from time to time be reasonably required to carry out the terms and provisions of the Contract Documents.
- **15.6.4** Any specific requirement in the Contract Documents or the Construction Services Agreement that the responsibilities or obligations of the Construction Manager also apply to a Subcontractor is added for emphasis and are also hereby deemed to include a Subcontractor of any tier. The omission of a reference to a

Subcontractor in connection with any of the Construction Manager's responsibilities or obligations shall not be construed to diminish, abrogate or limit any responsibilities or obligations of a Subcontractor of any tier under the Contract Documents or the applicable Subcontract.

### 15.6.5 [Reserved]

**15.6.6 Attorneys' Fees and Costs.** If any legal action, litigation or other proceeding, including arbitration, is brought for the enforcement of the Construction Services Agreement, the Design Services Agreement, or the Contract Documents and/or because of an alleged dispute, breach, default or misrepresentation in connection with any provision of the Construction Services Agreement, the Design Services Agreement or the Contract Documents, the prevailing Party shall be entitled to recover, in addition to all other amounts awarded, reasonable attorneys' fees all reasonable costs and fees associated with paralegal, experts, consultants, or others engaged for the purposes of the litigation or proceeding, in addition to court costs, the expenses of arbitration, and other reasonable costs bringing or defending the action, and, in addition, any other relief of which it may be entitled.

#### <u>ARTICLE 16</u> TERMINATION OR SUSPENSION OF THE CONTRACT

### 16.1 TERMINATION BY THE CONSTRUCTION MANAGER

- **16.1.1** The Construction Manager may terminate the Construction Services Agreement if the Work is stopped through no act or omission of the Construction Manager, a Subcontractor, Sub-Subcontractor, or any other persons performing portions of the Work under contract with them, for any of the following reasons:
  - .1 issuance of an Order of a court or other Governmental Authority that shuts the Project down completely for an uninterrupted period of one hundred eighty (180) days; or
  - .2 an act of a Government Authority, such as a declaration of national emergency, making material necessary to complete activities on the critical path of the Construction Schedule unavailable for an uninterrupted period of one hundred eighty (180) days;
  - .3 if repeated suspensions, delays, or interruptions of the Work by the Authority cause critical path delays to the Construction Schedule, which constitute in the aggregate more than one hundred percent (100%) of the total number of days shown in the initial Construction Schedule to be scheduled for completion, or one hundred eighty (180) days of critical path delay to the Construction Schedule in any 365-day period, whichever is less; or
  - .4 if all the Work on the Project is completely stopped for a period of one hundred eighty (180) days through no act or omission of the Construction Manager or a Subcontractor or any Person performing portions of the Work under contract with them, because the Authority has persistently failed to fulfill the Authority's obligations under the Contract Documents with respect to matters critical to the progress of the Work.
- 16.1.2 If one of the above reasons exists, the Construction Manager may provide the Authority and Architect a thirty (30) day written notice to cure, and if the Authority does not diligently begin and continue to cure the cause of the termination within said thirty (30) days, the Construction Manager may serve a second ten (10) day termination notice to the Authority, and if the cause of the termination remains uncured, the Construction Manager may terminate the Construction Services Agreement and the Construction Manager's sole and exclusive remedy shall be payment for the allowable Costs of the Work properly and timely performed prior to the effective date of termination and for items properly and timely fabricated off the Project Site, delivered, and stored in accordance with the Authority's instructions, and proven costs of termination for the Construction of the Construction Manager's Fee attributable to the Work actually executed. The Construction Manager hereby waives and releases all other Claims for payment and damages, including anticipated Fee, profits or lost overhead for portions of the unperformed Work. The Authority shall be credited for:
  - .1 payments previously made to the Construction Manager for the terminated portion of the Work;
  - .2 Claims that the Authority has against the Construction Manager under the Construction Services Agreement; and
  - .3 the value of the materials, supplies, equipment, or other items that are to be disposed of by the Construction Manager that are part of the GMP.
- 16.1.3 If the Authority has not issued a Certificate for Payment, or the Authority has not made payment on an approved Certificate of Payment, and the Construction Manager is entitled to stop the Work as set forth in Subparagraph 11.7.1, and if the Authority's failure to issue a Certificate for Payment or failure to make a payment on a Certificate of Payment continues uncured for one hundred twenty (120) days, the Construction

Manager may terminate the Construction Services Agreement upon ten (10) days additional written notice, and the Construction Manager's sole remedy shall be as set forth in **Subparagraph 16.1.2**.

### 16.2 TERMINATION BY THE AUTHORITY FOR CAUSE

- 16.2.1 The Authority may terminate the Construction Services Agreement if the Construction Manager:
  - .1 repeatedly refuses or fails to supply enough properly skilled workers or proper materials;
  - .2 fails to make payment to Subcontractors or Suppliers for materials or labor in accordance with the respective agreements between the Construction Manager and the Subcontractors;
  - .3 repeatedly disregards Applicable Laws or Orders of a Public Authority;
  - .4 otherwise is in substantial breach of a provision of the Contract Documents;
  - .5 becomes insolvent, files a Petition in Bankruptcy, or is adjudicated bankrupt, makes a general assignment to the benefit of its creditors, or cannot pay its debts as they generally become due; or
  - .6 suspends its business operations or otherwise fails to operate its business in the ordinary course.
- **16.2.2** When any of the above reasons exist, the Authority may, without prejudice to any other rights or remedies of the Authority and after giving the Construction Manager and the Construction Manager's surety, if any, seven days' written notice, terminate employment of the Construction Manager and may, subject to any prior rights of the surety:
  - .1 Exclude the Construction Manager from the site and take possession of all materials, equipment, tools, and construction equipment and machinery thereon owned by the Construction Manager;
  - .2 Accept assignment of Subcontracts pursuant to Paragraph 7.4 hereof; and
  - .3 Finish the Work by whatever reasonable method the Authority may deem expedient; provided, however, in the event such matters are cured by the Construction Manager within said seven (7) days to the satisfaction of the Authority, such termination notice shall, upon written notice from Authority, be rescinded.

Upon written request of the Construction Manager, the Authority shall furnish to the Construction Manager an accounting of the costs incurred by the Authority in finishing the Work after a termination for cause.

- **16.2.3** When the Authority terminates the Construction Services Agreement for one of the reasons stated in **Subparagraph 16.2.1** hereof, the Construction Manager shall not be entitled to receive further payment until the Work is finished.
- 16.2.4 If the unpaid balance of the GMP exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, and other damages and costs incurred by the Authority in finishing the Work and not expressly waived, such excess shall be paid to the Construction Manager to the extent the Construction Manager has incurred unpaid Costs of the Work. If such costs and damages of the Authority to finish the Work exceed the unpaid balance of the GMP, the Construction Manager shall pay the difference to the Authority. The obligation to pay and the amount to be paid to the Construction Manager or Authority, as the case may be, shall survive termination of the Construction Services Agreement.
- 16.2.5 If, after a termination for cause, it is determined that the Construction Manager was not in breach or default, the rights and obligations of the parties will be the same as if the termination had been issued for convenience pursuant to **Paragraph 16.4** hereof, and the Construction Manager's sole and exclusive remedy from the Authority shall then be any payment required by that **Paragraph 16.4**.

### 16.3 SUSPENSION BY THE AUTHORITY FOR CONVENIENCE

- **16.3.1** The Authority may, without cause, order the Construction Manager in writing to suspend, delay, or interrupt the Work in whole or in part for such period of time as the Authority may determine.
- 16.3.2 The GMP and Contract Time shall be adjusted for increases in allowable Cost of the Work and time caused by suspension, delay, or interruption as described in **Subparagraph 16.3.1** hereof. Adjustment of the GMP shall include Fee, which shall be limited to \_\_\_% of the allowable Costs of the Work associated with the suspension, delay, or interruption. Furthermore, no adjustment in the GMP or Contract Time shall be made to the extent:
  - .1 that performance is, was or would have been so suspended, delayed, or interrupted by any other cause for which the Construction Manager is responsible; or

- .2 that an equitable adjustment is made or denied under another provision of the Construction Services Agreement; or
- .3 the suspension, delay, or interruption does not delay, suspend or interrupt activities then currently on the critical of the Construction Schedule; or
- .4 the Construction Manager fails to provide a notice of its Claim for adjustment to the GMP and Contract Time as required by **Subparagraph 6.2.3** hereof.

# 16.4 TERMINATION BY THE AUTHORITY FOR CONVENIENCE

- **16.4.1** The Authority may, in its sole discretion and at any time, terminate the Construction Services Agreement for the Authority's convenience and without cause.
- **16.4.2** Upon receipt of written notice from the Authority of such termination for the Authority's convenience, the Construction Manager shall:
- .1 cease operations as directed by the Authority in the notice;
- .2 take actions necessary, or that the Authority may direct, for the protection and preservation of the Work; and
- .3 except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing Subcontracts and purchase orders and enter into no further Subcontracts and purchase orders.
- **16.4.3** In case of such termination for the Authority's convenience, the Construction Manager shall recover, as its sole remedy, payment for the allowable Costs of the Work properly performed prior to the effective date of termination and for items properly and timely fabricated off the Project Site, delivered, and stored in accordance with the Authority's instructions, and proven costs of termination for the Construction Manager such as, for example, restocking charges, non-cancelable material purchase orders, and that portion of the Construction Manager's Fee attributable to the Work actually executed. The Construction Manager hereby waives and releases all other Claims for payment and damages, including anticipated Fee, profits or lost overhead for portions of the unperformed Work. The Authority shall be credited for:
  - .1 payments previously made to the Construction Manager for the terminated portion of the Work;
  - .2 Claims that the Authority has against the Construction Manager under the Construction Services Agreement; and
  - .3 the value of the materials, supplies, equipment, or other items that are to be disposed of by the Construction Manager that are part of the GMP.

## 16.5 [Reserved]

16.5.1 [Reserved]

### GENERAL CONDITIONS OF THE CONSTRUCTION SERVICES AGREEMENT APPENDIX A

#### **DEFINITIONS**

Defined terms as used in the Contract Documents shall have the following meanings:

"<u>Act</u>" shall mean 2012 Minnesota Laws, Chapter 299, as enacted or hereafter amended or supplemented, and including any successor law, providing for, among other things, the construction, financing, operation, and long-term use of a multi-purpose stadium and related infrastructure as a venue for the National football League and a broad range of other civic, community, athletic, educational, cultural, and commercial activities in the City of Minneapolis, Minnesota.

"<u>ADA</u>" shall mean Title III of the Americans with Disabilities Act and the regulations and definitive guidelines issued thereunder by the United States Department of Justice concerning accessibility of places and public accommodation and commercial facilities, as well as any other Applicable Laws governing the same or similar subject matter.

"<u>Addenda</u>" shall mean shall written or graphic instruments issued prior to the execution of the Construction Services Agreement which modify or interpret the Contract Documents, including the Drawings and Specifications, by additions, deletions, clarifications or corrections.

"Additional Insured Parties" shall have the meaning set forth in Subparagraph 9.1.15 of the Design Services Agreement.

"<u>Additional Services</u>" shall mean services beyond the scope of Design Services that are authorized or approved in writing by Contract Revision by the Authority as described in detail in **Paragraph 2.2** of the Design Services Agreement.

"Additional Services Authorization" shall mean the Authority's written approval of an Additional Service proposed by the Architect, as described in Paragraph 2.2 of the Design Services Agreement.

"Adjacent Property" shall mean all land adjoining and surrounding the Stadium Site on which will be located any public streets, sidewalks, plazas, or bridges and any public or private parking facilities or other accoutrements to be developed by Authority or other parties in connection with the Project.

"<u>Affiliate</u>" or "Affiliated Entity" of a specified Person shall mean any entity, corporation, partnership, limited liability company, sole proprietorship or other Person that directly or indirectly, through one or more intermediaries controls, is controlled by, or is under common control with the Person specified. For purposes of the Design Services Agreement and Contract Documents the terms "controls," "controlled by," or "under common control" means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person.

"<u>Agreement</u>" shall mean Construction Services Agreement (in the case of the Construction Manager) or Design Services Agreement (in the case of the Architect).

"Alternate" shall mean a proposed possible change in the Work as described in the Contract Documents.

"<u>Applicable Laws</u>" shall mean any and all laws (including all statutory enactments and common law), constitutions, treaties, statutes, codes, ordinances, charters, resolutions, Orders, rules, regulations, guidelines, standards, Governmental Approvals, authorizations, or other directives or requirements of any Governmental Authority enacted, adopted, promulgated, entered, implemented, ordered or issued and in force or deemed applicable by or under the authority of any Governmental Authority having jurisdiction over a person (or the property of such person), including the Act and Environmental Laws, applicable to the State, County, City, Authority, Team, Architect, Construction Manager, and/or other applicable Persons in connection with the design, development, construction, equipping, commissioning, use, occupancy, possession, operation, maintenance and management of the Project. Notwithstanding the foregoing, Applicable Laws shall expressly include all requirements, regulations and administrative orders of the City, County and State, as same may be modified by variances and waivers issued in accordance with Applicable Laws.

"<u>Application for Payment</u>" shall mean the Construction Manager's or Architect's monthly requisition for payment which shall be submitted on the form attached to or identified in the Construction Services Agreement (in the case of the Construction Manager) or Design Services Agreement (in the case of the Architect).

"<u>Architect</u>" shall mean HKS, Inc., its principals, officers, employees, agents, and the permitted successors and assigns, or such other firm of licensed architects and/or engineers as may be designated by Authority from time to time. For the purpose of the Design Services Agreement and the Contract Documents, "Architect" also includes the Architect's Subconsultants, unless otherwise specified.

"Architect Indemnitees" shall mean the Architect, its affiliates, and the foregoing Person's officers, directors, shareholders, affiliates, subsidiaries, parent companies, members, owners, agents, representatives, lenders, legal counsel, employees, heirs and assigns.

"<u>Architect's Subconsultant Agreement</u>" shall mean the written agreement between the Architect and any of the Architect's Subconsultants covering performance by the Architect's Subconsultant of a portion of the Architect's Design Services under the Design Services Agreement.

"<u>As-Built Drawings</u>" shall mean CAD generated red-lined Construction Documents showing the Work as constructed, prepared by the Construction Manager and indicating actual locations of utilities and all changes and alterations made to the Work during construction. CAD files will be provided by the Construction Manager in a format acceptable to the Authority.

"<u>Authority</u>" shall mean the Minnesota Sports Facilities Authority, a public body and political subdivision of the State of Minnesota, and its successors, assigns and designees.

"Authority-Furnished Items" shall mean materials and other items selected and purchased by the Authority, outside of the Construction Services Agreement for installation in the Project by a Trade Contractor.

"Authorized Representative" shall mean an individual, designated in writing by each party, who shall be authorized to bind that party under the terms of the Contract Documents.

"Basic Services" or "Designated Services" shall mean Construction Manager's Work (in the case of the Construction Manager) or the Architect's Design Services (in the case of the Architect).

"<u>Bid Analysis</u>" shall mean a comparison of all of the bids received for a particular scope of Work and an analysis of the costs and benefits of accepting the bid or proposal of any Subcontractor or Supplier being recommended by the Construction Manager, including an analysis of the price and schedule of Subcontractor's or Supplier's bid or proposal and its capabilities and history of complete, proper, and timely performance on other projects.

"Bid Documents" shall mean the procedural documents and Bid Packages submitted to prospective Subcontractors in connection with the competitive bidding process.

"Bid Package" shall mean the collection of pertinent portions of the Construction Documents into a package suitable for bidding by prospective Subcontractors working in a specific trade.

"Bidder's List" shall mean a list of prospective Subcontractors or if approved by the Authority, in consultation with the Architect, prepared and pre-qualified by the Construction Manage

"Building Information Modeling" or "BIM" shall have the meaning set forth in Exhibit 1 to the Design Services Agreement.

"<u>Bulletin</u>" shall mean written or graphic instruments issued by the Architect after the execution of the Construction Services Agreement which request a proposal from the Construction Manager that, if approved by the Authority, will cause the execution of a Contract Revision to modify the Contract Documents.

"Business Day" or "Work Day" shall mean Monday, Tuesday, Wednesday, Thursday and Friday, excluding any day that banks are required to close in the State of Minnesota, and excluding Saturday and Sunday.

"<u>Certificate for Payment</u>" shall mean a written certificate executed by the Architect indicating the amount that the Construction Manager is entitled to be paid in connection with each Construction Manager Application for Payment.

"Change" shall mean an agreed-upon and materially significant modification to the size, quality, or complexity of the Work reflected or understood to be in the Construction Documents and that also affects the GMP and Contract Time.

"Change Order" shall mean a written order signed by the Construction Manager, authorizing a change in a Subcontractor's scope of Work.

"City" shall mean the City of Minneapolis, Minnesota.

"<u>Claim</u>" shall have the meaning set forth in **Subparagraph 6.2.1** of these General Conditions attached as <u>Exhibit 6</u> to the Construction Services Agreement.

"<u>Conceptual Design Documents</u>" shall mean the preliminary project work plan, programming report, and pre-design document, concept sketches and renderings illustrating the scale and relationship of the Project components.

"Conceptual Design Phase" shall mean the Phase during which the Conceptual Design Documents are being prepared by the Architect.

"Confidential Information" shall mean all information (whether or not specifically labeled or identified as confidential), in any form

or medium, which is disclosed to or learned by the Architect and/or the Architect's Subconsultants, the Construction Manager and/or the Construction Manager's Subcontractors, Trade Contractors or any other member of the Project Team in the performance of the Work required in the Contract Documents, or acquired directly or indirectly such as in the course of discussion or investigations by the Architect and the Architect's Subconsultants relating to the Project, and which relates to the Authority or Team's business, finances, marketing strategy, products, services, research or development, suppliers, clients, or customers, or which relates to similar information of a third party who has entrusted such information to the Authority including, without limitation, any specialized knowhow, technical or non-technical data, formula, pattern, plan, strategy, compilation, program, device, method, technique, drawing, process, financial or business information, models, novel analysis, work papers, studies or other documents that contain, reflect, or are based on such information.

"Construction Change Directive" shall have the meaning ascribed in Paragraph 9.3 of the General Conditions.

"<u>Construction Cost</u>" shall mean the total cost or, to the extent that the Project is not completed, the estimated cost to the Authority of all elements of the Work as designed or specified in the Construction Documents.

"<u>Construction Documents</u>" shall mean those deliverables to be provided by the Architect, which are necessary to complete the Work, including, but not limited to, the Drawings, Specifications, Addenda, Bulletins and the Project Manual.

"Construction Documents Phase" shall mean that Phase during which the Construction Documents are being prepared by the Architect.

"<u>Construction Management Plan</u>" shall mean a comprehensive document prepared by the Construction Manager, for review and approval by the Authority, setting forth in detail the Construction Manager's planning, administrative and management techniques to complete the Construction Manager's Work. The Construction Management Plan shall include the certified GMP, detailed Estimates of Construction Costs, the Construction Schedule, certified completion date, the Project Manual, the Schedule of Values and other information requested by the Authority.

"Construction Manager" shall mean the legal entity that has entered into the Construction Services Agreement with the Authority for completion of the Work on the Project.

"<u>Construction Manager Contingency</u>" or "<u>Contingency</u>" shall have the meaning set forth in <u>Exhibit 3</u> to the Construction Services Agreement.

"<u>Construction Phase</u>" shall mean that Phase of the Project which shall commence after the Authority provides the Construction Manager or Trade Contractor with written Notice to Proceed with the Construction Phase.

"<u>Construction Phase Services</u>" shall mean the Basic Services described in Article 2 and Paragraph 2.2 of <u>Exhibit 1</u> of the Construction Services Agreement.

"<u>Construction Schedule</u>" shall mean the schedule prepared by the Construction Manager in CPM format and approved by the **SDC Group** and with the written approval of the Authority, and all adjustments thereto approved by the Authority, that describes the sequence and timing of the Work on the Project.

"Construction Services Agreement" shall mean the written agreement between the Authority and the Construction Manager.

"<u>Consultant</u>" shall mean a Person engaged by the Authority or Team to perform a portion of the professional services associated with the Project under direct contract with the Authority or Team.

"Consultant's Services" shall mean the complete professional services to be provided by the Consultant and as described in a Consulting Services Agreement between the Authority or the Team and the Consultant, and includes all labor, materials, equipment and services to be provided by the Consultant to fulfill its obligations under the Contract Documents.

"<u>Consulting Services Agreement</u>" shall mean a contract between the Authority or the Team and a Consultant for the performance of professional services associated with the Project.

"<u>Contamination</u>" means the presence or release or threat of release of Regulated Substances in, on, under or emanating to or from the Stadium Site, which pursuant to Environmental Laws requires notification or reporting to any Governmental Authority, or which pursuant to Environmental Laws requires the identification, investigation, cleanup, removal, remediation, containment, control, abatement, monitoring of or other Response Action to such Regulated Substances or which otherwise constitutes a violation of Environmental Laws.

"<u>Contract Documents</u>" shall mean the complete body of contractual documents that define the Work of the Construction Manager under the Construction Services Agreement, the Work of a Trade Contractor under a Trade Contractor Agreement, or the Work of a Vendor under a Vendor Agreement, including all exhibits attached to such agreements, <u>Exhibit 6</u> to the Construction Services Agreement, the Design Documents, Specifications, Addenda, Bulletins, the Project Manual, the GMP, the Construction Schedule prepared and approved in accordance with the Construction Services Agreement and other required Submittals.

"Contract Request Form" shall mean a form on which changes to the Work or Contract Time are requested or on which approvals to use certain Subcontractors are requested by the Construction Manager.

"<u>Contract Revision</u>" shall mean a written instrument prepared by the Authority and signed by the Authority and Construction Manager (in the case of the Construction Services Agreement), or signed by the Authority and Architect (in the case of the Design Services Agreement), stating their agreement upon a change in the Construction Manager's Work; the amount of the adjustment in the GMP, if any; the extent of the adjustment in the Contract Time, if any, and the updated Schedule of Values in the case of the Construction Services Agreement, or stating their agreement upon a change in the Design Services Fee or Contract Time (in the case of the Design Services Agreement).

"<u>Contract Time</u>" shall be as provided in the Construction Services Agreement, Adjustments to the Contract Time may include adjustments to the Milestone Dates, the required date of Substantial Completion, or the required date of Final Completion, as applicable and as allowed by the Contract Documents.

"Coordination Drawings" shall mean drawings prepared by Subcontractors detailing the Work and the coordination of Work items among the various Subcontractors.

"<u>Cost of the Work</u>" shall mean the amount of payment Construction Manager is eligible to receive for the proper performance of the Work as provided in **Article 8** of the Construction Services Agreement.

"<u>Cost Control Report</u>" shall have the meaning set forth in <u>Exhibit 1</u> to the Construction Services Agreement.

"CPM" shall mean a critical path method format to be used for the Construction Schedule.

"<u>Critical Path</u>" shall mean those Work activities identified on the Construction Schedule which, if delayed, will cause a corresponding Delay in the Substantial Completion Date.

"Day" shall mean a calendar day, unless otherwise specifically designated.

"Delay" shall mean any delay or interruption in the progress of the Work as anticipated on the approved Construction Schedule.

<u>"Design Delivery Schedule"</u> shall mean the schedule covering Design Services to be performed by the Architect and the Architect's Consultants as set forth in Article 4 of the Design Services Agreement, and as the same may be modified by mutual agreement of the Parties

"Design Development Documents" shall mean the Drawings, Specifications and other documents prepared by the Architect that establish and describe the size and character of the Project as to architectural, civil, structural, landscape, mechanical and electrical systems, graphics and signage, and other elements, and which include typical construction details, equipment layouts and specifications that identify major materials and systems and as more specifically described in the Design Services Agreement.

"Design Development Phase" shall mean that Phase during which the Architect prepares the Design Development Documents.

"Design Document Works" shall have the meaning set forth in Paragraph 1.3 of the Design Services Agreement.

"Design Documents" shall mean, collectively and as applicable, the Concept Design Documents, Schematic Design Documents, the Design Development Documents and the Construction Documents prepared by the Architect.

"Design Phases" shall mean those Phases which are preparatory to the physical construction of the Project during which the Schematic Design Documents, Design Development Documents and Construction Documents are being prepared by the Architect.

"Design Services" shall mean the complete architectural design and engineering for the Project, and includes all labor, materials, equipment and services to be provided by the Architect to fulfill its obligations under the Design Services Agreement. The Design Services shall include all, of the labor, materials, equipment and services to be provided by the Subconsultants of any tier employed directly or indirectly by the Architect, and shall include all necessary and appropriate coordination and integration of the Consultant's Services to allow the Construction Manager, Vendors, and Trade Contractors to complete the Work in accordance with the Contract Documents. The Design Services shall be performed in strict accordance with all provisions of the Act.

"Design Services Contract Time" shall have the meaning ascribed to such term in Paragraph 4.1 of the Design Services Agreement.

"Design Services Fee" shall mean the total cost for the Architect to complete the Design Services in accordance with the Design Services Agreement, including, all costs, expenses, profit and overhead of the Architect, and its Subconsultants, to include all related

Direct Personnel Expenses and Reimbursable Expenses.

"Design Team" shall mean Architect and its Subconsultants as set forth in the Design Services Agreement.

"Designated Services" shall mean the Architect's Design Services provided in Exhibit 1, attached to the Design Services Agreement.

"Direct Personnel Expenses" shall mean all direct and indirect costs of employment including salaries, wages, insurance, customary and/or special benefits associated with the employment of a Person by the Architect for the Design Services.

"Division" shall mean one of the divisions of a Specification as defined by the Construction Specifications Institute.

"<u>Drawings</u>" shall mean graphic or pictorial portions of the Design Documents prepare by Architect, Subconsultants, and Consultants, wherever located and whenever issued, which show, among other things, the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules and diagrams.

"Early Release Bid Packages" shall mean those bid packages prepared by the Architect in connection with the Fast-Track Construction process, which are put out for bidding prior to final completion of the remainder of the Construction Documents.

"EAW" shall mean any environmental assessment worksheet required under the Applicable Laws.

"EIS" shall mean any environmental impact statement required under the Applicable Laws.

"<u>Electronic Data</u>" shall mean Design Documents and other Project, data transmitted in electronic format and as further defined in **Subparagraph 1.3.1** of the Design Services Agreement.

"Environmental Laws" shall mean all Applicable Laws, including any consent decrees, settlement agreements, judgments, or Orders, issued by, or entered into with, a Governmental Authority pertaining or relating to: (i) pollution or pollution control; (ii) protection of human health or the environment; (iii) the presence, use, management, generation, processing, treatment, recycling, transport, storage, collection, disposal or release or threat of release of Regulated Substances; (iv) the presence of Contamination; and (v) the protection of endangered or threatened species.

"Estimated Construction Costs" shall mean the estimated cost from the Construction Manager of all elements of the Work designed or specified by Architect, Subconsultants, and Consultants. Estimated Construction Costs will not include the compensation of Architect, Subconsultants, or Consultants, the cost of the acquisition of the Stadium Site (or any development rights or zoning entitlements from adjoining properties) or the costs incurred by Authority in connection with Authority's responsibilities under Article 3 of the Design Services Agreement. "Existing Stadium" shall mean the Mall of America Field at Hubert H. Humphrey Metrodome in Minneapolis, Minnesota.

"<u>Extraordinary Measures</u>" shall have the meaning set forth in **Subparagraph 4.10.7** of these General Conditions attached as <u>Exhibit 6</u> of the Construction Services Agreement.

"FF&E" shall mean, collectively, furniture, furnishings, fixtures, accessories, equipment and similar items designed or specified by the Architect that comprise a portion of the Work.

"FF&E Budget" shall mean a budget covering only FF&E. The FF&E Budget is separate from the Construction Cost.

"Final Completion" shall mean that the Construction Manager has satisfactorily completed all of Work in strict conformity with the requirements of the Contract Documents, the Work has been finally accepted by the Authority, and the Construction Manager's submission of and the Authority's approval of the Construction Manager's Final Application for Payment.

"Final Payment" shall mean the last payment to the Construction Manager, including retainage, in connection with the Work.

**"Fixed Construction Budget"** shall mean the maximum established cost for the Work, which amount is equal to \$690,000,000 as of February 1, 2013 based on the scope of Work described in **Exhibit 13**. The Fixed Construction Budget shall establish the basis for determining the maximum amount of the GMP in the Construction Services Agreement including any maximum amounts for specific line items. The Fixed Construction Budget for the Work is an amount to be determined by the Authority and Team, in their discretion.

"Force Majeure" shall mean an act of God, fire, tornado, hurricane, named storms, flood, earthquake, explosion, war, terrorism, embargoes, civil disturbance, unusually severe weather that is abnormal and unforeseeable for the time of year in question, or industry-wide labor strikes

"General Conditions" shall mean the General Conditions of the Construction Services Agreement.

"GMP Pricing Documents" shall mean the Design Development Documents and such other Construction Documents as may be

required by the SDC Group or the Authority to establish the GMP with the Construction Manager.

"GMP Proposal" shall mean the proposed GMP for the Work prepared by the Construction Manager as part of the Construction Management Plan.

"Governmental Approvals" shall mean all waivers, franchises, variances, permits, authorizations, certificates, registrations, licenses, and Orders of and from any Governmental Authority having jurisdiction over the Project, Work, Stadium Site, Adjacent Property, Authority, Team, Consultant, Architect, Subconsultant, Construction Manager, Subcontractor, Trade Contractor, Vendor City, County, State, and other applicable Persons in connection with the design, development, construction, equipping, commissioning, use, occupancy, possession, operation, maintenance and management of the Project or any Adjacent Property.

"Governmental Authoritv(ies)" shall mean any federal, state, county, municipal or other governmental department, entity, authority, commission, board, bureau, court, agency, or any instrumentality of any of them having jurisdiction with respect to the Project, Work, Stadium Site, Adjacent Property, Authority, Team, Consultant, Architect, Subconsultant, Construction Manager, Subcontractor, Trade Contractor, Vendor, City, County, State, and other applicable Persons in connection with the design, development, construction, equipping, commissioning, use, occupancy, possession, operation, maintenance and management of the Project or any Adjacent Property.

"Governmental Body" or "Governmental Bodies" shall have the meaning set forth in Paragraph 12.10 of the Design Services Agreement.

"Green Globes" shall mean the requirements, standards and rating system developed by the Green Building Initiative for environmental design.

"Guaranteed Maximum Price" or "GMP" shall mean, as applicable, the absolute not-to-exceed limit for the amount of money to be paid to the Construction Manager for its Work under the Construction Services Agreement, or to a Trade Contractor or Vendor under any Trade Contract or Vendor Contract.

"Hazardous Materials" shall mean: (1) any "hazardous waste" as defined by the Resource, Conservation and Recovery Act of 1976 (42 U.S.C. Section 6901, et seq.) as amended, and regulations promulgated thereunder; (2) any "hazardous, toxic of dangerous waste, substance or material" specifically defined as such in (or for the purposes of) the Comprehensive Environmental Response, Compensation and Liability Act of 1980 (42 U.S.C. Section 9601, et seq.), as amended, and regulations promulgated thereunder; (3) any "hazardous waste" or "hazardous substance" as defined by applicable Minnesota State laws and regulations), as amended, and regulations promulgated thereunder; and (4) any hazardous, toxic or dangerous waste, substance, or material as defined in any so— called "superfund" or "superlien" law or any other federal, state or local statute, law ordinance, code, rule, regulation, order or decree regulating, relating to or imposing liability or standards of conduct concerning such waste, substance or material.

"Including" shall encompass the phase "including without limitation" and be understood to indicate an example and not a limitation.

"Indemnitees" shall mean the Minnesota Sports Facilities Authority, Minnesota Vikings Football, LLC, Hammes Company Sports Development, Inc., ICON Venue Group, LLC., Architect, Lender(s), their Affiliated Parties and each of the foregoing Persons' elected officials, appointed officials, board members, directors, officers, shareholders, affiliates, subsidiaries, parent companies, members, owners, agents, representatives, legal counsel, and employees, together with the constituent partners, members, owners, shareholders, and heirs and estates of each of the foregoing, and the City and State.

"Key Personnel" shall mean those employees of the Architect and Construction Manager, who will have primary responsibility for implementing, the Architect's obligations under the Design Services Agreement and the Construction Manager's Work under the Construction Services Agreement, respectively.

"LEED" or "Leadership in Energy and Environmental Design (LEED)" shall mean the requirements, standards and rating system developed by the U.S. Green Building Council for environmental design.

"Legal Requirements" shall mean the requirements set forth in any Applicable Laws.

"<u>Lender</u>" shall mean any bank, insurance company, trust, corporation, association, firm, partnership, Person, or other entity that has loaned or agreed to lend or otherwise provide funds or credit enhancement to enable the development and construction of the Project.

"Master Project Budget" means the budget to be prepared by the Authority which incorporates all of the costs associated with the Project in accordance with the Act.

<u>"Master Project Schedule"</u> means the schedule to be prepared by the Authority which incorporates the Construction Schedule developed by the Construction Manager and the Design Delivery Schedule developed by the Architect and all other primary activities associated with the Project

"MEP" shall mean mechanical, electrical and/or plumbing fixtures or systems.

"MGDPA" shall have the meaning set forth in Paragraph 12.9 of the Design Services Agreement.

"Milestone Date" shall mean a date in the Construction Schedule identified for its importance in measuring progress of all or part of the Work.

"Minimum Design Standards" shall be the standards and criteria agreed to on Exhibit 9 to the Design Services Agreement.

"Modification" shall mean a writing changing the terms of a contract signed by the parties to that contract.

"<u>National Football League</u>" or "NFL" shall mean, collectively, the Office of the National Football League Commissioner, the National Football League Commissioner, the member clubs of the National Football League, the NFL owners, and/or any other Person appointed by any of the foregoing, or any successor substitute association or entity of which the Team is a member or joint owner and which engages in professional football in a manner comparable to the National Football League.

"NFL Rules and Regulations" shall mean, collectively, the constitution and bylaws, and the rules, guidelines, regulations and requirements of the NFL, and any other rules, guidelines, directive, advisory opinions, regulations or requirements of the Office of the Commissioner of the NFL, and/or any other person appointed by the foregoing, that are generally applicable to NFL franchises, all as the same now exist or may be amended or adopted in the future.

"Notice to Proceed" shall have the meaning set forth in Paragraph 2.3 of the Construction Services Agreement.

"<u>Order</u>" mean any judgment, award, decision, directive, consent decree, injunction (whether temporary, preliminary or permanent), ruling, or writ ordered adopted, enacted, implemented, promulgated, issued, entered or deemed applicable by or under the authority of any Governmental Authority or arbitrator that is binding on any person or its property under Applicable Laws.

"Party" or "Parties" shall mean the Authority and/or the Construction Manager as applicable.

"<u>Pending Change Report</u>" shall mean a summary of the status of all pending requests for Contract Revisions and Change Orders that the Authority may generate from time to time.

"<u>Person</u>" shall mean: (1) an individual, sole proprietorship, corporation, limited liability company, partnership, joint venture, joint stock company, estate, trust, limited liability association, unincorporated association or other entity or organization; (2) any Federal, State, county or municipal government (or any bureau, department, agency or instrumentality thereof); and (3) any fiduciary acting in such capacity on behalf of any of the foregoing.

"<u>Phase</u>" as used herein, shall mean any one of the various phases, or distinct, contiguous portions of time, of the Architect's Design Services or Construction Manager's Work, as applicable.

"<u>Pre-Construction Phase Services</u>" shall mean those services described in Article 2 of the Construction Services Agreement and in <u>Exhibit 1</u>, Paragraphs of the Construction Services Agreement <u>"Program"</u> shall mean a narrative description of the requirements desired by the Authority and Team and prepared by the Architect to be incorporated into the design of the Project, which shall form the basis for preparation by the Architect of the Design Documents for the Project.

"Program" shall mean a narrative description of the requirements desired by the Authority and Team and prepared by the Architect to be incorporated into the design of the Project, which shall form the basis for the Design Documents.

"<u>Project</u>" shall mean the development, design, construction and equipping of the Stadium and the Stadium Infrastructure and any related activities in accordance with the Act.

"Project Costs" shall mean all costs authorized under the Preliminary Development Agreement or Development Agreement by and between the Authority and the Team to be incurred in the furtherance of the Project, including the costs of acquiring the Stadium Site and certain temporary relocation costs of the Team.

"<u>Project Labor Agreement</u>" shall mean an agreement by which labor organizations agree to terms and conditions of employment, to cooperate in resolving labor disputes, and to maintain labor peace on the Project.

"Project Manual" shall mean a written volume assembled for the Work that includes specific instructions to the Project Team setting forth the requirements, policies and procedures for performance and execution of the Design Services and Work and that contains, among other things, the General and Supplementary Conditions and Specifications of the Construction Services Agreement.

"Project Participant" shall mean a member of the Project Team.

"Project Safety Program" shall be developed by Construction Manager pursuant to the Construction Services Agreement.

"Project Site" shall mean the site where the Project is to be constructed.

"Project Team" shall mean any Person involved in the Project and under a contract with the Authority or Team, including the Construction Manager and its Subcontractors, Sub-Subcontractors, Suppliers, Vendors, the Architect and its Subconsultants of any tier, Trade Contractors of all tiers contracted directly by the Authority and Consultants contracted directly by the Authority or Team. The Authority must be presented, and asked to provide written approval of, all agreements or contracts, consistent with the terms and conditions of the Contract Documents, with any member of the Project Team before any services or Work is to be provided or performed on the Project by any such member.

"Project Website" shall mean the website managed and maintained by the Architect.

"Punchlist" shall mean a list, compiled by the Construction Manager and reviewed by the Architect at Substantial Completion, and approved by the Authority, which identifies items of Work that remain to be completed or corrected prior to Final Payment to the Construction Manager.

"Purchase Order" shall have the meaning set forth in Paragraph 10.5 of the Construction Services Agreement.

"<u>Record Drawings</u>" shall mean a reproducible set of Construction Documents into which the Architect has incorporated: (1) clarifications, sketches and other modifications made by the Architect during the Construction Phase; and (2) significant changes in the Work made during construction as shown on the Construction Manager's As-Built Drawings.

"Regulated Substance" shall mean any substance, material or waste, regardless of its form or nature, defined under Environmental Laws as a "hazardous substance," "hazardous waste," "toxic substance," "extremely hazardous substance," "toxic chemical," "toxic waste," "solid waste," "industrial waste," "residual waste," "municipal waste," "special handling waste," "mixed waste," "infectious waste," "chemotherapeutic waste," "medical waste," "regulated substance," "pollutant" or "contaminant" or any other substance, material or waste, regardless of its form or nature, which otherwise is regulated by Environmental Laws.

"Reimbursable Expenses" shall mean actual, out—of—pocket expenses incurred by the Architect and the Architect's Subconsultants for the benefit of the Project.

<u>"Reimbursable Expense Budget"</u> shall mean the Architect's budget for Reimbursable Expenses as set forth in the Design Services Agreement.

"<u>Required Insurance</u>" shall mean the insurance required of the Construction Manager pursuant to the Construction Services Agreement, including Article 11 and Attachment B to this <u>Exhibit 6</u>.

"Response Action" shall mean the investigation, cleanup, removal, remediation, containment, control, abatement, monitoring of or any other response action to the presence of Regulated Substances or Contamination in, on, at, under or emanating from the Stadium Site, including the correction or abatement of any violation required pursuant to Environmental Laws or by a Governmental Authority.

"Retractable Feature" shall mean an alternative design option to a fixed-roof stadium that would provide for an operable roof or other open air options for the Stadium.

"<u>RFI</u>" shall mean a request for information or clarification directed by the Construction Manager and/or a Subcontractor to the Architect.

"Samples" shall mean physical examples that illustrate materials, items of workmanship, and which establish standards by which the Work will be judged.

"Schedule of Values" shall mean an allocation of the entire GMP among the various portions of the Work as required by the Authority and to be provided by the Construction Manager.

"Schematic Design Documents" means drawings prepared by the Architect that illustrate the scale and relationship of the various Project components and which also contain square footage and volume calculations for the building interior spaces, building exterior spaces, and major architectural and interior finishes.

"Schematic Design Phase" shall mean that Phase during which the Schematic Design Documents are being prepared by the Architect.

<u>"Self-Performed Work"</u> shall mean such Work (other than performance of General Conditions) in which a substantial portion thereof (i.e., more than 20% thereof) is performed directly by Construction Manager's own labor forces or the labor forces of any Affiliate of Construction Manager (including any joint venture partners of Construction Manager).

"Shop Drawings" shall mean drawings, diagrams, illustrations, schedules, performance checks and other data prepared by the

Construction Manager or any Subcontractor to illustrate how a specified portion of the Work will be constructed.

"Specifications" shall mean the written specifications prepared by Architect, Subconsultants and Consultants consisting of the written requirements for materials, equipment, technical requirements and construction systems, standards and workmanship for the Work, and performance of related services.

"Stadium" shall mean a multipurpose stadium suitable for NFL football and a broad range of other civic, community, athletic, educational, cultural, and commercial activities.

"<u>Stadium Developer</u>" shall mean, as between the Authority and the Team, which of the two parties shall be responsible for the management of the construction of the Stadium and Stadium Infrastructure.

"Stadium Infrastructure" shall mean plazas, including the Stadium Plaza, parking structures, rights of way, connectors, skyways and tunnels, and other such property, facilities, and improvements, owned by the Authority or determined by the Authority to be reasonably necessary to facilitate the use and development of the Stadium.

"Stadium Plaza" shall mean the open air portion of the Stadium Infrastructure adjacent to the Stadium.

"Stadium Site" shall mean all or portions of the current site of the Existing Stadium and adjacent areas, bounded generally by Park and Eleventh Avenues and Third and Sixth Streets in the City of Minneapolis, together with any vacated streets or other lands acquired for the development, construction and use of the Project.

"Standard Details" shall mean design details developed by the Architect and the Architect's Subconsultants over time, and used routinely in their respective practices.

"Standard of Care" in regard to the Architect shall mean that standard of professional care, skill, diligence and quality that prevail among national design firms engaged in the planning, design, and construction administration of large scale and complex projects of similar scope, function, size, quality, complexity and detail, including the design of similar NFL stadiums in comparable urban areas throughout the United States, and further including performance in accordance with Applicable Laws and the NFL Rules and Regulations.

"Standard of Care" in regard to the Construction Manager shall mean that that standard of professional care, skill, diligence and quality that prevail among national construction management firms engaged in the planning, estimating, scheduling, construction, and construction administration of large scale and complex projects of similar scope, function, size, quality, complexity and detail, including the design of similar NFL stadiums in comparable urban areas throughout the United States, and further including performance in accordance with Applicable Laws and the NFL Rules and Regulations

"State-of-the-Art" shall have the meaning set forth in Subparagraph 2.1.13 of the Design Services Agreement.

"Subconsultant" shall mean a Person engaged or to be engaged by the Architect to perform a portion of the Design Services. The term "Subconsultant" also includes any lower tier Sub-Subconsultant engaged by a Subconsultant.

"Subcontract" shall mean a contract between the Construction Manager and a Subcontractor for the performance of a portion of the Work.

"Subcontract Amount" shall mean the total amount stipulated in a Subcontract to be paid to the Subcontractor for the Subcontractor's performance of the Subcontract.

<u>"Subcontractor"</u> shall mean a Person engaged by the Construction Manager to perform a portion of the Work. The term "Subcontractor" also includes any lower tier Sub-Subcontractor engaged by a Subcontractor.

"Subcontractor Application for Payment" shall mean an application for payment submitted to the Construction Manager by a Subcontractor.

"Submittal" shall mean a Shop Drawing, Sample, catalog cut or similar item for specific portions of the Work as required by the Construction Documents.

"Submittal Log" shall mean a log maintained by the Architect indicating the dates of submission of Shop Drawings and other Submittals by the Construction Manager, and their return after review by the Architect.

"Submittal Schedule" shall have the meaning set forth in Subparagraph 4.12.5 of the General Conditions.

"Substantial Completion" shall have the meaning set forth in Subparagraph 11.8.1 of the General Conditions.

<u>"Substantial Completion Certificate</u>" shall mean the certificate issued by the Architect and approved by the Authority indicating the date upon which the Project (or a designated portion thereof) is Substantially Complete.

<u>"Substantial Completion Date" or "Date of Substantial Completion</u>" shall mean the date identified in the Architect's Substantial Completion Certificate when the Work (or a designated portion thereof) is Substantially Complete.

"Substitution" shall mean a replacement for or alternative to an item of material or item identified in the Construction Documents which is proposed by the Construction Manager and approved in writing by Authority and Architect.

"<u>Sub-Subcontractor</u>" shall mean a Person engaged by a Subcontractor to perform a portion of the Work. The term "Sub-Subcontractor" also includes any lower tier Persons engaged by Sub-Subcontractor to perform a portion of the Work.

"Supplier" or "Materialman" shall mean a Person who has an agreement with the Architect, Construction Manager, or any member of the Project Team or any of their Subconsultants, Subcontractors or Sub-subcontractors, of any tier, to supply by sale or lease, directly or indirectly, any materials or equipment for the Work.

"Suspension" shall mean a delay, re-sequencing, stoppage and/or interruption of the Work or the Architect's Services (in whole or in part), in response to a written directive from the Authority.

"Targeted Businesses" shall have the meaning set forth in Paragraph 10.8 of the Construction Services Agreement.

"Targeted Business Coordinator" shall have the meaning set forth in Paragraph 10.8 of the Construction Services Agreement.

"Targeted Business Enterprise Participation Plan" shall have the meaning set forth in Paragraph 10.8 of the Construction Services Agreement.

"Team" shall mean, the Minnesota Vikings Football, LLC, or any person who purchases or otherwise takes ownership or control of or reconstitutes the NFL team known as the Minnesota Vikings.

<u>"Termination for Convenience"</u> shall mean the termination of the Construction Services Agreement, the Design Services Agreement, a Trade Contract, or a Vendor Contract by the Authority without cause and for the convenience of the Authority as decided in its sole discretion.

"Termination for Default" shall mean the termination of a Party under the Contract Documents by the Authority for cause.

"Testing" shall mean, performing those tests and inspections of the Work to determine conformance with the Contract Documents.

"<u>Third-Party Suit</u>" shall mean a summons and complaint served by a third party who is not a member of the Project Team against a member of the Project Team, the Authority, or Team in a court of law.

"Trade Contract" shall mean a contract between the Authority and a Trade Contractor for the performance of a portion of the Project.

"Trade Contractor" shall mean a Person other than the Construction Manager or its Subcontractors that has a direct contract with the Authority to perform a portion of the Project. The term "Trade Contractor" also includes any lower tier contractor engaged by a Trade Contractor.

"Trade Contractor Work" or "Vendor Work" shall mean the complete and total construction or performance described in and required by the Contract Documents, including preconstruction services and construction services, whether completed or partially completed, and include all labor, materials, equipment and services required to be provided by a Trade Contactor or Vendor to fulfill Trade Contractor's or Vendor's obligations under its applicable Trade Contract or Vendor Contract. The Trade Contractor's or Vendor's Work may constitute the whole or part of the Project. Any Trade Contractor's or Vendor's Work includes all of the labor, materials, equipment and services, Sub-Subcontractors of any tier, material and equipment suppliers employed directly by the Trade Contractor or Vendor. The Trade Contractor's or Vendor's Work shall be performed in strict accordance with all provisions of the Act.

"Trademark" shall mean a trademark used by the Authority that is protected under United States or International Trademark Law. Without limiting the foregoing, the term Trademark shall also have the meaning set forth in <u>Paragraph 1.3</u> of the Design Services Agreement.

"Trade Secret" shall mean any and all information that comes into the possession, custody or control by, through, from, or on behalf of the Authority or Team without regard to form, including, without limitation, any technical or non-technical data, formula, pattern, compilation, program, device, method, technique, drawing, process, financial data, financial plan, product plan, list of actual or potential customers or suppliers, that is not commonly known by or available to the public and which information: (1) derives economic value, actual or potential, from not being generally known to, and not being readily ascertainable by proper means by, other persons who can obtain economic value from its disclosure or use; and (2) is the subject of efforts that are reasonable under the circumstances to maintain its secrecy.

"Unique Elements" shall have the meaning set forth in Paragraph 1.3 of the Design Services Agreement.

"Value Engineering" shall mean an analysis of the feasibility of alternative systems, construction, concepts, equipment and materials to identify such alternative systems, construction, concepts, equipment and materials of equivalent quality, and having equivalent characteristics and utility, to those specified in the Design Documents that can be fully specified, obtained and installed at a lower price or, in the sole judgment of the Authority and Team, more-desirable operating characteristics or greater functionality or any combination of these. Value Engineering may also include the process whereby the Design Documents are reviewed by the Project Team for the purpose of determining constructability and bringing estimated Construction Cost within the Fixed Construction Budget including reviewing Construction Manager proposals that are designed to reduce Construction Cost and enhance the Project. As part of this process, the Design Documents are revised by the Architect as necessary to maintain the Fixed Construction Budget.

"Vendor" shall mean a party supplying FF&E, equipment, material, or other product or service to the Project under a direct contract with the Authority.

"Vendor Contract" shall mean a contract between the Authority and a Vendor for the performance of a portion of the Project.

"<u>Work</u>" in regard to the Construction Manager and the Construction Services Agreement shall mean the complete and total construction as described in and required by the Contract Documents, including preconstruction services and construction services, whether completed or partially completed, and includes all other labor, materials, equipment and services to be provided by the Construction Manager to fulfill its obligations under the Contract Documents. The Work of the Construction Manager may constitute the whole or a part of the Project. The Work of the Construction Manager includes all of the labor, materials, equipment and services to be provided by the Subcontractors, Sub-Subcontractors of any tier, material and equipment suppliers employed directly by the Construction Manager. The Work of the Construction Manager shall be performed in strict accordance with all provisions of the Act.

### GENERAL CONDITIONS OF THE CONSTRUCTION SERVICES AGREEMENT

### APPENDIX B

### CONSTRUCTION MANAGER'S INSURANCE REQUIREMENTS

For the Construction Services Phase, the Construction Manager shall, unless otherwise approved in writing by the Authority, obtain and maintain throughout the duration of the Project (or as otherwise specified) insurance of the minimum types and in the minimum amounts described below ("Required Insurance"). And, for the Pre-Construction Services Phase, the Construction Manager shall, unless otherwise approved in writing by the Authority, obtain and maintain the Required Insurance of the minimum types and in the minimum amounts described in Paragraphs 1.1, 1.2, 1.3, 1.4 and 1.6 hereof.

### 1.1 <u>Commercial General Liability Insurance</u>.

#### 1.1.1 Combined single limit of not less than:

\$2,000,000	Each Occurrence
\$4,000,000	General Aggregate
\$2,000,000	Products/Completed Operations Aggregate
\$2,000,000	Personal and Advertising Injury
\$100,000	Fire Legal Liability
\$10,000	Medical Payments

### 1.1.2 Coverage Required:

- Per project general aggregate,
- Premises-operations,
- Explosion, collapse, underground,
- Products/completed operations (to be maintained for 12 years after Substantial Completion),
- Independent contractor,
- Independent Construction Manager's Liability,
- Blanket contractual liability, subject to industry standard terms and conditions,
- Personal injury liability,
- Employment Practice Liability (which can be provided through a separate policy)
- Electronic data liability property damage,
- General liability primary (non-contributory),
  - Additional insured by endorsement (CG 20 10 10/04 and CG 20 37 10 04 or their equivalent) in a form subject to approval by the Authority including waiver of subrogation in favor of the Authority and the Indemnitees (but with regard to the Architect Indemnitees, only as arising out of the acts, errors, or omissions of the Construction Manager) and persons entitled to indemnification from the Construction Manager pursuant to the terms of the Construction Services Agreement, including additional insured coverage for both ongoing and completed operations,
- Supplementary Payments coverage for the Indemnitees of the Construction Manager (but with regard to the Architect Indemnitees, only as arising out of the acts, errors, or omissions of the Construction Manager) for property damage, bodily injury, and personal and advertising injury,
- Construction Manager shall secure such endorsements or estoppel certificates from insurer as may be necessary to confirm that the Construction Services Agreement is an insured contract for purposes of the policy,
- Designated Construction Project(s) General Aggregate Limit endorsement (CG 25 03 03 97 or its equivalent),
- Operations within 50 feet of railroad,
- Defense in addition to limits of liability,
- Broad form Property Damage coverage,
- Definition of Bodily Injury to include Mental Injury and Mental Anguish,
- Mobile Equipment coverage,
  - Liability assumed under an insured contract including the tort liability of another assumed in a business contract
- Libel, slander, false arrest and invasion of privacy,
- Designated Location(s), and

Coverage shall be written on ISO form CG 00 01 12 04 or an equivalent form. Insurance is to be on an occurrence form insuring bodily injury and property damage against the hazards of: construction operations, subcontractor and independent contractor, and shall include an aggregate limit per job site endorsement. Products and completed operation insurance shall remain in force twelve (12) years following completion of project.

### 1.2 Business Auto Liability Insurance.

### 1.2.1 <u>Combined Single Limit of</u>:

\$2,000,000 each Occurrence Bodily Injury and Property Damage

**1.2.2** <u>Coverage Required</u>: Primary for all owned automobiles; including liability for all owned, leased, rented, hired/non-owned automobiles; medical payments, uninsured and underinsured motorists coverage. Coverage shall be written on ISO form CA 00 01, CA 00 05, CA 00 12, CA 00 20, or an equivalent form.

The Authority shall be included as additional insured by endorsement in a form subject to approval by the Authority. An MCS-90 endorsement (or its equivalent) shall be attached when operations require this coverage.

#### 1.3 Workers' Compensation Insurance.

1.3.1 Workers' Compensation Limits: Statutory

#### 1.3.2 Employers' Liability Limits:

Bodily Injury by Accident	\$1,000,000	Each Accident
Bodily Injury by Disease	\$1,000,000	Policy Limit
Bodily Injury by Disease	\$1,000,000	Each Employee

- **1.3.3** Workers Compensation Insurance shall include coverage for Borrowed Servant and Alternate Employer, Voluntary Compensation, unintentional errors or omissions, FELA (in case RR worker/employee injured due to operations of contractor or their agents), The Alternative Employer endorsement shall be attached showing the Authority in the schedule as the Alternate Employer
- **1.3.4** Waiver of Subrogation Endorsement in favor of the Authority

### 1.4 Excess Liability Insurance with Limits of Not Less Than:

\$100,000,000	Each Occurrence
\$100,000,000	Aggregate
\$100,000,000	Products/Completed Operations Aggregate
\$10,000	Retention

Coverage is materially following form of underlying general liability, automobile liability and employers' liability, coverage including per project aggregate, primary, additional insured/non-contributory, and waiver of subrogation. Coverage shall have "drop down" feature and "pay on behalf of" wording. Excess policy shall be primary and any applicable policy by the Authority, Additional Insureds, Indemnitees, or Subcontractors, shall be secondary and non-contributory.

For Pre-Construction Services, the Construction Manager will carry a minimum Excess Liability Insurance of \$25,000,000 each Occurrence and Aggregate and Products/Completed Operations Aggregate.

### Additional Coverage Required when C.M. is engaged to start Construction:

**1.5** <u>**Builder's Risk.**</u> All Risk Form. Builder's Risk coverage at limits agreed upon by the Parties may be provided by the Construction Manager or by the Authority at Authority's discretion.

\$Full Project Value	Limit of Insurance
\$TBD	Earth Movement
\$TBD	Flood
\$TBD	Temporary Storage
\$TBD	Transit
\$TBD	Deductible – Any one Occurrence
\$TBD	Earth Movement Deductible
\$TBD	Flood Deductible
\$TBD	Windstorm Deductible
\$TBD	Materials stored off the Project Site or being shipped to the
nt. 1	1 •

Project Site, on a replacement cost basis.

The Construction Manager shall be provided a copy of the Builder's Risk policy before the Construction Phase of the Work begins and the Construction Manager will be an additional named insured under said policy as its interest may appear at the time of loss.

### 1.6 Contractor's Professional Liability.

\$25,000,000	Limit of Insurance
\$25,000,000	Annual Aggregate
\$2500,000	Retention

Coverage shall be maintained uninterrupted for a minimum of twelve years after Substantial Completion. Coverage shall apply to negligent acts, errors or omissions arising from the Construction Manager's professional services defined to include architecture, engineering, land surveying, landscape architecture, and construction management (or other services if appropriate). The retro date will be prior to the commencement of work. Such policy shall include a protective professional indemnity part.

### 1.7 <u>Contractor's Pollution Liability</u>.

### 1.7.1 <u>Combined Single Limit of Not Less Than:</u>

\$25,000,000 each Claim/\$25,000,000 Aggregate

- **1.7.2** The Authority and all Indemnitees specified in the Construction Services Agreement shall be named as an additional insured by endorsement in a form subject to approval by the Authority.
- **1.7.3** Coverage shall apply to bodily injury or property damage arising from a pollution incident or event caused by the Construction Manager's activities. No exclusion for mold, fungus, or microbial matters of any kind.
- **1.7.4** Coverage must be maintained for twelve years following completion of the Project.
- 1.7.5 Coverage shall include sudden and accidental pollution and/or discharge of contaminants.

### 1.8 Property Insurance/Contractor's Equipment

Construction Manager shall insure its own property and equipment (owned, rented or borrowed) including but not limited to tools, materials, supplies, equipment, forms, scaffolding towers, staging, bunkhouses, and other temporary structures including their contents except for such contents as are to be included in and remain a part of the permanent construction. Subcontractors will similarly be required to insure its property and equipment per the Construction Manager's standard Subcontract Agreement.

- 2. The Authority does not represent or warrant that coverage and limits of the Required Insurance will be adequate to protect the Construction Manager, and such coverage and limits shall not be deemed as a limitation on the Construction Manager's liability under the indemnities provided to The Authority and Indemnitees in the Construction Services Agreement, or any other provision of the Contract Documents.
- 3. Prior to commencing the work, the Construction Manager shall furnish the Authority with a certificate(s) of insurance, executed by a duly authorized representative of each insurer, showing that the Construction Manager maintains all Required Insurance.
  - **3.1** All certificates shall contain a provision that coverages afforded under the policies will not be cancelled, non-renewed, materially changed, or allowed to expire until at least thirty (30) days' prior written notice has been given to the Authority. The Construction Manager shall immediately furnish the Authority copies of all endorsements that are subsequently issued amending coverage or limits.
  - **3.2** The words "endeavor to" and "but failure to mail such notice shall impose no obligation or liability of any kind upon the company, its agents or representatives" shall be deleted from the cancellation or material change of any insurance referred to therein.
  - **3.3** Failure of the Authority to demand such certificate or other evidence of full compliance with the Required Insurance or failure of the Authority to identify a deficiency from evidence that is provided shall not be construed as a waiver of the Construction Manager's obligation to maintain such insurance.
- 4. The Authority shall have the right, but not the obligation, to prohibit the Construction Manager from entering the Project site until such certificates or other evidence that the Required Insurance has been duly placed in compliance with this <u>Appendix B</u>. The Construction Manager shall be responsible for any delays arising out of the Construction Manager's failure to comply with this <u>Appendix B</u>.
- 5. If the Construction Manager fails to maintain any or all of the Required Insurance, then, in addition to any other rights or remedies available to the Authority under the Agreement, the Authority shall have the right, but shall not be obligated, to (a) terminate the Agreement, or (b) purchase such insurance for the Construction Manager in which event the Construction Manager shall reimburse the Authority or provide for the cost thereof, plus ten percent (10%) as an administrative charge.

- 6. With respect to the Required Insurance, which the Construction Manager is obligated to maintain after Substantial Completion, an additional certificate(s) evidencing such coverage shall be promptly provided to the Authority when requested.
- 7. Construction Manager shall provide certified copies of all insurance policies required above prior to commencement of the Work and thereafter within ten (10) days of the Authority's written request for said copies.
- 8. Construction Manager will require Subcontractors to provide insurance in accordance with the terms and conditions of its standard Subcontract Agreement. Construction Manager shall include the above additional insured and waiver of subrogation requirements in all of its Subcontracts, unless otherwise agreed to in writing by the Authority. The Construction Manager shall be responsible for collecting certificates of insurance and monitoring insurance coverage of its Subcontractors and all lower tier Sub-Subcontractors to verify that the required coverage is maintained as required. All Subcontractors providing professional services shall be required to provide professional liability insurance. In addition, all Subcontractors shall include the same indemnification of the Authority and Indemnitees as provided in the Construction Services Agreement.
- 9. Construction Manager shall secure all Required Insurance from domestic insurer(s) authorized to do business in the State of Minnesota and reasonably satisfactory to the Authority with: (a) a claims paying ability of not less than "AA" (or the equivalent) by Standard & Poors and one other Rating Agency satisfactory to the Authority; and (b) "A:X" or better financial strength rating by AM Best. Construction Manager shall promptly notify the Authority of any material changes in coverage for all insurance required herein. All coverages except Professional Liability insurance shall be occurrence based. Insurance shall contain such provisions as the Authority deems reasonably necessary or desirable to protect its interest, including endorsement stating that neither the Construction Manager, the Authority, nor any other party shall be deemed a co-insurer. The Construction Manager shall pay the premiums for all insurance when due and payable.

### 10. Owner Controlled Insurance Program (OCIP)

**10.1 Reservation.** The Authority may elect to implement an Owner Controlled Insurance Program on this Project. Under this type of program, the Authority will centralize the purchase of insurance for activities of the Construction Manager, Trade Contractors, Subcontractors, and all Sub-Subcontractors for Work performed at the Project Site. This consolidated purchasing of insurance is known as an Owner Controlled Insurance Program (OCIP) and will include Workers' Compensation, General Liability, and Excess/Umbrella Liability coverage. Builder's Risk and Project specific Professional Liability insurance coverage will be provided under a separate program. Participation in the OCIP will be mandatory for all tiers of Contractors and Subcontractors unless specifically exclude in writing by the Authority. The Authority also reserves the right to not implement an OCIP at any time.

**10.2 Insurance Credit.** The Authority and its OCIP Administrator will pursue and be entitled to receive from the Construction Manager an insurance deduction for the Construction Manager's, the Subcontractors' and Sub-Subcontractors' General Liability, Excess Liability and Workers' Compensation insurances which are provided under the OCIP, which shall be credited to the Authority against the GMP. This credit shall be known as the "Insurance Credit." The following Subparagraphs are a proposed outline of a potential OCIP program and are subject to modification as insurance is placed and during negotiation of the Construction Management Plan.

**10.2.1** The Construction Manager shall be responsible to provide and to require each Subcontractor and Sub-Subcontractor to separately provide an identification and designation in their bids or proposals of all of their General Liability, Excess Liability and Worker's Compensation insurance costs. A 15% factor for profit and overhead on all insurance credits will be applied as part each Subcontractor's and Sub-Subcontractor's overall insurance deduction credit. Deductible or self-retention credits or self-insured amounts may be identified in the designation required by this **Subparagraph 10.2.1** but will not be allowed for inclusion in the insurance deduction credit rates.

**10.2.2** The Construction Manager, and all Subcontractor and Sub-Subcontractor bidders shall submit their bids or proposals with their insurance costs included for this Project, including a designation of the costs to provide Workers' Compensation, General Liability, and Excess Liability for this Project.

**10.2.3** The GMP will be reduced by the Insurance Credit. The Insurance Credit initially shall include and be based on all insurance costs designated by the Construction Manager, Subcontractor and Sub-Subcontractors pursuant to **Subparagraph 10.2.1**. Before retainage is released to the Construction Manager, a Subcontractor, or a Sub-Subcontractor, and at such other times that the Authority may require, the Authority, with the assistance of the OCIP Administrator, shall conduct an audit of all payrolls, and other relevant records, which the Construction Manager, Subcontractor, and Sub-Subcontractor, and Sub-Subcontractor agree to produce upon demand made during normal business hours, to determine whether any additional insurance deductions are due and owing to the Authority as part of the Insurance Credit so that all of the Construction Manager's, Subcontractor's, and Sub-Subcontractor's insurance costs are captured in the Insurance Credit. The final Insurance Credit will be based on actual payrolls including any and all Contract Revisions for Changes in the Work. The Authority shall have the right to apply retainage against the Insurance Credit, and to the extent that determination of the final Insurance Credit results in monies being owed to the Authority, such shall be paid on demand.

**10.2.4** For the Construction Manager, and each Subcontractor, and Sub-Subcontractor, the Authority with the assistance of the OCIP Administrator shall determine an insurance credit rate that will apply to facilitate the calculation of the Insurance Credit due from the Construction Manager based on the Construction Manager's, and each Subcontractor's and Sub-Subcontractor's payroll. The Construction Manager, and each Subcontractor

and Sub-Subcontractor shall be required to provide their bid estimates to the Authority and OCIP Administrator so that the estimated man-hours used can be verified for purposes of determining the insurance credit rate. In addition,

.1 Calculations will be based upon the rates that were effective at the time of award, and shall include the Insurance Credit information for all tiers of Subcontractors, and in the absence of a known Subcontractor or Sub-Subcontractor for a particular scope of work, 3.0% of the bid or proposal value will be used to estimate the initial Insurance Credit. Deductible or self-retention credits or self-insured amounts may be identified but will not be allowed for inclusion in the insurance deduction rates. Policies using composite rates must show, at a minimum, the deductible/self-retention/self-insured amount and the rating breakdown. If the credit percentage is not available, minimum and maximum rates for the program must be identified. Corporate allocations will not be allowed. Any rate credits (other than credits for deductibles or self-retention or self-insured plans) or surcharges shown on the declaration and/or rating sheets will be used to verify actual cost and determine the insurance deduction necessary to compute the Insurance Credit.

.2 Once the insurance credit rate is determined, the Authority with the assistance of the OCIP Administrator will use that rate as the basis to determine the Insurance Credit for the duration of the contract, and such rate will not be modified.

.3 The Authority with the assistance of the OCIP Administrator will develop a loss cost allocation/rate based upon the Construction Manager's, Subcontractor's or Sub-Subcontractor's experience outside of the OCIP. In the event that the loss experience develops a loss rate that is less than 50% of the self-insured retention (deduction or SIR) it will be assumed that the full deductible has been included by line of coverage and this will be included in the insurance credit rate.

**10.2.4** The Construction Manager, Subcontractor, and Sub-Subcontractor shall comply with all requests for information made by the Authority and OCIP Administrator for the purpose of conducting the audit pursuant to **Subparagraph 10.2.2** or determining the credit rate pursuant to **Subparagraph 10.2.3**.

10.2.5 Each Construction Manager, Subcontractor and Sub-Subcontractor enrolled in the OCIP must provide the following documentation:

- .1 Workers' Compensation Policy Declarations Page;
- .2 Workers' Compensation Rating/Information Page(s);
- .3 Workers' Compensation Experience Modification Worksheet;
- .4 General Liability Policy Declarations Page;
- .5 General Liability Rating/Information Page;
- .6 Excess Liability Policy Declarations Page; and
- .7 Excess Liability Rating/Information Page(s).

The foregoing shall be subject to the Authority's review and approval, which shall not unreasonably be withheld. Insurance broker and agent summary letters will not be accepted as proof of insurance.

**10.2.6** Failure to comply with the procedures in this **Paragraph 10.2** may result in any or more of the following non-exclusive list of consequences:

- .1 The Authority may deny admission to the Project Site until compliance is obtained.
- .2 The Authority may withdraw or deny insurance provided by the OCIP.
- .3 The Authority may withhold progress payments until compliance is obtained.

.4 The Authority may apply standard premium rates for Worker's Compensation and General Liability and Excess Liability insurance for the particular trade description. An experience modification of 1.0 will be applied to the Worker's Compensation Rate to determine the Insurance Credit.

.5 The Authority may apply a minimum Insurance Credit of 3.0% of the GMP in the case of the Construction Manager, or 3% of the value of a Subcontract or Sub-subcontract.

## GENERAL CONDITIONS OF THE CONSTRUCTION SERVICES AGREEMENT

## APPENDIX C

## AUTHORITY'S INSURANCE COVERAGES AND LIMITS

[Reserved]

### GENERAL CONDITIONS OF THE CONSTRUCTION SERVICES AGREEMENT

### APPENDIX E

### WRITTEN NOTICE ADDRESSES

Every notice, demand, request, consent, approval or other communication in connection with breach, indemnity, suspension, termination and/or default, which either party hereto is required or desires to give or make to the other party hereto shall, notwithstanding any other provisions of the Agreement, be effective only if given in writing and delivered by hand and receipted for, or by registered or certified mail, postage-prepaid, return receipt requested, or by overnight mail as follows:

If to the Construction Manager addressed to:	[INSERT]
with a copy to:	[INSERT] [INSERT]
with a copy to:	[INSERT] [INSERT]
If to the Architect, addressed to:	HKS, Inc. 1919 McKinney Avenue Dallas, TX 75201 Attention: Craig Williams Fax: (214) 969-3397
with a copy to:	HKS, Inc. 1919 McKinney Avenue Dallas, TX 75201 Attention: John Hutchings Fax: (214) 969-3397
If to the Authority, addressed to:	Metropolitan Sports Facility Authorities 900 South Fifth Street Minneapolis, MN 55415 Attention: Ted Mondale CEO/Executive Director Fax: 612-332-8334
with a copy to:	Dorsey & Whitney, LLP Suite 1500 50 South Sixth Street Minneapolis, MN 55402-1498 Attention: Jay L. Lindgren Fax: 612-340-2868
with a copy to:	Fabyanske, Westra, Hart & Thomson, PA 800 LaSalle Avenue, Suite 1900 Minneapolis, MN 55402 Attention: Dean B. Thomson Fax: 612-359-7602

 If to the Team, addressed to:
 Minnesota Vikings Football, LLC

 9520 Viking Drive
 Eden Prairie, MN 55344

 Attention: Kevin Warren
 Vice President of Legal Affairs & Chief Administrative Officer

 Fax No.: 952-828-6514
 Briggs and Morgan

 with a copy to:
 Briggs and Morgan

 2200 IDS Center
 80 South Eighth Street

 Minneapolis, MN 55402
 Minneapolis, MN 55402

Or to such other address or addresses as Authority, Architect, and Construction Manager shall from time to time designate by notice given and delivered as aforesaid.

Fax No.: 612-977-8650

Attention: Brian Wenger, Mike Grimes, and Matt Slaven

## EXHIBIT 7 EQUITY PLAN FOR THE CONSTRUCTION SERVICES AGREEMENT

Exhibit 7 is attached.



# MINNESOTA MULTI-PURPOSE STADIUM CONSTRUCTION SERVICES AGREEMENT EQUITY PLAN

# Section 1. Purpose / Outline – Overall

- 1.1 <u>Introduction</u>. The State of Minnesota created the Minnesota Sports Facility Authority ("Authority") to build a state-of-the-art multipurpose facility called the "People's Stadium" ("Project"). The Authority is required by law to promote the involvement of women and members of minority communities in the construction of the Project, as more completely described in Minn. Stat. § 473J.12. The Authority hereby adopts this comprehensive Equity Plan for the construction phase of the Project. The purpose of this Equity Plan is to implement the Authority's statutory mandate to promote employment of women and members of minority communities, create an employment program, hold a job fair, establish goals for construction contracts to be awarded to women-owned and minority-owned businesses, and establish workforce utilization goals.
- 1.2 <u>The Team</u>. Minnesota Vikings Football, LLC (the "Team") fully supports this Equity Plan.
- 1.3 <u>Affirmative Action / Non-Discrimination</u>. Pursuant to Minn. Stat. § 473J.12, the Authority shall make every effort to employ, and cause the NFL team, the Construction Manager and other Subcontractors, vendors, and concessionaires to employ women and members of minority communities when hiring.
- 1.4 <u>Employment Assistance Firm</u>. Pursuant to Minn. Stat. § 473J.12, the Authority will issue a Request for Proposals for an employment assistance firm, preferably minority-owned, or owned by a disabled individual or a woman, to create an employment program to recruit, hire, and retain minorities for the Stadium facility. The Authority intends to supplement its statutory mandate and engage an employment assistance firm to identify, train, and facilitate the hiring and utilization of minorities, women, and veterans by the Construction Manager and its Subcontractors hired to construct the Project.
- 1.5 Job Fair. Pursuant to Minn. Stat. § 473J.12, the Authority shall hold a job fair and recruit and advertise at Minneapolis Urban League, Sabathani,



American Indian OIC, Youthbuild organizations, and other such organizations. To facilitate the Construction Manager's achievement of the Authority's workforce goals, the Authority may organize and host other similar fairs and events.

- 1.6 <u>Stadium Equity Oversight Committee / Construction Manager Review</u> <u>Panel</u>. The Authority recognizes that community involvement and support are crucial to a successful Project in general and this Equity Plan in particular. To foster and facilitate appropriate community involvement, the Authority will establish various committees, including the Stadium Equity Oversight Committee described in Section 3, and the Construction Manager Equity Review Panel described in Section 2.
- 1.7 <u>Targeted Business</u>. In Sections 4-11 of the Equity Plan, the Authority establishes a Targeted Business Program. The Targeted Business Program sets an 11% and 9% goal for construction contracts for the Project to be awarded to women- and minority-owned Minnesota-based business enterprises, respectively. The Targeted Business Program defines specifically how the goals may be met, the parameters of the Construction Manager's necessary and reasonable good faith efforts to meet the goals, the requirements for reporting and monitoring, the assistance the Authority will provide, and the consequences if the Construction Manager fails to meet the goals and fails to make good faith efforts to achieve the goals.
- 1.8 <u>Veterans</u>. In Section 21 of the Equity Plan, the Authority establishes a Veterans Inclusion Program. The Authority honors the service of our veterans and intends to supplement its statutory mandate by creating the Veterans Inclusion Program to ensure that our veterans have every opportunity to participate in the Project. The Veterans Inclusion Program will include efforts to include small veteran-owned businesses in the construction contracts to be awarded by the Construction Manager. The Veterans Inclusion Program will also include efforts to utilize veterans in the construction workforce.
- 1.9 <u>Workforce</u>. In Sections 12-18 of the Equity Plan, the Authority establishes a Workforce Program. The Workforce Program sets a 32% and 6% goal for workforce utilization for the Project of minorities and women, respectively. These goals were first established by the Minnesota Department of Human Rights. The City of Minneapolis then adopted these same goals. Achievement of these goals may be impacted by the nature of the Minnesota workforce, the number of individuals from



licensed trades needed for the Project, the number of other large construction projects that will compete for workers with the Project, and the availability issues these and other factors create. Despite the challenges, the Authority is committed to ensuring that the Construction Manager meets the goals or uses all necessary and reasonable good faith efforts to meet these goals. Furthermore, the Authority is confident that the level of participation the Project actually achieves and the training plan that will be implemented will leave a lasting legacy of a larger minority and woman construction workforce. The Workforce Program defines specifically how the goals may be met, the parameters of the Construction Manager's necessary and reasonable good faith efforts to meet the goals, the requirements for reporting and monitoring, the assistance the Authority will provide, and the consequences if the Construction Manager fails to meet the goals and fails to make good faith efforts to achieve the goals.

- 1.10 <u>MSFA Staff</u>. The Authority is committed to a successful Equity Plan and, to that end, the Authority has and will continue to make a significant investment in experienced staff to administer the Equity Plan. Among other things, the Authority has hired two "loaned executives" from the Metropolitan Council with deep experience on all of the issues that are part of this Equity Plan. These loaned executives will not only administer the Equity Plan, but they will also train other Authority staff to take over administration once the loaned executives' employment with the Authority ends.
- 1.11 <u>Significant Investment</u>. The Authority has and will continue to make significant investments in the success of the Equity Plan, including the loaned executive staff, other staff, and retention of an employment assistance firm to establish employment programs as described in Section 20.
- 1.12 <u>Scope / Flowdown</u>. As more specifically articulated in the Construction Services Agreement to be entered into between the Authority and the Construction Manager, this Equity Plan applies to the Construction Manager and the Construction Manager's Subcontractors and Suppliers of any tier.
- 1.13 <u>Monitoring and Compliance</u>. As more specifically described throughout the Equity Plan, the Authority has primary responsibility to monitor and audit the Construction Manager's compliance with the Targeted Business Program, the Workforce Program, and the Veterans Inclusion Program.



The Authority may, from time to time and in its sole discretion, request assistance from the City of Minneapolis, the Minnesota Department of Human Rights, and other appropriate agencies to assist the Authority in its efforts.

## Section 2. Construction Manager Equity Review Panel

- 2.1 The Authority shall establish a "Construction Manager Equity Review Panel." The Panel will, among other things, provide an advisory recommendation regarding potential Construction Managers' experience and commitment to targeted business and workforce utilization goals.
- 2.2 The Authority shall appoint ten members at large to the Panel, including:
  - (a) Two representatives from the Authority;
  - (b) Two representatives from Minnesota Vikings Football, LLC;
  - (c) One representative from the City of Minneapolis;
  - (d) One representative from the Minnesota Department of Labor and Industry;
  - (e) One representative from the Minnesota women's business community, such as, for example, the Association of Women Contractors;
  - (f) One representative from the Minnesota minority business community, such as, for example, the National Association of Minority Contractors;
  - (g) One representative from the Minnesota economic development community, such as Metropolitan Economic Development Association; and
  - (h) One representative from Minnesota organized labor, such as, for example, the Minneapolis Building and Construction Trades Council.
- 2.3 The Authority shall appoint one of the two Authority representatives to chair, administer, and attend each meeting of the Panel. The Authority



and its staff representatives are also responsible for facilitating communication to and from the Panel.

- 2.4 No member of the Panel may be an employee of a business or other organization that intends to solicit a contract or subcontract or otherwise be paid as part of the total cost of the work of the Project.
- 2.5 After the Authority receives proposals from potential construction managers, and after the Authority generates a "short list" of proposers, but before the Authority awards the Construction Services Agreement to a construction manager, the Panel shall request information from each short-listed Proposer and conduct an interview of each such Proposer lasting no longer than 60 minutes. The interview shall consist of a 30-minute presentation by the proposer with the remaining time to be devoted to questions by the Panel.
  - (a) Prior to the interview, the Panel will send an interview packet to each Proposer containing a request for information and a list of questions;
  - (b) The request for information shall ask the Proposer to include, at minimum, the following information:
    - The Proposer's experience and accomplishments on projects with targeted business and workforce utilization goals;
    - (ii) Resumes of the Proposer's staff or contract organizations responsible for successful execution of the Equity Plan;
    - (iii) References from public agencies, minority and women business organizations, and community workforce agencies who can provide information regarding the Proposer's past efforts on projects with targeted business and workforce utilization goals;
    - (iv) A detailed description of strategies for compliance with equity plans including without limitation detailed solicitation strategies that includes events, outreach activities, and innovative ideas to reduce barriers for minorities, women, and small minority- and women-owned



businesses and encourage, support, and obtain workforce participation by minorities and women on projects.

- 2.6 After all short-listed Proposers are interviewed, the Panel shall generate and submit a report to the Authority. The report shall contain, at minimum, a "Pass" or "Incomplete" designation. A "Pass" designation signifies the Panel recommends that the Proposer continue in the construction manager selection process. An "Incomplete" designation signifies that the Panel recommends the Proposer not continue to final negotiations unless and until the proposer addresses any noted deficiencies in the Panel's report.
- 2.7 The Panel's recommendation is advisory only. The Authority and Team retain complete and final discretion to award the Construction Services Agreement to the Construction Manager of its choice.

# Section 3. Stadium Equity Oversight Committee

- 3.1 To ensure an accountable and transparent process, the Authority shall establish a "Stadium Equity Oversight Committee." The Committee will, among other things, facilitate communication with the community regarding the Equity Plan and related issues associated with the Project.
- 3.2 The Committee shall hold periodic meetings, as determined to be appropriate by the Authority, for the lifetime of the construction of the Project, beginning in March of 2013. The meetings shall be open to the public.
- 3.3 The Authority shall appoint two permanent Authority representatives to chair, administer, and attend each meeting of the Committee. The Authority and its staff representatives are also responsible for facilitating communication to and from the Committee, including a report to the Authority following each such meeting.
- 3.4 The Authority shall specifically invite representatives of the following stakeholders to attend and participate in meetings of the Committee:
  - (a) Minnesota Vikings Football, LLC;
  - (b) The Employment Assistance Firm(s);
  - (c) The Construction Manager;



- (d) All members of the Construction Manager Equity Review Panel, as defined by Section 2;
- (e) City of Minneapolis Office of Civil Rights (Prevailing Wage);
- (f) Minnesota Department of Human Rights (Affirmative Action and Workforce Utilization Reporting);
- (g) Minnesota Department of Labor and Industry (Prevailing Wage and Apprenticeship);
- (h) Minnesota Department of Administration (Materials Management TGB Program);
- (i) City of St. Paul (CERT Program);
- (j) Minnesota Unified Certification Program (DBE); and
- (k) Military Action Group.
- 3.5 The Authority may specifically invite additional parties to attend and participate in meetings of the Committee in the Authority's sole discretion, but the Committee and its meetings will remain open to the public. Any member of the public may attend and participate in the Committee meetings.
- 3.6 A representative from the Construction Manager is required to attend each Committee meeting.
- 3.7 A representative from the employment assistance firm is required to attend each Committee meeting.
- 3.8 The primary purpose of the Committee is to facilitate communication with the community regarding the Equity Plan and related issues. Although the Committee may from time to time make recommendations to the Authority, the Authority retains the sole, complete, and final discretion on all Equity Plan decisions related to the Project.

# Section 4. Targeted Business Program



- 4.1 The Authority recognizes the legislature's mandate that the Authority establish and attempt to meet goals for construction contracts for the Project (the "Project") to be awarded to women- and minority-owned business enterprises ("Targeted Businesses") in a percentage at least equal to the minimum used for City of Minneapolis development projects. In furtherance of the intention of such mandate, the Authority also finds it appropriate to adopt this Targeted Business Program to establish participation goals for construction contracts entered into by the Authority's Construction Manager and its various Subcontractors and Suppliers of all tiers.
- 4.2 The Authority establishes this Targeted Business Program based on, among other things, the study completed by National Economic Research Associates, Inc. for the City of Minneapolis captioned "The State of Minority- and Women-Owned Business Enterprise: Evidence from Minneapolis" dated October 21, 2010 ("Disparity Study"), and supplemental reports. The Authority adopts this Targeted Business Program for purposes of, among other things, ensuring a "level playing field" for WBE/MBE firms, fostering equal opportunity for WBE/MBE firms in contracts with the Authority, and reducing the relevant disparities identified in the Disparity Study.
- 4.3 The Authority hereby adopts the participation goals for work included in the Construction Services Agreement, based on the Disparity Study, and provided to the Authority by the City of Minneapolis. These goals are project- and contract-specific goals created by the City of Minneapolis Department of Civil Rights for the work included in the Construction Services Agreement. The City calculated these goals based on (a) the Disparity Study, (b) specific availability by scope of work for Targeted Businesses certified as more specifically described in Section 5.1, and (c) using the relative local disparity based on the total number of construction-related firms reported in the 2010 census. The goals are appropriate for the Construction Services Agreement because the Project The goals, expressed as a is located in the City of Minneapolis. percentage of the "Cost of the Work" as that term is defined by the Construction Services Agreement are as follows:
  - (a) 9% participation by MBE firms, as defined below, and
  - (b) 11% participation by WBE firms, as defined below.



("Targeted Business Goal").

- 4.4 The Authority will require the Construction Manager to make every necessary and reasonable good faith effort to achieve the goals set forth in Section 4.3.
- 4.5 The Authority adopts this Targeted Business Program to define more specifically how the goals may be met, the parameters of the Construction Manager's necessary and reasonable good faith efforts to meet the goals, the requirements for reporting and monitoring, and the consequences if the Construction Manager fails to meet the goals and fails to make good faith efforts to achieve the goals.
- <u>Section 5.</u> <u>Definitions Targeted Business.</u> For purposes of the Targeted Business Program:
  - 5.1 "MBE" means a Minnesota-based business that meets one or more of the following criteria:
    - (a) Certified by the Minnesota Uniform Certification Program ("MUCP") as an MBE (as that term is used by MUCP) and found in the MUCP directory at: <u>http://mnucp.metc.state.mn.us/;</u>
    - (b) Certified by the Minnesota Department of Administration and classified as a Targeted Group (TG) business in the directory for the Minnesota Department of Administration Materials Management website at:
      <u>http://www.mmd.admin.state.mn.us/process/search/</u>. The business also must have one or more of the following designations used in the Department of Administration's website: (A), (B), (H), (I), or (E). (The legend at the bottom of the directory page of the website provides this information.) Firms identified solely by a (D), (L), (M), (R), or (W) (which represent disabled-owned firms; firms in economically disadvantaged geographic areas; and woman-owned firms) do not meet the definition of an MBE;
    - (c) Certified by the Central Certification Program (CERT) as an MBE (as that term is used by CERT) and found in the CERT directory. The CERT directory will be made available by the Authority.



- 5.2 "WBE" means a Minnesota-based business that that meets one or more of the following criteria:
  - (a) Certified by the Minnesota Uniform Certification Program ("MUCP") as a WBE (as that term is used by MUCP) and found in the MUCP directory at: <u>http://mnucp.metc.state.mn.us/</u>;
  - (b) Certified by the Minnesota Department of Administration and classified as a Targeted Group (TG) business in the directory for the Minnesota Department of Administration Materials Management website at:
     <u>http://www.mmd.admin.state.mn.us/process/search/</u>. The business also must have the (W) designation used in the Department of Administration's website, denoting a woman-owned business;
  - (c) Certified by the Central Certification Program (CERT) as an MBE (as that term is used by CERT) and found in the CERT directory. The CERT directory will be made available by the Authority.
- 5.3 "Targeted Business" means an MBE or WBE.
- 5.4 "Subcontractor" means any entity the Construction Manager contracts with for any part of the Project, including without limitation Subsubcontractors and Suppliers of any tier to the Construction Manager.
- 5.5 "Subcontract" means any contract entered into by the Construction Manager or a Subcontractor for any part of the Project.
- 5.6 "Good Faith Efforts" means the necessary and reasonable efforts to meet the goals of this Targeted Business Program, including without limitation the good faith efforts described in this Targeted Business Program.

# Section 6. Achievement of Participation Goals – Targeted Business

- 6.1 The Authority will require the Construction Manager to use Good Faith Efforts to achieve the Targeted Business Goals through the Subcontracts entered into between the Construction Manager and its Subcontractors, which in turn may enter into various agreements with lower-tier Subcontractors.
- 6.2 The Construction Manager is not required to meet the goals of this Targeted Business Program for each individual Subcontract it enters into.



Instead, subject to Section 7.3, the Construction Manager may meet, or make Good Faith Efforts to meet, the goals of this Targeted Business Program by aggregating participation from all of its Subcontractors. As part of its Good Faith Efforts, the Construction Manager may request participation at different levels from different Subcontractors.

- 6.3 The Construction Manager may count toward the Targeted Business Goal the value (or a percentage of the value, as discussed below) of the various Subcontractors' contracts for work performed on the Project only after the MBE or WBE is certified as such as described in Sections 5.1 and 5.2.
- 6.4 Whether the Construction Manager achieves the goals of this Targeted Business Program will be evaluated and determined as the Project progresses and at the end of the Project based on the total Cost of the Work as defined in the Construction Services Agreement. If the total dollar amount of MBE and WBE contracts meets or exceeds the established Targeted Business Goal, then the Authority will determine that the goals of this Targeted Business Program have been met.
- 6.5 The Authority will utilize the following guidelines, plus the guidelines in Appendix A, in determining the percentage of WBE/MBE participation that will be counted toward the Targeted Business Goal:
  - (a) The value of work performed by a firm after it has ceased to be certified as an MBE or WBE will not be counted toward the Targeted Business Goal.
  - (b) If an MBE or WBE's certification lapses, the value of work performed by a firm during any period of lapsed certification will not be counted toward the Targeted Business Goal.
  - (c) Only amounts paid to and performed by an MBE or WBE will be counted toward the Targeted Business Goal; participation of a Subcontractor will not be counted until the amount has been paid to the Subcontractor.

# Section 7. Compliance Monitoring and Reporting – Targeted Business

7.1 The Authority intends to assist and monitor the Construction Manager's efforts to achieve the Targeted Business Goal during all phases of the Project, from pre-solicitation through construction. The Construction



Manager will cooperate with the Authority's monitoring plan and requests as outlined in this section.

- 7.2 <u>Pre-Solicitation Phase Authority</u>. During the pre-solicitation phase (between the award of the Construction Services Agreement and the prebid meeting(s) for each Subcontract), the Authority will:
  - (a) Participate with various governmental agencies, including without limitation the U.S. Small Business Administration and the State of Minnesota Materials Management Division, along with these agencies' outreach efforts and programs designed to assist MBEs and WBEs to become certified;
  - (b) Work with non-profit business support organizations to develop strategies designed to reduce barriers for MBEs and WBEs and create effective communication activities;
  - (c) Attend meetings and explain the Targeted Business Program to interested attendees;
  - (d) Provide the Construction Manager and its Subcontractors of all tiers all appropriate resources and assistance to find WBE and MBE firms; and
  - (e) Work with the Construction Manager to host, plan and operate small business "meet and greet" sessions intended to facilitate the introduction of small businesses, including WBE and MBE firms, to the Construction Manager and its Subcontractors of all tiers. The small business "meet and greet" sessions will be conducted no later than one month prior to respective bid due dates.
- 7.3 <u>Pre-Solicitation Phase Construction Manager</u>. During the pre-solicitation phase (between the award of the Construction Services Agreement and the pre-bid meeting(s) for each Subcontract), the Construction Manager shall:
  - (a) Provide one point of contact to the Authority for the presolicitation phase of the Project;
  - (b) Comply with the Construction Manager's Targeted Business Enterprise Plan required by Section 10.8.1 of the Construction Services Agreement;



- (c) Have the option to set individual goals on individual Subcontracts consistent with the Targeted Business Enterprise Plan;
- (d) Work with the Authority to host, plan, and operate no less than two small business "meet and greet" sessions;
- (e) Actively recruit lower-tier Subcontractors to participate in pre-bid meetings and small business "meet and greet" sessions; and
- (f) Hold a pre-bid meeting for all Subcontracts prior to bid due date.
- 7.4 <u>Solicitation Phase Authority</u>. During the solicitation phase (between each Subcontract's pre-bid meeting and the bid due date), the Authority will:
  - (a) Assist the Construction Manager, bidders, and Targeted Businesses with any questions regarding the Targeted Business Program; and
  - (b) Provide, upon request, rulings (based upon information reported by the Construction Manager, bidders, and Targeted Businesses) regarding (i) whether an MBE or WBE will be performing a commercially useful function (defined below), and (ii) whether and how an MBE or WBE's work will be counted toward the Targeted Business Goal.
- 7.5 <u>Solicitation Phase Construction Manager</u>. During the solicitation phase (between each Subcontract's pre-bid meeting and the bid due date), the Construction Manager shall:
  - (a) Provide one point of contact to the Authority for the solicitation phase of the Project;
  - (b) Comply with the Construction Manager's Targeted Business Enterprise Plan required by Section 10.8.1 of the Construction Services Agreement; and
  - (c) Submit all documentation required by the Authority, including without limitation:
    - (i) The Targeted Business Commitment and Information form (attached hereto as <u>Exhibit A</u>) for each bid, which shall identify by name the WBE or MBE that is committed to be



used on the specific Subcontract; the scope of work; and the contract value or percentage of total Subcontract amount represented by the WBE or MBE; and

- (ii) The Subcontractor Information Form (attached hereto as <u>Exhibit B</u>), signed by the committed WBE or MBE and the bidder identifying the work and contract value at time of bid.
- 7.6 <u>Post-Bid Review Phase Authority</u>. During the post-bid review phase (between each Subcontract's bid due date and completion of <u>Exhibit A</u>, the Authority will:
  - (a) Request the Construction Manager to provide Good Faith Effort documents and a plan to carry out Good Faith Efforts during the construction phase;
  - (b) Reserve the ability to adjust applicable MBE and WBE contract values based on commercially useful function determinations, incorrect Targeted Business Goal value calculations, or Targeted Business certification rulings; and
  - (c) Review <u>Exhibit A</u> within 14 calendar days of each bid. The Authority will award a Pass/Award, Pass/Waiver, or Fail for each such <u>Exhibit A</u> submitted by the Construction Manager, based on the following criteria:
    - Pass/Award Bidder committed to contract with MBE and WBE firms that meet or exceed each respective goal for that contract. A Pass/Award signifies that the bidder is to continue in the contract award process;
    - Pass/Waiver Bidder did not meet the Targeted Business Goal, but demonstrated sufficient Good Faith Efforts. A Pass/Waiver signifies that the bidder is to continue in the contract award process; or
    - (iii) Fail Bidder did not meet the Targeted Business Goals and did not demonstrate sufficient Good Faith Efforts. A Fail signifies that the bidder is not recommended to continue in the contract award process.



- 7.7 <u>Post-Bid Review Phase Construction Manager</u>. During the post-bid review phase (between each Subcontract's bid due date and completion of the Authority's review of **Exhibit A**, the Construction Manager shall:
  - (a) Provide one point of contact to the Authority for the post-bid review phase of the Project;
  - (b) Provide to the Authority all requested Good Faith Effort documentation within 3 calendar days of the bid; and
  - (c) Provide to the Authority a Good Faith Efforts Plan from the Construction Manager and each selected Subcontractor for the construction phase within 10 calendar days of the bid. This Good Faith Efforts Plan shall address, among other things, how the Subcontractor and Construction Manager intend to address changes to anticipated Targeted Business participation including, for example, changes in participation counting made by the Authority during the bid review phase and changes in scope during the construction phase.
- 7.8 <u>Construction Phase Authority</u>. During the construction phase (between the award of each Subcontract and Final Payment to the Construction Manager), the Authority will:
  - (a) Work with the Construction Manager to identify, prevent, and resolve Targeted Business-related concerns on the Project.
- 7.9 <u>Construction Phase Construction Manager</u>. During the construction phase (between the award of each Subcontract and final payment), the Construction Manager shall:
  - (a) Provide one point of contact to the Authority for the construction phase of the Project;
  - (b) Actively participate in documenting Good Faith Efforts and monitoring;
  - (c) Work with the Authority to identify, prevent, and resolve contractrelated issues with Targeted Business concerns on the Project;



- (d) Identify all Targeted Business work for each monthly Application for Payment in a timely and efficient manner, as more specifically described in Section 7.9(f);
- (e) Enter applicable data, including payment data, in the LCP Tracker system operated by the City of Minneapolis Department of Human Rights or other system designated by the Authority; and
- (f) Cooperate with the Authority's ongoing monitoring of the Construction Manager's Good Faith Efforts and submit to the Authority on a monthly basis complete and accurate Targeted Business utilization data including, without limitation, the following:
  - (i) Total Project Cost of the Work;
  - (ii) Total Value of approved Contract Revisions;
  - (iii) Total Value of approved Change Orders to each Subcontractor and Targeted Business;
  - (iv) Time period for which the monthly report covers. This period shall begin on the first day of each month and end on the last day of each month;
  - Total of all Applications for Payments made by the Construction Manager to the Authority as of the last day of the reporting period;
  - (vi) Total payments received by the Construction Manager from the Authority as of the last day of the reporting period;
  - (vii) Name of each WBE, MBE, and VBE the Construction Manager and/or Subcontractor has committed to use at time of respective bid;
  - (viii) Identification of Subcontractors that have hired each WBE, MBE, and VBE;
  - (ix) Identification of each business as a WBE, MBE, and VBE;



- (x) Total contract value for each committed Subcontractor, WBE, MBE, and VBE;
- (xi) Changes, if applicable, to the total contract value for each committed Subcontractor, WBE, MBE, and VBE;
- (xii) Identification of WBE, MBE, and VBE as a Contractor, Supplier, or Broker;
- (xiii) Value of work or supplies claimed by the WBE, MBE, and VBE during the report period;
- (xiv) Value of work or supplies to be counted towards the respective WBE, MBE, and VBE goal during the report period; and
- (xv) Total value of work or supplies invoiced to date and paid to date for each WBE, MBE, and VBE,

which information the Authority may require the Construction Manager to submit to the Authority on a form substantially similar to the CSA Equity Plan Progress Report attached hereto as <u>Exhibit</u> <u>C</u>, which form the Authority may amend, from time to time, in its sole discretion.

# Section 8. Participation Counting – Targeted Business

8.1 Guidelines regarding how MBE and WBE participation is counted toward the Targeted Business Goal are described in <u>Appendix A</u>.

# Section 9. Commercially Useful Function – Targeted Business

9.1 The Authority will determine whether an MBE or WBE is performing a commercially useful function ("CUF") as described in <u>Appendix B</u>.

# Section 10. Good Faith Efforts – Targeted Business

10.1 The Construction Manager's responsibility to provide Good Faith Efforts documentation, a Good Faith Efforts Plan, and otherwise engage in Good



Faith Efforts as described by the Targeted Business Plan are independent obligations and are required even if the Targeted Business Goals are met.

10.2 The Authority will determine whether the Construction Manager has made Good Faith Efforts to achieve the Targeted Business Goal as described in <u>Appendix C</u>.

#### Section 11. Consequences of Failure to Meet Goals and Failure to Make Good Faith Efforts – Targeted Business

- 11.1 At the end of the Project, and before the Authority makes Final Payment to the Construction Manager, the Authority shall determine, in the Authority's sole discretion, whether the Construction Manager met the Targeted Business Goal. If the Construction Manager fails to meet these goals, the Authority shall also determine, in its sole discretion, whether the Construction Manager made Good Faith Efforts to meet the Targeted Business Goal.
- 11.2 If the Authority determines the Construction Manager failed to make Good Faith Efforts to meet the Targeted Business Goal, the Construction Manager shall be liable to the Authority for the Authority's actual damages for the Construction Manager's failure to make Good Faith Efforts. The Authority also shall withhold payment to the Construction Manager as follows:
  - (a) The Authority shall determine whether the Construction Manager failed to make Good Faith Efforts for the entire Project or for only a part of it.
  - (b) If the Authority determines the Construction Manager failed to make Good Faith Efforts for the entire Project, the Authority shall deduct payment to or assess damages against the Construction Manager in the amount of the difference between the level of Targeted Business participation had Good Faith Efforts been used and the actual Targeted Business participation.
  - (c) If the Authority determines the Construction Manager failed to make Good Faith Efforts for only part of the Project, the Authority shall withhold payment to or assess damages against the Construction Manager in the amount of the difference between the level of Targeted Business participation had Good Faith Efforts



been used for that aspect of the Project, and the actual Targeted Business participation for that aspect of the Project.

- (d) The Authority shall make all determinations in Section 11.2(a)-(c) in the Authority's sole discretion.
- 11.3 Intentionally or recklessly false reporting of Targeted Business data, Good Faith Efforts regarding efforts to achieve Targeted Business goals, or the commercially useful function of reported Targeted Business participation shall be subject to the Minnesota False Claims Act. This liability also flows down and applies to Subcontractors and Suppliers of all tiers to the extent they intentionally or recklessly report similar false data regarding Targeted Business participation on the Project.

#### Section 12. Workforce Program

- 12.1 The Authority recognizes the legislature's mandate that the Authority establish construction workforce goals for utilization of women and minorities during construction of the Project in a percentage at least equal to the current City of Minneapolis goals, and which also includes efforts to include workers from City of Minneapolis zip codes that have high rates of poverty and unemployment.
- 12.2 In furtherance of the intention of such mandate, the Authority finds it appropriate to adopt this Workforce Program to establish construction workforce utilization goals for the construction work performed by the Authority's Construction Manager and its various Subcontractors.
- 12.3 The Authority adopts the following construction workforce utilization goals that have been adopted by the City of Minneapolis and which are based on the underlying data and intent of the workforce goal program established by the Minnesota Civil Rights Act as implemented by the Minnesota Department of Human Rights:

32% minority

6% female

("Workforce Goals")



These Workforce Goals are subject to change in the Authority's sole discretion based on, for example, the availability of various categories of trade labor, the competition for qualified labor from other concurrent projects, and any changes in the understanding of the supporting data.

- 12.4 The Authority adopts these Workforce Goals because they are the goals adopted by the City of Minneapolis for development projects in the City during the time period 2013 2016. The City of Minneapolis, in turn, adopted these goals based on the goals promulgated by the Minnesota Department of Human Rights for Hennepin County.
- 12.5 These goals were first established by the Minnesota Department of Human The City of Minneapolis then adopted these same goals. Rights. Achievement of these goals may be impacted by the nature of the Minnesota workforce, the number of individuals from licensed trades needed for the Project, the number of other large construction projects that will compete for workers with the Project, and the availability issues these and other factors create. Despite the challenges, the Authority is committed to ensuring that the Construction Manager meets the goals or uses all necessary and reasonable good faith efforts to meet these goals. Furthermore, the Authority is confident that the level of participation the Project actually achieves and the training plan that will be implemented will leave a lasting legacy of a larger minority and woman construction workforce. The Workforce Program defines specifically how the goals may be met, the parameters of the Construction Manager's necessary and reasonable good faith efforts to meet the goals, the requirements for reporting and monitoring, the assistance the Authority will provide, and the consequences if the Construction Manager fails to meet the goals and fails to make good faith efforts to achieve the Workforce Goals.
- 12.6 The Construction Manager is required to make Good Faith Efforts (as defined below) set forth in Section 12.3 to achieve the Workforce Goals.
- 12.7 The Authority intends to administer the Workforce Program and may seek assistance from Authority Staff, the City of Minneapolis Civil Rights Department, and the Minnesota Department of Human Rights. The Authority may also seek advice from the City and the Minnesota Department of Human Rights to develop additional policies, procedures, and forms to implement the Workforce Program. The Authority retains sole and complete discretion regarding the policies, administrative procedures, and forms to be developed and adopted.



### <u>Section 13.</u> <u>Definitions – Workforce.</u> For purposes of the Workforce Program:

- 13.1 "Minority" has the same meaning as "minority" in the Minnesota Department of Human Rights regulations, as follows:
  - (a) Black, persons having origins of any of the Black African racial groups not of Hispanic origin;
  - (b) Hispanic, persons of Mexican, Puerto Rican, Cuban, Central American, South American, or other Spanish culture or origin, regardless of race;
  - (c) Asian and Pacific Islander, persons having origins in any of the original peoples of the Far East, Southeast Asia, the Indian subcontinent, or the Pacific Islands; and
  - (d) American Indian or Alaskan Native, persons having origins in any of the original peoples of North America and maintaining identifiable tribal affiliations through membership and participation or community identification.
- 13.2 "Subcontractor" means any entity the Construction Manager contracts with for any part of the Project performing work on the job-site. "Subcontractor" also includes Subcontractors of any tier to the Construction Manager.
- 13.3 "Good Faith Efforts" means the necessary and reasonable efforts to meet the Workforce Goals, including without limitation the Good Faith Efforts described in this Workforce Program and Section 17.

#### Section 14. Affirmative Action Certificate - Workforce.

14.1 The Construction Manager and all Subcontractors are required to submit affirmative action plans and obtain a certificate of compliance from the Authority, the City of Minneapolis Department of Human Rights, the Minnesota Department of Human Rights, or other governmental body as determined by the Authority in its sole discretion, pursuant to Minn. R. 5000.3560, subp. 1(A). The Construction Manager and all its Subcontractors must maintain certificates of compliance throughout the Project, regardless of any exemptions that might otherwise apply under the Minnesota Department of Human Rights regulations.



#### Section 15. Compliance & Counting- Workforce.

15.1 The extent to which hours worked on the Project by minority and female workers count toward the Workforce Goals shall be determined as provided by <u>Appendix D</u>.

#### Section 16. Monitoring and Reporting – Workforce.

- 16.1 The Construction Manager is responsible for accurately collecting data from its own workforce of minorities and women and those of its Subcontractors of all tiers and promptly and regularly reporting it to the City and other governmental entities as directed by the Authority. The Construction Manager shall collect and report the data in a manner and form prescribed by the City or other governmental entities as directed by the Authority.
- 16.2 The Authority may request certain compliance and auditing services from the City of Minneapolis and the Minnesota Department of Human Rights. These services may include, for example, spot checking and on-siteauditing of the workforce utilization numbers reported by the Construction Manager and its Subcontractors of all tiers.
- 16.3 The Construction Manager must provide, at minimum, a report containing the following information on a monthly and cumulative basis throughout the Project, separated by Subcontract, by trade, and supported by certified payrolls including every person who works on the Project jobsite as part of the Construction Services Agreement:
  - (a) total hours of employment on the Project;
  - (b) total hours of employment of women;
  - (c) total hours of employment of minorities; and
  - (d) employee zip codes.

#### Section 17. Good Faith Efforts – Workforce.

17.1 The Construction Manager's responsibility to use Good Faith Efforts as described by the Workforce Program is an independent obligation and is required even if the Workforce Goals are met.



17.2 The Authority will determine whether the Construction Manager has made Good Faith Efforts to achieve Workforce Goals as described in **Appendix E**.

#### Section 18. Consequences of Failure to Meet Goals and Failure to Make Good Faith Efforts – Workforce.

- 18.1 Intentionally or recklessly false reporting of workforce data, good faith efforts regarding efforts to achieve workforce goals, or the commercially useful function of reported workforce labor shall be subject to the Minnesota False Claims Act. This liability also flows down and applies to the Construction Manager's Subcontractors of all tiers to the extent they intentionally or recklessly report similar false data regarding workforce participation by minorities or women on the Project.
- 18.2 If the Construction Manager or its Subcontractors of any tier do not utilize Good Faith Efforts to achieve the Workforce Goal, then the Construction Manager (a) shall be liable for paying the Authority the costs of the Authority's compliance enforcement in making its Good Faith Efforts determination and (b) shall cure the Workforce Goal shortfall to the extent caused by its lack of Good Faith Efforts. If the Construction Manager cannot cure the shortfall caused by its lack of Good Faith Efforts due to the stage of construction and/or its schedule status on the Project, the Construction Manager shall pay the Authority the cost to train the number of minorities and women that reasonably would have been employed on the Project but for the Construction Manager's lack of Good Faith Efforts, so that they are trained and available to work on future projects. The cost of proper training shall be determined by the Authority in its sole discretion, which determination will be final and binding on the Construction Manager unless arbitrary and capricious.

#### Section 19. Data Collection and Reporting

- 19.1 The Construction Manager and Subcontractors shall promptly and regularly comply with the Authority's requests to submit data, in electronic form or otherwise as directed by the Authority, regarding the Targeted Business Program, the Workforce Program, and other aspects of this Equity Plan.
- 19.2 The Construction Manager and Subcontractors shall submit the requested data directly to the Authority, the City of Minneapolis, the Minnesota



Department of Human Rights, or other governmental agency as directed by the Authority.

19.3 The Authority, in its discretion, may coordinate with the City of Minneapolis, the Minnesota Department of Human Rights, or other appropriate governmental agency to manage and document Targeted Business Program activity and Workforce Program activity throughout the Project.

#### Section 20. Employment Assistance Firm

- 20.1 Pursuant to Minn. Stat. § 473J.12, the Authority will issue a Request for Proposals for an employment assistance firm, preferably minority-owned, or owned by a disabled individual or a woman, to create an employment program to recruit, hire, and retain minorities for the Stadium facility. The Authority intends to supplement its statutory mandate and engage an employment assistance firm to identify, train, and facilitate the hiring and utilization of minorities, women, and veterans by the Construction Manager and its Subcontractors hired to construct the Project.
- 20.2 <u>Construction</u>. The Authority will issue a Request for Proposal to create an employment program to identify, train, recruit, and facilitate the hiring and utilization of minorities, women, and veterans by the Construction Manager and its Subcontractors hired to construct the Project.
- 20.3 <u>Facility Operation</u>. The Authority will issue a Request for Proposal to create an employment program to identify, train, recruit, and facilitate the hiring and utilization of minorities, women, and veterans to work in the Stadium facility after it is constructed.

#### Section 21. Veterans Inclusion Program

- 21.1 For purposes of the Veterans Inclusion Program, a small veteran-owned business ("VBE") is a firm certified as a Service-Disabled Veteran Owned Small Business (SDVOSB) or a Veteran Owned Small Business (VOSB) by the United States Department of Veteran Affairs. Each VBE also must be a Minnesota-based business.
- 21.2 To encourage participation in the Project by VBEs, during the presolicitation phase (between the award of the Construction Services



Agreement and the pre-bid meeting(s) for each Subcontract), the Authority will:

- (a) work with business support organizations in the veteran community to ensure Minnesota-based VBEs are aware of opportunities on the Project;
- (b) participate in Project information session meetings designed for VBEs; and
- (c) solicit VBEs to participate in the Project's small "meet and greet" sessions.
- 21.3 To encourage participation in the Project by VBEs, during the presolicitation phase (between the award of the Construction Services Agreement and the pre-bid meeting(s) for each Subcontract), the Construction Manager shall:
  - (a) participate in Project information session meetings designed for VBEs; and
  - (b) solicit VBEs to participate in the Project's small "meet and greet" sessions.
- 21.4 During the construction phase (between the award of each Subcontract and final payment), the Construction Manager and the Authority will monitor, count, and report VBE participation in the Project in the same manner as the Targeted Business Program as outlined in the Equity Plan, including without limitation Section 7.9(f) and Appendices A, B, and C.
- 21.5 To encourage participation in the Project by veterans as part of the construction workforce, the Authority will collaborate with and communicate Project job and training opportunities to veterans organizations, such as Minnesota Department of Veteran Affairs, Helmets to Hardhats, and other veterans organizations that provide employment and training assistance.
- 21.6 To encourage participation in the Project by veterans as part of the construction workforce, the Construction Manager shall:



- (a) Collaborate with the employment assistance firm(s) and the employment program(s) described in Section 20 to increase recruitment and training of Minnesota veterans for the Project; and
- (b) Indentify veterans in the data, including payment data, entered into the LCP Tracker system operated by the City of Minneapolis Department of Civil Rights or other system designated by the Authority.



#### APPENDIX A - Targeted Business Participation Counting

- (a) When a Targeted Business participates in a contract, count only the value of the work actually performed by the Targeted Business toward the Targeted Business Goal.
  - (1) Count the entire amount of that portion of a contract that is performed by the Targeted Business's own forces. Include the cost of supplies and materials obtained by the Targeted Business for the work of the contract, including supplies purchased or equipment leased by the Targeted Business (except supplies and equipment the Targeted Business Subcontractor purchases or leases from the Construction Manager or its affiliate).
  - (2) Count the entire amount of fees or commissions charged by a Targeted Business firm for providing a bona fide service, such as professional, technical, consultant, or managerial services, or for providing bonds or insurance specifically required for the contract, toward Targeted Business Goal, provided the fee is reasonable and not excessive as compared with fees customarily allowed for similar services.
  - (3) When a Targeted Business subcontracts part of the work of its contract to another firm, the value of the subcontracted work may be counted toward Targeted Business goals only if the Targeted Business's Subcontractor is itself a Targeted Business. Work that a Targeted Business subcontracts to a non-Targeted Business firm does not count toward Targeted Business Goal.
- (b) When a Targeted Business performs as a participant in a joint venture, count a portion of the total dollar value of the contract equal to the distinct, clearly defined portion of the work of the contract that the Targeted Business performs with its own forces toward Targeted Business Goal.
- (c) Count expenditures to a Targeted Business contractor toward Targeted Business Goal only if the Targeted Business is performing a commercially useful function on that contract.
  - (1) A Targeted Business performs a commercially useful function when it is responsible for execution of the work of the contract and is carrying out its responsibilities by actually performing, managing, and supervising the work involved. To perform a commercially useful function, the Targeted Business must also be responsible, with respect to materials and supplies used on the contract, for negotiating price, determining quality and quantity, ordering the material, and installing (where applicable) and paying for the material itself. To determine whether a Targeted Business is performing a commercially useful function,



evaluate the amount of work subcontracted, industry practices, whether the amount the firm is to be paid under the contract is commensurate with the work it is actually performing and the Targeted Business credit claimed for its performance of the work, and other relevant factors.

- (2) A Targeted Business does not perform a commercially useful function if its role is limited to that of an extra participant in a transaction, contract, or project through which funds are passed in order to obtain the appearance of Targeted Business participation. In determining whether a Targeted Business is such an extra participant, examine similar transactions, particularly those in which Targeted Businesses do not participate.
- (3) If a Targeted Business does not perform or exercise responsibility for at least 30 percent of the total cost of its contract with its own work force, or the Targeted Business subcontracts a greater portion of the work of a contract than would be expected on the basis of normal industry practice for the type of work involved, presume that it is not performing a commercially useful function.
- (4) When a Targeted Business is presumed not to be performing a commercially useful function as provided in paragraph (c)(3) of this section, the Targeted Business may present evidence to rebut this presumption, including normal industry practices for the type of work involved.
- (d) Use the following factors in determining whether a Targeted Business trucking company is performing a commercially useful function:
  - (1) The Targeted Business must be responsible for the management and supervision of the entire trucking operation for which it is responsible on a particular contract, and there cannot be a contrived arrangement for the purpose of meeting Targeted Business goals.
  - (2) The Targeted Business must itself own and operate at least one fully licensed, insured, and operational truck used on the contract.
  - (3) The Targeted Business receives credit for the total value of the transportation services it provides on the contract using trucks it owns, insures, and operates using drivers it employs.
  - (4) The Targeted Business may lease trucks from another Targeted Business firm, including an owner-operator who is certified as a Targeted Business. The Targeted Business who leases trucks from another Targeted Business receives



credit for the total value of the transportation services the lessee Targeted Business provides on the contract.

(5) The Targeted Business may also lease trucks from a non-Targeted Business firm, including from an owner-operator. The Targeted Business who leases trucks from a non-Targeted Business is entitled to credit for the total value of transportation services provided by non-Targeted Business lessees not to exceed the value of transportation services provided by Targeted Business-owned trucks on the contract. Additional participation by non-Targeted Business lessees receives credit only for the fee or commission it receives as a result of the lease arrangement. \*The Construction Manager or Subcontractor is not required to obtain additional permission to use this "matching manner" method of counting participation described in this paragraph (d)(5).

Example to this paragraph (d)(5): Targeted Business Firm X uses two of its own trucks on a contract. It leases two trucks from Targeted Business Firm Y and six trucks from non–Targeted Business Firm Z. Targeted Business credit would be awarded for the total value of transportation services provided by Firm X and Firm Y, and may also be awarded for the total value of transportation services provided by four of the six trucks provided by Firm Z. In all, full credit would be allowed for the participation of eight trucks. With respect to the other two trucks provided by Firm Z, Targeted Business credit could be awarded only for the fees or commissions pertaining to those trucks Firm X receives as a result of the lease with Firm Z.

- (6) For purposes of this paragraph (d), a lease must indicate that the Targeted Business has exclusive use of and control over the truck. This does not preclude the leased truck from working for others during the term of the lease with the consent of the Targeted Business, so long as the lease gives the Targeted Business absolute priority for use of the leased truck. Leased trucks must display the name and identification number of the Targeted Business.
- (7) All Targeted Business suppliers and trucking firms on the Project are required to participate in a semi-annual Targeted Business Counting Check with the Authority to ensure accurate reporting.
- (e) Count expenditures with Targeted Businesses for materials or supplies toward Targeted Business goals as provided in the following:



- (1)
- (i) If the materials or supplies are obtained from a Targeted Business manufacturer, count 100 percent of the cost of the materials or supplies toward Targeted Business Goal.
- (ii) For purposes of this paragraph (e)(1), a manufacturer is a firm that operates or maintains a factory or establishment that produces, on the premises, the materials, supplies, articles, or equipment required under the contract and of the general character described by the specifications.
- (2)
- (i) If the materials or supplies are purchased from a Targeted Business regular dealer, count 60 percent of the cost of the materials or supplies toward Targeted Business Goal.
- (ii) For purposes of this section, a regular dealer is a firm that owns, operates, or maintains a store, warehouse, or other establishment in which the materials, supplies, articles or equipment of the general character described by the specifications and required under the contract are bought, kept in stock, and regularly sold or leased to the public in the usual course of business.
  - (A) To be a regular dealer, the firm must be an established, regular business that engages, as its principal business and under its own name, in the purchase and sale or lease of the products in question.
  - (B) A person may be a regular dealer in such bulk items as petroleum products, steel, cement, gravel, stone, or asphalt without owning, operating, or maintaining a place of business as provided in this paragraph (e)(2)(ii) if the person both owns and operates distribution equipment for the products. Any supplementing of regular dealers' own distribution equipment shall be by a long-term lease agreement and not on an ad hoc or contract-by-contract basis.
  - (C) Packagers, brokers, manufacturers' representatives, or other persons who arrange or expedite transactions are not regular dealers within the meaning of this paragraph (e)(2).



- (3) With respect to materials or supplies purchased from a Targeted Business which is neither a manufacturer nor a regular dealer, count the entire amount of fees or commissions charged for assistance in the procurement of the materials and supplies, or fees or transportation charges for the delivery of materials or supplies required on a job site, toward Targeted Business goals, provided the fees are reasonable and not excessive as compared with fees customarily allowed for similar services. Do not count any portion of the cost of the materials and supplies themselves toward Targeted Business goals.
- (4) Fill, sand, gravel and other natural materials are counted at the supplier value of 60% regardless of pit of origin ownership.
- (5) A Targeted Business certified as a supplier may perform both the supplier role and broker role under the same contract. Items purchased directly and solely by the Targeted Business supplier from the manufacturer will be counted at the supplier value of 60%. Items purchased directly and solely by the Targeted Business supplier from another supplier or wholesaler will only count at the broker rate of 5%.
- (6) Specialized or "to specification" equipment purchases are presumptively considered only able to be brokered.
- (7) A Target Business certified as a broker may only count at the broker rate of 5% for all project work.
- (8) The determination between supplier and broker action may change for Targeted Business suppliers from bid/commitment phase to construction phase.
- (9) A Targeted Business contractor who performs the installation, connection, application of materials may only count the cost of supplies or materials towards the goal if:
  - (i) The Targeted Business purchases the supplies or materials itself and
  - (ii) Industry practice supports and is structured for the supplies or materials to be purchased by the Targeted Business, including without limitation structural steel, rebar, and paint and other coatings.



(f) Participation by a woman-owned Targeted Business certified as an MBE counts only toward the Targeted Business Goal for MBEs, even if the MBE is certified as a WBE or is able to be certified as a WBE.



## <u>APPENDIX B – Targeted Business Commercially Useful Function ("CUF")</u>

#### CUF Site Visits

- (a) An Authority representative will perform at least one site visit for each Targeted Business performing under a contract on the Project for the purpose of determining whether the Targeted Business is performing a "commercially useful function" ("CUF") as described in Appendix A.
- (b) The Authority will conduct at a minimum one site visit for each Targeted Business to determine that the Targeted Business is providing a CUF. The Authority will document each site visit and applicable support documentations.
- (c) CUF site visits are intended to be performed on the construction site or applicable place of business. In situations when it is not feasible to perform the CUF site visit on location, the Targeted Business will provide all documentation to demonstrate CUF compliance.
- (d) CUF site visits documentation includes photographs, brief interviews with staff, correspondence, and copies of business transactions.
- (e) The CUF site visit will be documented in a standardized form and retained by the Authority.
- (f) Data from the CUF form will be retained by the Authority for auditing purposes.

#### CUF Review

- (g) The Authority may conduct a CUF Review if the Authority has concerns whether a Targeted Business is performing a commercially useful function.
- (h) The Authority performs a CUF Review by collecting data, including site visit reports, photographs, and interview information. The Authority then analyzes the data and produces a CUF Finding which may, in the Authority's discretion, provide adjustments to whether and how the Targeted Business' participation is counted toward the Targeted Business Goals.



### APPENDIX C – Targeted Business Good Faith Efforts

- (a) The Construction Manager must make Good Faith Efforts to achieve the Equity Plan's Targeted Business Goal. Many of the Good Faith Efforts are described in Section 7, including without limitation the monitoring and reporting activities in the various phases in Section 7. This Appendix identifies a non-exclusive list of factors the Authority may consider when determining whether the Construction Manager has exercised Good Faith Efforts.
- (b) The Authority shall determine if the Construction Manager has made adequate Good Faith Efforts. If the Construction Manager fails to make adequate Good Faith Efforts, the consequences are discussed in Section 11.
- (c) Because the Construction Manager is not a Targeted Business, the Construction Manager's Good Faith Efforts to achieve the Targeted Business Goal necessarily must focus on obtaining participation of Targeted Business firms hired by the Construction Manager as Subcontractors or qualified joint venture partner.
- (d) The Construction Manager must make the good Good Faith Efforts described herein beginning with its solicitation of potential Subcontractors.
- (e) <u>Documentation</u>. The Authority has developed the forms described in Section 7 and may develop such other forms, affidavits, and other documentation the Authority deems appropriate for the Construction Manager to document its Good Faith Efforts to meet the Targeted Business Goal and allow the Authority to determine whether Good Faith Efforts have been made. The Construction Manager and its Subcontractors shall complete the forms the Authority requests them to complete.
  - (1) The Authority shall determine what information is required from the Construction Manager, proposed Subcontractors, or others as the Authority deems appropriate to evaluate the Construction Manager's Good Faith Efforts, and shall determine what reviews, examinations and assessments of information are appropriate for such evaluation.
  - (2) Before the Construction Manager enters into a contract with any Subcontractor, the Construction Manager must submit to the Authority any forms that the Authority deems appropriate.
  - (3) The Construction Manager must require its Subcontractors to engage in similar Good Faith Efforts as required by the Authority, and to similarly flow down their Good Faith Efforts requirements to lower tier Subcontractors. References in these



requirements to the Construction Manager shall also be deemed to refer to its Subcontractors and such lower tier Subcontractors with respect to their obligations to use Good Faith Efforts.

#### (f) <u>Removal of MBE or WBE</u>

- (1) The Construction Manager shall notify the Authority before the Construction Manager terminates a contract with a Targeted Business, changes the scope of the Targeted Business's contract, or otherwise releases a Targeted Business from performing work on the Project ("Removal Action").
- (2) The Authority will permit the Construction Manager to take a Removal Action when:
  - (i) The work committed to the Targeted Business was eliminated or adjusted via approved change order.
  - (ii) The Authority has verified that the Targeted Business is no longer in business or is unable to perform acceptable work under the contract.
  - (iii) Any resolved legal situation that impacts the ability for the Targeted Business to perform work.
  - (iv) The Targeted Business has defaulted on the terms of its Subcontract and can no longer perform the work as required.
- (3) To the extent a proper Removal Action is taken by the Construction Manager, the Construction Manager must make Good Faith Efforts as described herein to replace those firms with other Targeted Business firms, or otherwise increase Targeted Business participation to offset the loss of Targeted Business participation.
- (g) <u>Non-Exclusive List of Factors</u>. In addition to compliance with the various reporting and monitoring requirements described in this Equity Plan or developed by the Authority, the following is a list of non-exclusive factors the Authority may consider in making a determination whether the Construction Manager has made adequate Good Faith Efforts to meet the Targeted Business Goal:
  - (1) Soliciting through all reasonable and available means (attendance at pre-bid meetings, advertising and/or written notices) the interest of as many Targeted



Business firms as reasonably possible presently certified in the scopes of work of the contract.

- (2) Soliciting Targeted Business firms in reasonably sufficient time prior to bid opening or the proposal deadline to allow Targeted Business firms to respond to solicitations, or allowing sufficient time for Targeted Business firms to respond to solicitations prior to finalizing selections of Subcontractors. The Construction Manager must determine with certainty if the Targeted Business firms are interested by taking and documenting appropriate steps to follow up on initial solicitations.
- (3) Selecting portions of the work to be performed by Targeted Business firms in order to increase the likelihood that the participation goals will be achieved. This includes, where appropriate, breaking contract work into smaller units to facilitate Targeted Business participation.
- (4) Providing interested Targeted Business firms with adequate information about the requirements of the contract in a timely manner to assist them in responding to a solicitation.
- (5) Negotiating in good faith with interested Targeted Business firms and providing written documentation of such negotiation with each such business. In determining whether the Construction Manager negotiated in good faith, the Authority may consider a number of factors including price, scheduling and capabilities as well as the contract goal.
- (6) The fact that there may be some additional costs involved in finding and using Targeted Business firms is not itself sufficient reason for the Construction Manager's failure to meet the participation goals as long as such costs are reasonable.
- (7) If requested by a solicited Targeted Business firm, the Construction Manager must make reasonable efforts to assist such Targeted Business firms in obtaining financing, training, or insurance as may be appropriate for their work on the Project, provided that the Construction Manager need not provide financial assistance toward this effort.
- (8) Effectively using the services of minority/woman community organizations; minority/woman contractors' groups; local, state and federal business assistance offices; and other organizations as allowed on a case-by-case basis to provide assistance in the solicitation and placement of Targeted Business firms.



#### **APPENDIX D – Workforce Compliance and Counting**

Pursuant to Section 12.7 of the Equity Plan, the Authority may seek advice from the City of Minneapolis and the Minnesota Department of Human Rights for the purpose of establishing additional policies, procedures, and forms for the workforce program, including compliance and counting.

Until the Authority, in its sole discretion, adopts specific policies, procedures, and forms regarding how workforce participation by minorities and women is counted toward the Workforce Goal, the Authority will use the following guidelines:

- (a) The Workforce Goal is expressed as a percentage of the total hours worked by individuals performing eligible work ("Workforce Work") as part of the Construction Services Agreement, whether the workers are employed by the Construction Manager or Subcontractors of any tier;
- (b) Workforce Work means work performed on the Project job-site; and
- (c) Every hour of Workforce Work performed by an eligible minority or woman counts toward the Workforce Goal.
- (d) Every hour of Workforce Work performed by an eligible minority woman counts toward both the minority Workforce Goal and the female Workforce Goal.



#### <u>APPENDIX E – Workforce Good Faith Efforts</u>

- (a) The Construction Manager must make Good Faith Efforts to achieve the Equity Plan's Workforce Goal. Many of the Good Faith Efforts are described in Section 16, including without limitation cooperation with the monitoring and reporting activities. This Appendix identifies certain minimum requirement for Good Faith Efforts as well as a list of additional non-exclusive factors the Authority may consider when determining whether the Construction Manager has exercised Good Faith Efforts.
- (b) The Construction Manager is required to take the following steps as part of its Good Faith Efforts:
  - (1) Make a good faith effort to maintain a working environment free of harassment, intimidation, and coercion at all sites, and in all facilities at which the Construction Manager's employees are assigned to work. The Construction Manager must specifically ensure that all lead supervisors, superintendents, and other on-site supervisory personnel are aware of and carry out the Construction Manager's obligation to maintain such a working environment, with specific attention to minority or female persons working at such sites or in such facilities.
  - (2) Establish and maintains a current list of minority and female recruitment sources, provide written notification to minority and female recruitment sources and to community organizations when the contractor or its unions have employment opportunities available, and maintain a record of the organizations' responses.
  - (3) Utilize the Construction Works database to maintain a current file of the names, addresses, and telephone numbers of each minority and female off-the-street applicant and minority or female referral from a union, a recruitment source, or community organization and of what action was taken with respect to each person. If the person was sent to the union hiring hall for referral and was not referred back to the Construction Manager by the union or, if referred, not employed by the Construction Manager, this must be documented in the file with the reason therefore, along with whatever additional actions the Construction Manager may have taken.
  - (4) Provide immediate written notification to the Authority when the union or unions with which the Construction Manager has a collective bargaining agreement has not referred to the Construction Manager a minority person or woman sent by the Construction Manager, or when the Construction Manager has other information that the union referral process has impeded the Construction Manager's efforts to meet its obligations.



- (5) Develop on-the-job training opportunities and/or participate in and pay for training programs for the areas which expressly include minorities and women, including upgrading programs and apprenticeship and trainee programs relevant to the contractor's employment needs, especially those programs funded or approved by the state of Minnesota. The Construction Manager must provide notice of these programs to the sources compiled under Paragraph (b)(2).
- (6) Disseminate the Construction Manager's equal employment opportunity policy by providing notice of the policy to unions and training programs and requesting their cooperation in assisting the Construction Manager in meeting its equal employment opportunity obligations; by including it in any policy manual and collective bargaining agreement; by publicizing it in the company newspaper, annual report, etc.; by specific review of the policy with all management personnel and with all minority and female employees at least once a year; and by posting the company equal employment opportunity policy on bulletin boards accessible to all employees at each location where construction work is performed.
- (7) Review, at least annually, the company's equal employment opportunity policy and affirmative action obligations under these specifications with all employees having any responsibility for hiring, assignment, layoff, termination, or other employment decisions including specific review of these items with on-site supervisory personnel such as superintendents, general lead supervisors, etc., prior to the first day of construction work at any job site. A written record must be made and maintained identifying the time and place of these meetings, persons attending, subject matter discussed, and disposition of the subject matter.
- (8) Disseminate the Construction Manager's equal employment opportunity policy externally by including it in any advertising in the news media, specifically including minority and female news media, and providing written notification to and discussing the Construction Manager's equal employment opportunity policy with other contractors and Subcontractors with whom the Construction Manager does or anticipates doing business.
- (9) Direct its recruitment efforts, both oral and written, to minority, female, and community organizations, to schools with minority and female students, and to minority and female recruitment and training organizations serving the contractor's recruitment area and employment needs, including zip codes in the City of Minneapolis with high rates of poverty and unemployment. Not later than one month prior to the date for the acceptance of applications for apprenticeship or other training by any recruitment source the Construction Manager must send



written notification to organizations such as the above, describing the openings, screening procedures, and tests to be used in the selection process.

- (10) Encourage present minority and female employees to recruit other minority persons and women.
- (11) Ensure that seniority practices, job classifications, work assignments, and other personnel practices do not have a discriminatory effect by continually monitoring all personnel and employment related activities to ensure that the equal employment opportunity policy and the Construction Manager's obligations under these specifications are being carried out.
- (12) Ensure that all facilities and company activities are nonsegregated except that separate or single-user toilet and necessary changing facilities shall be provided to assure privacy.
- (13) Document and maintain a record of all solicitations of offers for subcontracts from minority and female construction contractors and suppliers, including circulation of solicitations to minority and female contractor associations and other business associations.
- (14) Conduct a review, at least annually, of all supervisors' adherence to and performance under the Construction Manager's equal employment opportunity policies and affirmative action obligations.
- (c) The following is a list of additional non-exclusive factors the Authority may consider when determining whether the Construction Manager has exercised good faith efforts:
  - (1) Timely submission of compliance review reports;
  - (2) Construction Manager's cooperation with on-site compliance reviews and audits;
  - (3) Construction Manager's compliance with making available records or other information as required by the Workforce Program;
  - (4) Participating in voluntary associations which assist in fulfilling one or more of their good faith obligations in Paragraphs (b)(1) to (b)(14). The efforts of a contractor association, joint contractor-union, contractor-community, or other similar group of which the Construction Manager is a member and participant, may be asserted as fulfilling any one or more of its obligations provided that the Construction Manager actively participates in the group, makes every effort to assure that the group has a positive impact on the employment of minorities and



women in the industry, ensures that the concrete benefits of the program are reflected in the Construction Manager's minority and female workforce participation, makes a good faith effort to meet its individual goals and timetables, and can provide access to documentation which demonstrates the effectiveness of actions taken on behalf of the Construction Manager. The obligation to comply, however, is the Construction Manager's and failure of such a group to fulfill an obligation must not be defense for the Construction Manager's noncompliance.

- (d) The Authority also may consider whether the Construction Manager takes prompt corrective action if and when it becomes aware that any of the following conditions exist with regard to its workforce:
  - (1) underutilization of women or minorities in any job group;
  - (2) minority or female employees move laterally, vertically, at a lesser rate than nonminority or male employees;
  - (3) a selection process eliminates minorities or women at higher rate than nonminority or male employees;
  - (4) pre-employment inquiries and application forms do not satisfy state law requirements;
  - (5) descriptions of jobs do not accurately reflect functions involved;
  - (6) selection procedures are not valid predictors of job performance;
  - (7) disproportionately high rejection of women or minorities by hiring supervisors;
  - (8) women, minorities, and disabled persons who are not participating in companysponsored activities;
  - (9) segregation still exists at some facilities;
  - (10) disparities by minority group status or sex in terms of length of service and type of job held;
  - (11) managers, supervisors, or employees lack interest in company equal employment opportunity policies;
  - (12) underrepresentation of women or minorities in training or career improvement programs;



- (13) techniques for evaluating effectiveness of its equal employment opportunity programs have not been established; and
- (14) inadequate display of equal employment opportunity posters.

Exhibit A



CONSTRUCTION SERVICES AGREEMENT EQUITY PLAN

# TARGETED BUSINESS COMMITMENT AND INFORMATION FORM

Proposer Company Name: \_\_\_\_\_

## Check ONE of the following:

- \_\_\_\_ No Targeted Business participation is committed on this project
- The following Targeted Business (MBE & WBE) participation is committed on this project:

<b>Firm Name</b> (Legal business name used for Targeted Business certification)	WBE (Chec	MBE k one)	How will firm participate? (subcontractor, consortium, joint venture)	Description of work	Estimated dollar value of participation	Estimated percentage of total bid
				Total WBE %	Total MBE %	

(Form continued on next page. Use copies of page 1 of this form if additional space is needed to list committed Targeted Businesses and attach such copies to the form.)



# CONSTRUCTION SERVICES AGREEMENT EQUITY PLAN TARGETED BUSINESS COMMITMENT AND INFORMATION FORM

#### On behalf of the Proposer identified below, I certify that:

(Check ONE of the following)

\_\_\_\_ No Targeted Business (MBE or WBE) participation is committed on this project.

Proposer is committed to use the Targeted Business contractor(s) listed in this form on this project at the stated percentage(s).

I further certify that I have read the Targeted Business requirements found in the Construction Services Agreement Equity

Plan. I am authorized on behalf of the Proposer to submit this certification to the Minnesota Sports Facilities Authority.

This certification is a material representation of fact on which the Authority may rely in awarding the contract.

Name: \_\_\_\_\_

Title: \_\_\_\_\_

Exhibit B



### CONSTRUCTION SERVICES AGREEMENT EQUITY PLAN TARGETED BUSINESS INFORMATION FORM

Check ONE of the following:

- \_\_\_\_ No Targeted Business will be used by Proposer on this project.
- Targeted Businesses are proposed to be used on this project.

The following is

- 1) a list of Targeted Businesses proposed to be used on the project AND
- 2) a list of Targeted Businesses who were considered by the Proposer for the project but were not selected by the Proposer:

#### 1) TARGETED BUSINESS PROPOSED TO BE USED ON THE PROJECT:

Firm Name	Address	Telephone Number

#### 2) TARGETED BUSINESS WHO WERE CONSIDERED BUT WERE NOT SELECTED:

Firm Name	Address	Telephone Number

(Form continued on next page. Use copies of page 1 of this form if space is needed to list additional Targeted Businesses and attach such copies to the form.)



## CERTIFICATION

On behalf of the Proposer identified below, I certify that the information provided in this form is true and correct.

Proposer Name:	
By:	Date:
Name:	
Title:	

END OF DOCUMENT

Exhibit C

# **CSA Equity Plan Progress Report\***

1. Project #:		9. Original Contract Amount:		
2. Project Title: 1		10. Change Orders To Date:		
3. Construction Manager: 1		11. Revised Contract Amount:	\$0.00	
4. Type of Services:		12. Total Earned To Date:		
5. Contract #:		13. Contract Dollars Remaining:	\$0.00	
6. Contract Award Date:		14. Amount Paid to Date:		
7. Payment Claim #:		15. Percent Paid to Date:		
8. Progress Report #:				

PROJECT GOAL 11% WBE 9	9% MBE   CO	NTRACT CO	MITMENT	% WBE% MBE		Inter	im Report	
Reporting Period: From			to			Final Report		
A) W/MBE Subcont (SEE INSTRUCTIO		C) Non- f W/MBE Sub- contractor	D) \$ Amount to Non- W/MBE	E) \$ Amount in Change Orders to W/MBE Contract	F) Total W/MBE Contract \$\$ Amount	G) \$\$ Amount to be paid to W/MBE	H) \$\$ Amount Paid To W/MBE	I) W/MBE%
					\$0.00			#DIV/0!
					\$0.00			#DIV/0!
					\$0.00			#DIV/0!
					\$0.00			#DIV/0!
					\$0.00			#DIV/0!
					\$0.00			#DIV/0!
	\$ -		\$-	\$-	\$-	\$-	\$ -	
		Т	otal W/MBE	Contract \$\$ Amount as % of Total	Contract An	nount (#11):	#DIV/0!	
	\$	\$ Amount Pa	id to W/MBE	To Date as % of Total Contract Am	nount Paid To	o Date (#14)	#DIV/0!	#DIV/0!
Explanation if W/MBE Goal N	lot Being Met or Other Com	ments:					Total W/M	IBE %
Signature: Title:				Date:				

**INSTRUCTIONS:** 

1. List each W/MBE Subcontractor only once. Insert appropriate information in Columns A), B), E), G), and H).

2. Complete Columns C) and D) only for each non-W/MBE Subcontractor hired by the W/MBE Subcontractor.

3. Column D) shall include all W/MBE Change Order amounts passed along to Non-W/MBE.

Authority Representative Signature

Date

\*Note that formatting may change.

#### EXHIBIT 8 CONSTRUCTION MANAGER PARENT OR JOINT VENTURE COMPANY GUARANTY

The following Construction Manager Parent or Joint Venture Company Guaranty is included in this Construction Services Agreement as **Exhibit 8**:

Reference is made to that certain Construction Services Agreement dated \_\_\_\_\_\_, as amended and modified from time to time ("Construction Services Agreement"), between \_\_\_\_\_\_"), between \_\_\_\_\_") and the Minnesota Sports Facilities Authority ("Authority"), which Construction Services Agreement provides for the construction of a professional sports Stadium and Stadium Infrastructure in Minneapolis, Minnesota. \_\_\_\_\_\_\_ is an Affiliate or Joint Venture Partner of \_\_\_\_\_\_ ("Guarantor"), and Guarantor has an interest in the completion of the Construction Services Agreement.

In consideration of the Construction Services Agreement, and such other and further good and valuable consideration, receipt of which Guarantor acknowledges, Guarantor hereby undertakes and guarantees to the Authority that \_\_\_\_\_\_ will fulfill its obligations to the Authority under the Construction Services Agreement and, if \_\_\_\_\_\_ does not do so, Guarantor hereby declares that it will perform those obligations as though they were its own, it being explicitly understood and agreed, however, that, notwithstanding any other provision contained in this Guarantee or at law or otherwise, under no circumstances shall Guarantor's liability or obligations hereunder exceed or be different from the liability or obligations of \_\_\_\_\_\_ under the Construction Services Agreement and that Guarantor's liability hereunder shall only become effective if, and to the extent that \_\_\_\_\_\_\_ has not fulfilled its obligations under the Construction Services Agreement.

Guarantor agrees that its liability under this Guarantee shall not be affected by any amendment, modification (including but not limited to Contract Revisions or Change Orders), extension or waiver of any of the terms of the Construction Services Agreement and that Guarantor's liability hereunder shall apply to the Construction Services Agreement as so amended, modified, or extended.

Guarantor agrees that its obligations under this Guarantee are irrevocable, independent and continuing, subject to the conditions and limitations stated herein and in the Construction Services Agreement. Subject to the provisions hereof, Guarantor shall satisfy its obligations hereunder in a timely manner in accordance with the obligations of \_\_\_\_\_\_ under the Construction Services Agreement after demand therefor is properly made in writing by the Authority, and such demand shall be conclusively deemed to have been effectually made if made, with necessary modification, in accordance with the notice provisions contained herein.

This Guarantee shall inure to the benefit of the Authority, its successors, and assigns and shall be binding upon Guarantor and its permitted successors and assigns. Nothing in this Guarantee, whether expressed or implied, is intended to confer upon any person other than the Authority and Guarantor, and their permitted successors and assigns, any rights or remedies of any nature or kind whatsoever. Guarantor shall not assign this Guarantee or its rights or obligations hereunder, whether by operation of law or otherwise, without the Authority's prior written consent.

Any notice, demand, document or other communication required or permitted to be given herein shall be in writing and shall be sufficiently given if delivered to the applicable Party at its address below, or if sent by international express mail or facsimile, addressed as follows:

To Authority:

Attention:	
Facsimile:	

With a copy to:

Attention:\_\_\_\_\_ Facsimile:\_\_\_\_\_

To Guarantor:

Attention:\_\_\_\_\_ Facsimile:\_\_\_\_\_

With a copy to:

Attention:\_\_\_\_\_ Facsimile:\_\_\_\_\_

Any such notice, if mailed, shall be deemed to have been given on the third day following such mailing or, if delivered, shall be effective on the day of delivery. Any notice sent by facsimile shall be deemed to have been given on the business day next following the date of transmission. For the purposes hereof, a "business day" shall be a day other than Saturday or Sunday and which is neither a statutory nor civil holiday in the state or province or country of the addressee. Each of the Authority and Guarantor shall be entitled to specify a different address by giving notice in accordance with the terms hereof to the other.

This Guarantee shall be governed by and construed in accordance with the same laws and definitions of terms that are applicable to the Construction Services Agreement. For the purpose of this Guarantee, but for no further purpose, Guarantor hereby agrees to dispute resolution as more specifically provided for by the Construction Services Agreement.

By:\_\_\_\_\_\_ Its:\_\_\_\_\_ Date:\_\_\_\_\_\_

MINNESOTA SPORTS FACILITIES AUTHORITY

By:			-
Its:			
Date:			-

#### MINNESOTA SPORTS FACILITIES AUTHORITY

By:			
Its:			
Date:			

#### EXHIBIT 9 CONSTRUCTION MANAGER'S BONDS

The Payment and Performance Bond forms that the Construction Manager is required to provide are included in this Construction Services Agreement as  $\underline{Exhibit 9}$ :

[INSERT]

#### EXHIBIT 10 CONSTRUCTION MANAGER'S WARRANTY

The following Construction Manager's Warranty is included in this Construction Services Agreement as **Exhibit 10**:

#### CONSTRUCTION MANAGER'S WARRANTY OF CONSTRUCTION

Pursuant to the Construction Services Agreement between the Minnesota Sports Facilities Authority ("Authority") and [INSERT] ("Construction Manager"), Construction Manager hereby warrants and guarantees that all of the Work performed under the Construction Services Agreement will be of new and of good quality, will be free of defects except for those inherent in the quality of the Work allowed by the Contract Documents, and will conform to the requirements of the Contract Documents. If the Work does not conform to this warranty, it shall be considered defective, and Construction Manager shall remedy at its own expense any such defective Work (including the costs that the Authority or Architect incur in dealing with or as a result of the defective Work) so that the Work conforms to the Contract Documents. The Construction Manager's warranty shall extend for a period of two (2) year after final acceptance by Authority. Where guarantees or warranties are required in the Contract Documents for a period of more than two (2) years, such longer terms shall apply. All Suppliers' warranties and guarantees, express or implied, respecting any part of the Work and any materials used therein are hereby assigned by the Construction Manager to the Authority. This Warranty of Construction shall supplement, and not supersede, warranties and guarantees given by Construction Manager under the terms of the Contract Documents.

	CONSTRUCTION MANAGER: [INSERT]
WITNESS:	
	Title:
	Date:
THE STATE OF	
COUNTY OF	

GIVEN UNDER MY HAND AND SEAL OF OFFICE this \_\_\_\_\_day of \_\_\_\_\_\_, 20\_\_\_.

NOTARY PUBLIC

SEAL

MY TERM EXPIRES

### EXHIBIT 11 CONTRACT REVISION AND CHANGE ORDER FORMS

Forms that Construction Manager must use to request Contract Revisions and Change Orders will be added to the Construction Services Agreement in this <u>Exhibit 11.</u>

### EXHIBIT 12 ENVIRONMENTAL AND OTHER RESPONSIBILITIES DURING CONSTRUCTION

The Construction Manager is responsible to administer and perform the Work in accordance with the Environmental and Other Responsibilities During Construction as outlined in this **Exhibit 12**:

[INSERT]

#### EXHIBIT 13 MINNESOTA MULTIPURPOSE STADIUM MINNEAPOLIS, MINNESOTA NOTES OF CLARIFICATION CONSTRUCTION SERVICES AGREEMENT

The following section of the Notes of Clarification contains information which describes the format and content of the Cost Estimate to be prepared by the Construction Manager.

- A. Base Estimates and Alternates
- B. Divisional Breakdown / Schedule of Values Format
- C. Detailed Notes To The Budget

#### A. BASE ESTIMATES AND ALTERNATES

The Conceptual Design Package includes two different bowl structures, using either concrete or steel. The base estimates should assume a concrete bowl, with alternate pricing for using a steel structure. The Conceptual Design Package also includes three different approaches to the exterior enclosure of the Stadium, with multiple variations of each approach with respect to cladding and operability of the roof and window walls. The Preliminary Construction Estimate must include three complete baseline estimates and alternate pricing, as follows:

- 1. Scheme 1 Baseline Estimate: Roof structure of Scheme 1.1, all clear ETFE roof material and solid curtainwall. Provide alternates for the following:
  - a. Operable roof structures 1.2
  - b. Operable roof structure 1.3
  - c. Unit cost pricing for using fritted ETFE, PTFE and metal deck roof material,
  - d. Sliding operable walls
  - e. Pivoting operable walls
- 2. Scheme 2 Baseline Estimate: Roof structure of Scheme 2.1, all clear ETFE roof material and solid curtainwall. Provide alternate pricing for the following:
  - a. Operable roof structure 2.2
  - b. Unit cost pricing for using fritted ETFE, PTFE and metal deck roof material
  - c. Sliding operable walls as shown on page 3.95
  - d. Pivoting operable walls as shown on page 3.95
- 3. Scheme 3 Baseline Estimate: Roof structure of Scheme 3.1, all clear ETFE roof material and solid curtainwall. Provide alternate pricing for the following:
  - a. Operable roof structure 3.21
  - b. Unit cost pricing for using fritted ETFE, PTFE and metal deck roof material
  - c. Sliding operable window walls on both ends of the stadium
- 4. For each scheme, identify the value of all Construction Manager and Subcontractor costs for Liability Insurance, Worker's Compensation Insurance and Bonds.

#### **B.** DIVISIONAL BREAKDOWN AND SCHEDULE OF VALUES

Each item below is a line item on the Schedule of Values to be used by the Construction Manager. All estimates and resulting Schedules of Values, as well as applications for payments, are to follow the following organization.

01 00 00	General Requirements
02 00 00	Existing Conditions

0.0 0.0 0.0	
03 00 00	Concrete
04 00 00	Masonry
05 00 00	Metals
06 00 00	Wood, Plastics & Composites
07 00 00	Thermal & Moisture Protection
08 00 00	Openings
09 00 00	Finishes
10 00 00	Specialties
11 00 00	Equipment
12 00 00	Furnishings
13 00 00	Special Construction
14 00 00	Conveying Equipment
21 00 00	Fire Suppression
22 00 00	Plumbing
23 00 00	Heating Ventilating and Air Conditioning
26 00 00	Electrical
27 00 00	Communications
28 00 00	Electronic Safety and Security
31 00 00	Earthwork
32 00 00	Exterior Improvements
33 00 00	Utilities
	Construction Manager's Fee & Staffing (Preconstruction)
	Construction Manager's Fee & Staffing (Construction)
	Performance & Payment Bonds (CM and Subcontractors)
	Insurance
	Building Permit
	Construction Contingency

# C. DETAILED NOTES TO THE BUDGET IN ADDITION TO THOSE INCLUDED WITHIN ADDENDUM 4 DATED 1/17/13 TO THE CM RFP.

#### 01 00 00 General Requirements

This is the cost of performing the work for the Construction Manager that is not covered by other divisions. The Construction Manager's staffing costs and fees are not included in General Requirements. Each subcontractor is required by the Construction Manager to cover as much of these types of costs as possible, but some are picked up by the Construction Manager. Costs may include the following: Fencing, Temporary Partitions, Man Lifts, Clean-up, Trash Disposal, Safety, Snow Removal etc. See the Construction Services Agreement for additional information. The following line items must be included in the General Requirements.

- 1. Two (2) year warranty on all work.
- 2. Project is sales tax exempt.
- 3. Safety program to be approved by Authority and Authority's Risk Management Consultant Project Manager.
- 4. Temporary enclosure weather protection, temporary heating.
- 5. Manage and Coordinate Food Service Installation.
- 6. Manage and Coordinate fixed seating installation.
- 7. Manage and Coordinate furniture installation.
- 8. Manage and coordinate scoreboard and scoreboard control room installation.
- 9. Coordinate with "Trade Contractors" performing scoreboard and sound installations.
- 10. Manage and Coordinate work of Trade Contractors, utility companies, authority staff and Team staff and other Project Participants as it relates to the work of the project.

#### 02 00 00 Existing Conditions

Existing Conditions work includes: clearing, site demolition, building demolition and selective demolition. The following work must be included in Existing Conditions:

- 1. Demolition of the existing H.H.H. Metrodome.
- 2. Site clearing and demolition of all areas included within the red line on Page 2.14.
- 3. Temporary site logistics to facilitate events at the Metrodome until the final game of the 2013 NFL season.
- 4. Include shoring and retaining walls as necessary.
- 5. Removal of two 1000 gallon fuel storage tanks and associated piping.

#### 03 00 00 Concrete

Concrete includes concrete foundations, cast-in-place concrete as well as pre-cast concrete. The following line items must be included in Concrete.

- 1. Base estimates should include a cast in place concrete seating bowl with precast treads and risers.
- 2. Include concrete caisson foundation.
- 3. Fill for metal pan stairs.
- 4. Concrete steps at aisles.
- 5. Precast concrete vomitories in seating areas.

#### 04 00 00 Masonry

Masonry is to include brick, block and stone. The following line items must be included in Masonry:

1. Include block walls for all field level areas as well as public concourse areas.

#### 05 00 00 Metals

Metals are to include structural steel, ornamental steel, miscellaneous metals, metal fabrications as well as expansion joint covers. The following line items must be included in Metals:

- 1. Steel roof structure(s).
- 2. Metal pan stairs.
- 3. Handrails and railings.
- 4. Expansion joints and expansion joint covers.
- 5. Catwalk and ladder system within scoreboard and roof structure.
- 6. Connection details to be designed by Thornton Tomasetti (TT). Some done within Tekla model. Some minor detailing required to be designed by Construction Manager for TT's approval.

#### 06 00 00 Wood, Plastics & Composites

Wood, Plastics & Composites is to include rough carpentry, finish carpentry, wood trim, millwork and cabinetry. The following line items must be included in Woods, Plastics and Composites:

- 1. Blocking for television brackets, toilet accessories, signage, artwork, etc. on entire project.
- 2. Wood trim in club and suite areas.
- 3. Lockers and millwork.

#### 07 00 00 Thermal & Moisture Protection

Thermal and Moisture Protection is to include roofing systems, waterproof and traffic coatings, sprayed on insulation and fireproofing. The following line items must be included in Thermal & Moisture Protection:

1. Fully adhered thermoplastic polyolefin (TPO) roofing on all roofs (open to above and/or within unenclosed spaces).

- 2. Exterior balconies to utilize double waterproofing "sandwich slab". Structural slab to be covered with membrane and insulation. Topping slab over membrane.
- 3. Install waterproof coating system (i.e. Sonoguard Self-Priming Base Coat etc.) at seating bowl and concourses.
- 4. All exterior partition walls to be insulated with vapor barrier.
- 5. Spray-on insulation at structure over and/or under exterior open areas.
- 6. Sub-roofs.
- 7. Metal panels.

#### 08 00 00 Openings

Openings includes wood and metal doors and frames, windows, lites, glazing, storefront, operable grills, roll-up doors and grills, curtainwall and associated hardware. The following line items must be included in Openings:

- 1. Hollow metal doors and welded frames (not "knock-down") in utility areas.
- 2. Manually operable roll-up doors at concession stands.
- 3. Exterior curtain wall system to be insulated vision glass with low E, high performance color tinting.
- 4. Glass doors at entrances to Indoor Clubs.
- 5. Master keying system.
- 6. Removable glass system at all suites.
- 7. Ticket windows at ticket office with audio system.

### 09 00 00 Finishes

Finishes includes all of the traditional division 09000 disciplines and trades. These include drywall, acoustical ceilings, paint, floor and wall coverings. The following line items must be included in Finishes:

- 1. "Utility" areas to have painted CMU walls, sealed concrete floors.
- 2. Food service areas to have painted drywall ceilings or mylar facing tiles in drop ceiling, frp walls, and stainless steel wall covering at grilles and fryers as required by Governmental Authorities.
- 3. Food preparation areas (kitchens) to have tile flooring as required by Governmental Authorities.
- 4. Club and suite areas to have upgraded finishes.
- 5. Exclude finishes for team store.
- 6. Exclude finishes for hall of fame.
- 7. Exclude finishes for restaurant.
- 8. Acoustical treatment for the seating bowl.

#### 10 00 00 Specialties

Specialties include traditional carpentry specialties including: lockers, brackets, flagpoles, toilet partition and accessories etc. The following line items must be included in Specialties:

- 1. Three flag poles with internal halyards to be mounted adjacent to external plaza.
- 2. Suspended metal toilet partitions with lockable doors and urinal screens in all restrooms .
- 3. Most toilet accessories to be furnished and installed by Construction Manager (grab bars, mirrors, drink shelf, feminine napkin sales, purse shelf, and disposal etc.).
- 4. All outside corners of walls to have corner guards.
- 5. Toilet paper and paper towel dispensers to be furnished by the Authority and installed by the Construction Manager.
- 6. Signage by others.

#### 11 00 00 Equipment

Equipment is for built-in and loose equipment supplied and installed by the Construction Manager including: refrigerators, televisions, food service equipment etc. See FF&E matrix. The following line items must be included in Equipment:

- 1. Include supply and installation of Televisions, Television Brackets and any required blocking.
- 2. Concession counters to be stainless steel.
- 3. Exclude supply and installation of Food service equipment.
- 4. Trash and recycling chutes.
- 5. Laundry Equipment
- 6. Safety tiebacks for window washing.
- 7. Parking control equipment
- 8. Loading dock equipment
- 9. Residential appliances
- 10. Athletic Equipment
- 11. Defibrillators
- 12. Exclude x-ray equipment

#### 12 00 00 Furnishing

Furnishings are manufactured products such as permanent seating. The following line items must be included in Furnishings:

- 1. 19" plastic self rising seat at regular seating areas.
- 2. 21" padded self rising seats at clubs.22" fully upholstered seating at suites.
- 3. Cupholders for all seating
- 4. Include supply and installation of signage
- 5. Include half house curtain
- 6. Include recessed aluminum floor mats at all entrances

#### 13 00 00 Special Construction

This category is for work not included in divisions above such as athletic surfaces. The following line item must be included in Special Construction:

1. None

#### 14 00 00 Conveying Systems

Elevators, escalators and dumb waiters. The following line items must be included in Conveying Systems:

- 1. Sixteen (16) 5,000 lb passenger elevators.
- 2. Two (2) 12,000 lb freight elevator.
- 3. Twenty (20) escalators.
- 4. Speakers and traveling cable in elevator cabs.
- 5. Capacity (traveling cable) for televisions in elevator cabs.

#### 21 00 00 Fire Suppression

Dry and wet fire suppression sprinkler systems. The following work is to be included in Fire Suppression:

- 1. Fire suppression sprinkler distribution and heads in all occupied spaces as required by Governmental Authorities (wet system utilized).
- 2. Connect food service equipment rough-in and final connect.
- 3. Connections required to fire alarm system.
- 4. Hangers, supports, insulation and hangers to be in compliance with SMACNA standards.
- 5. Hangers and supports to be in compliance with seismic standards for Zone 4.
- 6. Firestopping of all penetrations to meet UL rating of penetrated assembly.
- 7. Include Ansul systems as required in cooking areas.

#### 22 00 00 Plumbing

The following work must be included in Plumbing:

- 1. Waste, vent and supply piping.
- 2. Automatic flush valves at all fixtures.

- 3. Water closets.
- 4. Urinals.
- 5. Lavatories and bar sinks.
- 6. Janitors' sinks.
- 7. Assume use of district heating and cooling system.
- 8. Seating bowl drainage system.
- 9. Toilets and food preparation areas (including concessions) are to have hot and cold running water.
- 10. Electric water coolers.
- 11. Natural gas distribution for cooking.
- 12. Connect food service equipment rough-in and final connect.
- 13. Provide carbon dioxide distribution system to serve all concession stands, bars and pantries etc.
- 14. Provide beer distribution system to serve all concession stands, bars and pantries etc.
- 15. Beer Conduit to connect beer coolers to bars and concession stands.
- 16. Soft drink distribution systems to be from "bag and box" locations near or within concessions, bars, pantries etc. on all levels.
- 17. Supports, insulation and hangers to be in compliance with SMACNA standards.
- 18. Hangers and supports to be in compliance with seismic standards for Zone 4.
- 19. Firestopping of all penetrations to meet UL rating of penetrated assembly.
- 20. Include grease interceptors.

#### 23 00 00 Heating Ventilation and Air Conditioning (HVAC)

The following line items must be included in Heating Ventilation and Air Conditioning:

- 1. Heating Ventilation and Air Conditioning (HVAC) System to provide heating and cooling for peak loads for ASHRAE 0.5% values for Minneapolis, MN.
- 2. Duct work, hangers, supports, insulation and hangers to be in compliance with SMACNA standards.
- 3. Hangers and supports to be in compliance with seismic standards for Zone 4.
- 4. All interior spaces are to be heated and cooled. Include a 4 pipe system fed from the district heating and cooling system.
- 5. Smoke management system. (Interlocking with fire alarm system and door controls required.)
- 6. Connect food service equipment rough-in and final connect.
- 7. Provide grease scrubbers for kitchens and cooking concession stands.
- 8. All HVAC equipment shall be controlled by an Energy Management and Control System (EMCS Temperature Control).
- 9. Firestopping of all penetrations to meet UL rating of penetrated assembly.
- 10. Includes hoods and exhaust duct for cooking areas.
- 11. Include a humidity control system for the team areas at the field level.
- 12. Include carbon monoxide detection system at interior parking areas.

#### 26 00 00 Electrical

Distribution, power, lighting, lightning protection etc. The following line items must be included in Electrical:

- 1. Requirements for power distribution (primary cabling and transformation) and new redundant electrical service.
- 2. Include raceways, conduits, cable trays, buss ducts, duct banks junction boxes etc. as required for all electrical work (20-26-000, 20-27-000, 20-28-000, EMCS etc.) in compliance with NEC and to the requirements of Governmental Authorities.
- 3. Include all cable, wire, connections and terminations for the following work. (Cable, wire connections and terminations for 20-27-000, 20-28-000 and EMCS to be by others. Include pull string and labeling in all raceway and conduit installed for others.)

- 4. Include all lighting systems (including emergency light as required by Governmental Authorities).
- 5. Include exterior building lighting systems.
- 6. Include temporary lighting and power connections for project. Utility cost to be paid by the Authority.
- 7. Connect food service equipment rough-in and final connect. Provide disconnects as required.
- 8. Connect power for HVAC fans, pumps and equipment etc.– rough-in and final connect. Provide disconnects as required.
- 9. Connect power to equipment and electrical signage installed by the Construction Manager, Trade Contractors, Authority and Team.
- 10. Power outlets at each Point of Sale (POS) outlet.
- 11. Provide heat trace for all drain and supply pipes and lines open to the environment (not in tempered or climate controlled spaces).
- 12. Provide convenience outlets for cleaning and maintenance operations. Distribution, voltage and quantity of outlets to be similar to similar portions of the existing building.
- 13. Raceways for television (network) camera positions.
- 14. Electrical connections for smoke management system.
- 15. Lighting control system.
- 16. Lightning protection and grounding system. Ground bus risers and a building perimeter ground loop shall be provided. Completed system to be tested and certified by independent testing agency.
- 17. Firestopping of all penetrations to meet UL rating of penetrated assembly.
- 18. Hangers and supports to be in compliance with seismic standards for Zone 4.
- 19. Include sports lighting for football and baseball.

#### 27 00 00 Communications

This category includes electronic systems for distributed television, closed circuit television (CCTV) and sound - including public address (PA) systems. The following line items must be included in Communications:

- 1. Exclude scoreboards, ribbon boards and video control room.
- 2. Include all cable, wire, devices, connections and terminations for the following work. (Raceways to be provided by 20-26-000 Contractor).
- 3. New telephone and data systems. Includes the accommodation of point of sale (POS) system in concession stands. POS to include "home-runs" from each outlet to data closet.
- 4. Optical fiber back-bone.
- 5. Include distributed sound system.
- 6. Include broadcast television and radio infrastructure and cabling. Truck parking and uplink areas to include power connectivity as necessary.
- 7. Include provisions for Skycam and Cable cam systems.
- 8. Distributed Audio System
- 9. MATV
- 10. Broadcast Distribution and Camera Locations
- 11. Communication Systems/Information Management
- 12. Security and Surveillance
- 13. Include Hi density wifi and neutral host Distributed Antenna System for all areas of the building.
- 14. Include IPTV system for entire building.

#### 28 00 00 Electronic Safety and Security

Electronic systems such as fire alarm, surveillance and access control. The following line items must be included in Electronic Safety and Security:

- 1. Include all cable, wire, devices, connections and terminations for the following work. (Raceways to be provided by 20-26-000 Contractor).
- 2. Addressable fire alarm system.
- 3. Security system (electric strikes, cameras and intercoms).

#### 31 00 00 Earthwork

Site work includes: clearing, excavation and retention system. The following line items must be included in Earthwork:

- 1. Foundation excavation as required for expansion.
- 2. Backfill to fill Metrodome excavation as required.

#### **32 00 00** Exterior Improvements

Exterior Improvements work includes: paving, hardscaping and landscaping. The following line items must be included in Exterior Improvements:

- 1. New hardscaping and landscaping depicted within the red site boundary on page 2.14 of the Conceptual Design Package dated 12/10/12.
- 2. Precast bollards and barriers.
- 3. Concrete paving.
- 4. Asphalt paving.
- 5. Exclude exterior public art.
- 6. Exclude Parking Garages and Skyways.

#### 33 00 00 Utilities

Utilities are work five feet from the building's foundation to, and including, connection to utility providers. The following line items must be included in Utilities:

- 1. Gas service.
- 2. Sanitary sewer service.
- 3. Storm sewer service, main and structures.
- 4. Water main hydrant and hydrant lead.
- 5. Extension and connection to existing district heating and cooling system.

#### **Construction Manager Fee & Staffing**

Construction Manager fee is to include all home office expenses. Staffing costs are only for staff assigned to the project site. See Cost of Work matrix in Appendix A for details. The following line items must be included separately in Construction Manager Fee & Staffing:

- Construction Manager's fee for Preconstruction
   Construction Manager's staffing cost for Preconstruction
   Construction Manager's fee for project.
- 4. Construction Manager's staffing cost for project.

#### Performance & Payment Bond

The Construction Manager is required to provide a performance and payment bond on their work. The following line item must be included in Performance & Payment Bond:

1. Construction Manager to provide performance and payment bonds.

### Insurance

The Construction Manager is required to include costs for Liability Insurance and Workers Compensation insurance for the Construction Manager and all of its subcontractors. The following line items must be included within Insurance:

- 1. Construction Manager liability insurance.
- 2. Construction Manager worker's compensation insurance.
- 3. Subcontractor liability insurance
- 4. Subcontractor worker's compensation insurance.
- 5. Exclude Builder's Risk insurance.

#### **Building Permit**

The Construction Manager is required to include costs for all permits required of the Construction Manager and its subcontractors for the Project. The following line items must be included within Building Permit:

- 1. Building Permit Fees
- 2. Plan Check Fees
- 3. Trailer Permits
- 4. Tap Fees
- 5. Zoning Fees
- 6. Parking Fees
- 7. Jobsite Signage Fees
- 8. Temporary Lane Closure Permits

#### **Construction Contingency**

Construction Contingency is to cover foreseen, as well as unforeseen, costs the Construction Manager incurs in the process of constructing the project. This includes acceleration of the work to ensure on time completion, as well as the completion of contract documents by the Architect. The Construction Manager cannot spend Construction Contingency without prior written approval from the Authority. The following line items must be included in the Construction Contingency:

1. Construction Contingency pursuant to **Exhibit 3** of the Construction Services Agreement.

#### EXHIBIT 14 PROJECT COST IDENTIFICATION MATRIX

Description	Basic Fee	General Conditions	Cost	A/E Cost	Authority Cost	N/
A - Construction Services		34 <sup>1</sup>		3° - 33		
Description (See rate schedule attached)	Basic Fee	General Conditions	Direct	A/E Cost	Authority	N/
Corporate Executives	X					
Principle in Charge	-	\$1,269,080		12 0		
Operations Manager	8	\$100,000		2		-
Project Executive		\$1,102,926		1.3		_
Sr. Project Manager	16 16	\$2,404,370		8 - 6		_
Project Manager		\$3,123,648				
Assistant Project Manager		\$254,716		12 0		
General Superintendent	- 1	\$974,908		12		
Superintendent		\$2,361,659		18		
Superintendent		Included in A.9		0 0		
Superintendent		included in A.9				
Superintendent		Included in A.9				_
Superintendent	- 1	Included in A.9		12		
Assistant Superintendent		Included in A.9		14 3		_
Assistant Superintendent		Included in A.9		5 - 6		
Senior Project Engineer		Included in A.17				
Project Engineer		\$2,863,451		- 0		
Project Engineer	3 a 3 a	Included in A.17		14 3		
Project Engineer	1.0	Included in A.17		- K		
Project Engineer	1 E E	Included in A.17		- S		
Project Engineer (3-D modelling) (includes all VDC/BIM)		\$1,024,215				
Assistant Engineer		Included in A.17		- 0		-
Field Engineer (Site Layout)		Included in A.17		14		
QA/QC Manager		\$559,721		16.00		
QA/QC Engineer		\$448,607		15 0		
Safety Officer / Manager		\$540,673				
Safety Engineer	- 1 K	\$366,111		- Q ()		
M.E.P. Coordinator		\$722,688		- 12		-
Sports Consultant		Included in A.17				
Permit Coordinator		Included in A.17		12 0		
Secretarial (On Site) (Include admin support and office mgr)		\$702,970				
E.E.O. Officer		\$548,351		- Q ()		2
Scheduling		\$470,842		12 0		-
Project Phasing		Same and the second		- 10		
Project Budget Estimating		Included in Precon Fee		12 0		
Estimating		Included in Precon Fee		- 8 6j		
Chief / Lead Estimator		Included in Precon Fee				
Assistant Estimator		Included in Precon Fee		12 0		
Assistant Estimator	- 1 - E	Included in Precon Fee				
Mechanical / Plumbing Estimator		Included in Precon Fee		10 3		
Electrical Estimator		Included in Precon Fee		-10		
Cost Engineer(s)		\$613,290				
Purchasing				-12 O		
Value Engineering	- 1 A	\$110,668				
Life-cycle Analysis		Contraction of the				
Production Engineering	10			- 80 - 80		
Punch List Manager						

## Exhibit 14

The Peoples Stadium

Exhibit	14

Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
48 Building Commissioning Manager	2	\$292,556				-
49 Legal Services Allowance		\$100,000		1 - V		
50 Accounting		\$765,866		4 8		
51 System Development				2 <u>8</u>		
52 Drafting Detailer						
53 Secretarial (Home Office)	2 2	Second and a second		8		
54 Off-site Staff Travel Costs		\$207,500		1		
55 Off-site Staff Transportation		\$6,534		3		-
56 Project Staff Moving/Subsistence		\$75,000		P		8
57 Community Outreach Expenses (Includes)	directors/coordinator)	\$75,000				
58 Professional Services Agreements		\$10,000				
CONSTRUCTION SUB-TOTAL		\$ 22,095,351				
Pre-Construction			)			
Pre-Construction Executive	2 2		\$1,000,000	<u> </u>		_
2 Pre-Construction Manager			In line above	<u> </u>		
3 Project Manager			In line above	9		
4 Estimator		- <u>3</u>	in line above	<u>(</u> )		
<sup>6</sup> Estimator			in line above	2		
6 Estimator	2 3		in line above			
7 Estimator			in line above	1		
* Purchasing			in line above	8		
<sup>9</sup> Purchasing			in line above	<u> </u>		
10 Clerical			in line above	1		
11 Preconstruction Services Reimbursible Ex	benses	-	In line above			-
PRE-CONSTRUCTION SUB-TOTAL		\$ -	\$ 1,000,000			
TOTAL I Category A		\$ 22,095,351				

The Peoples Stadium

Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
B - Safety, Security, and Services						
Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
Safety Equipment		\$195,000	10000	12	()	
First Aid Supplies		included in B.01 above				
Handralis & Toeboards	0	Q	X	- 2		
Opening Protection		in the second	X	8 8		
Fire Extinguishers / Fire Watch		\$37,340		- S - S	1 2	
Security Guards / Flagman	Ű.	\$532,480		3	8 8	
Weekly Cleanup			X			
Final Clean (Interior/exterior)	<u>0</u>	and and and a second	X	- 2	2	
Temporary Fencing		\$315,054	- 0.1%.	- S - S	. S	24.0
Covered Walkways		S. Strategy	201	- 8		X
Safety Nets & Apparatus'			X	- 23 - 13	6	
Debris Haul / Removal		\$710,375				
Traffic Control		\$200,000		- 2	S	
Roadway Maintenance		00051061-63	X	- Q 1	. S	
Dust Control (all)		a nor received a	X	- A	1	
2-way Radio & Communication		\$97,190			6 S	
Trash Chute & Hoppers						X
Snow / ice Removal			X		S	
Safety Signage		\$25,000		- 2	( S	
Drug Testing	10 million (1997)	\$15,000		- 16	S	
Street Cleaning			X	1	. S.	
Safety Awards / Incentive Program		2	X	6		
Subtotal Category B		\$2,127,439				

#### C - Facilities, Equipment, and Services

Description	Basic Fee	General Conditions	Direct	A/E Cost	Authority Cost	N/A
Contractor Office Trailer(s) Rental		in an and the	1.1	8	. 3	
Temp Office Power & Heating Connections		\$871,500		- <u>1</u>		
Temp Office Power & Heating Utility Usage		Included in C.1 above				
Temp Office Phone / Internet Connections	8	Included in C.1 above		- 8	S	
Temp Office Phone / Internet Usage Fees		\$10,500		8		
Temp Office Drinking Water		\$9,430			6 8	
Temp Office Furniture		\$82,500				
Temp Office Equipment (Copiers, Phones, Fax, etc.)	8	\$63,675		- 8 8	E	
Temp Office Computers		\$185,452		- S - S		
Temp Office Supplies	1. Contract (1. Co	\$130,000				

The Peoples Stadium

Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
A/E Office Trailer(s) Rental	- 2	and the second				
Temp Office Power & Heating Connections, Permits & Fees		\$135,200		38 - S		
1 Temp Office Power & Heating Utility Usage	- S	Included in C.10 above		10 - 2		-
2 Temp Office Phone / Internet Connections, Permits & Fees		Included in C.10 above				
<sup>3</sup> Temp Office Phone / Internet Usage Fees		\$3,472		3 1 2		
Temp Office Drinking Water	- S S.	\$1,600		1 8		
5 Temp Office Furniture	3. S	\$15,000		3 1 2		-
Temp Office Equipment (Copiers, Phones, Fax, etc.)		\$13,720				
7 Temp Office Computers	121	\$13,500		18 - S		
<sup>8</sup> Temp Office Supplies		\$8,000		14		
Owner's Rep. Office Trailer(s) Rental	- 24			3P		
<sup>9</sup> Temp Office Power & Heating Connections, Permits & Fees		\$135,200				
Temp Office Power & Heating Utility Usage	- 31	included in C.19 above		11 8		_
1 Temp Office Phone / Internet Connections, Permits & Fees		Included in C.19 above		1 8		
2 Temp Office Phone / Internet Usage Fees		\$3,472				
Temp Office Drinking Water	3	\$1,600		1		
4 Temp Office Fumiture	- 12 - 12 - 12 - 12 - 12 - 12 - 12 - 12	\$15,000		- S		
Temp Office Equipment (Copiers, Phones, Fax, etc.)	- S S.	\$13,720		10 - 20		
<sup>8</sup> Temp Office Computers		\$13,500				
7 Temp Office Supplies	- 9	\$8,000		30		
8 Contractor Tool / Utility Trailer(s) Rental	- Sie	\$11,557		1: 3		_
Contractor Material Storage Trailer(s)		\$14,872		1		
Project Site				14 8		
Temporary Electrical Service Connections, Permits & Fees		By Xcel Energy				
1 Temporary Electrical Utility Usage	3 3	\$2,000,000		1 8		
2 Temporary Generators		Trade Specific costs	X			
3 Generator Fuel Usage & Maintenance	- 2	Trade Specific costs	X	3 S S		
Temporary Water Service Connections, Permits & Fees	3. 3	By MSFA		10 8	X	
5 Temporary Water Utility Usage		\$64,665		1 2		
Temporary Heating Service Connections, Permits & Fees		Included in C.35 above				

The Peoples Stadium

1	Description	Basic Fee	General Conditions	Direct	A/E Cost	Authority Cost	N/A
r i	Temporary Heat Utility/ Fuel	1	included in C.41 below		- C		2
	Temporary Heating / Cooling Equipment		\$428,111		0		<
0	Temporary Heating / Cooling		Included in C.38 above				
	Temporary Heating / Cooling Enclosures		\$602,145		0		3
	Temporary Heat Utility/ Fuel		\$746,126		12		( – –
	Temporary Tollets / Sewer Connect		\$446,189				_
	Temporary Decks, Partitions, Stairs and enclosures (exclusive of winter conditions)		\$175,000		0		
4	Project Signs / Bulletin Boards		\$15,000		- C		<u>.</u>
5	Temporary Roads and Walkways ONLY		\$6,078				
е,	Air Compressors / Generators		Included in C.48		- 0		8
7	Dewatering Equipment (Intermittant)		\$37,115		1		ý. – –
	Miscellaneous Equipment		\$32,333		- 0		
9	Misc. Fuel / Repairs / Maintenance		\$10,000				
	Construction Temporary Lighting		In Electrical Bid Peckage	X	- 12 2		X
	Smail Tools & Materials		\$32,591	326	- 0 0		1
16	Construction Holsts - Material			x	- Q	-	<u> </u>
	Construction Hoists - Labor	2 1		X	12 3		
	Staff Vehicles, Gas & Maintenance		\$284,715				
	Tower Crane(s) *		Trade Specific cests	X	-		X
	Tower Crane Operator "		Trade Specific costs	X	- C		X
	Tower Crane Freight and Set Up *		Trade Specific costa	X			X
	Erosion Control		1		- 12 2		
	Silt Fending			X	1. 1		
	Inlet Protection Temporary Seeding		-				-
	Misc. & Maintenance			X	12 5		-
	Temporary Seeding			x	10 0		_
ł	Temporary Seeding			Â	÷ (		8
1							-
E							
L		Subtotal Category C	\$6,640,537		12 3		

Exhibit 14

The Peoples Stadium

Description	Basic Fee	General Conditions	Direct	A/E Cost	Authority Cost	NIA
D - Reproduction and Printing (Pre-Constr	uction Phase)	<i>ii</i> 12.		33 0	C 18	
Description	Basic Fee	General Conditions	Direct	A/E Cost	Authority Cost	NIA
Design Development Documents		Included in Preconstruction Reimbursible Expenses Ibern 11				
Bid Package & Specs		See above		1		
Construction Documents		See above				
Postage & Express Costs	2	See above		12	e 31	
		<u>8</u>				
	0 19					
2	Subtotal Category D	\$0				

#### E - Reproduction and Printing (Construction Phase)

Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
All Forms	S	\$6,995		- 8	( S	
Presentation Charts/Graphics	S	0.00000000		- S - S		
Value Analysis Studies	S			8 6	6 S	
Reference Materials						
Duplication Expenses (Misc.)	Q			- 42	S	
Shop Drawing Printing	3	\$50,000		- 12	( S)	
Special Forms	( ) ( ) ( ) ( ) ( ) ( ) ( ) ( ) ( ) ( )	198890628		1	1	
Posting and Delivery Exp.	2	\$55,125		- 18 - 11	( <u>3</u>	
As-Built Documents (draft/print)		\$35,000				
Subtotal	Category E	\$147,120		9	8	

The Peoples Stadium

Description	Basic Fee	Conditions	Cost	A/E Cost	Authority Cost	NA
F- Quality Control		<u> </u>	402	- 10 - 13 - 13 - 13 - 13 - 13 - 13 - 13		2
Description	Basic Fee	Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
Field Inspector			0.000	3. 3	X	0
Special Inspection Consultants				0 3	X	-
Inspectors Transportation					X	
Inspectors Equipment				- 12	X	
Special Inspection Consultants				- C - 2	X	<u> </u>
Masonry Testing		3		1.2	X	
Concrete Testing				- C - S	X	
Compaction Testing	3.4	0		- Q	X	-
Weiding Testing					X	
Soils Investigation & Testing				- C	X	1
Special Testing Consultants		an an an an an an Arriente Arr		10	X	
Field Office Supplies / Materials		Included in C.9		- C		÷
Project Photographs	3.4	\$52,500		2 3	( ) ( ) ( ) ( ) ( ) ( ) ( ) ( ) ( ) ( )	
Air and Water Balance			X			
Operator On-site Training		53	X	- D	1	2
Prepare Operations Manuals			X	- 3		
Prepare Maintenance Manuals			X	- 3		<u>.</u>
Building Commissioning		8	X	- 2 - 3		
Surveying			X			
		33		- 10		2
				1.1		
				- 15 - E		3
		3		12 3		
				12		_
	Subtotal Category F	\$52,500				

	Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
1	Offsite Storage Yard Rental				- ()		X
2	Parking Lot Rentals		STREET, MARKEN		18 19		X
3	Parking Fees (Staff and Crew)		\$105,200		- S		5 - M
4	Road Closure Fees *		\$1,100,000				
6	Curb & Gutter Permits						X
6	Sign Permits (Jobsite Directional)		and the second se		- C		X
7	Building Permits (Based on \$690 million)		\$6,041,675		12 2		8 - 6 M
8	Plan check Fees If Applicable		included, item G.7		- S		2
1	After Hours / Noise Permits		\$109,500		- 12		
	SWPPP Fee		\$1,200				
1	Temporary Contractor Trailer(s)				- 12		
9	Trailer Permit	1			- 8 - B		X
10	Water Connection *				18 3		X
11	Sanitary Connection ."	1	8		- 6		X
	Gas Service Charge"						X
13	Special Tap Fees *				- 10		X

The Peoples Stadium

Ex	h	ib	it	1	4	
----	---	----	----	---	---	--

Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
Temporary A/E Trailer(s)			8	8	12	100
4 Trailer Permit			5	3	2 1	X
<sup>6</sup> Water Connection *				16	S	X
8 Sanitary Connection ."						X
7 Gas Service Charge*		2		38	2 2	X
8 Special Tap Fees "			5	10	S	X
Temporary Owner's Rep. Traller(s)		2	2	35	2	
Trailer Permit				1	S 1	X
Water Connection *						X
Sanitary Connection ."						X
2 Gas Service Charge"	3			3.	S	X
<sup>23</sup> Special Tap Fees *	2. 2.		Q	5		X
Temporary Job Site Services				1	8	
4 Water Connection *						X
Sanitary Connection ."		1		- R	8	X
* Storm Water Connection *			1	5 a	8 1	X
Gas Service Charge"	2		6	1	S	X
Power Service Charge *	0 0		2		8	X
9 Steam / Chilled Water Charge"						X
<sup>19</sup> Steam / Chilled Water Charge" 10 Special Tap Fees "		-		1	8 - A	X
Permanent Services			8	1	Second 1	
Water Connection *	20		6	1	X	
2 Sanitary Connection ."			5	1	X	
Storm Water Connection *					X	
4 Gas Service Charge*		4			X	
<sup>6</sup> Power Service Charge *	0			51	X	
Steam / Chilled Water Charge"	11.		(		X	
7 Special Tap Fees *	0			1	X	
Zoning Fees / Consultants *	5. SI			16	X	
Professional Surveying - ALTA					X	
0			2	1	S 11	
1						
	Subtotal Category G	\$7,357,575		1	I	

\* If applicable, represent final/connection cost, not temporary.

The Peoples Stadium

Description	Basic Fee	General	Direct	A/E Cost	Authority	N//
		Conditions	Cost		Cost	
H - Insurance						
Description	Basic Fee	General	Direct	A/E Cost	Authority	N/A
Description	Daole 1 66	Conditions	Cost	ALC COOL	Cost	
General Liability			\$4,319,000			
Completed Operations Liability *						X
Excess Liability Coverage			Included in H.3			
Off-site Insurance			Included in H.3			
Automobile			Included in H.3			
Worker's Compensation (any additional above the cost included Section A above)	in		\$23,539			
All Insurance based on a total construction value of \$500 million						
	ibtotal Category H		\$4,342,939			

#### I - Miscellaneous & Other Costs

-	2	Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
I (	3							
1 4	4							
I (	5							
I (	6							
I (	7							
I (	8							
I (	9							
I 1	0							
I	- [							
I	E							
		Subtotal	I Category I	\$0				

TOTAL GENERAL CONDITION COSTS (A to I) \$

\$ 38,420,523

\* If Applicable.

1

The Peoples Stadium

Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A		
J - Bonds								
Builders Risk insurance	\$2.764 per \$1,000 for a Fixed Roof Scheme. \$3.631 per \$1,000 for a Operable Roof Scheme. Attach separate sheet indicating rate schedule per \$1,000.00 of Hard Construct Costs							
Payment & Performance Bond	\$9.4352 per \$1,000 for a Warranty Period of 24 months. Attach separate sheet indicating rate schedule per \$1,000.00 of Hard Construct Costs							

2	Assume total Construction Cost (Including fee and general conditions) to total not more than	\$12,500,000
	\$740M	
	Total Basic Fee	\$12,500,000

NOTES:

Items defined as Direct Costs are included within the Fixed Construction Budget of \$690,000,000,000.00
 Permit Costs only cover GMP costs up to \$690,000,000.00. Additional permit costs for work above \$690,000,000.00 will be the responsibility of the Authority which will be in addition to the GCGMP.

The Peoples Stadium

**Project Cost Identification Matrix** 

Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A

### **A - Construction Services**

Princip Opera Projec Sr. Projec Assista Gener Superi Superi Superi Superi Assista Assista	prate Executives ple in Charge ations Manager ct Executive roject Manager ct Manager tant Project Manager ral Superintendent rintendent rintendent rintendent rintendent rintendent tant Superintendent tant Superintendent	\$1,269,080 \$100,000 \$1,102,926 \$2,404,370 \$3,123,648 \$254,716 \$974,908 \$2,361,659 Included in A.9 Included in A.9 Included in A.9			
Opera Projec Sr. Projec Assista Gener Superi Superi Superi Superi Assista Assista	ations Manager ct Executive roject Manager ct Manager tant Project Manager ral Superintendent rintendent rintendent rintendent rintendent rintendent tant Superintendent tant Superintendent tant Superintendent	\$100,000 \$1,102,926 \$2,404,370 \$3,123,648 \$254,716 \$974,908 \$2,361,659 Included in A.9 Included in A.9			
Projec Sr. Projec Assista Gener Superi Superi Superi Superi Assista Assista	ct Executive roject Manager ct Manager tant Project Manager ral Superintendent rintendent rintendent rintendent rintendent rintendent tant Superintendent tant Superintendent	\$1,102,926 \$2,404,370 \$3,123,648 \$254,716 \$974,908 \$2,361,659 Included in A.9 Included in A.9			
Projec Sr. Projec Assista Gener Superi Superi Superi Superi Assista Assista	ct Executive roject Manager ct Manager tant Project Manager ral Superintendent rintendent rintendent rintendent rintendent rintendent tant Superintendent tant Superintendent	\$1,102,926 \$2,404,370 \$3,123,648 \$254,716 \$974,908 \$2,361,659 Included in A.9 Included in A.9			
Sr. Projec Projec Assista Gener Superi Superi Superi Assista Assista	roject Manager ct Manager tant Project Manager ral Superintendent rintendent rintendent rintendent rintendent rintendent tant Superintendent tant Superintendent	\$2,404,370 \$3,123,648 \$254,716 \$974,908 \$2,361,659 Included in A.9 Included in A.9			
Project Assista Gener Superi Superi Superi Assista Assista	ct Manager tant Project Manager ral Superintendent rintendent rintendent rintendent rintendent rintendent tant Superintendent tant Superintendent	\$3,123,648 \$254,716 \$974,908 \$2,361,659 Included in A.9 Included in A.9			
Assista Gener Superi Superi Superi Superi Assista	tant Project Manager ral Superintendent rintendent rintendent rintendent rintendent rintendent tant Superintendent tant Superintendent	\$254,716 \$974,908 \$2,361,659 Included in A.9 Included in A.9			
Gener Superi Superi Superi Superi Assista	ral Superintendent rintendent rintendent rintendent rintendent rintendent tant Superintendent tant Superintendent	\$974,908 \$2,361,659 Included in A.9 Included in A.9			
Superi Superi Superi Superi Assista	rintendent rintendent rintendent rintendent rintendent tant Superintendent tant Superintendent	\$2,361,659 Included in A.9 Included in A.9			
Superi Superi Superi Assista	rintendent rintendent rintendent rintendent tant Superintendent tant Superintendent	Included in A.9 Included in A.9			
Superi Superi Superi Assista	rintendent rintendent rintendent tant Superintendent tant Superintendent				
Superi Superi Assista Assista	rintendent rintendent tant Superintendent tant Superintendent	Included in A.9			
Superi Assista Assista	rintendent tant Superintendent tant Superintendent				
Assista Assista	tant Superintendent tant Superintendent	Included in A.9			
Assista	tant Superintendent	Included in A.9		<u> </u>	
		 Included in A.9	1	<u>├</u> ───┼	
Senior		 Included in A.17	-	<u>                                     </u>	
	ct Engineer	\$2.863.451			
	ct Engineer	Included in A.17			
	ct Engineer	Included in A.17			
	ct Engineer	Included in A.17			
	ct Engineer ct Engineer (3-D modelling) (includes all VDC/BIM)	\$1.024.215			
	tant Engineer	<b>۵۱,024,215</b> Included in A.17		<u> </u>	
	Engineer (Site Layout)	Included in A.17		<u> </u>	
	C Manager	\$559.721			
	C Engineer	\$448,607		<u> </u>	
	y Officer / Manager				
		\$540,673			
	y Engineer	\$366,111			
	P. Coordinator	\$722,688 Included in A.17			
	s Consultant	Included in A.17			
	it Coordinator				
	etarial (On Site) (include admin support and office mgr)	\$702,970			
	D. Officer	 \$548,351			
Sched		\$470,842			
	ct Phasing	 Included in D		<b>├</b> ──── <b>├</b>	
	ct Budget Estimating	 Included in Precon Fee		├	
Estima		Included in Precon Fee		ļļ	
	/ Lead Estimator	Included in Precon Fee		ļļ	
	tant Estimator	Included in Precon Fee			
	tant Estimator	Included in Precon Fee			
	anical / Plumbing Estimator	Included in Precon Fee			
	rical Estimator	Included in Precon Fee			
	Engineer(s)	\$613,290			
Purcha					
	Engineering	\$110,668			
	ycle Analysis				
	uction Engineering h List Manager				

## **Project Cost Identification Matrix**

Description	Basic Fee	General	Direct	A/E Cost	Authority	N/A
		Conditions	Cost		Cost	
<sup>8</sup> Building Commissioning Manager		\$292,556				
<sup>19</sup> Legal Services Allowance		\$100,000				
<sup>50</sup> Accounting		\$765,866				
51 System Development						
<sup>52</sup> Drafting Detailer						
<sup>53</sup> Secretarial (Home Office)						
<sup>54</sup> Off-site Staff Travel Costs		\$207,500				
<sup>55</sup> Off-site Staff Transportation		\$6,534				
<sup>56</sup> Project Staff Moving/Subsistence		\$75,000				
7 Communtiy Outreach Expenses (includes directors/coordinator)		\$75,000				
8 Professional Services Agreements		\$10,000				
CONSTRUCTION SUB-TOTAL		\$ 22,095,351				
Pre-Construction						
<sup>1</sup> Pre-Construction Executive			\$1,000,000			
<sup>2</sup> Pre-Construction Manager			In line above			
<sup>3</sup> Project Manager			In line above			
<sup>4</sup> Estimator			In line above			
<sup>5</sup> Estimator			In line above			
<sup>6</sup> Estimator			In line above			
7 Estimator			In line above			
8 Purchasing			In line above			
9 Purchasing			In line above			
<sup>10</sup> Clerical			In line above			
Preconstruction Services Reimbursible Expenses			In line above			
PRE-CONSTRUCTION SUB-TOTAL		\$-	\$ 1,000,000			
TOTA	L I Category A	\$ 22.095.351				
1014	L I Calegory A	\$ 22,095,351				

**Project Cost Identification Matrix** 

Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
-------------	-----------	-----------------------	----------------	----------	-------------------	-----

## B - Safety, Security, and Services

Description	Basic Fee	General	Direct	A/E Cost	Authority	N/A
		Conditions	Cost		Cost	
<sup>1</sup> Safety Equipment		\$195,000				
<sup>2</sup> First Aid Supplies		included in B.01 above				
<sup>3</sup> Handrails & Toeboards			Х			
<sup>4</sup> Opening Protection			Х			
<sup>5</sup> Fire Extinguishers / Fire Watch		\$37,340				
<sup>6</sup> Security Guards / Flagman		\$532,480				
<sup>7</sup> Weekly Cleanup			Х			
<sup>8</sup> Final Clean (interior/exterior)			Х			
<sup>9</sup> Temporary Fencing		\$315,054				
<sup>10</sup> Covered Walkways						Х
<sup>11</sup> Safety Nets & Apparatus'			Х			
<sup>12</sup> Debris Haul / Removal		\$710,375				
<sup>13</sup> Traffic Control		\$200,000				
<sup>14</sup> Roadway Maintenance			Х			
<sup>15</sup> Dust Control (all)			Х			
<sup>16</sup> 2-way Radio & Communication		\$97,190				
<sup>17</sup> Trash Chute & Hoppers						Х
<sup>18</sup> Snow / Ice Removal			Х			
<sup>19</sup> Safety Signage		\$25,000				
<sup>20</sup> Drug Testing		\$15,000				
<sup>21</sup> Street Cleaning			Х			
<sup>22</sup> Safety Awards / Incentive Program			Х			
Subtotal Category B		¢0.407.400				
Subiolal Calegory B		\$2,127,439				

## C - Facilities, Equipment, and Services

Description	Basic Fee	General	Direct	A/E Cost	Authority	N/A
		Conditions	Cost		Cost	
Contractor Office Trailer(s) Rental						
Temp Office Power & Heating Connections		\$871,500				
Temp Office Power & Heating Utility Usage		Included in C.1 above				
Temp Office Phone / Internet Connections		Included in C.1 above				
Temp Office Phone / Internet Usage Fees		\$10,500				
Temp Office Drinking Water		\$9,430				
Temp Office Furniture		\$82,500				
Temp Office Equipment (Copiers, Phones, Fax, etc.)		\$63,675				
Temp Office Computers		\$185,452				
Temp Office Supplies		\$130,000				

**Project Cost Identification Matrix** 

	Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
	A/E Office Trailer(s) Rental						
1(	<sup>0</sup> Temp Office Power & Heating Connections, Permits & Fees		\$135,200				
1	1 Temp Office Power & Heating Utility Usage		Included in C.10 above				
12	<sup>2</sup> Temp Office Phone / Internet Connections, Permits & Fees		Included in C.10 above				
1:	<sup>3</sup> Temp Office Phone / Internet Usage Fees		\$3,472				
14	<sup>4</sup> Temp Office Drinking Water		\$1,600				
	<sup>5</sup> Temp Office Furniture		\$15,000				
10	<sup>6</sup> Temp Office Equipment (Copiers, Phones, Fax, etc.)		\$13,720				
1	7 Temp Office Computers		\$13,500				
18	<sup>8</sup> Temp Office Supplies		\$8,000				
	Owner's Rep. Office Trailer(s) Rental						
19	<sup>9</sup> Temp Office Power & Heating Connections, Permits & Fees		\$135,200				
20	<sup>0</sup> Temp Office Power & Heating Utility Usage		Included in C.19 above				
2	<sup>1</sup> Temp Office Phone / Internet Connections, Permits & Fees		Included in C.19 above				
2	<sup>2</sup> Temp Office Phone / Internet Usage Fees		\$3,472				
2	<sup>3</sup> Temp Office Drinking Water		\$1,600				
24	<sup>4</sup> Temp Office Furniture		\$15,000				
2	<sup>5</sup> Temp Office Equipment (Copiers, Phones, Fax, etc.)		\$13,720				
20	<sup>6</sup> Temp Office Computers		\$13,500				
	7 Temp Office Supplies		\$8,000				
	<sup>8</sup> Contractor Tool / Utility Trailer(s) Rental		\$11,557				
29	<sup>9</sup> Contractor Material Storage Trailer(s)		\$14,872				
	Project Site						
30	<sup>0</sup> Temporary Electrical Service Connections, Permits & Fees		By Xcel Energy				
3	<sup>1</sup> Temporary Electrical Utility Usage		\$2,000,000				
32	<sup>2</sup> Temporary Generators		Trade Specific costs	Х			
3	<sup>3</sup> Generator Fuel Usage & Maintenance		Trade Specific costs	Х			
34	<sup>4</sup> Temporary Water Service Connections, Permits & Fees		By MSFA			X	
3	<sup>5</sup> Temporary Water Utility Usage		\$64,665				
36	<sup>6</sup> Temporary Heating Service Connections, Permits & Fees		Included in C.35 above				

**Project Cost Identification Matrix** 

		Description	Basic Fee	General	Direct	A/E Cost		N/A
				Conditions	Cost		Cost	
С	37	Temporary Heat Utility/ Fuel		Included in C.41 below				
С	38	Temporary Heating / Cooling Equipment		\$428,111				
С	39	Temporary Heating / Cooling		Included in C.38 above				
С	40	Temporary Heating / Cooling Enclosures		\$602,145				
С	41	Temporary Heat Utility/ Fuel		\$746,126				
С	42	Temporary Toilets / Sewer Connect		\$446,189				
С	43	Temporary Decks, Partitions, Stairs and enclosures		\$175,000				
		(exclusive of winter conditions)						
С	44	Project Signs / Bulletin Boards		\$15,000				
С	45	Temporary Roads and Walkways ONLY		\$6,078				
С	46	Air Compressors / Generators		Included in C.48				
С	47	Dewatering Equipment (Intermittant)		\$37,115				
С		Miscellaneous Equipment		\$32,333				
С	49	Misc. Fuel / Repairs / Maintenance		\$10,000				
С		Construction Temporary Lighting		In Electrical Bid Package	Х			Х
С		Small Tools & Materials		\$32,591				
С		Construction Hoists - Material			Х			
С	ŀ	Construction Hoists - Labor			Х			
С		Staff Vehicles, Gas & Maintenance		\$284,715				
С		Tower Crane(s) *		Trade Specific costs	Х			Х
С		Tower Crane Operator *		Trade Specific costs	Х			Х
С	57	Tower Crane Freight and Set Up *		Trade Specific costs	Х			Х
С		Erosion Control						
С		Silt Fencing			Х			
С		Inlet Protection			Х			
С		Temporary Seeding			Х			
С		Misc. & Maintenance			Х			
С	62	Temporary Seeding			X			
С								
С								
C								
C								
C C	ŀ							
C C								
C	L	Subtat	al Category C	\$6,640,537				
		Subtor	a salegory o	<b>Φ</b> 0,040,337				

## **Project Cost Identification Matrix**

D

D D D D

D D D D

Description	Basic Fee	General	Direct	A/E Cost	Authority	N/A
		Conditions	Cost		Cost	
D - Reproduction and Printing (Pre-Construction	Phase)					
Description	Basic Fee	General	Direct	A/E Cost	Authority	N/A
		Conditions	Cost		Cost	
Design Development Documents		Included in Preconstruction Reimbursible Expenses Item 11				
Bid Package & Specs		See above				
Construction Documents		See above				
Postage & Express Costs		See above				
	Subtotal Category D	\$0				

### E - Reproduction and Printing (Construction Phase)

Description	Basic Fee	General	Direct	A/E Cost	Authority	N/A
-		Conditions	Cost		Cost	
All Forms		\$6,995				
Presentation Charts/Graphics						
Value Analysis Studies						
Reference Materials						
Duplication Expenses (Misc.)						
Shop Drawing Printing		\$50,000				
Special Forms						
Posting and Delivery Exp.		\$55,125				
As-Built Documents (draft/print)		\$35,000				
	Subtotal Category E	\$147,120				

## **Project Cost Identification Matrix**

Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N//
F- Quality Control						
Description	Basic Fee	General	Direct	A/E Cost	Authority	N/.
		Conditions	Cost		Cost	
Field Inspector					X	
Special Inspection Consultants					X	
Inspectors Transportation					X	
Inspectors Equipment					X	
Special Inspection Consultants					X	
Masonry Testing					X	
Concrete Testing					Х	
Compaction Testing					X	
Welding Testing					X	
Soils Investigation & Testing					Х	
Special Testing Consultants					X	
Field Office Supplies / Materials		Included in C.9				
Project Photographs		\$52,500				
Air and Water Balance			Х			
Operator On-site Training			Х			
Prepare Operations Manuals			Х			
Prepare Maintenance Manuals			Х			
Building Commissioning			Х			
Surveying			Х			
2						
3						
5						
	Subtotal Category F	\$52,500				

### G - Permits & Special Fees

Description	Basic Fee	General	Direct	A/E Cost	Authority	N/A
		Conditions	Cost		Cost	
Offsite Storage Yard Rental						Х
Parking Lot Rentals						Х
Parking Fees (Staff and Crew)		\$105,200				
<sup>1</sup> Road Closure Fees *		\$1,100,000				
Curb & Gutter Permits						Х
ີ່ Sign Permits (Jobsite Directional)						Х
Building Permits (Based on \$690 million)		\$6,041,675				
Plan check Fees If Applicable		Included, Item G.7				
After Hours / Noise Permits		\$109,500				
SWPPP Fee		\$1,200				
Temporary Contractor Trailer(s)						
Trailer Permit*						Х
Water Connection *						Х
Sanitary Connection .*						Х
2 Gas Service Charge*						Х
<sup>3</sup> Special Tap Fees *						Х

## **Project Cost Identification Matrix**

	Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
	Temporary A/E Trailer(s)						
1	14 Trailer Permit*						Х
1	<sup>15</sup> Water Connection *						Х
1	16 Sanitary Connection .*						Х
1	17 Gas Service Charge*						Х
1	18 Special Tap Fees *						Х
	Temporary Owner's Rep. Trailer(s)						
1	<sup>19</sup> Trailer Permit*						Х
2	<sup>20</sup> Water Connection *						Х
2	<sup>21</sup> Sanitary Connection .*						Х
2	<sup>22</sup> Gas Service Charge*						Х
2	<sup>23</sup> Special Tap Fees *						Х
	Temporary Job Site Services						
2	<sup>24</sup> Water Connection *						Х
2	<sup>25</sup> Sanitary Connection .*						Х
2	<sup>26</sup> Storm Water Connection *						Х
2	<sup>27</sup> Gas Service Charge*						Х
2	<sup>28</sup> Power Service Charge *						Х
	<sup>29</sup> Steam / Chilled Water Charge*						Х
3	<sup>30</sup> Special Tap Fees *						Х
	Permanent Services						
3	<sup>31</sup> Water Connection *					Х	
	<sup>32</sup> Sanitary Connection .*					Х	
3	<sup>33</sup> Storm Water Connection *					X	
3	<sup>34</sup> Gas Service Charge*					Х	
3	<sup>35</sup> Power Service Charge *					Х	
3	<sup>36</sup> Steam / Chilled Water Charge*					X	
	<sup>37</sup> Special Tap Fees *					Х	
	<sup>38</sup> Zoning Fees / Consultants *					X	
	<sup>39</sup> Professional Surveying - ALTA					Х	
4	40						
	41						
4	42						
		Subtotal Category G	\$7,357,575				

\* If applicable, represent final/connection cost, not temporary.

**Project Cost Identification Matrix** 

Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A
H - Insurance						

#### 1 <sup>2</sup> Description Basic Fee General Direct A/E Cost Authority N/A Conditions Cost Cost <sup>3</sup> General Liability \$4,319,000 Н Н <sup>4</sup> Completed Operations Liability \* Х Included in H.3 Н <sup>5</sup> Excess Liability Coverage Included in H.3 <sup>6</sup> Off-site Insurance Н 7 Automobile Included in H.3 Н Н <sup>8</sup> Worker's Compensation (any additional above the cost included in \$23,539 Section A above) Н 9 Н All insurance based on a total construction value of \$690 million Н \$4,342,939 Subtotal Category H

#### I - Miscellaneous & Other Costs

Description	Basic Fee	General	Direct	A/E Cost	Authority	N/A
		Conditions	Cost		Cost	
3						
L I I I I I I I I I I I I I I I I I I I						
5						
7						
3						
	Subtotal Category I	\$0				

TOTAL GENERAL CONDITION COSTS (A to I) \$

\$ 38,420,523

\* If Applicable.

1

## **Project Cost Identification Matrix**

Description	Basic Fee	General Conditions	Direct Cost	A/E Cost	Authority Cost	N/A	
-------------	-----------	-----------------------	----------------	----------	-------------------	-----	--

## J - Bonds

J

J

Builders Risk Insurance	\$2.784 per \$1,000 for a Fixed Roof Scheme.
	\$3.631 per \$1,000 for a Operable Roof Scheme.
	Attach separate sheet indicating rate schedule per \$1,000.00 of Hard
	Construct Costs
Payment & Performance Bond	\$9.4352 per \$1,000 for a Warranty Period of 24 months.
	Attach separate sheet indicating rate schedule per \$1,000.00 of Hard
	Construct Costs

### K - Basic Fee

 Assume total Construction Cost (including fee and general conditions) to total not more than \$740M	\$12,500,000
Total Basic Fee	\$12,500,000

#### NOTES:

1. Items defined as Direct Costs are included within the Fixed Construction Budget of \$690,000,000.00

2. Permit Costs only cover GMP costs up to \$690,000,000.00. Additional permit costs for work above \$690,000,000.00 will be the responsibility of the Authority which will be in addition to the GCGMP.